

# Energy and Seismic Renovation Strategies for Sustainable Cities

Edited by Giuseppe Margani Printed Edition of the Special Issue Published in Sustainability



www.mdpi.com/journal/sustainability

# **Energy and Seismic Renovation Strategies for Sustainable Cities**

## **Energy and Seismic Renovation Strategies for Sustainable Cities**

Special Issue Editor Giuseppe Margani

MDPI • Basel • Beijing • Wuhan • Barcelona • Belgrade



Special Issue Editor Giuseppe Margani University of Catania Italy

Editorial Office MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

This is a reprint of articles from the Special Issue published online in the open access journal *Sustainability* (ISSN 2071-1050) from 2018 to 2019 (available at: https://www.mdpi.com/journal/sustainability/special\_issues/strategies\_sustainable\_cities)

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

LastName, A.A.; LastName, B.B.; LastName, C.C. Article Title. *Journal Name* Year, Article Number, Page Range.

ISBN 978-3-03897-944-9 (Pbk) ISBN 978-3-03897-945-6 (PDF)

Cover image courtesy of Sebastiano D'Urso and Bruno Cicero.

© 2019 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license, which allows users to download, copy and build upon published articles, as long as the author and publisher are properly credited, which ensures maximum dissemination and a wider impact of our publications.

The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons license CC BY-NC-ND.

## Contents

About the Special Issue Editor
Preface to "Energy and Seismic Renovation Strategies for Sustainable Cities" ix
Paolo La Greca and Giuseppe MarganiSeismic and Energy Renovation Measures for Sustainable Cities: A Critical Analysis of theItalian ScenarioReprinted from: Sustainability 2018, 10, 254, doi:10.3390/su100102541
Emil-Sever Georgescu, Mihaela Stela Georgescu, Zina Macri, Edoardo Michele Marino, Giuseppe Margani, Vasile Meita, Radu Pana, Santi Maria Cascone, Horia Petran, Pier Paolo Rossi, Vincenzo Sapienza and Marius Voica Seismic and Energy Renovation: A Review of the Code Requirements and Solutions in Italy and Romania Reprinted from: Sustainability 2018, 10, 1561, doi:10.3390/su1005156120
Sebastiano D'Urso and Bruno CiceroFrom the Efficiency of Nature to Parametric Design. A Holistic Approach for SustainableBuilding Renovation in Seismic RegionsReprinted from: Sustainability 2019, 11, 1227, doi:10.3390/su1105122756
Riccardo Privitera and Daniele La RosaReducing Seismic Vulnerability and Energy Demand of Cities through Green InfrastructureReprinted from: Sustainability 2018, 10, 2591, doi:10.3390/su1008259176
Annarita Ferrante, Giovanni Mochi, Giorgia Predari, Lorenzo Badini, Anastasia Fotopoulou, Riccardo Gulli and Giovanni SempriniA European Project for Safer and Energy Efficient Buildings: Pro-GET-onE (Proactive Synergy of inteGrated Efficient Technologies on Buildings' Envelopes)Reprinted from: Sustainability 2018, 10, .812, doi:10.3390/su1003081297
Tiziano Dalla Mora, Maria Pinamonti, Lorenzo Teso, Giosuè Boscato, Fabio Peron and Piercarlo Romagnoni         Renovation of a School Building: Energy Retrofit and Seismic Upgrade in a School Building in Motta Di Livenza         Reprinted from: Sustainability 2018, 10, 969, doi:10.3390/su10040969
Mariangela De Vita, Antonio Mannella, Antonio Sabino and Alessio Marchetti Seismic Retrofit Measures for Masonry Walls of Historical Buildings, from an Energy Saving Perspective Reprinted from: <i>Sustainability</i> <b>2018</b> , <i>10</i> , 984, doi:10.3390/su10040984
<b>Tiziana Basiricò and Daniele Enea</b> Seismic and Energy Retrofit of the Historic Urban Fabric of Enna (Italy)Reprinted from: Sustainability 2018, 10, 1138, doi:10.3390/su10041138
Angela Moschella, Antonio Gagliano, Alessandro Lo Faro, Attilio Mondello, Angelo Salemi and Giulia Sanfilippo A Methodology for an Integrated Approach for Seismic and Energy Refurbishment of Historic Buildings in Mediterranean Area Reprinted from: <i>Sustainability</i> <b>2018</b> , <i>10</i> , 2448, doi:10.3390/su10072448

Marina Fumo, Antonio Formisano, Giulia Sibilio and Antonella Violano	
Energy and Seismic Recovering of Ancient Hamlets: the Case of Baia e Latina	
Reprinted from: Sustainability 2018, 10, 2831, doi:10.3390/su10082831	206
Rossella Corrao	
Mechanical Tests on Innovative BIPV Façade Components for Energy, Seismic, and Aesthetic	
Renovation of High-Rise Buildings	
Reprinted from: Sustainability 2018, 10, 4523, doi:10.3390/su10124523	224

## About the Special Issue Editor

**Giuseppe Margani**, MEng, Ph.D., is associate professor of Building Construction at the Department of Civil Engineering and Architecture (DICAr), University of Catania, Italy. He received his Ph.D. in "Architectural Engineering: design, production and renovation techniques" from the University of Palermo in 2000. His research interests include sustainable architecture, energy and seismic renovation, integration of renewable energy systems in the building envelope, innovation in building technologies, restoration of historical buildings, history of construction technologies. He is the author of 6 books and the author or co-author of over 60 papers published on national and international peer-reviewed journals or conference proceedings. He has a sound experience in the development and direction of national and international research projects and he is an independent expert of the EC for research and innovation programs. He is serving as guest editor, editorial board member and reviewer of over 10 international journals. As a licensed architectural engineer, he has over 25 years of work experience in detailed design, building refurbishment, and non-destructive tests on cultural assets.

## Preface to "Energy and Seismic Renovation Strategies for Sustainable Cities"

Sustainability has become a fundamental requirement for the future of our cities.

This requirement is mostly associated with environmental issues, and a great effort has been made in the past years to build a low-carbon society. However, sustainability must also be associated with safety.

As a consequence, in seismic countries, sustainable cities must be not only low-carbon-emitting bust also earthquake-safe.

This concept represents the basic premise of this book.

According to this premise, in earthquake-prone nations like Italy—where most of the building stock is both highly earthquake-vulnerable and energy-consuming—energy renovation actions should be combined with seismic upgrades. Nevertheless, many barriers significantly limit the real possibility of undertaking combined retrofit measures, especially in the case of multi-owner housing and high-rise buildings. These barriers are of different kinds: economic/financial (high renovation costs, insufficient incentives and subsidies, landlord-tenant dilemmas, etc.), technical (ineffectiveness of conventional upgrade solutions, need of regulatory simplification, etc.), organizational (occupants' dislocation and disruption, consensus to the retrofit expenditure by condominium ownerships, excessive time for getting construction permits, etc.), and cultural/social (insufficient information and skills, lack of adequate policy measures for promoting renovation actions).

This book aims to overcome these barriers and to bridge the gap between sustainability and safety, so to conserve both human and environmental resources. It brings together 11 contributions on different seismic and energy renovation measures, proposing technical solutions at district, building, and component level, for both historic and modern case studies.

Finally, I would like to thank the editorial team of *Sustainability* for inviting me to guest-edit the Special Issue on "Energy and Seismic Renovation Strategies for Sustainable Cities" which has been transformed into this book. I also thank all the authors and reviewers for their fundamental contributions.

Giuseppe Margani Special Issue Editor





## Article Seismic and Energy Renovation Measures for Sustainable Cities: A Critical Analysis of the Italian Scenario

#### Paolo La Greca and Giuseppe Margani \*

Department of Civil Engineering and Architecture, University of Catania, Via Santa Sofia 64, 95123 Catania, Italy; paolo.lagreca@unict.it

\* Correspondence: margani@unict.it; Tel.: +39-095-738-2509

Received: 13 December 2017; Accepted: 15 January 2018; Published: 19 January 2018

Abstract: One of the main challenges of the twenty-first century is to increase the sustainability level of our cities. However, a town, to be considered sustainable, must, above all, be safe, particularly against natural hazards, which in Europe are mostly related to climate changes (e.g., hurricanes, floods, storms, and landslides) and seismic events (earthquakes). Unfortunately, sustainability is still not a prerogative of most European cities, especially those placed in seismic countries such as Italy, where at least 50% of the residential stock is earthquake-prone, while over 80% of the same stock is highly energy-consuming and carbon dioxide-emitting, thus contributing to trigger hazards related to climate changes. In this context, renovation actions, which combine both energy and seismic issues are strongly needed. Nevertheless, several technical, organizational and financial barriers considerably limit the real possibility to extensively undertake this kind of renovation. This study analyzes such barriers, with particular reference to the Italian scenario, suggesting and discussing possible solutions and underlining the advantages of increasing energy and seismic performances at the same time. The proposed solutions may be effectively extended to many other countries with similar socio-economic scenarios.

Keywords: seismic retrofit; energy retrofit; sustainability; safety; policy measures; apartment blocks

#### 1. Introduction

Sustainability was not an explicit value until the last quarter of the 20th century; therefore, sustainability performances were not requested in the recent past.

The approach has changed in connection with the intensification of climate change, environmental degradation, overconsumption of natural resources, population growth and pursuit of incessant economic rise. Today, sustainability is instead considered a fundamental quality and a prerogative in any socio-economic context.

According to the definition of the 2005 United Nations World Summit, sustainability is based on three main pillars: environment, society and economy [1]. However, the social dimension, and in particular safety, has often been neglected, especially in relation to the vulnerability of the building stock.

In the last years, the EU has produced big financial efforts to increase the sustainability level of our cities and, in the 2014–2020 budget, over 5% of the European Regional Development Fund has been allocated to sustainable urban development [2–4].

These resources have mainly been driven towards energy efficiency and low-carbon measures, to reduce the energy bills and the hazard risks related to climate-change (e.g., hurricanes, floods, storms, landslides, desertification, melting of glaciers and sea level rise) that may cause significant damages and life losses. Special attention has been paid to existing buildings, which are responsible

for about 40% of the final energy demand and therefore represent a great opportunity for energy saving and decarbonization [4–6]. In particular, renovation activities have been privileged over new constructions to limit urban-sprawl and soil consumption, according to the European aim to achieve no net land take by 2050 [7].

However, less effort has been made to reduce the seismic vulnerability of the existing real estate, mostly due to the high rate of European countries that are not listed as earthquake-prone (Figure 1). Consequently, the sustainability level of towns placed in seismic areas remains inadequate, since most buildings and infrastructures are unsafe, i.e., not sufficiently earthquake-resistant.



Figure 1. European-Mediterranean seismic hazard map [8].

In such areas, energy renovation actions must be synergistically combined with seismic retrofitting, for two main reasons: (a) to prevent life losses and damages caused by earthquakes; and (b) to avoid several costs otherwise duplicated, for instance those for building-site setup and scaffolds, as well as for cladding, plasters and other finishing [9]. As illustrated by Belleri and Marini, in the case of energy refurbishment only, the risk of a building located in seismic regions can be equalized to an additional annual embodied equivalent  $CO_2$  that almost equals its annual operational  $CO_2$  [10].

Nevertheless, several barriers considerably limit the real possibility to extensively undertake combined retrofit actions, especially for multi-owner housing and high-rise buildings. These barriers are of different kinds: (i) technical (e.g., unfeasibility and/or ineffectiveness of conventional retrofit solutions, and need of regulatory simplification); (ii) financial (e.g., high renovation costs, "split-incentive"/"landlord-tenant dilemma", and insufficient incentives and subsidies); (iii) organizational (e.g., temporary alternate accommodation for occupants, consensus to the retrofit expenditure by condominium ownerships, and excessive time to obtain building permits); and (iv) cultural/social (insufficient information and skills, and lack of adequate policy measures to promote renovation actions).

According to this general premise, this study intends to fill the gap existing in the scientific literature regarding combined seismic and energy renovation strategies and, in particular, aims to: (a) outline the scenario of seismic vulnerability and energy performance of the Italian residential

building stock, with particular reference to that built during 1950–1990; (b) review and discuss the barriers which limit the concrete possibility to extensively undertake combined seismic and energy retrofitting interventions; and (c) suggest and discuss possible countermeasures to overcome such barriers and promote combined renovation actions.

The proposed countermeasures, which will be discussed with specific reference to the Italian reality, may be effectively applied also in other countries with similar hazard scenarios and socio-economic backgrounds, contributing to effectively enhance the sustainability level of their towns.

#### 2. Seismic Vulnerability of the Italian Building Stock and Current Renovation Strategies

#### 2.1. Seismic Vulnerability

Along with Greece, Turkey, Bulgaria, Romania, Iceland and most Balkan states, Italy is listed among the most earthquake-prone European nations (Figure 1). In these countries, the seismicity is not as high as in the Pacific coast of the Americas or in some Asian regions such as Japan, Indonesia, Philippines, Mongolia, Pakistan, Nepal, or China. Nevertheless, telluric shakes often produce in Europe significant damages for the high vulnerability of the building stock and the considerable value of that large portion classified as historic architectural heritage.

The seismicity of the Italian peninsula is due to its geographic position at the convergence of the Eurasian and African tectonic plates. The relative movement between these plates causes energy accumulation and deformations that are occasionally released in the form of earthquakes of different magnitude. The highest seismicity is concentrated in the central and southern part of the country, along the Apennine chain, in Calabria and Sicily, as well as in some northern regions, such as Friuli-Venezia Giulia.

Since 1974, Italy has been progressively classified into seismic zones, based on past earthquakes intensity and frequency. The last seismic classification map, updated in 2015, has catalogued 44% of the territory and 36% of the municipalities as highly hazardous, i.e., with a peak ground acceleration (PGA) value > 0.15 g (zones 1, 2, 2A, 2B of Figure 2) and a 10% chance of being exceeded in 50 years [11]. More in detail, nearly 3 million people live inside areas exposed to a very high-hazard level (zone 1; 0.25 g < PGA  $\leq$  0.35 g), while 18.8 million in areas exposed to a high-hazard level (zone 2, 2A, 2B; 0.15 g < PGA  $\leq$  0.25 g), i.e., globally 21.8 million people live in highly earthquake-prone municipalities, that is around 36% of the whole Italian population (Figure 3).



Figure 2. Seismic classification map of Italy (2015), with indication of seismic areas according to PGA values [12].



Figure 3. Distribution of the Italian population, area and municipalities over different hazard levels/seismic zones [11].

According to the 2011 census of the Italian national statistical institute (ISTAT), around 2/3 of the existing residential stock was built before 1974 (Figure 4), i.e., before the enforcement of Law 64/1974 [13], which represents the first specific and extensive code for earthquake-resistant buildings in Italy. This code applied only to new edifices included in the seismic classification map that has been progressively updated by releasing revised versions. Therefore, after 1974, a great number of Italian municipalities, now included in the map, were not yet classified as seismically vulnerable. Consequently, even after 1974, a significant number of edifices have been designed neglecting the above-mentioned code.



Figure 4. Historical growth of the residential building stock in Italy (data from 2011 ISTAT census [14]).

The same census also reported that nearly 1.8 million residential buildings erected before 1974 (i.e., 15% of the overall stock) lie in a poor or mediocre state of conservation, as shown in Table 1.

 Table 1. Number of residential buildings by age and state of conservation (data from 2011
 ISTAT census [14]).

	Poor	Mediocre	Good	Very Good	Total
Before 1919	74,561	441,737	896,196	420,010	1,832,504
1919-1945	53,159	348,766	672,771	252,311	1,327,007
1946-1960	36,389	375,174	940,919	348,354	1,700,836
1961-1970	20,126	320,106	1,209,616	500,985	2,050,833
1971-1980	11,533	221,145	1,254,545	630,428	2,117,651
1981-1990	5422	104,265	800,786	552,294	1,462,767
1991-2000	1743	25,896	334,992	508,386	871,017
2001-2005	542	6718	108,670	349,174	465,104
2005-2011	566	3960	46,791	308,662	359,979
Total	204,041	1,847,767	6,265,286	3,870,604	12,187,698

Based on this premise, estimating that at least 8% of the post-1974 building stock has been realized out of the seismic codes and considering that, according to Figure 3, around 70% of the Italian territory is included in areas exposed from a very high to an intermediate-hazard level (zones 1, 2 and 3), today over 50% of the Italian residential buildings (i.e., nearly 6.4 million) are not earthquake-safe and need urgent actions to improve their seismic resilience.

In the last 2500 years, Italy has been hit by over 30,000 earthquakes of medium-high intensity (>IV–V degree of the Mercalli scale), and approximately 560 of VIII intensity or more (nearly one every 4.5 years). The last century has witnessed seven earthquakes with a maximum Moment Magnitude Scale (MMS) greater than or equal to 6.5 (i.e., between X and XI degrees of the Mercalli scale) [15]. In comparison with other countries, in Italy, the ratio between the damage caused and the energy released by earthquakes is particularly high, due to the high population density and the vulnerability of the building stock.

To perceive the socio-economic consequences of the seismic vulnerability of Italian cities, it must be highlighted that, in the last 50 years, earthquakes have caused around 5000 deaths and over  $\in$ 150 billion damage [16]. To these figures, one should also add significant psychological consequences, whose effects might be overcome by the whole community only after several decades.

In this sense, the earthquake that struck the region of Abruzzo in April 2009 is emblematic: with a maximum MMS of 6.3, this earthquake caused 308 deaths, over 1500 injuries, nearly 70,000 homeless, and over  $\notin$ 10 billion damages [17]. Today, seven years after, the beautiful and beloved historic center of L'Aquila—capital of the Abruzzo region—is still ruined, since both State and private investors have not yet found sufficient financial resources for its reconstruction. In similar conditions are many other neighboring towns that have been hit by the same earthquake. Recently, from August 2016 to January 2017, a new series of earthquakes with a maximum MMS between 6.0 and 6.5 hit several regions of central Italy, killing 333 people and devastating entire towns, with more than  $\notin$ 23 billion damages [18].

Considering the priorities, it must be pointed out that the financial efforts for seismic retrofit interventions have to be primarily allocated to infrastructures and strategic public buildings, but it is also extremely important to upgrade the residential stock, because it is generally more continuously inhabited and, consequently, produces the highest number of deaths in case of seismic events.

According to this scenario and considering the high frequency of earthquakes in Italy, the seismic retrofit of the existing building stock represents an imperative, since it allows to consistently reduce the extent of damage and the number of victims.

#### 2.2. Current Seismic Retrofit Strategies

Today, there are two main types of seismic retrofit strategies for edifices with reinforced concrete (RC) structure [19], which in Italy represents the building standard since the 1950s:

- strengthening of the existing structure with conventional techniques (jacketing with RC, steel, or fiber-reinforced polymers) and/or addition of extra structural members (pillars, shear walls, beams, foundations) [20,21]; and
- installation of base isolators and/or energy dissipation devices [22–25].

Another possible renovation measure consists in installing an external steel braced frame, which partially wraps the building and reduces its oscillations through energy dissipation devices applied between the nodes of the old and the new structure. Nevertheless, the adoption of this solution is restricted to limited cases, since it is mostly suitable for isolated edifices and generally requires a considerable strengthening of the existing RC frame too.

Historic and listed buildings, which in Europe account for around 30% of the current stock [4], need dedicated solutions, due to conservation issues and the variety of building fabric [26]. These edifices in Italy generally have unreinforced masonry bearing walls and, according to their fabric and their cultural value, they request different seismic-retrofit interventions, such as application

of anchoring and tying devices, mortar injection, overlay of RC or fiber-reinforced polymer layers, bracing (e.g., steel sections, reinforced masonry, buttresses), insertion of internal or external frames, post-tensioning of unreinforced masonry walls.

A detailed overview and description of the above-mentioned strategies goes beyond the scope of this work.

#### 3. Energy Performance of the Italian Building Stock and Current Renovation Interventions

#### 3.1. Energy Performance

As for many other countries, in Italy, the household and tertiary sector is the most energy-consuming one, accounting for 37.7% of the overall demand (Figure 5). To reduce the energy need of this sector, in the last decade the Directive on the Energy Performance of Buildings (EPBD) has effectively promoted the increase of the energy performance especially for new buildings.



Figure 5. Final energy consumption by sector in Italy (Data from Italian Energy Balance 2016 [27]).

However, new buildings have little influence on the overall demand, since their incidence on the whole building stock is very low. Nowadays, new edifices increase the existing stock on average by less than 1.5% every year (Figure 6), while the demolition rate is estimated around 0.2–0.5% per year [4,28]. Therefore, nearly 85% of European building stock predicted for 2030 has already been built.



Figure 6. Annual growth rate of new buildings added to the existing stock by nation [28].

These data suggest the need to improve, above all, the performance of the existing building stock to reduce energy consumptions and greenhouse gas emissions. This applies particularly to Italy, whose edifices are, in general, performing badly. Indeed, here the first regulation concerning the reduction of energy consumption for buildings was issued in 1976 [29], but it was low-restrictive and often neglected, due to insufficient controls. The first effective and comprehensive legislation was only introduced in 1991 [30], when 86% of the current residential stock had been already built (Figure 4). For this reason, today, most Italian edifices are characterized by an annual heating demand ranging from 140 to 220 kWh/m<sup>2</sup> (Figure 7) [31–33], i.e., significantly over the limits set by current laws. Consequently, they need urgent energy retrofitting measures.



Figure 7. Heating demand by different building types and age groups in Italy (kWh/m<sup>2</sup>/year) [33].

According to the National Energy Balance 2016, in Italy, 76% of the energy demand is covered by fossil fuels (Figure 8), which are mostly imported. This overdependence, along with bad energy performances, lead not only to high energy bills, but also to excessive greenhouse gas and pollutant emissions. Of course, excessive greenhouse gas emissions contribute to global warming and climate changes, which may cause the aforementioned hazardous events. These events, like earthquakes, may produce consistent damage and life losses, thus any effort to reduce their incidence should be promoted.



Figure 8. Energy mix in Italy (data from Italian Energy Balance 2016 [27]).

The decarbonization potentiality of energy retrofitting is undoubtedly relevant, both in Italy and in Europe, as highlighted by studies of the Fraunhofer ISI, which have predicted that energy-upgrade

measures in the household and tertiary sector might reduce the overall final energy demand by 25% in the year 2050 [34–36].

Apart from environmental issues, the inability of households to adequately heat or cool their homes, due to continuously rising energy costs, may also produce significant social consequences. Houses with poor thermal comfort have a strong impact on health and consequently on healthcare expenditure. Hence, the social return on investment from energy retrofitting can be relevant too.

Energy improvements are today mandatory in Italy when major renovations are made. In particular, when refurbishment activities involve more than 25% of the building envelope, the thermal transmittance (U) of the envelope and the efficiency of the heating and cooling system must be considerably improved [37]. If the retrofitting involves the entire building envelope, the current regulation also imposes the installation of renewable energy source (RES) systems [38]. Anyway, it is necessary to respect specific U-values of the envelope components in order to take advantage of the actual fiscal incentives [39].

#### 3.2. Current Energy Retrofit Interventions

Nowadays, there are two main solutions for enhancing the energy performances of buildings [40]:

- to reduce energy consumptions; and
- to promote the energy production on site through Renewable Energy Source (RES) systems.

The first solution is generally addressed by increasing the thermal resistance of the building envelope (e.g., application of insulating layers on walls and roofs, and replacement of the existing windows with high-performing ones), by providing sun shading devices, by improving the air-sealing (in particular for cold climates), by exploiting bioclimatic resources (solar radiation, night ventilation, etc.), by improving the efficiency of Heating, Ventilating and Air Conditioning (HVAC) equipment, and by changing operational schedules.

The second strategy, which is often combined with the former one, is mainly accomplished by installing solar panels (photovoltaic, PV, and/or solar thermal, ST), which today represent one of the most cost-effective solutions for energy production on site, especially in southern and central Europe, where sun-based RES systems turn out to be quite efficient [41–43].

Both strategies may also be optimized and interconnected by a Building Energy Management System (BEMS), i.e., a computer-based control device that supervises and monitors the mechanical and electrical equipment of the building (e.g., HVAC, RES, lighting and power systems), according to comfort requirements and occupancy regimes. BEMS are currently used mostly for commercial and industrial buildings, but they may be effectively installed also in residential ones.

In addition, in this case, specific solutions should be considered for historic or listed edifices, which present valuable facades not suitable for external insulation application or conventional RES system installation [44,45]. These buildings generally need non-invasive retrofit techniques, such as insulating and/or phase-change-material plasters, internal wall insulation [46–51], specific solutions for PV or ST integration, etc.

As in Section 2.2, a detailed analysis of the energy retrofit strategies is here omitted.

#### 4. Barriers to the Seismic and Energy Renovation of Residential Buildings

This section will now consider the technical, financial, bureaucratic, cultural and organizational barriers that might affect combined seismic and energy retrofitting strategies.

The analysis will focus on the Italian residential stock realized from the 1950s to the 1980s, which accounts for around 60% of the current real estate (Figure 4). Historic or listed edifices are here not included, since for them the retrofit interventions are too closely connected to each specific case and the relative costs are hardly predictable and generalizable.

The principal potential obstacles that may affect this kind of strategies are listed and discussed below.

#### 4.1. Technical Feasibility of Renovation Interventions

The principle of sustainability generally leads to prefer renovation activities over demolition and reconstruction practices [52–55], since renovation may keep and reuse many building components (e.g., foundations, structural frame, walls, floor slabs, etc.)—saving resources and reducing waste—and limit urban-sprawl and soil consumption.

However, in particular, seismic renovation can sometimes be technically unfeasible or not recommendable. This applies especially to the edifices in a very poor state of conservation, with weak and carbonated RC and/or affected by significant design or construction errors (e.g., inadequate load-bearing structure, faulty concrete composition and/or compaction). For such buildings, any retrofitting measure may turn out to be ineffective, both from the technical and the economic point of view. In this case, the most sensible solution would be to demolish and reconstruct the edifice.

#### 4.2. Cost of Seismic and Energy Renovation

Costs have usually a key role in undertaking renovation actions. Retrofitting expenditures strongly depend on many variables, such as state of conservation, type of selected intervention, number of stories, total floor area, plan irregularities, presence of adjacent buildings, local seismicity, soil type, local prices for materials and labor, etc.

Recent studies have calculated the costs for the combined energy and seismic retrofit of apartment blocks, which represent one of the most frequent building types constructed in Italian urban areas in the considered period (1950–1990) and, above all, the most inhabited one [33]. With reference to the renovation strategies described in Sections 2.2 and 3.2, this cost currently ranges from 100 to  $230 \text{ €/m}^3$ , i.e., between 30,000 and 70,000 € for a 100-m<sup>2</sup> apartment [56–60].

The main contribution to these renovation costs is due to the seismic component that ranges from about 50 to  $150 \text{ C/m}^3$ . High expenditures, along with a difficult access to capital and an unwillingness to incur debt, often discourage building owners from supporting seismic renovation practices, especially taking into account that it is uncertain when and where telluric shakes will occur. Therefore, owners might tend to believe that earthquakes will spare their families and properties and consequently to repress eventual prevention activities.

Moreover, families with low incomes have often dwellings with poor seismic and energy performance. This circumstance reduces the opportunity of undertaking renovation actions even for the buildings with the highest seismic vulnerability and decarbonization potential.

#### 4.3. Temporary Alternate Accommodation for Occupants

In many cases, renovation activities imply the necessity of emptying and leaving the housing during retrofitting works, which may last for several months. This entails a relevant disruption to the occupants, additional rental costs for an alternate accommodation, a stressful interruption of everyday routines (especially for elderly and disabled people), as well as psychological concerns about the real and timely conclusion of the refurbishment works.

#### 4.4. Insufficient Awareness and Skills

Building owners are often unaware of the real seismic vulnerability and energy performance of their dwellings.

In particular, with specific reference to the seismic vulnerability, as stated before, they might tend to assume that earthquakes are unlikely events, which will occur in a distant future, after their death. Therefore, unless they are driven by the emotional push of a recent devastating seismic event, they are often inclined to neglect the relevant efforts necessary for a seismic renovation. In addition, energy performance issues are also usually ignored by building owners, who often do not monitor their energy consumption and costs, do not fully comprehend the effectiveness of specific retrofitting technologies, and might not be keen in learning about renovation options. Moreover, seismic and energy retrofitting works and related financial or fiscal incentives require a specific knowledge and expertise, while there is a lack of advice agencies, skilled professionals (architects, engineers, auditors) and qualified constructors [61,62].

There is also a strong need of simple and reliable decision-making tools to compare different seismic and energy retrofitting scenarios and select the best option in terms of costs, available incentives and financial aids, improved seismic and energy performance, thermal comfort, increased property value, and reduced disruption to the residents.

#### 4.5. Consensus to Retrofit Expenditure by Condominium Ownership

This barrier may represent the most relevant one, particularly in countries like Italy, where the property of multifamily buildings is largely fractioned. In this case, it will be difficult to find a consensus for expensive and demanding renovation initiatives among all the owners, especially for demolition and reconstruction scenarios, which entail risks and concerns in switching to a new solution. In addition, personal dislikes related to past disputes may further complicate decision making.

This issue becomes even more relevant when buildings involve different occupant typologies, i.e., both owner-occupants and tenants.

Moreover, in the case of a short decision time-frame, owners might neglect the best choice for the building performance and instead opt for solutions that suit their personal situation. For instance, elderly people are often not willing to engage in renovations, and the same may occur to tenants or owners who expect to move soon elsewhere [63].

Even if the approval of renovation works in Italy is legally insured by absolute majority (i.e., >50%, see Table 2), substantial practical difficulties arise every time that one or more owners dissent, especially if they do not have sufficient financial resources. Dissenters may severely delay decision making or even jeopardize technically necessary retrofitting activities.

Table 2. Share of required majorities for decisions on renovations [63].

	Austria	Bulgaria	Czech Republic	Germany	Finland	France	Italy	Romania	Spain
Required majority for decision on renovation	>50% of share, but minority rules	>67% of area	>75% of votes	>75% of shares	>50% of shares	>50% of shares	>50% of shares	>67%	>50% of shares

Far easier is the circumstance of single-owned multifamily buildings, which do not present consensus concerns.

#### 4.6. "Split-Incentive Barrier"

In the case of rented properties, the most relevant issue is represented by the so-called "split-incentive barrier" or "landlord-tenant dilemma": energy costs are paid directly by tenants and landlords are not driven to invest in efficient building systems; conversely, if landlords pay energy expenses (gross leases), tenants will have little incentive to save energy in their leased space [32,63–65]. Moreover, if the building is rented out, also specific seismic renovation interventions will often be neglected, since landlords are not driven to invest money for supporting tenants' safety. In this case, the government, which is responsible for financing healthcare and reconstruction activities due to catastrophic events, is the subject that mostly benefits from preventing and limiting earthquake consequences.

However, in countries such as Italy, Spain, Romania and Bulgaria, where more than 70% of the families live in property homes (Table 3), "split-incentive" issues are less relevant in comparison with countries such as Austria or Germany, whose share of owner-occupancy in multifamily housing is below 25%.

Table 3. Share of owner-occupancy in multifamily housing by nation [62,66].

Austria	Bulgaria	Czech Republic	Germany	Finland	France	Italy	Romania	Spain
23%	90%	79%	24%	50%	26%	72%	96%	86%

#### 4.7. Bureaucratic Obstacles

Italian building owners and professionals also have to face an endemic and structural problem: bureaucratic obstacles. Several months frequently elapse to obtain a building permit, especially for renovating protected or listed edifices. This is mostly due to confusing regulations, which often overcomplicate the procedure, to the fragmentation of competences among many different agencies (responsible for architectural design, structures, listed buildings, etc.), and to the inertia of the offices in charge of releasing permits.

Moreover, the access to fiscal incentives, which will be illustrated in Section 5.1, is for some aspects complicated. For instance, it is now necessary to follow two distinct and parallel procedures for seismic and energy renovation, increasing the likelihood of making mistakes and consequently of missing the incentives.

Over the last years, some simplifications have been adopted to accelerate procedures, but the results are still unsatisfactory.

#### 5. Possible Countermeasures and Discussion

In this paragraph, possible solutions for overcoming the barriers considered in the previous section will be reviewed (in case of existing measures), suggested (in case of novel ones) and discussed.

#### 5.1. Financial and Fiscal Incentives

According to the prices previously indicated, nowadays the combined seismic and energy retrofitting of a standard  $100\text{-m}^2$  apartment in Italy costs on average around  $50,000 \in$  (i.e., about  $165 \notin /\text{m}^3$ ). This is a relevant amount that turns out to be unaffordable for most people; thus, costs often represent the most relevant barrier. Hence, a solution that in the past years has shown satisfactory results in Italy consists in granting fiscal incentives, such as tax credit and VAT reduction. These incentives represent supporting measures, which are usually preferable and more effective than coercive regulations that would impose renovation actions without meeting the real needs and the public acceptance.

In particular, in Italy, since 1998, the government has been offering tax credits, allowing subtracting 36–65% (with a gradual increase over the years) of refurbishment costs (project management included) from the tax due, with deductions equally distributed over 5 or 10 years (according to the beneficiary's age) [67–70]. In 2017, these shares have been consistently enhanced until 2021, especially for apartment blocks: 75–85% for seismic upgrades (according to the reached seismic vulnerability), with deductions distributed over five years [71], and 70–75% for energy upgrades (according to the reached energy performance), with deductions distributed over 10 years [69]. For all these costs, the current regulation also allows reducing VAT from 22% to 10% [72].

Moreover, the Italian government has recently introduced a further useful incentive: if a beneficiary has a low income and consequently his tax credit turns out to be higher than the tax due, he will able to transfer this credit to third parties, such as construction companies or banks. Actually, with the introduction of this incentive, it is advisable that the deductions will always be distributed over 5 years, instead of 10, avoiding to penalize the assignee. Furthermore, it is desirable that all the fiscal incentives will be extended sine die, confirming the current highest shares (70–85%).

As an alternative to the tax credit and only for energy retrofitting, it is possible to benefit from subsidies to produce thermal energy from RES and to increase the energy efficiency [73]. These subsidies cover 40% of the eligible expenditure, with specific limits for the unit and total

costs of each type of intervention. For private buildings, the eligible costs will be refunded in five annual rates.

Taking advantage of these incentives, in particular the tax credit, as well as the reduced energy bill after renovation, the investment for combined seismic and energy retrofitting may be repaid within 10–11 years, as highlighted in previous studies [60]. The investment cost will considerably increase if the installation of RES systems is considered, but according to recent investigations, in mild climates the PBT may reach similar values [74].

A simple PBT of 11 years could be sufficiently attractive from an economic point of view. Anyway, due to the relevant cost of the initial investment, these fiscal incentives alone are often not sufficient to concretely promote both energy and seismic retrofits.

Consequently, public-supported financial measures should also be fostered for low-income people, such as subsidies and/or long-term and low-interest loans. To grant these loans easily, the energy cost savings generated by the energy renovation could also be accepted as a form of collateral [62]. Another financial incentive is represented by feed-in tariffs, which should be further fostered in case of refurbishments that include electricity production from RES.

Nevertheless, it is important to underline that although governments should underpin and stimulate renovation with financial incentives, public funds can only cover a small part of the necessary investment. Therefore, it is necessary to develop sustainable, income-based and attractive schemes to facilitate the engagement of building owners in retrofitting activities, without flooding the market with subsidies.

#### 5.2. Reconstruction Scenarios for Buildings in Poor Conditions

Another significant obstacle concerns the ineffectiveness of any seismic retrofit intervention for buildings in poor conditions. As mentioned in Section 4.1, in such cases the only possible solution would be to demolish and reconstruct the edifice, facing costs that in Italy account for around  $400 \ m^3$  for apartment blocks, i.e., nearly 2.5 times more than those necessary for refurbishment solutions. Hence, high costs and the necessity of a temporary alternate accommodation for occupants represent here the trickiest issue.

Regarding the alternate accommodation, householders without personal opportunities (second homes, accommodation by relatives or friends) could benefit from public vacant buildings, which should be refurbished (if necessary) and cheaply leased for this purpose. Once this cycle has been concluded, these buildings could be allocated to social housing. Of course, specific guarantees should be promoted to safeguard property owners against unsuccessful or delayed delivery of their dwellings. Otherwise, if the occupants do not request to live in the same building, it is possible to move them into a new one specifically built in another plot, possibly made available from the government, while the degraded one will be demolished and reconstructed on site for the same purpose, triggering a virtuous rotating process that leads the progressive renovation of the building stock.

With reference to the high investment costs, a possible solution consists in creating surplus apartments, whose sale may considerably reduce the final expenditure. For example, it may be destined for sale 30% of the built volume that can be obtained as follows: (a) each owner renounces to 10% of the net surface of his property; (b) the government allows increasing the building volume, for instance by 10%; and (c) the story height is decreased by 10% (in Italy, apartments built during 1950–1990 often have a net floor height of 3 m or even higher, which can be reduced to 2.7 m according to the current regulations). Such measures may contribute to enhance the economic appeal of the operation, to reach consensus in case of multi-owner housing, and to encourage an active involvement of general contractors.

An increase of the building volume, namely a volumetric bonus, is currently allowed under specific circumstances by many Italian municipalities in case of demolition and reconstruction activities that address explicit sustainability targets [75]. An interesting further option consists in the possibility to sell the volumetric bonus to third parties, such as building contractors and construction

companies. Moreover, in Italy, the reconstruction of an earthquake-prone building in seismic zone 1 recently benefits also from tax credits, namely 75–85% (according to the reached seismic safety) calculated over a maximum investment of 96,000  $\in$  [76]. Nevertheless, negative consequences may arise from the resulting increase of urban density; therefore, municipalities should provide new areas for complementary urban services (schools, parking lots, green areas, etc.).

#### 5.3. Information and Engagement Campaigns

Information and engagement campaigns are recommended to achieve a behavioral inclination towards more sustainable choices and decisional strategies for energy efficiency and seismic safety. These campaigns should be encouraged by municipalities and involve housing associations and administrators, which require a preliminary support and training for suggesting appropriate retrofit scenarios for the buildings they represent or manage. Schools also play a relevant role in creating safer environment and should be actively involved to obtain an effective and widespread educational and awareness process.

Another useful information and engagement tool consists in promoting free visits of demonstration buildings (e.g., through guided tours or open-door days), which can play as a virtuous model for a seismic and energy renovation.

As an alternative to public actions, contractors could directly include a decision-support package in their service, even if they might not be perceived as impartial and clients could be unwilling to pay for this extra assistance.

It is also crucial to develop simplified and user-friendly decision-support tools for assessing the seismic vulnerability and the energy performance of buildings and for choosing the best alternative, as already specified in Section 4.4. To this purpose, the EU has already promoted several calls for proposals within the Horizon 2020 funding initiative, while, in some countries (e.g., Finland, Germany, France, and Switzerland), financial aids have been specifically offered for supporting energy audits [61].

#### 5.4. Regulatory Instruments

An effective way for promoting building renovation is the imposition of a seismic label to rate the seismic safety of an edifice. This label has been recently adopted in Italy [71] and works likewise the energy one, which has been already imposed by the EPBD to rate the energy performance.

However, thus far, the energy performance certificate for the sale or rental of buildings has still had little effect on the market price of the real estate [77]. To increase the value of the renovated stock, the seismic and energy label should also be supported by a new taxation for real estate transactions, which should be indexed according to the reached performance.

Moreover, governments should promote mandatory insurances to cover damage from natural hazards, with premiums based again on the same label. Considering the recent history, natural hazards are not extraordinary events (see Section 2.1) and they must be insured, like usual accidents. The risks faced by the insurance company in case of earthquakes or other natural hazards can be opportunely alleviated through new security tools, like the catastrophe bonds (namely "cat bonds"), which allow transferring some of these risk from the insurance company to investors [78]. More in detail, an insurance company issues bonds and sell them to investors through an investment bank. If no catastrophe occurs, the company will pay a coupon to the investors. On the contrary, if a catastrophe does occur, then the investors will lose part or all of their principal, which is used to pay the claim-holders.

An additional useful measure is represented by a compulsory establishment of a renovation fund, in order to collect money for future retrofitting activities. In Germany, a renovation fund as high as 1% of the building value has been already activated [62], but this rate is usually too little to cover expensive interventions like seismic upgrades. Fund rates should be determined according to the seismic vulnerability and energy performance of the considered edifice.

Finally, particularly in Italy, a substantial regulatory simplification, both for seismic and energy renovation actions, is necessary to avoid bureaucratic trammels and to accelerate the procedure to obtain the building permit. In the light of this simplification, also the access to fiscal incentives has to be streamlined, developing a unified procedure for combined seismic and energy renovations.

#### 5.5. Consensus to the Retrofit Expenditure for Multi-Owner Housing

As indicated in Section 4.5, for multi-owner housing, it is often difficult to find consensus for renovation works, since many residents are not well informed, do not attend assemblies regularly and have conflicting interests. Moreover, condominium assemblies do not only involve financial, technical and legal issues, but also imply interpersonal and psychological problems.

In such uncertain situations, a possible solution may consist in engaging external parties such as municipal agencies, housing associations, structural and energy consultants—to support and speed up decision making. In fact, condominiums may take advantage of a step-by-step technical and organizational supporting process, which can be moderated by external and impartial professionals.

To reach consensus, a useful contribution can be given by solutions that minimize the disruption to the occupants during the renovation works. This may be achieved trying to operate mainly from the outside of the building, promoting the use of prefabricated components [79], and concentrating the inner interventions in a short period. The works must be organized by proceeding from the foundation level to the top of the edifice, operating progressively floor by floor.

#### 5.6. "Split-Incentive Barrier"

In the case of both renovation and reconstruction, the "split-incentive barrier" plays a significant role, as illustrated in Section 4.6. A possible countermeasure to this problem consists in revising contracts to permit landlords to raise the rent of the retrofitted or rebuilt property, with an increase commensurate with the reduced energy bill paid by tenants and the enhanced seismic performance.

Anyway, apart from a possible contract review, it should be underlined that, due to the renovation activities, landlords will nevertheless benefit from an asset of greater value and able to survive catastrophic events, as well as from the possible exploitation of tax incentives. Moreover, the money saved by tenants on energy costs will leave more money left over for rent, reducing defaulting circumstances. Finally, especially in a competitive rental market, a seismic-safe, low-energy and thermally-comfortable building will have better chances to be well rented or sold.

Consequently, the "split-incentive barrier" might be overcome simply through appropriate and accurate information campaigns, which illustrate all the benefits related to seismic and energy renovation actions.

#### 6. Conclusions

Seismic and energy renovation of buildings represents today a prevention action that is becoming more and more necessary to increase the sustainability level of our towns. It will allow reaching very relevant benefits, at environmental, social and economic levels. In particular, in the case of earthquakes or natural hazards related to climate change, this action will decrease the number of deaths, injuries, and disabilities, as well as the tragic social and psychological consequences for those who will lose their beloved ones; it will also consistently reduce damages and the related economic efforts for repair and reconstruction activities. Furthermore, it will contribute to lower the carbon dioxide emissions, increase the property value, decrease the energy bill, improve the dwelling comfort and healthiness, refresh the architectural image of towns, boost the building market and the employment rate, and consequently increase consumption, gross national product and tax revenue. A positive psychological sensation of life security and welfare may also be taken into account.

Nevertheless, today private owners are often not sufficiently motivated to undertake seismic and energy renovations, mostly due to the significant economic effort, occupants' disruption, and problematic decision-making that this kind of intervention usually implies. Hence, national governments must directly intervene, especially with specific supporting measures, which are generally more preferable than coercive ones.

First, national governments must lead by example upgrading public buildings, adopting state-of-the-practice standards that safeguard the edifices as well as the environment, and providing substantive, performance-based guidelines. Moreover, several further measures have here been suggested, which can be summarized as follows:

- Extend sine die the fiscal incentives (tax credit, VAT reduction, and feed-in tariffs), with the current highest shares (70–85%) and with transferable deductions distributed over five years;
- Foster long-term, low-interest loans and subsidizes for low-income people.
- Grant transferable volumetric bonuses and fiscal incentives, in case of demolition and reconstruction actions, extending these benefits also to other seismic zones.
- Provide permanent or temporary public alternate accommodations in case of demolition and reconstruction.
- Develop easily accessible information about seismic vulnerability and energy performance of buildings, as well as user-friendly decision-support tools to select the best renovation option.
- Index the taxation of the real estate market according to the seismic and energy label.
- Promote mandatory insurances for covering damage from natural hazards, with premiums based on the previous label.
- Establish a reserve fund for renovation.
- Simplify regulations to accelerate the implementation of renovation activities.
- Develop a single procedure to access fiscal incentives for combined seismic and energy upgrades.
- Engage external parties to support and speed up decision making for the refurbishment of multi-owner housing.
- Develop renovation techniques and methods that minimize occupants' disruption.

Prevention is essentially a matter of mindset and culture [80]. Since European countries have a great tradition and culture, the basic premises for developing a prevention attitude and reaching proper economic and technical solutions are all there.

However, wide engagement actions, at both local and European level, are fundamental to raise awareness of the social, environmental and cultural relevance of prevention actions, and to achieve consensus and behavioral change towards decisional strategies for both energy efficiency and seismic safety.

In this context, schools, universities and research institutes play a crucial role, stimulating institutions and political forces to strongly promote and encourage the upgrade of the building stock.

This virtuous circle is possible, as well shown by the movement for the restoration of historic cities that has originated in Europe and afterwards has reached a correct conservative profile, producing brilliant results of urban rebirth, which are clearly evident in Italy as well as in many other countries.

Today, similar results could also be obtained for the renovation of modern buildings, without aggravating the public debt that has now reached unsustainable levels, not only in Italy but also in Europe and in most other developed countries. Hence, it will also be essential that private householders invest personal capital and that the Government act as guarantor, regulating the subject and, as mentioned above, granting substantial financial and fiscal incentives.

Author Contributions: P.L.G. wrote Sections 1, 5.2 and 5.3. G.M. conceived and designed the research and wrote Sections 2–4, 5.1, 5.4–5.6 and 6.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- UN General Assembly. 2005 World Summit Outcome: Resolution Adopted by the General Assembly on 16 September 2005; A/RES/60/1; UN General Assembly: New York, NY, USA, 24 October 2005; pp. 1–38.
- Policy and Background. Available online: http://ec.europa.eu/environment/europeangreencapital/aboutthe-award/policy-guidance/ (accessed on 25 November 2017).
- 3. Available Budget 2014–2020. Available online: http://ec.europa.eu/regional\_policy/en/funding/availablebudget/ (accessed on 25 November 2017).
- 4. European Commission. *Energy-Efficient Buildings PPP: Multi-Annual Roadmap and Longer Term Strategy;* Publications Office of the European Union: Luxemburg, 2010; ISBN 978-92-79-15228-3.
- 5. Balaras, C.A.; Droutsa, K.; Dascalaki, E.; Kontoyiannidis, S. Heating energy consumption and resulting environmental impact of European apartment buildings. *Energy Build*. **2005**, *37*, 429–442. [CrossRef]
- 6. Uihlein, A.; Eder, P. Policy options towards an energy efficient residential building stock in the EU-27. *Energy Build.* **2010**, *42*, 791–798. [CrossRef]
- European Commission. Living Well, within the Limits of Our Planet: General Union Environment Action Programme to 2020; Publications Office of the European Union: Luxemburg, 2014; ISBN 978-92-79-34724-5.
- Institut Cartogràfic de Catalunya. The European-Mediterranean Seismic Hazard Map; Institut Cartogràfic de Catalunya: Barcelona, Spain, 2013; Available online: http://www.preventionweb.net/files/10049\_ 10049ESCSESAMEposterA41.jpg (accessed on 25 November 2017).
- Federico, M.; Zuccaro, G. Seismic and energy retrofitting of residential buildings: A simulation— Based approach. UPLanD 2016, 1, 11–25.
- Belleri, A.; Marini, A. Does seismic risk affect the environmental impact of existing buildings? *Energy Build.* 2016, 110, 149–158. [CrossRef]
- ANCE/Cresme. Primo Rapporto ANCE/CRESME—Lo Stato del Territorio Italiano 2012; ANCE/Cresme: Rome, Italy, 2012; Available online: http://www.camera.it/temiap/temi16/CRESME\_rischiosismico.pdf (accessed on 25 November 2017). (In Italian)
- Italian Civil Protection Agency. Seismic Classification. Available online: http://www.protezionecivile.gov.it/ jcms/en/classificazione.wp;jsessionid=94C4BFB09911106F854EE344B3E87E00.worker1?request\_locale=en (accessed on 3 January 2018).
- Cruciani, C. Law 64/1974. In Provvedimenti per le Costruzioni con Particolari Prescrizioni per le Zone Sismiche; Gazzetta Ufficiale della Repubblica Italiana n. 76 del 2.2.1974: Rome, Italy, 2011. (In Italian)
- Italian National Statistical Institute. Population Housing Census. Available online: http: //dati-censimentopopolazione.istat.it/Index.aspx?lang=en&SubSessionId=f51c0068-9a7d-42c5-abe4-9c3ef54656a0&themetreeid=-200 (accessed on 25 November 2017).
- 15. Italian Civil Protection Agency. Emergencies in Italy. Available online: http://www.protezionecivile. gov.it/jcms/en/emerg\_it\_sismico.wp;jsessionid=A4DC4349AA99D6AFF6143008CDD0AA08.worker2? contentId=RIS12254 (accessed on 3 January 2018).
- 16. Parducci, A. La Sfida Dell'isolamento Sismico; Il Prato Casa Editrice: Saponara, Italy, 2007. (In Italian)
- Dipartimento Protezione Civile. Emergenza Sisma Abruzzo. In *Relazione di Esecuzione delle Spese Sostenute* a Valere sul Contributo del Fondo di Solidarietà Dell'unione Europea (FSUE); Dipartimento Protezione Civile: Rome, Italy, 2009. Available online: http://www.protezionecivile.gov.it/resources/cms/documents/ RelazioneFSUE.pdf (accessed on 25 November 2017). (In Italian)
- Dipartimento Protezione Civile. Terremoto Centro Italia. Available online: http://www.protezionecivile. gov.it/jcms/it/terremoto\_centro\_italia\_2016.wp (accessed on 3 January 18). (In Italian)
- 19. Fardis, M.N. Seismic Design, Assessment and Retrofitting of Concrete Buildings; Springer: Dordrecht, The Netherlands, 2009; ISBN 978-1-4020-9842-0.
- 20. Oliveto, G.; Decanini, L.D. Repair and retrofit of a six storey reinforced concrete building damaged by the earthquake in south-east Sicily on the 13th December 1990. *Soil Dyn. Earthq. Eng.* **1998**, *17*, 57–71. [CrossRef]
- Foraboschi, P. Versatility of steel in correcting construction deficiencies and in seismic retrofitting of RC buildings. J. Build. Eng. 2016, 8, 107–122. [CrossRef]
- 22. Cardone, D.; Flora, A. Direct displacement loss assessment of existing RC buildings pre- and post-seismic retrofitting: A case study. *Soil Dyn. Earthq. Eng.* **2014**, *64*, 38–49. [CrossRef]

- 23. Mazza, F.; Pucci, D. Static vulnerability of an existing r.c. structure and seismic retrofitting by CFRP and base-isolation: A case study. *Soil Dyn. Earthq. Eng.* **2016**, *84*, 1–12. [CrossRef]
- Almeida, A.; Ferreira, R.; Proença, J.M.; Gago, A.S. Seismic retrofit of RC building structures with Buckling Restrained Braces. *Eng. Struct.* 2017, 130, 14–22. [CrossRef]
- Moon, K.H.; Han, S.W.; Lee, C.S. Seismic retrofit design method using friction damping systems for old lowand mid-rise regular reinforced concrete buildings. *Eng. Struct.* 2017, 146, 105–117. [CrossRef]
- Ascione, F.; Ceroni, F.; De Masi, R.F.; de'Rossi, F.; Pecce, M.R. Historical buildings: Multidisciplinary approach to structural/energy diagnosis and performance assessment. *Appl. Energy* 2017, 185. [CrossRef]
- Ministero dello Sviluppo Economico. *Bilancio Energetico Nazional 2016;* Ministero dello Sviluppo Economico: Roma, Italy, 2016. Available online: http://dgsaie.mise.gov.it/dgerm/ben.asp (accessed on 3 January 2018). (In Italian)
- Baek, C.H.; Park, S.H. Changes in renovation policies in the era of sustainability. *Energy Build.* 2012, 47, 485–496. [CrossRef]
- 29. Law 373/1976. In Norme per il Contenimento del Consumo Energetico per usi Termici Negli Edifici; Gazzetta Ufficiale della Repubblica Italiana n. 148 del 7.6.1976: Rome, Italy, 2015. (In Italian)
- Law 10/1991. In Norme per L'attuazione del Piano Energetico Nazionale in Materia di Uso Razionale Dell'energia, di Risparmio Energetico e di Sviluppo Delle Fonti Rinnovabili di Energia; Gazzetta Ufficiale della Repubblica Italiana s.o. n. 13 del 16.1.1991: Rome, Italy, 1991. (In Italian)
- ENEA—Agenzia Nazionale Nuove Tecnologie. Rapporto Annuale Efficienza Energetica 2015; Agenzia Nazionale Nuove Tecnologie: Roma, Italy, 2015; Available online: http://www.enea.it/it/pubblicazioni/ pdf-volumi/raee-2015.pdf (accessed on 25 November 2017). (In Italian)
- Economidou, M.; Laustsen, J.; Ruyssevelt, P.; Staniaszek, D. Europe's Buildings under the Microscope; BPIE: Brussels, Berlgium, 2011; ISBN 9789491143014.
- 33. Buildings Performance Institute Europe (BPIE). Data Hub for the Energy Performance of Buildings. Available online: http://www.buildingsdata.eu/ (accessed on 20 September 2016).
- Eichhammer, W.; Fleiter, T.; Schlomann, B.; Faberi, S.; Fioretto, M.; Piccioni, N.; Lechtenbohmer, S.; Schuring, A.; Resch, G. Study on the Energy Savings Potentials in EU Member States, Candidate Countries and EEA Countries: Final Report 2011; European Commission: Brussels, Belgium, 2009. Available online: https://ec.europa.eu/energy/sites/ener/files/documents/2009\_03\_15\_esd\_efficiency\_ potentials\_final\_report.pdf (accessed on 25 November 2017).
- Wesselink, B.; Harmsen, R.; Eichhammer, W. Energy Savings 2020—How to Triple the Impact of Energy Saving Policies in Europe; Ecofys: Utrecht, The Netherlands; Fraunhofer ISI: Karlsruhe, Germany, 2010.
- Boßmann, T.; Eichhammer, W.; Elsland, R. Policy Report—Contribution of Energy Efficiency Measures to Climate Protection within the European Union until 2050; Federal Ministry for the Environment, Nature Conservation and Nuclear Safety: Berlin, Germany, 2012.
- Inter-Ministerial Decree 26.6.2015. In Adeguamento Linee Guida Nazionali per la Certificazione Energetica Degli Edifici; Allegato 1; Gazzetta Ufficiale della Repubblica Italiana n. 162 del 15.7.2015: Rome, Italy, 2015. (In Italian)
- Legislative Decree 28/2011. In Attuazione della Direttiva 2009/28/CE Sulla Promozione Dell'uso Dell'energia da Fonti Rinnovabili, Recante Modifica e Successiva Abrogazione Delle Direttive 2001/77/CE e 2003/30/CE; Gazzetta Ufficiale della Repubblica Italiana s.o. n. 71 del 28.3.2011: Rome, Italy, 2011. (In Italian)
- Decree 26.1.2010. In Aggiornamento del Decreto 11 Marzo 2008 in Materia di Riqualificazione Energetica Degli Edifici; Gazzetta Ufficiale della Repubblica Italiana n. 35 del 12.2.2010: Rome, Italy, 2010. (In Italian)
- Ma, Z.; Cooper, P.; Daly, D.; Ledo, L. Existing building retrofits: Methodology and state-of-the-art. Energy Build. 2012, 55, 889–902. [CrossRef]
- 41. Baljit, S.S.S.; Chan, H.-Y.; Sopian, K. Review of building integrated applications of photovoltaic and solar thermal systems. *J. Clean. Prod.* 2016, *137*, 677–689. [CrossRef]
- 42. Tripathy, M.; Sadhu, P.K.; Panda, S.K. A critical review on building integrated photovoltaic products and their applications. *Renew. Sustain. Energy Rev.* 2016, *61*, 451–465. [CrossRef]
- Maurer, C.; Cappel, C.; Kuhn, T.E. Progress in building-integrated solar thermal systems. Sol. Energy 2017, 154, 158–186. [CrossRef]
- Webb, A.L. Energy retrofits in historic and traditional buildings: A review of problems and methods. *Renew. Sustain. Energy Rev.* 2017, 77, 748–759. [CrossRef]

- Mazzarella, L. Energy retrofit of historic and existing buildings. The legislative and regulatory point of view. Energy Build. 2015, 95, 23–31. [CrossRef]
- Bianco, L.; Serra, V.; Fantucci, S.; Dutto, M.; Massolino, M. Thermal insulating plaster as a solution for refurbishing historic building envelopes: First experimental results. *Energy Build*. 2015, 95. [CrossRef]
- 47. Zagorskas, J.; Zavadskas, E.K.; Turskis, Z.; Burinskiene, M.; Blumberga, A.; Blumberga, D. Thermal insulation alternatives of historic brick buildings in Baltic Sea Region. *Energy Build*. **2014**, *78*, 35–42. [CrossRef]
- 48. Vereecken, E.; Van Gelder, L.; Janssen, H.; Roels, S. Interior insulation for wall retrofitting—A probabilistic analysis of energy savings and hygrothermal risks. *Energy Build.* **2015**, *89*, 231–244. [CrossRef]
- Walker, R.; Pavía, S. Thermal performance of a selection of insulation materials suitable for historic buildings. Build. Environ. 2015, 94, 155–165. [CrossRef]
- 50. Troi, A.; Bastian, Z. Energy Efficiency Solutions for Historic Buildings: A Handbook; Walter de Gruyter GmbH: Berlin, Germany, 2015.
- Cirami, S.; Evola, G.; Gagliano, A.; Margani, G. Thermal and Economic Analysis of Renovation Strategies for a Historic Building in Mediterranean Area. *Buildings* 2017, 7, 60. [CrossRef]
- Power, A. Housing and sustainability: Demolition or refurbishment? *Proc. ICE Urban Des. Plan.* 2010, 163, 205–216. [CrossRef]
- 53. Ireland, D. New Tricks with Old Bricks: How Reusing Old Buildings Can Cut Carbon Emissions; The Empty Homes Agency: London, UK, 2008.
- 54. Ding, G. Demolish or refurbish—Environmental benefits of housing conservation. *Australas. J. Constr. Econ. Build.* 2013, 13, 18–34. [CrossRef]
- 55. Munarim, U.; Ghisi, E. Environmental feasibility of heritage buildings rehabilitation. *Renew. Sustain. Energy Rev.* **2016**, *58*, 235–249. [CrossRef]
- Kappos, A.J.; Dimitrakopoulos, E.G. Feasibility of pre-earthquake strengthening of buildings based on cost-benefit and life-cycle cost analysis, with the aid of fragility curves. *Nat. Hazards* 2008, 45, 33–54. [CrossRef]
- 57. ANCE Catania. Sicurezza Strutturale per gli Edifici Residenziali in Cemento Armato Costruiti Prima Dell'entrata in Vigore della Normativa Sismica; ANCE: Catania, Italy, 2012; Available online: http: //www.consultaingegnerisicilia.it/download/Studio%20adeguamento%20antisismico.pdf (accessed on 25 November 2017). (In Italian)
- Caponetto, R. Rehabilitation versus demolition and re-building. In *Changing Needs, Adaptive Buildings,* Smart Cities, Proceedings of the 39th International Association for Housing Science (IAHS) World Congress, Milan, Italy, 17–20 September 2013; Ural, O., Pizzi, E., Croce, S., Eds.; PoliScript: Milan, Italy, 2013; Volume 1, pp. 933–940.
- Margani, G. La Riqualificazione Dell'edilizia Residenziale del Secondo Dopoguerra; La "Zona a Mare" di Catania; Aracne: Rome, Italy, 2014; ISBN 978-88-548-7585-2. (In Italian)
- D'Agata, G.; Margani, G.; Pettinato, W. Cost evaluation of seismic and energy retrofit for apartment blocks in southern Italy. In *Demolition or Reconstruction? Proceedings of the Colloqui.AT.e* 2017, *Ancona, Italy, 28–29* September 2017; Bernardini, G., Di Giuseppe, E., Eds.; EdicomEdizioni: Monfalcone, Italy, 2017; pp. 870–881.
- Huber, A.; Mayer, I.; Beillan, V.; Goater, A.; Trotignon, R.; Battaglini, E. Refurbishing residential buildings: A socio-economic analysis of retrofitting projects in five European countries. In Proceedings of the World Sustainable Energy Days, Wels, Austria, 2–4 March 2011; pp. 1–14.
- Matschoss, K.; Heiskanen, E.; Atanasiu, B.; Kranzl, L. Energy renovations of EU multifamily buildings: Do current policies target the real problems? In *Rethink, Renew, Restart, Proceedings of the ECEEE 2013 Summer Study, Stockholm, Sweden, 3–8 June 2013*; Berg Publishers: Stockholm, Sweden, 2013; pp. 1485–1496.
- Heiskanen, E.; Matschoss, K.; Kuusi, H.; Kranzl, L.; Lapillone, B.; Sebi, C.; Mairet, N.; Zahradník, P.; Atanasiu, B.; Zangheri, P.; et al. Working Paper: Literature Review of Key Stakeholders, Users and Investors; European Union: Brussels, Belgium, 2012; Available online: http://www.entranze.eu/files/downloads/ D2\_4/D2\_4\_Complete\_FINAL3.pdf (accessed on 25 November 2017).
- 64. Sorrell, S.; O'Malley, E.; Schleich, J.; Scott, S. *The Economics of Energy Efficiency: Barriers to Cost-Effective Investment;* Edward Elgar: Cheltenham, UK, 2004; ISBN 978-1-84064-889-8.
- Dyson, C. The Split Incentive Barrier: Theory or Practice in the Multifamily Sector? In *The Climate for* Efficiency is Now, Proceedings of the 2010 ACEEE Summer Study on Energy Efficiency in Buildings, Pacific Grove, CA, USA, 15–20 August 2010; Berg Publishers: Stockholm, Sweden, 2010; pp. 64–75.

- Italian National Institute of Statistics (ISTAT). Censimento Delle Abitazioni; Italian National Institute of Statistics: Rome, Italy, 2013. Available online: http://www.istat.it/it/files/2013/12/Nota-diffusione\_delle\_ abitazioni20122013.pdf (accessed on 25 November 2017). (In Italian)
- Italian Ragioneria Generale dello Stato. Law 449/1997. In Misure per la Stabilizzazione della Finanza Pubblica; Gazzetta Ufficiale della Repubblica Italiana n. 302 del 30.12.1997; Istituto poligrafico e Zecca dello Stato: Rome, Italy, 1998. (In Italian)
- Law 296/2006. In Disposizioni per la Formazione del Bilancio annuale e Pluriennale dello Stato (Legge Finanziaria 2007); Gazzetta Ufficiale della Repubblica Italiana n. 299 del 27.12.2006: Rome, Italy, 2006. (In Italian)
- Law 232/2016. In Bilancio di Previsione dello Stato per L'anno Finanziario 2017 e Bilancio Pluriennale per il Triennio 2017–2019; Gazzetta Ufficiale della Repubblica Italiana n. 297 del 21.12.2016: Rome, Italy, 2016. (In Italian)
- Law 208/2015. In Disposizioni per la Formazione del Bilancio Annuale e Pluriennale dello Stato (Legge di Stabilità 2016); Gazzetta Ufficiale della Repubblica Italiana n. 302 del 30.12.2015: Rome, Italy, 2015. (In Italian)
- 71. Italian Ministero delle Infrastrutture e dei Trasporti. Ministerial Decree 58/2017. In Sisma Bonus—Linee Guida per la Classificazione del Rischio Sismico delle Costruzioni Nonché le Modalità per L'attestazione, da Parte di Professionisti Abilitati, Dell'efficacia Degli Interventi Effettuati; Ministero delle Infrastrutture e dei Trasporti: Rome, Italy, 28 February 2017. (In Italian)
- Ministero Dell'economia e delle Finanze. Decree of the President of the Republic 633/1972. In *Istituzione e Disciplina Dell'imposta sul Valore Aggiunto;* Gazzetta Ufficiale della Repubblica Italiana n. 292 del 11.11.1972: Rome, Italy, 1972. (In Italian)
- 73. Ministerial Decree 16.02.2016. In Aggiornamento della Disciplina per L'incentivazione di Interventi di Piccole Dimensioni per L'incremento Dell'efficienza Energetica e per la Produzione di Energia Termica da Fonti Rinnovabili; Gazzetta Ufficiale della Repubblica Italiana No. 51 del 2.3.2016: Rome, Italy, 2016. (In Italian)
- 74. Evola, G.; Margani, G. Renovation of apartment blocks with BIPV: Energy and economic evaluation in temperate climate. *Energy Build*. **2016**, *130*, 794–810. [CrossRef]
- 75. Unified Conference, Provision 1.4.2009. In Intesa, ai Sensi Dell'articolo 8, Comma 6, della Legge 5 Giugno 2003, n. 131, tra Stato, Regioni e gli enti Locali, Sull'atto Concernente Misure per il Rilancio Dell'economia Attraverso L'attività Edilizia; Gazzetta Ufficiale della Repubblica Italiana n. 98 del 29.4.2009: Rome, Italy. (In Italian)
- 76. Law 96/2017. In Conversione in Legge, con Modificazioni, del Decreto-Legge 24 Aprile 2017, n. 50, Recante Disposizioni urgenti in Materia Finanziaria, Iniziative a Favore Degli enti Territoriali, ULTERIORI interventi per le Zone Colpite da Eventi Sismici e Misure per lo Sviluppo; Gazzetta Ufficiale della Repubblica Italiana s.o. n. 31 del 23.6.2017: Rome, Italy, 2017. (In Italian)
- Baek, C.; Park, S. Policy measures to overcome barriers to energy renovation of existing buildings. *Renew. Sustain. Energy Rev.* 2012, 16, 3939–3947. [CrossRef]
- 78. Edesess, M. Catastrophe Bonds: An Important New Financial Instrument; EDHEC-Risk Institute: Nice, France, 2014.
- Oostra, M. Towards user-oriented plug & play facades. Upgrading the energy performance of row houses. In Proceedings of the Advanced Building Skins, Bern, Switzerland, 3–4 November 2015; pp. 437–451.
- Douglas, M.; Wildavsky, A. Risk and Culture: An Essay on the Selection of Technical and Environmental Dangers; University of California Press: Berkeley, CA, USA, 1983.



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



Review



## Seismic and Energy Renovation: A Review of the Code Requirements and Solutions in Italy and Romania

Emil-Sever Georgescu<sup>1</sup>, Mihaela Stela Georgescu<sup>2</sup>, Zina Macri<sup>2</sup>, Edoardo Michele Marino<sup>3</sup>, Giuseppe Margani<sup>3</sup>, Vasile Meita<sup>1</sup>, Radu Pana<sup>1</sup>, Santi Maria Cascone<sup>3</sup>, Horia Petran<sup>1</sup>, Pier Paolo Rossi<sup>3</sup>, Vincenzo Sapienza<sup>3,\*</sup> and Marius Voica<sup>2</sup>

- <sup>1</sup> NIRD URBAN—INCERC, Soseaua Pantelimon, nr. 266, 021652, Sector 2, 021652 Bucharest, Romania; emilsevergeorgescu@gmail.com (E.-S.G.); vasile.meita@gmail.com (V.M.); pana.radu@gmail.com (R.P.); hp@incerc2004.ro (H.P.)
- <sup>2</sup> University of Architecture and Urbanism "Ion Mincu," 010014 Bucharest, Romania; mihaelastela.georgescu@yahoo.com (M.S.G.); zina\_and@yahoo.com (Z.M.); mvoika@yahoo.com (M.V.)
- <sup>3</sup> Department of Civil Engineering and Architecture—University of Catania, via Santa Sofia 64, 95125 Catania, Italy; emarino@dica.unict.it (E.M.M.); margani@unict.it (G.M.); santimariacascone@gmail.com (S.M.C.); prossi@dica.unict.it (P.P.R.)
- \* Correspondence: vsapienza@unict.it; Tel.: +39-0957382531

Received: 14 March 2018; Accepted: 1 May 2018; Published: 14 May 2018



Abstract: Most European cities are characterized by very large areas, often formed by buildings of low quality, from a series of perspectives (architectural, technological, materials, technical systems, etc.). The possibility of renovating them is strategic to improve both the quality of life and to the possibility of economic recovery for building companies. In the last decades, the attention of the scientific community has been addressed to the energy renovation, thanks to the strong activities of the European Community in this field. However, since a relevant part of the EC territory is at risk of earthquake, the possibility to combine both energy and seismic renovation actions may be strategic for many countries. In particular, Italy and Romania are linked by a common social tradition that springs from the Roman Empire. Nowadays, this link is stronger, thanks to common interests in social, cultural and business fields. Therefore, the investigation of possible synergies for seismic and energy renovation strategies may be really interesting for both countries. In this paper, after an overview of regulations and common practices for buildings with reinforced concrete structures, in both states, some key combined renovation interventions will be described and discussed, as well as advantages and perspectives of integrated renovation approaches. The outcomes of this work are to show the way to transform existing energy-consuming and seismic-prone buildings into energy-efficient and seismic-resistant ones.

Keywords: building rehabilitation; energy efficiency; seismic reinforcement

#### 1. Introduction

Most European cities are characterized by large urban areas, built after the II World War and formed by edifices that often show low standards of quality. The renovation of these districts represents a strategic issue to improve the quality of life and to foster the recovery of the building sector.

In the last decades, the attention of the scientific community has been driven mainly to energy retrofitting, thanks to the strong activities of the European Community in this field. However, a significant number of EU regions is earthquake-prone, as it has been unfortunately shown even in recent time in Greece (2008, 6.4 Mw and 2015, 6.5 Mw), in Serbia (2010, 5.3Mw) and in Spain (2011,

5.1 Mw) [1], as well as in Italy and Romania, as described in detail in Section 2. So, the opportunity to combine energy and seismic renovation turns out to be crucial for many countries.

Italy and Romania, whose real estate is highly energy-consuming [2] and seismically vulnerable [3], may play a breakthrough role.

This paper will review both traditional and innovative renovation interventions devoted in Italy and Romania to enhance the seismic and energy performance of recent buildings, that is, erected from the 1950s through the 1980s, which are generally characterized by reinforced concrete (RC) or steel load bearing structures. The goal is to show the way to transform existing energy-consuming and seismic-prone buildings into energy-efficient and seismic-resistant ones.

Historic edifices [4], that is, built before 1950, need specific measures [5–7] and are not considered in this paper.

#### 2. State of Art

#### 2.1. Overview of Technical Regulations in Italy in Relationship with Earthquake Vulnerability

The classification of the Italian territory in areas with different levels of seismic hazard started in 1909, after the devastating earthquake that hit the cities of Messina and Reggio Calabria in 1908. The district of Messina and the whole Calabria region were classified as earthquake prone areas and the explicit consideration of the seismic excitation in the design of buildings located in these areas became mandatory [8]. The effect of the earthquake was simulated by equivalent horizontal forces [9]. The level of seismic resistance to be provided to the structure was set to avoid the collapse on the occurrence strong ground motions but damage of structural members caused by such extreme events was admitted. The classification of new seismic areas generally followed the catastrophic events that struck the country over time; for instance, after the Avezzano earthquake in 1915, a portion of the central area in Italy was classified as earthquake prone [10]. Other important seismic events that anticipated the classification of new seismic areas are those that occurred in Abruzzo and the southern area of Marche (1943), the central area of Calabria (1947), Carnia (1959), Velina valley (1961), Irpinia (1962), Monti Nebrodi (1967), Belice valley (1968), Tuscania (1971), Friuli (1976), the southern area of Calabria, Patti gulf (1978), Valnerina (1979) and Irpinia-Basilicata (1980). Today, the whole country is considered earthquake prone and the expected level of seismic excitation is given based on the geographic location of the site (Figure 1). Unfortunately, the most considerable growth of the building stock in Italy took place in seventies when the application of seismic provisions was not mandatory in most of the country. As a consequence, most of the existing buildings supported by a RC frame, which represents the most common technology in Italy [11], have been designed without considering seismic provisions and are affected by important structural deficiencies. In these buildings, resisting elements are mainly arranged in one direction, which makes the structure weak and flexible in the orthogonal direction. Furthermore, the structure if often low ductile at both global and local level, because it tends to form story collapse mechanism and ductile detailings of structural members are missing. Finally, the situation may be aggravated by the use of low quality or time degradation of materials. The seismic upgrading of these buildings is of paramount importance for the safety of the population and the resilience against earthquakes [11].

The Italian Building code provides specific sections for existing buildings. More in detail, provisions for seismic assessment and retrofitting of existing buildings are reported in Section 8 of the "Norme Tecniche per le Costruzioni" (Technical rules for constructions, NTC08) enforced in 2008 [12] and in the relevant section of the associated Commentary [13]. In addition, Section 11 of the Ordinance n. 3431 [14] emanated in 2005 still applies when not in contrast with NTC08. The Italian code is fully consistent with the European building code, which provides the regulations on existing buildings in Eurocode 8 part 3 (EN 1998-3) [15]. Finally, EN 1998 as implemented by means of the relevant National Annexes [16] is also applicable in Italy. The code provisions for existing buildings in force in Italy (i) regulate the procedure for the identification of the structural system and its geometric/mechanical

features, (ii) define the methods of analysis and their limits of application and (iii) provide instructions for the execution of the safety verifications.



**Figure 1.** Italian seismic hazard map for probability of exceedance of 10% in 50 years, Peak Ground Acceleration (PGA) in g.

#### 2.2. Overview of Technical Regulations in Italy in Relationship with Energy Performance Requirements

In Italy, the firs regulation for the reduction of the energy consumption in buildings was issued in 1976 (Law 373/1976) [17] but it was low-restrictive an often unattended, due to inadequate controls. The first effective code addressing thermal performance criteria was issued only in 1991 (Law 10/1991) [18], when over 80% of the current residential stock had been already built [19]. Consequently, most of the Italian real estate is highly energy-consuming [20].

According with its geographic characteristics, Italy has a large variety of climatic conditions (Figure 2) and so, the approach to the energy renovation problems, is very different, along the peninsula. As a simplification, in the northern regions, where the heating demand is prevailing, it is necessary to highly insulate the building envelope [21], while in the southern ones, where conversely the cooling demand is predominant, it is important to take advantage of natural ventilation, external sun-shading devices and massive walls [22,23].



Figure 2. Italian climatic zones, according to the heating degree-days (HDD).

Today, a recent directive regulates the energy efficiency of new and existing buildings: the Inter-ministerial Decree 26/6/2015 [24].

With particular reference to renovation actions, this directive classifies three levels of interventions:

- a. *relevant renovations of first level*, which involves more than 50% of the building envelope, as well as the upgrade of the heating and/or cooling system;
- b. *relevant renovations of second level*, which involves 25/50% of the building envelope and may involve also the heating and/or cooling system;
- c. *energy requalification*, which involves less than 25% of the building envelope and/or the upgrade of the heating and/or cooling system.

In case of interventions (a), the renovated building must have the same energy performance of a new building. In particular, the energy performance must be higher than that of a theoretical so-called "reference building" that has the same location, orientation, use and geometry (shape, volume, surfaces, etc.) of the real building but is characterized by the following U-values for each envelope component (Tables 1 and 2):

Table 1. Thermal transmittance values  $[W/m^2K]$  for the envelope components of the "reference building," currently in force.

Puilding Components		Cli	matic Z	one	
bunding Components	A, B	С	D	Ε	F
Exterior walls	0.45	0.38	0.34	0.30	0.28
Windows	3.20	2.40	2.00	1.80	1.50
Roofs	0.38	0.36	0.30	0.25	0.23
Ground slab	0.46	0.40	0.32	0.30	0.28

**Table 2.** Thermal transmittance values  $[W/m^2K]$  for the envelope components of the "reference building," in force from 1 January 2019 for public buildings and from 1 January 2021 for all the other buildings.

Building Components		Cli	matic Z	one	
building components	A, B	С	D	Ε	F
Exterior walls	0.43	0.34	0.29	0.26	0.24
Windows	3.00	2.20	1.80	1.40	1.10
Roofs	0.35	0.33	0.26	0.22	0.20
Ground slab	0.44	0.38	0.29	0.26	0.24

For interventions (b) and (c) the energy performance requirements to be verified regard mainly the thermal characteristics of the portion of the building envelope interested by the renovation works and/or the efficiency of the upgraded systems.

In particular, for (c), the g-values of the windows oriented from East to West (passing through South) must be not higher than 0.35, while thermal transmittance of the renovated components of the building envelope must be not higher than the following values (Tables 3 and 4):

Table 3. Maximum values of the thermal transmittance  $\left[W/m^2K\right]$  of the building envelope components, currently in force.

Puilding Components		Cli	matic Z	one	
building Components	A, B	С	D	Ε	F
Exterior walls	0.45	0.40	0.36	0.30	0.28
Windows	3.20	2.40	2.10	1.90	1.70
Roofs	0.34	0.34	0.28	0.26	0.24
Ground slab	0.48	0.42	0.36	0.31	0.30

Puilding Components	Climatic Zone					
building Components	A, B	С	D	Е	F	
Exterior walls	0.40	0.36	0.32	0.28	0.26	
Windows	3.00	2.00	1.80	1.40	1.00	
Roofs	0.32	0.32	0.26	0.24	0.22	
Ground slab	0.42	0.38	0.32	0.29	0.28	

Table 4. Maximum values of the thermal transmittance  $[W/m^2K]$  of the building envelope components, in force from 1 January 2019.

For (b), it is necessary to address the limits set for (c) and, in addition, the transmission heat loss coefficient  $H'_T$  of the renovated portion of the building envelope must be not higher than the following values (Table 5):

Table 5. Maximum values of the transmission heat loss coefficient  $H'_T$  [W/m<sup>2</sup>K].

Climatic Zone						
A, B	С	D	E	F		
0.73	0.70	0.68	0.65	0.62		

Moreover, according to the Legislative Decree 28/2011 [25], if the energy renovation involves the entire envelope (external walls, windows, roofs and ground slab) of buildings with a net floor area over 1000 m<sup>2</sup>, at least 50% of the energy demand for heating, cooling, Domestic Hot Water (DHW) and electricity must be covered by Renewable Energy Sources (RES).

Finally, according to Law 90/2013 [26], from 2019 all new public buildings and from 2021 all new buildings must reach the nearly Zero Energy Building (nZEB) standard. Of course, this applies also to relevant renovations of first level (a).

#### 2.3. Overview of Technical Regulations in Romania in Relationship with Earthquake Vulnerability

The seismic hazard of Romania (Figure 3) is dominated by the Vrancea intermediate depth earthquakes in south-east of the country. This source affects with high intensities ca. 50% of the territory and is felt with quite important damaging effect on very large areas in neighbouring countries, at each strong event. Romania is a country that is periodically subject to such destructive Vrancea earthquakes, as most recently in 1940 and 1977 [27]. Other crustal (shallow) earthquakes can generate locally very high intensities, with a strong tendency of concentration in west and north.

The November 10, 1940 Vrancea earthquake had a magnitude  $M_{G-R} = 7.4$  (converted at present at  $M_w = 7.6-7.7$ ) with an epicentral intensity assessed as 10 MCS. The March 4, 1977 Vrancea earthquake had a magnitude  $M_{G-R} = 7.2$  (converted at present at  $M_w = 7.5$ ) and caused damage to a large area and to a variety of buildings. The Building Research Institute (INCERC) seismic record of 4 March 1977 pointed out, for the first time, the spectral content of long period seismic motions of Vrancea earthquakes, the duration, cycle number and higher values of actual accelerations versus code ones, with important effects of overloading upon flexible structures [27–30]. As a consequence of the INCERC record, the spectral curve (the dynamic coefficient  $\beta_r$ ), from the P13-1970 Code (which became P100-78), as well as the seismic zonation map of Romanian Standard STAS 2923-63 have been radically changed, with increase of base shear forces. Since then, all areas are seismic.

The new seismic design codes P100-1991, revised 1992, introduced a seismic zonation with two maps, one in terms of seismic coefficient (related to PGA) and another in terms of corner period of the design spectrum, with 3 values. The return period of the map was different in function of the source type.

The procedures for the harmonization between national and European regulations in the field of civil engineering started in Romania in the mid-1990s. Prior to its accession to the European Union (on 1 January 2007), Romania has also followed tightly the programme for the adoption of Eurocodes as national standards [31–33]. The Romanian Seismic Design Code P100-1/2006 has prepared the adoption, starting from 2011, of the homologous EN 1998-1, as the Romanian standard SR EN 1998-1, together with its National Annex for Romania and has represented an essential factor in the transition to European norms. The P100-1/2006 Code implements important elements of progress with respect to its previous version, P100-92. The zoning map was set with a return periods of 100 years [31]. The P100-1/2013 Code, introduced a map with 225 years return period. Currently Peak Ground Acceleration (PGA) from 0.10 g to 0.40 g are compulsory by the zonation map (Figure 3) [32]. A zonation map with 475 years return period map (EC 8 level) was not yet endorsed, being considered necessary an interval for stakeholders' adaptation.



**Figure 3. Left**: The seismic zoning map of Romania (PGA) for a mean recurrence interval of 225 years and 20% exceedance probability in 50 years. **Right**: The seismic zoning map of Romania in terms of Tc—corner periods of the response spectrum (Code P100-1/2013, UTCB [32]).

The Romanian code for the seismic assessment of existing buildings, P100-3/2008, includes several notions and concepts from its European homologue, EN 1998-3. However, Code P100-3 preserves a quantitative approach, based on seismic risk indices, as in code, P100-92 (chapters 11 and 12) and is based on a three-tier approach, similar to that of the ASCE standards [33,34]. The relevant chapters of this code [35] are:

- Generalities; Performance requirements and qualifying criteria;
- Seismic assessment of structures and Non-Structural Components (NSC);
- Collecting the information for structural assessment; Levels of Knowledge (KL1, KL 2, KL3);
- Qualitative assessment; Assessment by calculation (Level 1, 2, 3); Assessment of foundations;
- Final assessment and conclusions;
- Annex A—Performance based seismic assessment of existing buildings;
- Annex B—Reinforced concrete structures; Annex C—Steel structures; Annex D—Masonry structures; Annex E—non-structural components (NSC);
- Annex F (informative)—Guide for seismic rehabilitation of existing buildings (for different materials, energy dissipation systems and base isolation).

#### 2.4. Overview of Technical Regulations in Romania in Relationship with Energy Performance Requirements

Technical regulations for the calculation of the thermal protection of the building envelope have been developed since 1961, with standard STAS 6472-61, revised in 1968, 1973, 1975, 1984 (when there is a major change in the insulation requirements of envelope elements—by normative NP-84), 1989.

In 1997, the technical regulation C107-1997 was developed, based on the European and International CEN ISO standards, revised afterwards [36–38]. It introduced the calculation of the thermal resistance values of the envelope elements (1) taking into account the correction due to the effect of thermal bridges (R', which is the inverse thermal transmittance, U-value, of a building element),

evaluated by the linear thermal transmittance  $\psi$  and point thermal transmittance  $\chi$  (Figure 4). R is the thermal resistance of flat, unlimited elements and A is the area of the considered building element.

$$\mathbf{R}' = \frac{1}{\mathbf{U}'} = \frac{1}{\frac{1}{\mathbf{R}} + \frac{\sum \psi \cdot \ell}{\mathbf{A}} + \frac{\sum \mathbf{X}}{\mathbf{A}}} \tag{1}$$

The calculation of a global heat loss coefficient G of buildings has been introduced, whereby the thermal performance of a building envelope can be assessed by imposing a GN norm ( $G \le GN$ ). The heating energy requirement for buildings in Romania can be judged by the global thermal insulation coefficient G and the average G, as follows:

- 1950–1985—1.00 W/m<sup>3</sup>K
- 1986–1997–0.80 W/m<sup>3</sup>K
- 1998–2010–0.55 W/m<sup>3</sup>K



**Figure 4.** Extract of a thermal bridges catalogue, showing the linear thermal transmittances  $\psi$  of a corner with an embedded, uninsulated reinforced concrete (RC) member (column or shear wall) [39].

The climate zonation map in force has 5 zones with design temperatures for Winter (from  $-12 \degree C$  to  $-24 \degree C$ ) (Figure 5 [40]).

According to 2011 Census, the existing buildings stock of Romania has some 5.3 million buildings, including 8.7 million conventional dwellings. Standardized apartment blocks have a share of up to 70% of the existing housing stock in some urban areas.

Existing apartment blocks with large panel structure (over 35% of the total number of blocks until the 1990s) that according to the level of the achieved thermal protection are divided into two categories:

- apartment blocks built according to standard projects until 1985 (approximately 30% of the built stock) mostly with 5 or 9 stories, having an average G-coefficient of about 1 W/(m<sup>3</sup>K)/average thermal resistances of only 0.6–0.5 m<sup>2</sup>K/W, which have to be thermally insulated as first measure in energy renovation;
- apartment blocks with 5 stories and 9 stories erected after 1985 according to standard projects (about 7% of the total built stock) based on the provisions of Decree 256 and NP 15 Normative with a medium thermal resistance increased to about 0.9 m<sup>2</sup>K/W, characterized by an average global thermal insulation coefficient G of about 0.8 W/m<sup>3</sup>K.



Figure 5. Climatic zonation map of Romania—Winter design temperatures [40].

Structures made of masonry predominate numerically in the dwelling buildings; even in Bucharest. Meanwhile, their number has increased throughout the country.

The level of thermal protection of buildings from the existing building stock corresponds, independently of the structural system used, to the specifications and exigencies imposed during each period, by the technical regulations for the calculation of the thermal performance of the envelope elements. According to each generation of the technical regulations, as well as to the technological level specific to the period, there are groups of buildings with the same level of thermal protection, regardless of the materials used to build the building envelope.

The share of energy consumption in the annual energy balance of an average apartment built between 1970 and 1985 is: heating energy 55.5%; DHW 9.5%; drinking water 1.4%; consumption of natural gas for the preparation of food 9.7%; electricity consumption for lighting 13.9%. Out of the annual energy consumption of a building irrespective of its destination, the heating energy and DHW production represents the main annual energy consumption of about 75% [40].

The implementation of the European Parliament's Energy Performance of Buildings Directive (EPBD 2002/29/EC, EPBD 2010/31/EC) is also being carried out in Romania in compliance with the provisions of Law no. 372/2005 modified and completed later. A calculation methodology for energy performance of buildings (including the energy certification of buildings and energy audit) was initially based on several norms (2001), later on incorporated in a comprehensive one 2006 [41] and is currently under review. Some other technical guidelines for buildings thermal rehabilitation have been developed, as well as a Catalogue of thermal bridges [39].

For all new buildings it is already mandatory, since 2007, to present the energy performance certificate at the reception of the executed work. For existing buildings that are being rehabilitated, expanded or upgraded, such a certificate must be drawn up. For apartments in residential buildings, the energy performance certificate of the building is mandatory for sale-purchase or lease, from 2010. For public buildings, the energy performance certificate must be exposed at the entrance.

The minimum thermal resistance— $R'_{min}$  and thermal transmissions— $U'_{max}$  of the building envelope elements (taking into account the effect of thermal bridges), on the whole of the dwelling

buildings, designed on the basis of the design contracts concluded after 1 January 2011 [42] is presented in Table 6.

**Table 6.** The minimum thermal resistance— $R'_{min}$  and thermal transmissions— $U'_{max}$  of the building elements, on the whole of the dwelling buildings, designed on the basis of the design contracts concluded after 1 January 2011 [42].

		<b>Residential Buildings</b>		
Nr. Crt.	Building Components	R' <sub>min</sub> [m <sup>2</sup> K/W]	U' <sub>max</sub> [W/m <sup>2</sup> K]	
1	Exterior walls (excluding glazed surfaces, including adjoining walls of open joints)	1.80	0.56	
2	Windows	0.77	1.30	
3	Top slabs above the last level, under terraces or attics	5.00	0.20	
4	Bottom slab over unheated basements and cellars	2.90	0.35	
5	Walls adjacent to closed joints	1.10	0.90	
6	Slabs that delimit the building at the bottom, from the outside (in the bow-windows, passage gangs, etc.)	4.50	0.22	
7	Slabs on the ground (over ground level)	4.50	0.22	
8	Slabs at the bottom of heated semi-basement or basements (under ground level)	4.80	0.21	
9	External walls, under ground level, of heated semi-basement or basements (under ground level)	2.90	0.35	

The Energy Performance of Buildings Directive, issued in 2002 and revised in 2010 (EPBD 2010/31/EC) and the European Directive on the use of RES (RESD 2009/28/EC), were the basis for the drafting of country strategies and government policies, transposed into national laws. The National Plan [43,44] includes the long-term energy efficiency strategy at national level, based on existing legal framework and programs establishing the contribution of the state, the local administration and the owners and specifies the necessary documentation and eligible type of actions. The definition of nZEB in Romania was detailed by the MDRAP Order 386/2016 by officially specifying the performance levels in terms of the maximum admissible level of primary energy from fossil sources and of CO<sub>2</sub> emissions resulting from the operation of buildings-by building types and winter climate zones in Romania (Figure 5). The levels will be applied mandatory for all new buildings starting from 2021. The maximum allowed value of the primary energy use (thermal and electric energy supply processes), determined by cost optimal calculations based on reference buildings vary according to the winter climate zoning of Romania (values between brackets correspond to the average climate zone for Romania): 98 to 217 (111) kWh/m<sup>2</sup>yr for single-family residential buildings, 93 to 135 (100) kWh/m<sup>2</sup>yr for multi-family apartment buildings, 45 to 89 (57) kWh/m<sup>2</sup>yr for office buildings. In order to ensure the total energy use of a nZEB, energy from RES shall account for at least 10% of the total calculated primary energy of the building.

As intermediary performance values until the enforcement of nZEB obligations for new buildings, the MDRAPFE Order No. 2641/2017 recently established (for the first time in Romania) minimum requirements in terms of maximum primary energy consumption for heating (only). The limits (max primary energy for heating in kWh/m<sup>2</sup>yr) vary from 60 for office buildings to 117 (large residential), 149 (hospital) and 153 (small residential). This time, the requirements are applied both for newly built and existing buildings undergoing major renovation.

#### 3. Seismic Resistance Assessment and Structural Strengthening

#### 3.1. Italian Relevant Seismic Codes and Expertise on Seismic Strengthening of the Existing Building Stock

The identification of the structural system is of paramount importance for a proper assessment of the structure and for the design of retrofit interventions. In fact, usually designers do not know the mechanical features of the materials, size and arrangement of the structural members and quality of structural detailing. Hence, information on these aspects have to be collected from the available documentation, analysis of the codes in force at the time of construction, field investigations and in-situ and/or laboratory measurements. Based on the completeness and reliability of the collected
information, the structure is assigned to one of three "knowledge levels": limited knowledge (KL1), normal knowledge (KL2) and full knowledge (KL3). The knowledge level of the structure determines method of analysis and numerical model that can be used for the evaluation of the structural response and the values to be adopted for the confidence factors [45] in the safety verifications.

Linear or nonlinear methods of analysis and modelling can be used for the evaluation of the seismic response of the structure. The choice depends on the achieved knowledge level of the structure. Linear structural analysis model (Linear numerical Model analysed by Lateral Force Analysis or Multi-modal Response Spectrum Analysis [46,47]) must be used for knowledge level KL1, while nonlinear structural analysis model (Nonlinear numerical Model analysed by Nonlinear Static Analysis [48,49] or Nonlinear Time-History Analysis [50–52]) may be used if the knowledge level achieved by means of the preliminarily inspection of the structure is KL2 or KL3. Since the collapse mechanism of an existing structure is not known a priori, any member may yield and there is no distinction between dissipative and non-dissipative members. Hence, nonlinear models and methods of analysis, which are able to detect the collapse mechanism and predict the inelastic demand of the members explicitly, should be preferred to the linear ones when applicable.

Criteria for safety verification are given separately for RC, steel, composite and masonry structures. Furthermore, for each type of structure, the Italian code provides two sets of criteria: the first set applies if the seismic response to be assessed has been determined by linear methods of analysis, whereas the second one is used in combination with nonlinear methods of analysis. Criteria for linear methods of analysis are rather conservative to compensate for the low level of accuracy of these methods. Furthermore, the conservativism of the criteria for safety verification is also controlled by the values of the confidence factors, which modify the strength of the members to be considered in the verification. A lower knowledge level means a higher degree of uncertainty of the features of the structure, which is compensated by the use of larger values of confidence factors.

If the seismic response of the structure does not fulfil the standard required by the code, the gap should be filled by seismic upgrading [53]. Increasing the safety level of the building against collapse is the primary goal of retrofit interventions. However, a reduced damage of structural and non-structural elements in the occurrence of moderate earthquakes and a limited disturbance to the occupants of the building during the realization of the retrofit interventions should be considered as further important goals in the choice and application of the retrofit technique. Many strategies may be embraced to pursue these objectives. The most classical way is to increase the strength and/or ductility capacity of the structure at global or local level. For instance, the (global) lateral strength of the structure can be increased by adding new seismic-resistant elements, for example, RC shear walls or steel braced frames. The new resisting elements should be properly connected to the existing structure and stiff enough to draw part of the seismic force. Alternatively, the strength can be increased locally by interventions on individual structural elements; for example, concrete or steel jacketing [54] as well as Fibre Reinforced Polymer (FRP) plating and wrapping [55] can be used to improve the flexural and/or shear strength of RC members. Furthermore, these interventions also allow the enhancement of the ductility capacity of RC members through the confinement of concrete.

An alternative retrofitting solution is to reduce the seismic demand of the structure by base isolation [56–59]. First, columns are cut off from foundation; then isolators are installed between column bases and foundation. These measures elongate the fundamental period of the structure and thus drastically reduce the seismic force on the structure. This strategy is very effective but cannot be used when the building to be upgraded is contiguous with other buildings. Furthermore, its effectiveness decreases with the aspect ratio. Seismic demand can also be reduced by using dampers [60]. Many kinds of dampers (hysteretic, viscous, viscoelastic, etc.) [61–65] are available on the market and can be embedded in the structure [66–68] or inserted between the structure to be retrofitted and an external reaction structure [69]. Dampers dissipate part of the input energy provided by the earthquake, thereby reducing the displacement demand of the structure. As an example, Figures 6 and 7 show two retrofit interventions on schools in Italy. In particular, Figure 6

shows the retrofit intervention carried out at the school Cappuccini located in Ramacca at about 50 km from Catania by means of Buckling Restrained Braces (BRBs). The school was condemned owing to a severe crack pattern present in the non-structural elements after the earthquake that stroke the eastern part of Sicily in 2002. The building was built in the 1970s and endowed with RC moment resisting frames designed for gravity loads only. The BRBs were inserted in the early 2000s to reduce the seismic action effects on the existing structure and dissipate part of the input seismic energy. The BRBs were inserted within some frames located in the central part of the building [70]. Figure 7 instead shows the retrofit intervention carried out at the school Gentile-Fermi located in Fabriano at about 70 km from Ancona by means of viscoelastic dampers [71]. The building was built in the 1950s and endowed with RC moment resisting frames designed for gravity loads only. The school was condemned after an earthquake occurred in 1997. Like the previous structure, that of the school Gentile-Fermi had insufficient resistance to seismic actions and insufficient local and global ductility. Viscoelastic dampers were chosen to dissipate part of the seismic input energy and limit the increase of the lateral stiffness consequent to the introduction of braces. To reduce the interruption of the functionality of the construction, the viscoelastic dampers were inserted within some frames located on the perimeter of the building.



Figure 6. Retrofit intervention by means of buckling restrained braces (BRBs) in the school Cappuccini in Ramacca (Catania).



**Figure 7.** Retrofit intervention by means of viscoelastic dampers in the school Gentile-Fermi in Fabriano (Ancona) [71].

The retrofit intervention is selected based on the analysis of the structural deficiencies. Often, two or more techniques are combined together to achieve the seismic upgrading of the building [72–79]. The great variety of techniques and the need to combine them together represent a further source of difficulty with respect to the case of new structures, for which the designer needs to select just one structural type among a limited number. As an example of a retrofit intervention in which more techniques are combined, Figure 8 shows the retrofit intervention carried out at the school Varano located in Camerino at about 30 km from Macerata by means of rigid rocking towers and viscous

dampers patented by Alessandro Balducci [80]. The school was built in the 1960s and consists of two separate constructions. Both are endowed of RC framed structures. The retrofit intervention was completed in 2012 (in about 7 months) without interruption of the school activities. The intervention aimed at dissipating a large amount of energy by means of the viscous dampers at the base of the towers and leave the RC structural elements to be elastic under moderate intensity earthquakes. The two steel braced towers were connected to the decks by means of steel trusses.



**Figure 8.** Retrofit intervention by means of dissipative towers and viscous dampers in Camerino (Macerata) [80].

The regulations given in the code are mostly for traditional retrofit interventions, while innovative techniques are not covered by code provisions. This is a gap in the code that will be hopefully filled in the future. Research devoted to standardizing the method of application of these techniques for the seismic upgrading of structures and to formulate design methods for retrofitting interventions able to achieve the performance objectives stipulated in the Italian code are of paramount importance. This research activity could create the background for the new generation of building codes and make innovative technique immediately available. In fact, the Italian building code has been developed in a performance-based framework. The performance objectives are defined through three limit states, namely Near Collapse (NC), Significant Damage (SD) and Damage Limitation (DL). Details on the seismic excitation level that should be considered for the verification of each limit state and acceptance criteria may be found in the relevant part of the Italian building code [13,14] and EN 1998 Part 3 [15]. This allows the use of techniques and design methods that ensure the minimum level of structural safety required by the code, even if these techniques are not explicitly considered in the code.

## 3.2. Romanian Relevant Seismic Codes and Expertise on Seismic Strengthening of the Existing Building Stock

The experience of 1977 Vrancea earthquake is relevant for the situation of existing built stock in terms of features, vulnerability and risk. The most heavily affected category in 1977 was that of the old tall buildings. This situation may be understood mainly in connection with some features of their structural design, which was without or with reduced seismic requirements. Some urban and architectural planning patterns played a role (irregular shape in plane and on vertical). The damage of 1940 earthquake were not repaired, while from 1940 to 1977 the buildings maintenance was neglected. The old, relatively low and stiff, bearing masonry buildings (Figure 9) have shown, as a rule, a better performance, especially in Bucharest and collapses were noticed in isolated cases. The spectral specific of strong motion was an important factor of overloading or reduced loading [27,28].



Figure 9. Low-rise and mid-rise masonry buildings, after some 80–100 years.

The new apartment buildings, built after 1950, present a wide diversity of architectural planning and of structural solutions, with many solutions used in standardized design for low-rise buildings (up to 5 stories) and high-rise buildings (8 to 18 stories, the most frequent being that of 10–11 stories).

Before 1990, various construction technologies and structural systems, included prefabrication and industrialized forms for cast-in-place concrete. The large panel standardized apartment blocks represented an increasingly important share of the new construction. The performance of these buildings was good or fair in almost all zones for five-story as well as for eight- or nine-story buildings. It is important to say that most of IPCT solutions were tested in INCERC.

The cast-in-place, RC shear wall buildings that present the greatest share among the structural solutions in seismic zones, especially for high-rise buildings, have shown various behaviour patterns, depending on their overall number of stories, on the quality of workmanship and structural solution, as well as on the intensity of shaking. Under such circumstances, five-story buildings have shown a good or fair behaviour, independent of the structural solution adopted (smaller or large intervals between shear walls).

The new RC framed structures, with five or eleven to twelve stories, for which a regular pattern of columns and beams has been provided, have shown generally a much better performance than old buildings (pre-1940) with RC framed structures, for which the place of columns has been an irregular one.

After 4 March 1977, the specialists of INCERC, as well as the great number of civil engineers existing in 1977 in numerous design institutes provided immediate technical guidelines for repair and strengthening. From structural point of view, the most damaged were pre-1940 structures with columns and beams but not moment resisting frames and all other structures designed after 1940's with some seismic code but at more reduced base shear force, as it was the knowledge for each period. But on 30 March 1977 the Government of that time ordered that the existing structures shall be maintained or rehabilitated, nominally, at the initial strength level. On 4 July 1977, it was ordered to make only local repairs. Thus, political orders left Bucharest with a large stock of high-occupancy, high-rise residential buildings that have been damaged by the 1977 (and possibly the 1940) Vrancea earthquake. Most of these buildings, still in use, are at significant risk from future Vrancea earthquakes [30].

The new seismic design codes P100-1991, revised 1992 and 1997, introduced in chapters 11 and 12 the obligation to evaluate buildings and indicate classes of risk, as it is also in Code P100-3/2008, in force, and, if required, to rehabilitate the existing buildings, with some public financing. Buildings of the first class of risk are labelled with a red dot.

By Law, that is, Government Ordinance on Existing Buildings Risk Reduction [81], evaluation of residential buildings resistance (in terms of Code P100-92, later on in Code P100-3/2008) was provided for free, while for design and strengthening works the owner may receive a bank credit at 5% interest up to 20 years, paid by Government; the apartment owners in buildings of first class of risk, with an income under the country average, may receive full subsidies.

For high-rise buildings erected before 1940 (Figure 10), conceived without seismic design, the strengthening means a rather general jacketing of existing frames, from foundations to the top, to become moment-resisting ones, ductile and able to withstand lateral forces. This implies local or general evacuation and relocation of occupants for 1 to 2 years. For other 1950–1977 (Figure 11) structural types, designed for lower seismic loads, interventions may be on some specific members or zones, or on several stories. In such cases, some advanced techniques, as frame bracing could be applied, with less need of evacuation.



**Figure 10.** High-rise apartment building, with RC columns and beams, infilled with masonry, erected before 1940. **Left side**: a building without visible strengthening or renovation. **Right side**: with local repair, that is, limited and inadequate jacketing of columns, after 4 March 1977 earthquake.



Figure 11. High-rise apartment building erected in the 1970s, having commercial spaces at ground-floor, with visible damages from 4 March 1977 earthquake, before renovation.

#### 4. Current Technologies for Energy Efficiency

#### 4.1. Current Technologies for Energy Efficiency

The current retrofit technologies used in Italy for enhancing the energy efficiency of buildings can be categorized into three main categories:

- (i) heating and cooling demand reduction;
- (ii) upgrade of the Heating, Ventilating and Air Conditioning (HVAC) equipment;
- (iii) installation of RES technologies.

To category (i) belong all the interventions that involve the building fabric: envelope insulation (roof, wall, ground floor) [82], windows retrofits (thermal-break frames, multiple glazing, inert gas filling, low-emission coatings, external sun-shading systems, etc.), cool roofs and coating and increased air tightness. To this group one may also add bioclimatic technologies for exploiting natural resources: solar radiation, natural ventilation, evaporative cooling, etc.

Category (ii) includes mostly the improvement of the heating and cooling systems, for example, by installing high efficiency heat pumps, biomass boilers, geothermal power systems, etc.

Finally, category (iii) embraces the integration of different RES systems, such as solar thermal (ST) collectors, PhotoVoltaic (PV) or hybrid PhotoVoltaic and Thermal (PV/T) panels, wind power micro-turbines, etc. [83–86].

In case of relevant renovations of first level (see (a) in Section 2.2.), in Italy, from 2019 (for public buildings) or 2021 (for all other buildings), it will be mandatory to reach the nZEB standard. In this case further strategies should be necessary in some cases, such as the integration into the building envelope of Phase Change Materials (PCM), the use of low conductivity insulation materials (e.g., aerogel, vacuum insulated panels), as well as the use of Building Energy Management Systems (BEMS) [87–90], that is, computer-based control devices that supervise and monitor the mechanical and electrical equipment (e.g., HVAC, RES, household appliances, lighting and power systems) [91,92], according to comfort requirements, occupancy regimes, energy demand and current electricity price.

For each renovation intervention, the implementation cost and the potential benefits are different. Figure 12 shows the intervention costs versus environmental benefits (reduction of  $CO_2$  emission). It can be observed, that the installation of RES systems may require high investment costs and low environmental benefits. This is mostly due to the high environmental impact of the disposal of some RES systems (e.g., PV, PV/T) [93].



Figure 12. Intervention cost versus environmental benefits.

In Romania, the insufficient thermal insulation of the existing buildings compared to the requirements stipulated in the current thermal regulations is added to the limited seismic resistance of some structural types, as well as a state of degradation of the built part or an advanced wear of the installations. Therefore, many of the buildings requiring energy rehabilitation also need the assessment of the structural resistance and of strengthening scenarios.

The implementation of the Energy Performance Building Directive (2002/91/CE followed by the recast version 2010/31/UE) in Romania, has determined an upgrading of the minimum standards of energy performance of the buildings. It was introduced a methodology for energy balance calculation and an energy performance certificate, both for new and for existing buildings at their rehabilitation moment (Figure 13). Special norms for the inspection of the heating, cooling and ventilation devices

were elaborated. Several successive National Programs for the rehabilitation/modernization of the multi-story residential buildings were applied in the last decades. Currently, the attention is directed to public buildings rehabilitation and certification.



Figure 13. Content of the Romanian Building Energy Performance Certificate [42].

As for the building envelope (opaque components and fenestration), till now, it was applied additional thermal insulation layers using polystyrene or mineral wool plates with thermal conductivity ( $\lambda$ ) between 0.030 and 0.045 W/mK (on external walls, roofs, ground slab), in many apartment blocks built during 1950–1990. Maximal values of the thermal transmittance U, for external walls, were U' = 0.70 W/m<sup>2</sup>K till 2010 and then U' = 0.55 W/m<sup>2</sup>K, taking into account the effect of thermal bridges for each envelope element.

The old types of windows (with wood frames and double glass without any coating) were replaced, in retrofitting programs, with high efficiency double glazing windows with low emissivity coatings, tinted and/or gas filled, having especially PVC frames (U =  $2 \text{ W/m}^2\text{K}$  till 2010 and since 2011 U =  $1.3 \text{ W/m}^2\text{K}$ ). In the last 2–3 windows with triple glazing and low emissivity coatings are provided, having U =  $1.1 \text{ W/m}^2\text{K}$ . In general, thermal bridges are taken into account in the calculations and in designing the renovation measures. However, sensitive connections remain with high values of the thermal bridge coefficient, in particular remaining penetration of RC elements in the insulation layer at basement or building foundation and installation of insulating windows in the structural layer (rather than in line with the insulation layer).

With the implementation of the first projects under the National Rehabilitation Program of the Housing Blocks—Condominiums, coordinated and funded by MDRAPFE and the Local Councils of City Halls, a number of new issues emerged.

The choice of technical and architectural solutions to improve the thermal performance of vertical envelopes must be done, in the future, using the optimal cost method and taking into account and controlling all important aspects.

The new approaches, which can be effectively adopted both in Italy and in Romania, are indicated in the following list:

- Ventilated facades (opaque ventilated façades, double skin glass façades, hybrid façades—wall/glass) were provided only for special buildings (offices, hospitals, public buildings, etc.) due to their initial high cost. The curtain walls using special glass were used for office buildings.
- Green walls and green roofs were used sometimes. Solar shading devices—external or internal, were provided without a detailed analysis. Passive solar energy systems like solar greenhouse were studied and provided in some cases.
- Active solar energy systems as PV panels, ST collectors and mixed systems were provided on some demonstration buildings but Building Integrated Photovoltaics (BIPV) in building envelopes are not usual, although used in pilot buildings.

Acoustic performance for the external thermal insulating systems is under study.

For roofs (flat or sloped), technologies to improve ventilation (single or double ventilation layers for sloped roofs), passive cooling (cool roofs), thermal inertia and waterproof (green roofs or cool roofs), RES use, etc., must be provided.

In Figures 14 and 15 some of the main solutions for thermal energy renovation are presented. The legend is common for the 3 figures and the relative numbers indicate: 1—external wall, possibly plastered; 2—efficient thermal insulation layer (EPS, mineral wool, etc.); 3—insulation protection layer (mineral additives mortar, with fibre glass reinforcement); 4—thin external finishing; 5—protective layer against wind; 6—ventilated layer; 7—external cladding with closed joints; 8—damp-proof course; 9—interior gypsum boards/dry plastering.



**Figure 14.** Main solutions for thermal energy renovation of external walls in current field, with external insulation. **Left**: External Thermal Insulation Composite System (ETICS) composite compact structure with protection layer of thin rendering with fiberglass reinforcement. **Right**: ventilated layer structure with protection layer and external cladding fixed trough a metallic frame [39].



**Figure 15.** Main solutions for thermal energy renovation of external walls in current field, with internal insulation and finishing of gypsum boards [39].

For the future buildings (nZEB), the most promising and interesting insulating materials are: Aerogel, PCM, Transparent Insulation Materials (TIM), Vacuum Insulating Panels (VIP), organic materials (cork, sheep-wool, etc.), high efficiency windows.

In Romania, the nZEB concept does not seem to be easily applicable yet, in particular in the case of the renovation of existing buildings [94]. Besides the required investments and optimal integration of the technologies suitable for the construction and/or renovation of buildings at nZEB level, one of the main barriers for this consists in the skills gaps experienced by the building sector.

Some projects [95–98] approach this barrier by developing a roadmap for construction workforce qualification to achieve the sustainable energy policy objectives set for Romania for 2020. Thanks to them, mechanisms to supporting the national implementation of large-scale and long-term qualification schemes for thermal insulation systems and high thermal performances windows installers are defined (Figure 16).

Aspects such as the arrangement of the ventilation of the spaces in order to obtain adequate indoor air quality, the repair of the sidewalks, the removal of moisture and mould, the rehabilitation of the waterproofing, the modernization of the balconies and loggias were subsequently explicitly included in the current legislation. The arrangement of extensions and mansard spaces is of interest to a number of investors. The approach aims to provide RES, in order to achieve the cost-optimal energy performance of a building or, rather, of a district of buildings and also the highest indoor comfort by a good thermal insulation, restricting the heating load, the use of air conditioning units and artificial lights.



**Figure 16.** Practical training facility for nearly Zero Energy Building (nZEB) in the Building Knowledge Hub (Bucharest).

# 4.2. Common Practice and Current Projects of Energy Renovation in Romania

The compound Prietenia of Sfantu Gheorghe City, consisting of 18 apartment buildings of five stories made of large panels, with 410 apartments, was under energy renovation from 2008 to 2009 (Figures 17–20). The project was done with the consultancy of UAUIM and IPCT Instalatii, in an integrated approach, to address also the architectural and landscape image as local urban issues.

The Prietenia Project allowed an energy reduction of ca. 60% and for this goal the following solutions and technologies have been applied:

- for terrace—EPS 120 of 16 cm thickness;
- for external walls—EPS 80 of 10 cm thickness (XPS 8 cm at socle);
- for slab over basement—EPS 70 of 8 cm thickness;
- for internal walls and slabs of entrance hall EPS 80 of 10 cm thickness;
- thermostatic valves and mixing water faucets, with reduced consumption.



Figure 17. The apartment buildings before renovation, Prietenia compound, Sfantu Gheorghe.



Figure 18. The apartment buildings under renovation, Prietenia compound, Sfantu Gheorghe.



**Figure 19.** The apartment buildings under renovation, Prietenia compound, Sfantu Gheorghe. Community members were able to choose the colours of renovated facades.



**Figure 20.** The apartment buildings and adjacent spaces, including playing grounds for children, at the end of renovation, Prietenia compound, Sfantu Gheorghe.

The mass renovation programs in Bucharest (Figures 21–24) and other large cities were quite successful and popular in the view of communities, especially because and when the cost was fully supported by local authorities or from European funds. However, from the point of view of architectural and technical quality the high speed of works and the lack of workers training caused some critics.



**Figure 21.** Large high-rise apartment buildings in Bucharest, erected in the 1970s, before and after energy renovation.

It is worth to mention that the shear wall structural types of 1970s, although designed to forces lower than current code, behaved satisfactorily in 1977 earthquake and thus they were included in energy renovation projects without other structural upgrading.



Figure 22. Large high-rise apartment building in Bucharest, erected in the 1970s, during energy renovation.



Figure 23. Large high-rise apartment building in Bucharest, erected in the 1970s, after energy renovation.



**Figure 24.** The UTCB Lacul Tei Students Dormitory building after energy renovation (2007). The first public building with displayed Energy Performance Certificate.

The IR images presented in Figure 25 illustrate the qualitative impact of applying external insulation to the Students Dormitory building, while the previously replaced windows (double glazing aluminium profile, moderate thermal performance) were not improved in the energy renovation.



Figure 25. The UTCB Lacul Tei Students Dormitory building. IR thermography images before and after energy renovation (2007).

Based on the JRC study, Romania has one of the 10 exemplary Renovation Strategies (as required art. 4 of the Directive 2012/27/EU), ranked the 3rd at EU level, mainly for having detailed estimations, 4 scenarios, multiple benefits assessed, cost benefit analysis. The biggest source of funding for energy efficiency comes from EU Cohesion Policy Funds, while between 2014 and 2020, Romania intends to allocate more than €1.25 billion to building renovation of residential and public building. The majority of the funds will be used as grants for the building owner/association, covering (some of) the cost of the thermal renovations. However, deep renovations rate remains among the lowest in Europe, while the energy renovation of individual houses is not effectively supported by public funding nor stimulated by actual incentives. Moreover, the energy performance level of current deep renovation is far from the nZEB levels both in terms of design specifications and performance of works (limited insulation, attention to thermal bridges, no mandatory airtightness levels or mechanical ventilation with heat recovery, etc.). Nevertheless, minimum requirements in terms of maximum primary energy use for space heating are in force for major renovation of existing buildings since 2017. One could envisage that this is a first step towards building renovation at nZEB levels in Romania, still regulation has to be improved, nZEB technology market growth has to be stimulated and private investments have to be attracted by innovative and sustained communication campaign along the public funding energy renovation programs applied for demonstration buildings.

## 4.3. Common Practice and Current Projects of Energy Renovation in Italy

The requalification of the multi-storey RC buildings is one of the hardest achievement [99]. Recently, a specific research study has verified the possibility to transform this kind of buildings into nZEB, with special attention to the Southern regions of the country [100]. The target of this study was also to contain the Pay Back Time (PBT). To get these aims, the authors have defined the most economical way to insulate the whole building envelope and to integrate PV panels in the façades. The simulations were carried on two multi-storey building of Librino, a popular neighbourhood of Catania, designed in the Seventies by Japanese architect Kenzo Tange [101].

In the first case study (Figure 26), the design actions have been limited, in order to contain the PBT in around 10 years (taking advantage of the current fiscal incentives). The U-value of the envelope has been reduced by applying an external insulation (Table 7). The installation of a new electrically powered air-water reversible heat pump has been considered for heating and cooling purposes. The calculation of the primary energy balance before and after renovation has shown a reduction of 37.1% (Table 8). Copper Indium Gallium Selenide (CIGS) PV panels integrated in the façades will be responsible for the electricity production onsite (Figure 26). In particular, in spring and fall the electricity production exceeds the electricity demand (Figure 27); in winter, external electricity contributions are limited, while in summer the electricity demand turns out to be higher than the electricity production.

Sustainability 2018, 10, 1561



Figure 26. Energy renovation of an apartment block in Librino—Catania [100].



Figure 27. Comparison between electricity production from photovoltaic (PV) and electricity consumption [100].

Puilding Component	Before	After	
Bunding Component	[W/m <sup>2</sup> K]		
External walls (RC)	1.71	0.35	
External walls (clay bricks)	0.51	0.23	
Roof	0.44	0.44	
Slab over the porch	0.90	0.46	
Ground slab	2.09	0.37	

Table 7. U-value of the building envelope before and after renovation.

**Table 8.** Primary energy balance before and after renovation (without renewables) (H=heating, C = cooling, W = DHW, L = lighting.

[kWh/m <sup>2</sup> a]	PE <sub>H</sub>	PE <sub>C</sub>	PE <sub>W</sub>	PEL	PE
Before After	61.99 38.31	15.43 10.27	20.27 0.00	75.36 60.29	173.05 108.87
Reduction	38.2%	33.4%	100.0%	20.0%	37.1%

In the second case study, Cadmium Telluride (CdTe) PV panels were specifically dimensioned to get the nZEB standard, covering large part of the façade (more than 60%, excluding the northern side). In this case the energy renovation becomes also an opportunity to redesign the façades (Figure 28), changing the building architectural image. The energy demand has been reduced by insulating the whole envelope from the outside and by providing an air-water reversible heat pump. Table 9 shows the improvement of the thermal insulation of the building envelope after the proposed intervention. According to the calculation of the primary energy balance before and after renovation (Table 10), the primary energy needs decrease by 62%. The PBT turned out to be 9 years (with fiscal incentives).



Figure 28. Comparison between the PV energy production and energy consumptions [100].

Building Component	Before	After	
Bunding Component	[W/m <sup>2</sup> K]		
External walls (RC)	4.10	0.40	
External walls (lightweight concrete)	1.96	0,39	
Windows	5.80	1.60	
Roof	3.85	0.30	
Slab over the porch	3.14	0.38	

Table 9. U-value of the building envelope before and after renovation.

**Table 10.** Primary energy balance before and after renovation without renewables) (H = heating, C = cooling, W = DHW, L = lighting).

[kWh/m <sup>2</sup> a]	PE <sub>H</sub>	PE <sub>C</sub>	PE <sub>W</sub>	PEL	PE
Before After	153.42 20.79	15.13 21.78	19.26 0.00	66.85 54.92	256.46 97.49
Reduction	86.4%	-43.9%	100.0%	20.0%	62.0%

The exploitation of the façades for PV installation turned out to be essential, since the roof surface of apartment blocks with more than 4 stories is generally insufficient to host all the panels required to reach the nZEB standard.

#### 5. Seismic versus Energy Renovation: Technical Solutions, between Opportunities and Constraints

In Italy and Romania, a large number of RC structures need seismic strengthening. While for some recent generations the base isolation and bracing is possible and feasible, for most of cases a considerable amount of new structural members is required. Many traditional techniques are available and provisions that regulate their use are given in the relevant parts of the Italian [12–14], Romanian [35] and European [15] codes. Some examples of possible solutions are presented below, along with comments on their implications on the energy renovation solutions.

Figure 29 shows the strengthening of columns by concrete jackets. The concrete cover added to most of columns involves also the cutting and/or removal of some envelope material. The final design must take into account the adequate detailing and calculation in order to ensure a continuous thermal insulation for control of thermal bridges impact on performances.



Figure 29. Strengthening solution for existing columns, using concrete jacketing (adapted after [34]).

In Figure 30 the new shear wall, when and if it is added at exterior, will change completely the envelope thermal parameters therefore the energy renovation design shall take into account the new details and thermal transmittance values. In Figures 31 and 32 the jacketing of shear walls is necessary

on the edges. In case of external walls, the need of special care for energy renovation detailing may be limited to those areas.



Figure 30. Strengthening solution using new structural walls, as shear walls (adapted after [34]).







**Figure 32.** Strengthening solution for structural walls, with steel jacketing interventions on the edges (adapted after [34]).

However, when looking in details of such solutions, as in Figure 33, the building of a new and greater flange-column of a shear wall also changes the envelope situation, with the need of careful thermal bridges analysis. In Figure 34, the renovation with steel elements is much easier to apply but steel has a greater heat transfer capacity and also changes the envelope situation, with the need of careful thermal bridges analysis.



**Figure 33.** Detail of strengthening solution for structural walls, with RC jacketing interventions on the edges (adapted after [34]).



Figure 34. Detail of strengthening solution for structural walls, with steel jacketing interventions on the edges (adapted after [34]).

The examples herein presented point out that many solutions are available in literature to upgrade existing structures by enhancing its stiffness, strength and ductility capacity. However, these interventions may have a negative impact on the thermal performance of the building, thus requiring additional measures. On the other hand, any intervention devoted only to the energy requalification of RC buildings could be nullified by the effect of the earthquake if the building is not seismic-resistant. The interventions depicted in the previous figures (Figures 29–34) are intended to add lacking properties, in terms of resistance and ductility, to structural members in existing buildings. The cases referred to shall ensure such qualities by adding stronger or more ductile flanges to shear-walls, of inserting a web between columns, as to become structural ones. They shall be associated with other reinforcement measures. Based on this consideration, existing buildings should always be assessed in terms of both seismic and energy deficiencies and the new challenge will be to ideate, design and realize interventions that integrate seismic upgrading and energy renovation. An example that fits well this philosophy is the intervention by Takeuchi et al. [102,103] on the Midorigaoka-1st building of Tokyo Institute of Technology, a 6-story RC building designed in 1966 before the revision of Building Code of Japan in 1971. The main deficiency of the structure was the low ductility capacity of the columns. Furthermore, the thermal performance of the building envelope did not satisfy the current standard in Japan. The solution proposed by Takeuchi et al. is an integrated facade that includes glasses, louvers and a steel BRB frame (Figure 35). This multi-skin exoskeleton improves both the seismic and the thermal performance of the building. In particular, the steel frame equipped with BRBs is firmly attached to the ground and to the building facade. In occurrence of ground motions, the BRBs act as hysteretic dampers, yield and absorb input seismic energy before that the RC structure is damaged, thus reducing the drifts and protecting the structure. The system of glasses and louvers properly oriented can mitigate the inner temperature both in summer and winter and reduce the consumption of energy needed for heating and refreshing, respectively.



Figure 35. Midorigaoka-1st building after retrofit [103] (photographs by Mamoru Ishiguro).

There are also some recent and interesting research works that propose and/or assess combined seismic ad energy renovation scenarios for recent buildings. For instance, Leone and Zuccaro [104] have developed, within the EU-FP7 CRISMA project, a multi-criteria decision support system to select the optimal integrated retrofitting scenario, taking into account technical, financial and economic aspects. The proposed tool aims at enabling decision makers and local authorities to implement policies and large-scale programs devoted to the sustainable improvement of the existing residential stock. Calvi et al. [105] have proposed a "green and resilient indicator" to evaluate the earthquake resilience and energy efficiency of the exiting building stock. Manfredi and Masi [106] have proposed an integrated retrofitting intervention that consists in replacing the existing masonry infill walls with hollow clay blocks that are able to increase both seismic and thermal performance. For mid-low hazard areas, this technique could determine a full rehabilitation with regards to both seismic and thermal requirements, in compliance with the corresponding codes. La Greca and Margani [19] have highlighted the advantages of combined renovation actions, indicating the barriers that currently limit such actions and suggesting possible countermeasures. Moreover, the H2020 ProGETone project [107] is currently developing retrofitting solutions based on the implementation of multi-skin exoskeletons to enhance the seismic resilience and the energy performance.

# 6. Discussion

The seismic and energy renovation combined approach has strong and weak points, both in Italy and in Romania.

The recent Italian legislative has introduced different incentives, in form of tax credit (so-called 'SismaBonus'), which, especially from 2017, considerably reduce the economic investment for the renovation of earthquake-prone buildings. In particular, the building owners in seismic areas may receive a tax credit distributed over 5 years that covers up to 85% of the cost of the retrofit intervention, with a cost limit of €96,000 per apartment. Simple guidelines are provided to assess the seismic risk class of buildings in compliance with the current seismic technical code for constructions. Hence, the amount of the tax deduction is related to the class increase due to the strengthening intervention.

In addition, in Italy also the energy efficient renovation benefits from different fiscal incentives, which since 1998 have considerably encouraged this kind of intervention. Nowadays, energy retrofitting investments take advantage of VAT reductions (10% instead of 22%) and tax credits that range from 36% to 75% of the renovation cost, with a limit of €40,000 per apartment (so-called 'EcoBonus'). In particular, interventions on single family houses benefit from fiscal incentives that allow to write off 36% of costs on taxes, with deductions equally distributed over 10 years. This share increases from 70% up to 75% for apartments buildings, according to the reached energy performance. If required, the tax credit can be assigned to third parties, such as construction companies. Alternatively, it is possible to benefit from incentives governed by the Ministerial Decree of 16 February 2016 (also known as 'Conto Termico 2.0') that

provides subsidies for the production of thermal energy from RES and the increase in energy efficiency. These subsidies cover 40% of the eligible expenditure, with specific limits for the unit and total costs of each type of interventions. They will be refunded either in five annual rates or in a single solution in the case of public administrations or Energy Service Companies (ESCOs).

From January 2018 [108], it is also possible to benefit from tax credits that cover up to 85% of the cost for combined seismic and energy renovation actions of apartment blocks, with a limit of  $\notin$ 136,000 per apartment and deductions distributed over 10 years (so-called 'SuperBonus').

The Italian residential real estate has a very high level of fragmentation. It is usually quite difficult to reach a consensus among the building owners for the implementation of renovation works.

In Romania, since 69% of residential buildings existing in 2011 have been erected before 1977, many dwellings may have insufficient earthquake protection. The technical aspects of assessment, design and solutions for seismic strengthening are solved in relevant codes and laws starting during the1990s, while the energy rehabilitation has also a pertinent evolution, fully correlated with the EU Directives.

The problem of funding seismic strengthening is legally and financially solved since 1994 but the key issue is that of relationship between the funding provided by MDRAPFE and actual management of seismic strengthening projects which is done by local authorities. In Bucharest, there is a large list of pre-1940 buildings ranked at seismic risk but the number of buildings that were strengthened is relatively low.

Presently in Romania, is in force a yearly National Program for seismic risk reduction (based on Ordinance n. 20/1994) [81] correlated with Code P100-3/2008 [35] and EC8 provisions. For the time being, the National Program for thermal and energy rehabilitation [80] is a more successful social project, because it was applied on buildings with low or any seismic risk. The cost of energy rehabilitation was born in most cases by local authorities, from own budgets or European Programs.

In both countries, the main reason that makes the energy rehabilitation to be one step in front of structural interventions is that seismic strengthening involves the structure and is operated mainly from the inside of the building, while the energy renovation involves, above all, the envelope and is operated mostly from the outside. The energy renovation works are speedy, while for seismic strengthening owners are often afraid of high costs and related mortgaging; many of them are rather old and low-income, absentees or just do not want to de disturbed by evacuation and long-term noisy works. On the other hand, the funding is from separate sources and under separate legal framework, thus the full renovation approach is difficult.

From the energy renovation perspective, the level of ambition in terms of performance follows the nZEB target. Even if these performances are made mandatory only for new buildings authorized in very short time from now, the EU policies are rapidly advancing towards the renovation of existing building stock at nearly zero energy levels. This implicates a very high insulation level, high performance (and usually heavy) windows (installed in line with the applied insulation, that is, outside the mechanical resistance layer) and with minimized thermal bridges. This could be a very difficult task for the design of energy renovation technical solutions, especially in respect to the highly enforced design provisions due to the high seismic risk. In current practice, most of the building envelope renovation details (e.g., joining different building components) are designed with greater focus on mechanical resistance issues than heat transfer concept (e.g., avoiding thermal bridges), the result being usually an interrupted insulating envelope. In all cases, mechanical resistance has to be ensured but if penetration of the insulating layer is unavoidable, then the thermal conductivity of the penetrating material should be as low as possible and insulation layers at building component connections should merge into each other over the entire surface without interruption.

In this respect, best-practice guidelines and detailed specification for the building envelope have to be developed in order to support suitable solutions for nZEBs, tested in order to comply with the rigorous seismic design provisions. The aim is to inform and to shape the internal market in order to increase the current level of technical knowledge and technologies before the legal requirements under Directive 2010/31/EU cause a blockage of the local construction industry due to the inability to fulfil the Directive.

#### 7. Conclusions

The European and National Standards and Directives have been promoting, from different sustainability reasons, seismic and energy renovation actions to mitigate disasters and climate changes impact. In Italy, over 60% of the current building stock was built between the 1950s and the 1980s, when seismic safety and energy efficiency regulations were absent or mandatory only in few regions. As a consequence, most of the real estate is earthquake-prone and highly energy-consuming [19]. The combined seismic and energy renovation of Italian buildings is of paramount importance for the safety of the population against earthquakes and other natural hazards related to climate change.

In Romania, there is a comprehensive framework of laws and technical regulations, both for earthquakes and energy, implemented over the last five decades for new and existing buildings. In this country, the wide impact of the 1940 and 1977 earthquake is a landmark, while the probability of a large earthquake in the near future is increasing, thus the seismic risk reduction is a national priority. The National Program for thermal and energy rehabilitation of buildings is a separate endeavour, also a priority, pushed by the impact of climate changes and the related European Union Directives. In order to achieve the ambitious and very demanding objectives of Eurocodes and European Energy Performance Directives, under Romanian laws, it is necessary to mobilize all the involved stakeholders, to identify and face the positive and negative aspects and barriers, as well as the deficiencies that exist or may occur in the future.

In both countries, the policy of renovation of the real estate is hindered especially by high costs and disruptions for users. The integration of structural, architectural and urban planning approaches may improve the effectiveness and diffusion of these actions. However, steps like structural analysis, energy audits, public tender procedures, quality of projects and execution, regulatory framework, financial and fiscal incentives, quality control, training of designers and builders and workers' qualification still represent critical issues.

A variety of techniques for seismic upgrading and energy renovation of buildings are already available. Some of these techniques are known to structural designers and construction companies for a long time now and provisions for their application are provided in the current technical regulations. Other techniques, which are classified as innovative, are also applicable because studies that demonstrate their effectiveness and the relevant design guidelines are available in the technical literature. However, each of these techniques aims at improving only one aspect of the building performance: reduction of the risk of damage/failure caused by earthquakes or reduction of the energy consumption during the building operation. Based on the framework depicted above, it emerges that often seismic upgrading interventions can have no effect or may be even detrimental on the energy performance of the building and vice versa. Hence, the main challenge of the next years is to develop approaches that integrate together techniques for energy and seismic renovation. Important aspect to be deepened are the study of the compatibility of the integrated techniques and the definition of interventions that improve both seismic energy performance. The target of these approach should be the transformation of the existing energy-consuming and seismic-prone buildings in nZEB seismic-resistant ones. The availability of such integrated approaches for seismic-energy upgrading can promote the renovation of existing buildings by pursuing multiple targets: (i) to benefit from fiscal and financial incentives for both seismic and energy renovation, (ii) to maximize the positive effect gained by the intervention on the building (two performance objectives are reached with a single intervention) and (iii) to preserve in time the value of the investment (the investment for energy renovation could be nullified by earthquakes if seismic upgrading has not been pursued too). To this purpose, the exchange of experience among countries with similar issues, such as Italy and Romania, plays a crucial role to share best practices and to define correct skill and adequate competences.

Author Contributions: Italian side–G.M. and V.S.: conceptualization and energy issues; S.M.C.: case-studies of the energy renovation; E.M.M. and P.P.R.: seismic issues. Romanian side–E.-S.G., M.S.G., H.P. and V.M.: conceptualization, state-of-the-art, codes review; Z.M., R.P. and M.V.: case-studies.

**Funding:** The authors of DICAR acknowledge the results based on the financing provided by the National Research Plan of the Italian Ministers of University and Scientific Research through the Departmental Research Projects of which they are responsible. The authors of NIRD URBAN-INCERC acknowledge the research framework and financial support provided by the Ministry of Research and Innovation under the Nucleu Program CRESC. The authors of University of Architecture and Urbanism "Ion Mincu" acknowledge the results based on the financing provided by Romanian Ministry of Research and innovation in the following contracts: (1) UAUIM Contract no. 23/2008, 2008–2010, PNCDI 2, n. 1511, Program—INNOVATION, module I: Product-system development, Project category: PDP, project type: CDI Oriented: "Passive houses suitable for climatic conditions in Romania—acronym Passive House"; (2) UAUIM Contract no. 21/2008-2011, contract PNCDI 2, no. 22120, 2008–2011, Program 4—partnerships in the priority areas, research direction: D3, domain 2: "Integrated solutions systems for the rehabilitation of buildings / residential neighborhoods—Acronym SIR." (3) The project "Train-to-nZEB: The Building Knowledge Hubs" is funded by the European Union's Horizon 2020 research and innovation programme under grant agreement n. 649810. (4) The project "Fit-to-nZEB: The Building Knowledge Hubs" is funded by the European Union's Horizon 2020 research and innovation programme under grant agreement n. 754059.

Acknowledgments: The present review paper describes some results of the International Research titled "Seismic and Energy Renovation Strategies for Sustainable Buildings". This research has been carried out within the Agreement of Cooperation signed in July 2017 between the University of Catania, the "Ion Mincu" University of Architecture and Urban Planning (Bucharest) and URBAN INCD INCERC—National Institute for Research and Development in Construction, Urban Planning and Sustainable Spatial Development (Bucharest).

Conflicts of Interest: The authors declare no conflict of interest.

#### Acronyms

BEMS	Building Energy Management Systems
BIPV	Building Integrated Photovoltaics
BRB	Buckling Restrained Braces
CdTe	Cadmium Telluride
CIGS	Copper Indium Gallium Selenide
DHW	Domestic Hot Water
DL	Damage Limitation
EN 1988	Eurocode 8
ESCO	Energy Service Company
ETICS	External Thermal Insulation Composite System
FRP	Fibre Reinforced Polymer
HDD	Heating Degree Days
H′T	transmission heat loss coefficient
HVAC	Heating, Ventilating and Air Conditioning
NC	Near Collapse
NTC08	"Norme Tecniche per le Costruzioni" enforced in 2008
KL	Knowledge Levels
NSC	Non-Structural Components
nZEB	nearly Zero Energy Building
PBT	Pay Back Time
PCM	Phase Change Materials
PV	PhotoVoltaic
PV/T	PhotoVoltaic and Thermal
PGA	Peak Ground Acceleration
R	Thermal Resistance
RC	Reinforced Concrete
RES	Renewable Energy Sources
SD	Significant Damage
ST	Solar Thermal
TIM	Transparent Insulation Materials
U	Thermal Trasmittance
VIP	Vacuum Insulating Panels

# References

- 1. Wikipedia List of 21st-Century Earthuakes. Available online: https://en.wikipedia.org/wiki/List\_of\_21st-century\_earthquakes#2009 (accessed on 23 May 2018).
- Pérez-Lombard, L.; Ortiz, J.; Pout, C. A review on buildings energy consumption information. *Energy Build.* 2008, 40. [CrossRef]
- 3. Calvi, G.M.; Pinho, R.; Magenes, G.; Bommer, J.J.; Restrepo-Vélez, L.F.; Crowley, H. Development of seismic vulnerability assessment methodologies over the past 30 years. *ISET J. Earthq. Technol.* **2006**, *43*, 75–104.
- 4. Mazzarella, L. Energy retrofit of historic and existing buildings. The legislative and regulatory point of view. *Energy Build.* **2015**, *95*, 23–31. [CrossRef]
- Salemi, A.; Patania, F.; Cascone, S.; Gagliano, G.; Lo Faro, A.; Lombardo, G.; Moschella, A.; Sapienza, V. Thermal performance of existing buildings in Mediterranean climate area. In *Changing Needs, Adaptive Buildings, Smart Cities*; POLIScript: Milan, Italy, 2013; pp. 1017–1024.
- Detommaso, M.; Gagliano, A.; Nocera, F.; Patania, F.; Sapienza, V. Deploy Energy-efficient Technologies in the Restoration of a Traditional Building in the Historical Center of Catania (Italy). *Energy Procedia* 2014, 62, 62–71. [CrossRef]
- Sapienza, V. Spontaneous architecture and energetic sustainability: The Aeolian homes of Filicudi Isle. In Proceedings of the ZEMCH-12, Glasgow, UK, 20–22 August 2012; pp. 340–350.
- Royal Decree n. 193, 18/04/1909. Mandatory Technical and Hygienical Rules for Repair of Existing and New Buildings in the Areas Struck by the Earthquake of 28 December 1908 and others Listed in the Royal Decree of 15/04/1909; Gazzetta Ufficiale n. 95, 22/04/1909; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 1909. (In Italian)
- Report of the Commission in Charge for Studying and Proposing Mandatory Building Codes for the Municipalities Struck by the Earthquake of 28 December 1908 and by Others Earlier; Stabilimento Tipo-Litografico del Genio Civile: Rome, Italy, 1909. (In Italian)
- Royal Decree n. 573, 29/04/1915. Mandatory Technical and Hygienical Rules in the Areas Struck by the Earthquake of 13 January 1915; Gazzetta ufficiale n. 117, 11/05/1915; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 1915. (In Italian)
- 11. Ricci, P.; De Luca, F.; Verderame, G.M. 6th April 2009 L'Aquila earthquake, Italy: Reinforced concrete building performance. *Bull. Earthq. Eng.* 2011, *9*, 285–305. [CrossRef]
- Italian Ministry of Public Works, Ministry Decree. 14/01/2008. Technical Regulations for Constructions; Suppl. Ord. Gazzetta Ufficiale n. 29, 4/02/2008; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2008. (In Italian)
- Italian Ministry of Public Works: Circolare n. 617. 2/02/2009. Commentary on Technical Regulations for Constructions; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2009. (In Italian)
- Presidence of the Council of Ministers of the Italian Republic: Ordinance n. 3431, 3/05/2005. Technical Regulations for Design, Assessment and Seismic Upgrade of Buildings; Suppl. Ord. Gazzetta Ufficiale n. 85, 10/05/2005; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2005. (In Italian)
- CEN EN 1998-3, EuroCode 8. Design of Structures for Earthquake Resistance—Part 3: Assessment and Retrofitting of Buildings; European Committee for Standardization: Bruxelles, Belgium, 2005.
- Italian Ministry of Infrastructure and Transport, Ministry Decree. 21/07/2012. Ratification of National Annexes with Parameters Ascribed for Application of Eurocodes; Suppl. Ord. Gazzetta Ufficiale n. 73, 27/03/2013; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2013. (In Italian)
- Law n. 373/1976. Norme per il Contenimento del Consumo Energetico per usi Termici Negli Edifici; Gazzetta Ufficiale Della Repubblica Italiana n. 148 del 7.6.1976; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2015. (In Italian)
- Law n. 10/1991. Norme per L'attuazione del Piano Energetico Nazionale in Materia di Uso Razionale Dell'energia, di Risparmio Energetico e di Sviluppo Delle Fonti Rinnovabili di Energia; Gazzetta Ufficiale della Repubblica Italiana s.o. n. 13 del 16.1.1991; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 1991. (In Italian)
- 19. La Greca, P.; Margani, G. Seismic and energy renovation measures for sustainable cities: A critical analysis of the Italian scenario. *Sustainability* **2018**, *10*, 254. [CrossRef]
- Economidou, M.; Laustsen, J.; Ruyssevelt, P.; Staniaszek, D. Europe's Buildings under the Microscope; BPIE: Brussels, Berlgium, 2011; ISBN 9789491143014.

- Finocchiaro, L.; Wågø, S.I. Architectural design and aesthetics of Zero Emission Buildings: An analysis of perceived architectural qualities in the ZEB Living LAB in Trondheim. In Proceedings of the 33rd PLEA International Conference: Design to Thrive, Edinburgh, UK, 3–5 July 2017.
- 22. Cicero, C.; Lombardo, G. Building envelope and energy. J. Civ. Eng. Arch. 2011, 8, 986–995.
- 23. Margani, G. L'edificio passivo nel clima mediterraneo. *Costruire in laterizio* **2011**, *141*, 46–49.
- Inter-Ministerial Decree, 26/6/2015. Application of the Calculation Methodology of the Energy Performances and Definition of the Prescription and of the Minimum Required for Buildings; Minister of Economic Development: Rome, Italy, 2015.
- Legislative Decree n. 28, 3/3/2011. Attuazione Della Direttiva 2009/28/CE Sulla Promozione Dell'uso Dell'energia da Fonti Rinnovabili, Recante Modifica e Successiva Abrogazione Delle Direttive 2001/77/CE e 2003/30/CE; Gazzetta Ufficiale della Repubblica Italiana s.o. n. 81 del 28.3.2011; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2011. (In Italian)
- Law 90, 3/8/2013. Conversione in Legge, con Modificazioni, del Decreto-Legge 4 Giugno 2013, n. 63, Recante Disposizioni Urgenti per il Recepimento Della Direttiva 2010/31/UE del Parlamento Europeo e del Consiglio del 19 Maggio 2010; Gazzetta Ufficiale della Repubblica Italiana n. 181 del 3.8.2013; Istituto Poligrafico e Zecca dello Stato: Rome, Italy, 2013. (In Italian)
- 27. Balan, S.; Cristescu, V.; Cornea, I. *The Romania Earthquake of 4 March 1977*; Editura Academiei: Bucharest, Romania, 1982. (In Romanian)
- Georgescu, E.S. Earthquake Engineering Development before and after the 4 March 1977, Vrancea, Romania Earthquake. In Proceedings of the 25 years of Research in Earth Physics, Bucharest, Romania, 25–27 September 2002; pp. 93–107.
- Berg, G.V.; Bolt, B.A.; Sozen, M.A.; Rojahn, C. Earthquake in Romania March 4, 1977. An Engineering Report; National Academy Press: Washington, DC, USA, 1980.
- 30. Georgescu, E.S.; Pomonis, A. The Romanian Earthquake of 4 March 1977; Revisited: New Insights into its Territorial, Economic and Social Impacts and their Bearing on the Preparedness for the Future. In Proceedings of the 14th World Conference on Earthquake Engineering, Beijing, China, 12–17 October 2008.
- Code P100-1/2006. Seismic Design Code, Part I—Provisions for the Design of Buildings. Indicative P100-1; Elaborated by UTCB, Endorsed by MDRL; Part I, n. 803 and 803 bis; Official Journal of Romania: Bucharest, Romania, 2006. (In Romanian)
- Code P100-1/2013. Seismic Design Code—Part I—Provisions for the Design of Buildings. Indicative P100-1 (In Romanian) Elaborated by UTCB, Endorsed by MDRAP; Part I, n. 558 of 03. 09. 2013; Official Journal of Romania: Bucharest, Romania, 2013. (In Romanian)
- 33. Craifaleanu, I. Romanian Seismic Design Code: Benchmarking Analyses with Reference to International Codes and Research Needs for Future Development. *Constructii* **2011**, *12*, 55–64.
- 34. Craifaleanu, I. Eurocode 8, Part 3 and the Romanian Seismic Code for the Assessment of Existing Buildings, P100-3: Similarities and Differences. Presented at the Workshop Implementation of the EC 8-Part 3: 2005: Assessment and Interventions, on Buildings in Earthquake Prone Areas, Athens, Greece, 12 April 2013.
- Code P100-3/2008. Seismic Design Code—Part III—Provisions for Seismic Assessment of Existing Buildings, Indicative P 100-3/2008. Elaborated by UTCB, Endorsed by MDRL; n. 704; Official Journal of Romania: Bucharest, Romania, 2008. (In Romanian)
- 36. C107/2016. Normative Regarding the Thermal Calculation of Building Elements of Buildings, Indicative C 107-2005, Approved by Order of the Minister of Transports, Constructions and Tourism; n. 2055/2005 Annex A: Climate Zoning of Romania, Annex B: Energy Requirement for Buildings Whose Energy Consumption is Almost Equal to Zero, Modified and Completed by Order of Minister of Regional Development and Public Administration n. 386/28.03.2016; Official Journal of Romania: Bucharest, Romania, 2016.
- C107/6-2002 General Norm for Calculating the Mass Transfer (Humidity) through the Building Elements; Bulletin of Constructions no. 14/2002; INCERC: Bucharest, Romania, 2002.
- C107/7-02 Norm for Thermal Stability Design of Building Closures; Bulletin of Constructions no. 8/2003; INCERC: Bucharest, Romania, 2003.

- 39. Catalogue of Thermal Bridges SPECIFIC to Buildings; Annex K (Informative) to Order n. 1590/24.08.2012 for the Modification and Completion of Part 3—Normative Regarding the Calculation of Thermal Energy Performance of Building Elements of Buildings, Indicative C107/3 of the Technical Regulation "Normative Regarding the Thermo-Technical Calculation of Building Elements of Buildings" 2005, Approved by the Order of the Minister of Transports, Constructions and Tourism no. 2055/2005; Official Journal of Romania: Bucharest, Romania, 2012.
- 40. Order of Minister of Regional Development and Public; Administration n. 386/28.03.2016; Official Journal of Romania: Bucharest, Romania, 2016. (In Romanian)
- Mc 001/2006. Methodology of Buildings Energy Performance Calculation; MTCT Order n. 157, with Annexes 1, 2 and 3, Official Journal of Romania Part I, n. 126 bis, 21.02.2007; Romanian Government: Bucharest, Romania, 2007.
- C107/2010. Normative Regarding the Thermo-Technical Calculation of Building Elements of Buildings; Indicative C107-2005 Modified by Order n. 2513 of 22 November 2010 for the Amendment of Official Journal of Romania, Part I, n. 820 of 18 December 2010; Romanian Government: Bucharest, Romania, 2010.
- Possibilities for Reduction of Energy Use for Space Heating of Residential Buildings; IPCT Contract n. 336/1992 (for the Ministry of Public Works), 1992–1994; Ministry of Public Works: Bucharest, Romania, 1992.
- 44. Decision No. 122/25.02.2016 for Approval of National Plan of Action in the Field of Energy Efficiency; Part I, nr. 169 bis of 11.03.2015; Monitorul Oficial/Official Journal of Romania: Bucharest, Romania, 2016. (In Romanian)
- 45. Franchin, P.; Pinto, P.E.; Rajeev, P. Confidence factors? J. Earthq. Eng. 2010, 14, 989–1007. [CrossRef]
- Chopra, A.K. Dynamics of Structures: Theory and Applications to Earthquake Engineering; Prentice Hall: Upper Saddle River, NJ, USA, 2001; ISBN 0-13-086973-2.
- 47. CEN EN 1998-1, EuroCode 8. Design of Structures for Earthquake Resistance—Part 1: General Rules, Seismic actions and Rules for Buildings; European Committee for Standardization: Bruxelles, Belgium, 2005.
- Bosco, M.; Ghersi, A.; Marino, E.M. On the evaluation of seismic response of structures by nonlinear static methods. *Earthq. Eng. Struct. Dyn.* 2009, *38*, 1465–1482. [CrossRef]
- 49. Fajfar, P. Capacity spectrum method based on inelastic demand spectra. *Earthq. Eng. Struct. Dyn.* **1999**, *28*, 979–993. [CrossRef]
- Iervolino, I.; Galasso, C.; Cosenza, E. REXEL: Computer aided record selection for code-based seismic structural analysis. *Bull. Earthq. Eng.* 2010, *8*, 339–362. [CrossRef]
- Tanganelli, M.; Viti, S.; Mariani, V.; Pianigiani, M. Seismic assessment of existing RC buildings under alternative ground motion ensembles compatible to EC8 and NTC 2008. *Bull. Earthq. Eng.* 2017, 15, 1375–1396. [CrossRef]
- 52. Vamvatsikos, D.; Cornell, C.A. Incremental dynamic analysis. *Earthq. Eng. Struct. Dyn.* 2002, 31, 491–514. [CrossRef]
- Garevski, M.; Ansal, A. Earthquake Engineering in Europe. Geotechnical, Geological and Earthquake Enigneering; Springer: Berlin, Germany, 2010; ISBN 978-90-481-9543-5.
- Dell'umbria, R. Guidelines for Post-Seismic Rehabilitation and Reconstruction of Buildings; Tipografia del Genio Civile: Rome, Italy, 1999; ISBN 88-7722-460-6. (In Italian)
- 55. Bakis, C.E.; Bank, L.C.; Brown, V.L.; Cosenza, E.; Rizkalla, S.H.; Triantafillou, T.C. Fiber-reinforced polymer composites for construction—State-of-the-art review. *J. Compos. Constr.* **2002**, *6*, 73–87. [CrossRef]
- Barbagallo, F.; Hamashima, I.; Hu, H.; Kurata, M.; Nakashima, M. Base shear capping buildings with graphite-lubricated bases for collapse prevention in extreme earthquakes. *Earthq. Eng. Struct. Dyn.* 2017, 46, 1003–1021. [CrossRef]
- 57. Fenz, D.M.; Constantinou, M.C. Behaviour of the double concave Friction Pendulum bearing. *Earthq. Eng. Struct. Dyn.* **2006**, *35*, 1403–1424. [CrossRef]
- Naeim, F.; Kelly, J.M. Design of Seismic Isolated Structures: From Theory to Practice; Wiley & Sons: New York, NY, USA, 1999; ISBN 978-0-471-14921-7.
- Kumar, M.; Whittaker, A.S.; Constantinou, M.C. Characterizing friction in sliding isolation bearings. *Earthq. Eng. Struct. Dyn.* 2015, 44, 1409–1425. [CrossRef]
- Christopoulos, C.; Filiatrault, A. Principles of Passive Supplemental Damping and Seismic Isolation; IUSS Press: Pavia, Italy, 2006; ISBN 8873580378.
- Deng, K.; Pan, P.; Nie, X.; Xu, X.; Feng, P.; Ye, L. Study of GFRP Steel Buckling Restraint Braces. J. Compos. Constr. 2015, 19, 04015009. [CrossRef]

- 62. Lavan, O. Optimal Design of Viscous Dampers and Their Supporting Members for the Seismic Retrofitting of 3D Irregular Frame Structures. J. Struct. Eng. 2015, 141. [CrossRef]
- 63. Lee, K.S.; Fan, C.P.; Sause, R.; Ricles, J. Simplified design procedure for frame buildings with viscoelastic or elastomeric structural dampers. *Earthq. Eng. Struct. Dyn.* **2005**, *34*, 1271–1284. [CrossRef]
- 64. Rossi, P.P. Importance of isotropic hardening in the modeling of buckling restrained braces. J. Struct. Eng. 2015, 141. [CrossRef]
- 65. Xie, Q. State of the Art of Buckling-Restrained Braces in Asia. J. Constr. Steel Res. 2005, 61, 727–748. [CrossRef]
- Barbagallo, F.; Bosco, M.; Marino, E.M.; Rossi, P.P.; Stramondo, P.R. A multi-performance design method for seismic upgrading of existing RC frames by BRBs. *Earthq. Eng. Struct. Dyn.* 2017, 46, 1099–1119. [CrossRef]
- 67. Formisano, A.; Castaldo, C.; Chiumiento, G. Optimal seismic upgrading of a reinforced concrete school building with metal-based devices using an efficient multi-criteria decision-making method. *Struct. Infrastruct. Eng.* **2017**, *13*, 1373–1389. [CrossRef]
- 68. Pan, P.; Li, W.; Nie, X.; Deng, K.; Sun, J. Seismic performance of a reinforced concrete frame equipped with a double-stage yield buckling restrained brace. *Struct. Des. Tall Spec. Build.* **2017**, *26*, e1335. [CrossRef]
- 69. Impollonia, N.; Palmeri, A. Seismic performance of buildings retrofitted with nonlinear viscous dampers and adjacent reaction towers. *Earthq. Eng. Struct. Dyn.* **2018**. [CrossRef]
- Neri, F.; La Guzza, M.; Russo, A. Use of Dissipative Braces for Seismic Protection of the School Building Cappuccini in Ramacca (CT). On-line Journal GLISNews 2008. Available online: http://www.assisiantiseismicsystems.org/territorial/glis/glisnews/gn11/gn11\_cappuccini.pdf (accessed on 29 September 2017). (In Italian)
- 71. FIP Industriale. Available online: https://www.fipindustriale.it/index.php?area=108&menu=99&page=228 (accessed on 29 September 2017).
- Sorace, S.; Terenzi, G.; Magonette, G.; Molina, F.J. Experimental investigation on a base isolation system incorporating steel–Teflon sliders and pressurized fluid viscous spring dampers. *Earthq. Eng. Struct. Dyn.* 2008, 37, 225–242. [CrossRef]
- 73. Marshall, J.D.; Charney, F.A. Seismic response of steel frame structures with hybrid passive control systems. *Earthq. Eng. Struct. Dyn.* 2012, 41, 715–733. [CrossRef]
- Wolff, E.D.; Ipek, C.; Constantinou, M.C.; Tapan, M. Effect of viscous damping devices on the response of seismically isolated structures. *Earthq. Eng. Struct. Dyn.* 2015, 44, 185–198. [CrossRef]
- 75. De Domenico, D.; Ricciardi, G. An enhanced base isolation system equipped with optimal tuned mass damper inerter (TMDI). *Earthq. Eng. Struct. Dyn.* **2017**, 1–24. [CrossRef]
- 76. Qu, Z.; Wada, A.; Motoyui, S.; Sakata, H.; Kishiki, S. Pin-supported walls for enhancing the seismic performance of building structures. *Earthq. Eng. Struct. Dyn.* **2012**, *41*, 2075–2091. [CrossRef]
- 77. Toranzo-Dianderas, L.A.; Restrepo, J.I.; Carr, A.J.; Mander, J.B. Rocking confined masonry walls with hysteretic energy dissipators and shake-table validation. In Proceedings of the 13th World Conference on Earthquake Engineering, Vancouver, BC, Canada, 1–6 August 2004.
- Panian, L.; Steyer, M.; Tipping, S. An innovative approach to earthquake safety and concrete construction in buildings. J. Post-Tens. Inst. 2007, 5, 7–16.
- Stevenson, M.; Panian, L.; Korolyk, M. Post-tensioned concrete walls and frames for seismic-resistance—A case study of the David Brower Center. In Proceedings of the SEAOC Annual Convention, HI, USA, 23–27 September 2008.
- Balducci, A.; Castellano, M.G. Seismic retrofitting of a RC school building through the "Damping Towers" system. *Progettazione Sismica* 2015, 6. (In Italian) [CrossRef]
- Ordinance n. 20/1994. Government Ordinance on Existing Buildings Risk Reduction; Part I, n. 863, Revised, Republished in 19.11.2015; Monitorul Oficial/Official Journal of Romania: Bucharest, Romania, 2015. (In Romanian)
- Stazi, F.; Vegliò, A.; Di Perna, C.; Munafò, P. Experimental comparison between 3 different traditional wall constructions and dynamic simulations to identify optimal thermal insulation strategies. *Energy Build.* 2013, 60, 429–441.
- Baljit, S.S.S.; Chan, H.-Y.; Sopian, K. Review of building integrated applications of photovoltaic and solar thermal systems. J. Clean. Prod. 2016, 137, 677–689. [CrossRef]
- 84. Tripathy, M.; Sadhu, P.K.; Panda, S.K. A critical review on building integrated photovoltaic products and their applications. *Renew. Sustain. Energy Rev.* 2016, *61*, 451–465. [CrossRef]

- Maurer, C.; Cappel, C.; Kuhn, T.E. Progress in building-integrated solar thermal systems. Sol. Energy 2017, 154, 158–186. [CrossRef]
- Gagliano, A.; Nocera, F.; Patania, F.; Capizzi, A. Assessment of micro-wind turbines performance in the urban environments: An aided methodology through geographical information systems. *Int. J. Energy Environ. Eng.* 2013, *4*, 1–14. [CrossRef]
- 87. Özkan, H.A. A new real time home power management system. Energy Build. 2015, 97, 56-64. [CrossRef]
- 88. Sapienza, V.; Rodonò, G. Intesa Sanpaolo Tower. Arketipo 2016, 100, 104-115.
- Sapienza, V. The new chamber building of Malta: Architectural image and building technology. *Techne* 2014, 8, 219–227. [CrossRef]
- 90. Lu, Y.; Wang, S.; Shan, K. Design optimization and optimal control of grid-connected and standalone nearly/net zero energy buildings. *Appl. Energy* **2015**, *155*, 463–477. [CrossRef]
- 91. Gagliano, A.; Patania, F.; Nocera, F.; Capizzi, A.; Galesi, A. GIS-based decision support for solar photovoltaic planning in urban environment Smart Innovation. *Syst. Technol.* **2013**, *22*, 865–874. [CrossRef]
- Caponetto, R.; Caponetto, R.; Cecere, C.; D'Urso, S.; Habib, E.; Margani, G.; Sapienza, V. Automated collecting building envelopes. In *ICT, Automation and Industry of the Built Environment: From the Information Exchange to the Field Management*; Maggioli Editore: Rimini, Italy, 2013; pp. 340–357, ISBN 88-387-6258-1.
- Ma, Z.; Cooper, P.; Daly, D.; Ledo, L. Existing building retrofits: Methodology and state-of-the-art. Energy Build. 2012, 55, 889–902. [CrossRef]
- Complex, Multicriterial, Integrated Rehabilitation of Urban and Housing Complexes. Certification of Sustainable Buildings; "Ion Mincu" University Publishing House: Bucharest, Romania, 2011; ISBN 978-606-638-002-7.
- 95. Building Knowledge Hubs. Available online: http://www.train-to-nzeb.com (accessed on 30 September 2017).
- SC 007-2013. Framework Solutions for the Thermo-Hygro-Energetic Rehabilitation of Existing Building Envelope; Part I, no. 540 bis/27.08.2013; Monitorul Oficial/Official Journal of Romania: Bucharest, Romania, 2013.
- 97. BUILD UP Skills Project Romania (ROBUST). Available online: http://www.iee-robust.ro (accessed on 30 September 2017).
- BUILD UP Skills QualiShell Project. Available online: http://www.iee-robust.ro/qualishell (accessed on 29 September 2017).
- Castellazzi, L.; Zangheri, P.; Paci, D. Synthesis Report on the assessment of Member States' building renovation strategies. In *Publications Office of the European Union*; UE: Bruxelles, Belgium, 2016; ISBN 978-92-79-54703-4.
- Caponetto, R.; Margani, G.; Sapienza, V. La riqualificazione energetica dell'edilizia residenziale. *Il Progetto Sostenibile* 2016, *38*, 58–73. (In Italian)
- 101. Zappalà, S. Librino between district and town. Spazio e Società 1990, 52, 118-121.
- 102. Takeuchi, T.; Yasuda, K.; Iwata, M. Studies on Integrated Building Facade Engineering with High-Performance Structural Elements. *IABSE Symp. Rep.* **2006**, *92*, 33–40. [CrossRef]
- 103. Takeuchi, T.; Yasuda, K.; Iwata, M. Seismic retrofitting using energy dissipation façades. In Proceedings of the ATC-SEI09, San Francisco, CA, USA, 10–12 September 2009.
- 104. Leone, M.F.; Zuccaro, G. Seismic and energy retrofitting of residential buildings: A simulation—Based approach. *UPLanD* 2016, 1, 11–25.
- 105. Calvi, G.M.; Sousa, L.; Ruggeri, C. Energy Efficiency and Seismic Resilience: A Common Approach. In *Multi-Hazard Approaches to Civil Infrastructure Engineering*; Springer International Publishing: Cham, Switzerland, 2016; pp. 165–208.
- Manfredi, V.; Masi, A. Seismic Strengthening and Energy Efficiency: Towards an Integrated Approach for the Rehabilitation of Existing RC Buildings. *Buildings* 2018, *8*, 36. [CrossRef]
- 107. ProGETone. Available online: https://www.progetone.eu/ (accessed on 29 September 2017).
- Law 205/2017. Bilancio di Previsione Dello STATO per L'anno Finanziario 2018 e Bilancio Pluriennale per il Triennio 2018–2020; Gazzetta Ufficiale della Repubblica Italiana s.o. n. 302 del 29.12.2017; Gazzetta Ufficiale della Repubblica Italiana: Rome, Italy, 2017. (In Italian)



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).



MDPI

# Article From the Efficiency of Nature to Parametric Design. A Holistic Approach for Sustainable Building Renovation in Seismic Regions

# Sebastiano D'Urso <sup>1,\*</sup> and Bruno Cicero <sup>2</sup>

- <sup>1</sup> Department of Civil Engineering and Architecture, University of Catania, Via Santa Sofia 64, 95123 Catania, Italy
- <sup>2</sup> Architect-engineer, Via Mariannina Schininà 177b, 97100 Ragusa, Italy; bru.cicero@gmail.com
- \* Correspondence: sdurso@dau.unict.it; Tel.: +39-349-4787490

Received: 7 December 2018; Accepted: 22 February 2019; Published: 26 February 2019



**Abstract:** Cities are growing dramatically. At the same time, we are witnessing the obsolescence of the existing building stock due to its low performance in terms of structural stability, energy efficiency and, last but not least, beauty. Especially in Italy, a highly seismic country, most of the buildings erected between the 1950s and the 1980s are not only earthquake-prone but also aesthetically unpleasant. In this perspective, the urgency of improving the existing building stock in terms of seismic vulnerability opens up the opportunity to also work on its architectural image. This article draws from the assumption that the search for beauty represents an important and often neglected dimension of the search for sustainability. In particular, the presented study suggests and combines the use of parametric design and the structural shape of steel exoskeletons to renovate a typical earthquake-prone apartment block from the 1960s in Italy. The results show that the proposed parametric approach can provide and select different effective renovation solutions.

**Keywords:** sustainability and aesthetics; architectural image; parametric design; exoskeleton; seismic renovation; apartment blocks

# 1. Introduction

Despite the unprecedented post-second world war growth of cities [1], little attention has been paid to the issue of sustainability. In Italy, in particular, real-estate developments between the 1950s and the 1980s [2] were characterized by some local conditions that favored the spread of buildings that are far from being earthquake-resistant, as they usually failed to take into account the geographical, geological and topological features of the territory. Italy is a highly seismic territory. Politicians however, have, for many years, ignored this hazard. The first map that classifies, using a scientific approach, the Italian territory into seismic areas was in fact only produced between 1981 and 1984 (Ministerial Decree March 7, 1981), after the 1976 earthquake in the region of Friuli. Until then, most buildings located in highly seismic areas had been built without taking into account horizontal seismic actions [3]. Moreover, according to the Italian National Institute of Statistics (ISTAT) 2011 census [2,4] more than 60% of the existing buildings predate 1974, which is the year the first specific and extensive Italian code for earthquake-resistant buildings was issued (law 64/1974) [3]. This means that a large part of the current building stock was realized without anti-seismic restrictions and norms: about 10 million homes are located in the areas most at risk of which 1.4 million homes are located in zone 1 (high seismicity: Peak Ground Acceleration PGA > 0.25) and about 9 million homes in zone 2 (middle seismicity: 0.15 < PGA < 0.25); while 8.5 million homes are in zone 3 (low seismicity: 0.05 < PGA < 0.15) and 9.4 million ones in zone 4 (very low seismicity:  $ag \leq PGA$ ) [5].

#### Sustainability 2019, 11, 1227

Lack of careful planning and the urban speculation allowed Italian real estate developers after the second world war to build poor quality structures in terms of materials, construction details and in terms of architectural image as well [6–8]. It is also worth noting that most Italian buildings erected in compliance with the seismic regulations implemented between the 1950s and the 1990s are incompatible with current regulations, which feature much higher standards of anti-seismic efficiency. Hence, as stated by La Greca and Margani, over 50% of the Italian residential constructions are earthquake-prone but we have to highlight that they are also undeniably inadequate in terms of architectural aesthetics and design if compared to the beautiful Italian historic districts (Figure 1).



**Figure 1.** An example of a reinforced concrete building that does not take into account the aesthetic value of its urban landscape nor the seismic risk of the area in the Baroque city of Noto, Italy.

Therefore, a policy of renovation is urgent not only to meet the demands for higher standards of anti-seismic and energy efficiency, but globally. Nevertheless, there are a number of limits to a global renovation of the existing building stock: (i) technical limits due to the difficulties and often ineffectiveness of conventional solutions when it comes to renovating existing buildings; (ii) organizational limits related to providing accommodation to the building's residents during renovation: in fact, the current seismic and energy interventions generally mean relocating people and furniture to temporary accommodation throughout the duration of the work, which may last up to a year or more; (iii) economic limits due to the high costs of renovation. In addition to these limits, one must also take into account the current economic crisis [2].

Research should, therefore, explore both different solutions to the problems described above and the tools to be used to overcome such limits. Parametric design, the tool proposed and used for this work, makes it possible to overcome various building renovation limits. Indeed, this type of design tool can control many variables at once to obtain a set of solutions among which it is possible to choose the most effective and appropriate ones. This paper also aims to underline that the buildings' aesthetics and design are critical elements that should be linked to the concept of sustainability, as they affect all aspects of human life and experience. Nature has thus become our criterion in this study because it works taking into consideration all variables at play in the realm of sustainability, and parametric design has been used to retrofit buildings to better withstand seismic events while to date, literature shows only cases in which parametric design has been adopted for energy efficiency or architectural solutions. However, the main purpose of this paper is to show how this method works considering different variables, such as geometry, weight, materials, energy efficiency and structures. The relationship between sustainability, nature and the pursuit of beauty will be discussed in Section 2. Section 3 introduces steel exoskeletons as an optimal combination of shape and structure in nature as well as in architecture.

Section 4 presents our approach to the use of exoskeletons, and, in particular, introduces our prototyping algorithm.

Section 5 shows the creation of exoskeletons.

Section 6 illustrates the application of our methodology, tools and software (Figure 2) to a case study. We chose a typical case study representative of the Italian residential stock. The same case study was analyzed in a previous research project and this work is a development of that research. Unlike the latter, here we focused on architectural shape, both in terms of seismic performance and the relationship with the urban context. Finally, Section 7 presents our analysis and results.



Figure 2. Scheme of the proposed methodology, tools and software.

# 2. Aesthetics and Nature as a Reference for Parametric Design

To highlight the unsustainability of unaesthetic architecture and its consequences on society, we should be aware of the beneficial effects that beauty can have on human beings, as proposed by Wohlwill [9]. To do so we draw on the theories of James Hillman, who believes that the quest for beauty and aesthetics may give an answer to many of the problems of contemporary civilization [10]. The search for sustainability should also be the pursuit of beauty. Sustainability, when working in a holistic way with a building, aims at recreating as much of a harmonious relationship as possible with the environment. Good architecture is always a complex activity that establishes a relationship with the environment. In fact, the more relationships with the environment architecture is capable of establishing, the more sustainable it is. Lack of aesthetics and unsustainability, on the other hand, derive from overlooking the critical relationship with the environment. Moreover, not considering environmental issues means neglecting the local seismicity and climate, ignoring the integration of cultural traditions, overlooking the local economy and its anthropological specifications as well as the sense of the living experience in one's own context. Sustainability, like beauty [11–13], depends on the level of complexity that can be achieved through design and it can be seen as the result of a truly holistic approach. Using the concept of sustainability as a measure of beauty does not presuppose a change of paradigm of aesthetics [14,15] but it is an important element of good design. Since design is an intrinsically complex and multidimensional activity, the designer normally tends to reduce complexity in smaller chunks that can be more easily managed. Through parametric design, that tends to imitate the generative algorithms of nature, it is possible to consider the multiple dimensions of design. The goal is to enhance the designing process and optimize results, aware that variables such as aesthetic value, given their intrinsic nature, lend themselves to subjective interpretations. However, through the concept of aesthetics as a value of sustainability, even when the results are not optimal because there is no formula that guarantees optimization, we can address the dilemma between performance and shape and between genotype and phenotype in architecture. The parametric architecture thus conceived is of great help in terms of optimization in relation to the technical parameters pertaining to the sustainability of structural stability [16], energy efficiency, thermal and visual comfort, cost-effectiveness and architectural image.

In our case study and more generally in the context of structural optimization of buildings, we can proceed with three basic types of parametric designs that mimic the modus operandi of nature: the first intervenes on the dimensions of the structural elements, having fixed the shape of the building a priori; the second researches the formal configuration that best resists according to their relative shape; the third intervenes on the arrangement of the structural elements and on the connections between them.

The procedures nature implements, depending on each and every particular situation, develop one or more optimization solutions starting from their genotype. The genotypes nature offers as a reference are manifold. In our case study, exoskeletons seem to be the ones that promise more possibilities for effective solutions.

#### 3. The Exoskeleton: Shape and Structure

The steel exoskeleton, when the characteristics of the building make it possible, is installed outside the building. This is the ideal choice in term of mechanical strength and quantity of materials, which are key features for the construction of new buildings and to retrofit existing ones to improve their anti-seismic performance.

As already stated, among the many limitations highlighted by conventional systems of anti-seismic solutions, emerges the difficulty of enhancing buildings also from an architectural point of view, reducing at the same time the inconveniences residents experience during renovation. For these reasons, the model of the exoskeleton, besides offering a more effective anti-seismic solution is also a support for energy-efficient devices and, we would emphasize, a way redesign the architectural image of the building [17]. Furthermore, the external installation of an exoskeleton does not imply an interruption of the building's operation during the construction works, avoiding temporary relocating costs [18] and other inconveniences caused by large demolition activities, thus contributing to stimulating building owners to retrofit their buildings. As exoskeletons require less major structural intervention and reconstruction activities this is also in accordance with the principles of environmental sustainability and eco-efficiency. In fact, the use of new construction materials and the production of waste are minimized. However, some preliminary requirements are necessary to apply an exoskeleton to an existing building: the presence of free external spaces in front of the buildings' façades is a mandatory requirement for the installation of exoskeletons and their foundations; moreover, the installation of the exoskeleton must be often preceded by local interventions to reinforce the existing beam-column joints, to which the new steel structure will be connected.

Nevertheless, even by using exoskeletons, if we do not proceed by considering the problem of renovation in a global way and we consider only one aspect, we could only produce structural prostheses that do not regard aesthetics in terms of sustainability (Figure 3).



Figure 3. Examples of seismic retrofitting with steel exoskeletons.

Even the example of the exoskeleton used for the seismic retrofit of the Magneti Marelli headquarters in Crevalcore is typical of a non-holistic approach [19] (Figure 4).



Figure 4. Steps of the seismic retrofit of the Magneti Marelli headquarters in Crevalcore, Italy.

Here the adopted exoskeleton converts an office construction built in the 1970s into an earthquake-proof edifice, but this new structure does not integrate with the architecture it intends to strengthen and presents itself as a prosthesis that meets only its structural needs [8]. Moreover, much of the building was demolished and then rebuilt. In fact, engineers spared only the existing reinforced concrete structure and repaired and adapted it to accommodate an external dissipative-type steel frame.

An interesting example in holistic terms is the one by European research project Pro-GET-onE (acronym of *Proactive Sinergy of integrate Efficient Technologies on Buildings' Envelopes*) [8] (Figure 5). The set of solutions tested in this case is part of a holistic view of the problem. The holistic vision promoted by Pro-GET-onE is based on the integration of different technologies with the aim of increasing building performance in terms of energy efficiency, seismic safety and social and economic sustainability. In this case too, the objective is to rely on such external envelopes, which not only make the building earthquake-proof, but also support technologies for renewable energy supplies, especially solar and wind energy. As a result, the architectural transformation, delivers, among the others, a renewed and harmonious image of the case study selected for the verification of the theoretical approach discussed here. In this case, the original reinforced concrete structure was not affected by repairs and betterments.



Figure 5. Case study of the Pro-GET-onE research project.

Among the most used exoskeletons, the diagrid system stands out: a two-dimensional structural system, made up of triangular modules capable of resisting both horizontal and vertical loads without the need for additional elements [20].

From an architectural point of view, the ability of this system to adapt to any three-dimensional shape, even if not straight, offers the possibility to try an infinite number of configurations. On the external structure it is possible to obtain multiple solutions for the façade, from the simplest to the complex technological options available. Moreover, the geometry of the exoskeleton allows for a better management of the relationship between the full and empty spaces of the façade. The choice of materials plays a critical role in the diagrid solution. This choice often falls on steel

because of the advantages it offers, but reinforced concrete diagrids are available too. In fact, steel exoskeletons provide the following advantages: high levels of prefabrication and dry assembly standards, which involve shorter construction times and, therefore, less disruption in urban areas; use of recyclable materials; and easy maintenance and replacement in case of seismic damage or breakage. Finally, a diagrid-type structure requires the use of a smaller amount of material, with significant economic advantages despite the initial higher financial costs. The advantages of using an exoskeleton system are relevant in the case of reinforced concrete constructions built after World War II, which have no historical and artistic constraints thus representing a viable solution, as it allows them to operate, as much as possible from the outside, through the application of a three-dimensional external structure that envelops the existing building, structurally linking itself to it as an exoskeleton. On the other hand, the application of exoskeletons on existing structures may have other disadvantages, for example the exoskeleton rods, if not properly designed, could become an obstacle to natural lighting and ventilation.

Among the advantages, using a diagrid system, or an exoskeleton in general, the only elements subject to demolition and reconstruction are the balconies. This could be, however, the input for the implementation of energy retrofit interventions [8] with the consequent reduction of the thermal bridges, which would be limited only to the structural connections between the building and the exoskeleton. The exoskeleton could be the support for a second envelope which, if properly distanced from the façades of the existing building, can work as ventilated shading walls. The exoskeleton could also provide support for renewable energy production devices, increasingly used as elements for façade composition, as well as vertical gardens that contribute to passive cooling, and solar shadings of any kind for direct radiation control and natural lighting.

Finally, structural advantages are primarily linked to the ability to adapt to the shape of existing structures. The exoskeletons could also be combined with other structural seismic retrofit solutions, but even on their own they can solve the renovation issue. Furthermore, this type of structure ensures a better distribution of loads, thanks to the redundant character provided by the mesh, whose basic modules' geometry is of fundamental importance.

The size of the modules making up the diagrid is usually calculated by dividing in equal parts the height of the building on which it is applied. The number of modules, and consequently the angle in relation to the horizontal plane of the diagonal elements, has a great influence on the general response of the system, and often represents one of the parameters of structural optimization. The basic form of a diagrid usually has a rhomboidal shape. However, in the composition of exoskeletons with diagrid geometry, triangular meshes are preferred for the various advantages they offer both in terms of design and construction phases. The triangle is in fact a non-deformable figure and this gives the structure an intrinsic stability. The designer can also think of the nodes as hinges, which result in less stress on the rods due to the absence of bending moments. A division into triangles, moreover, returns only flat faces, which are more compatible with a possible coating of the structure.

The theme of the renovation of apartment blocks has been addressed taking into account all of the above with an integrated parametric design approach, which is illustrated below. While some studies on parametric design aimed at energy rehabilitation are present in the scientific literature [21], the use of this discipline in the field of seismic retrofit has remained unexplored until now.

## 4. The Prototyping Algorithm

The increasing diffusion of algorithms in design practice has led to the creation of user-friendly tools that can be exploited by all the professional figures involved in this field, regardless of the knowledge of complex programming languages. These tools are, in practice, software designed as graphic editors of algorithms.

The most widespread among the professionals in the field of architecture is Grasshopper [22], capable of interfacing with the three-dimensional modeling software Rhinoceros, and equipped with a vast community that has led to the development of countless faces to increase power and versatility.

Within this software it is possible to build your own script simply by connecting graphic elements (components), which represent the nodes of the parametric diagram and which are characterized by certain inputs and outputs, depending on the function that they perform [23]. The connections are made through wires, graphically characterized according to the structure of the data transported.

In the present work, the Grasshopper software was used to create a procedural algorithm aimed at computation generation and optimization through genetic algorithms of structural steel exoskeletons. Applied externally to reinforced concrete buildings designed to withstand vertical actions alone, they aim to improve the seismic response of the latter, while at the same time characterizing them from an architectural point of view.

The structural treatment within Grasshopper was carried out thanks to the Karamba3d plug-in [24], which uses finite element modeling to perform interactive static analyses on two-dimensional or three-dimensional systems, composed of both rods and shells. This plug-in does not present itself as a "full optional" structural calculation program for carrying out detailed engineering analyses but aims to be a support tool for architects and engineers, guiding the design choices from the initial stages of the process.

The procedural algorithm developed is presented as in Figure 6. This was divided into modules, marked with different colors, each containing the components that perform a specific function. These modules are related to: the geometry of the existing structures (1), the geometry of the exoskeleton (2), the variables of the optimization, i.e., the variations of the exoskeleton geometry, obtained through the optimization process (3), the creation of the finite element model (FEM) (4), structural analysis (5), study of the results and definition of the fitness function for optimization (6), export to SAP2000 (7). The modules concerning the geometry of the exoskeleton and the variables of the optimization represent the "variable" part of the script and therefore are closely linked to the project idea. Their treatment is then referred to in paragraph 6. The remaining modules remain constant and compose the algorithm to develop the shapes of the exoskeleton that have not yet been optimized. Most relevant modules (4, 5, 6) are described in detail in Figure 7.



Figure 6. The prototyping algorithm within Grasshopper.



**Figure 7.** Description of the main modules of the algorithm: finite element model assembly, linear structural analysis and optimization.

First of all, the geometries of the structure are modeled. The inputs to be specified at this stage are the planimetric dimensions, the number of floors, the floor height and the span between columns in both directions. The definition of these parameters allows extending the use of the proposed algorithm to most buildings with reinforced concrete structure. The structure of the existing building (Figure 8a) is not entirely modeled, but is approximated to a series of vertical lines, placed one on top of the other, in a number equal to that of the levels. Each of these lines represents the axis of a column with equivalent stiffness to that calculated on each storey. At the top of each of these lines lies a surface that represents the floor slab, surrounded by a system of linear elements, later made non-deformable, useful to avoid inconvenient deformation and torsional phenomena. Within this module it is also necessary to obtain the geometries related to the beam-pillar nodes of the existing structure, which represent the points of connection with the new steel exoskeleton.

The outputs of the geometric modeling flow into the module for the creation of the finite element model, visible in relation to the existing structure in Figure 8b. This section is divided into three groups. The first has the task of converting the geometries into structural elements. In particular, we need linear elements as starting geometries for the creation of beams, of two-dimensional mesh elements

for the creation of shells. In this phase, information relating to the materials is included: concrete is assigned to the existing structure, while steel is assigned to the exoskeleton. Here, also the cross sections assigned to the pillars of the existing structure are defined, which are useful to simulate the equivalent stiffness rate provided by the reinforced concrete building and drawn backwards from an FEM built in the SAP2000 environment.

The other two groups of the module in question are aimed at defining the supports, for creating the joints at the base and the carriages (useful to simulate a deformed beam attributable to a "shear-type" frame (Figure 8c) placed at the top of the pillars with equivalent stiffness, and loads.



Figure 8. Simplified modeling of the existing reinforced concrete structure. (a) geometries, (b) finite element model, (c) deformed beam attributable to a "shear type" frame.

Although related to a seismic problem, the evaluations related to the stresses, in this phase, have been dealt with by adopting static simplifications. Despite this, the algorithm was not created to optimize an eternally static condition: the results of this process, in fact, will then be used in a dynamic treatment. The loads involved are, therefore, represented by equivalent seismic actions acting on each individual floor of the structure, obtained through a script created ad hoc, according to the project spectrum and to the fundamental period of the structure. These last two parameters must be inserted as inputs and vary according to the geographical location of the existing building and its structural characteristics.

Once the finite element model is assembled, it is subject to a linear static structural analysis, carried out within Grasshopper thanks to the Karamba3d plug-in. In this step, simultaneously with the structural analysis, a first process of optimization of the cross sections of the structural elements is carried out: on each beam of the exoskeleton, to which the minimum section, chosen in a range of user-defined profiles, is assigned according to a verification of resistance in the elastic field, combined, in the case of elements subjected to compression, on an instability test. Here, the solver is required to draw from a range of commercial tubular profiles with a diameter of approximately 30 cm. If none of the profiles contained in the chosen range are verified, the algorithm returns an error message.

The results of the linear static structural analysis, expressed in terms of nodal displacements and stresses on the rods, flow into the optimization module. The optimization process is carried out using genetic algorithms, that represent a tool that use Darwinian theories of evolution to solve problems related to parametric design [25]. The optimization is based on a fitness function, built by the user, which returns a numerical value that defines the objectives of the process. In this case, the solver, the Octopus plug-in, starts a solution generation routine obtained by varying the intrinsic parameters of the model, located in the module related to variables definition. Among these solutions, only those that return a better fitness value than the previous ones survive.

The considered algorithm is aimed at minimizing the weight of steel used, keeping the totality of the structural elements of the exoskeleton under verification conditions and maintaining the relative drifts between the levels of the existing reinforced concrete building, below the limits, reduced to take into account the life safeguard limit [26]. In order to pursue these objectives, the fitness function is obtained by increasing to the actual weight of the exoskeleton steel, of suitably calibrated coefficients,
which are different from zero only in cases where there are drift values beyond the limits, or unverified beams. In this way, the configurations in which these conditions occur, tend to be discarded by the solver (the Octopus plug-in in this case), due to the higher fitness value that characterizes them.

The last of the modules that make up the algorithm is related to the export of the FEM model from Grasshopper to SAP2000, for the development of detailed dynamic analyses. For this operation to be possible, the GeometryGym plug-in [27] is used. The dialogue between the software handled in this work, exploits the API interface integrated in SAP2000 [28], able to provide complete and efficient access to all the tools of the structural software, allowing a bidirectional transfer of the models and control on the execution of the analysis and on the management of the data. A dialogue of this type between the software allows one to bypass the conventional user interface once again, and to speed up the export and import phases, especially in case of repeated modifications to the model.

#### 5. The Creation of Structural Exoskeletons

Summing up, once you have modeled the geometry related to the structure equivalent to the existing building, you can move on to create the elements related to the exoskeleton and to the definition of the optimization variables. These are nothing but the possible morphological transformations that the model can undergo during the search for the solution capable of using the lowest amount of steel. It should, therefore, be noted that the geometry of the exoskeleton modeled in this phase does not yet represent the final shape of the latter, but a starting configuration that will be manipulated automatically during the optimization phase.

Together with the vectors inherent to the equivalent seismic actions, the totality of the geometries represents one of the inputs for the assembly of the finite element model, on which a linear static analysis is then carried out. The results of this become the parameters that make up the fitness function, built by the user to define the objectives of the process. In order to achieve a fitness value congruent with the objective, the solver will make changes to the geometry of the exoskeleton (allowed by the user via the upstream imposition of the variables), and then re-analyze the variant created. In this way a routine is generated will end only when the numerical value of the fitness function is suitable to the objectives of the optimization. The process of generating structural exoskeletons is summarized in Figure 9.



Figure 9. Flowchart that summarizes the process of generating structural exoskeletons.

It should be noted that the algorithm developed during the present work is not aimed at the seismic improvement of a specific building but, due to the parametric character that distinguishes it, its use can be extended to any existing multi-storey building with a framed structure in reinforced concrete. By varying the size of the sections assigned to the pillars that simulate the existing structure, it is possible to manage the stiffness equivalent to the real one, while the number and shape of the slabs, and the positioning of the beam-pillar nodes, can be adapted to the building by modifying the geometric values that represent the input for their creation. Likewise, the forces applied in assembling the FEM vary depending on the geographic location of the building and its natural oscillation period.

#### 6. Application to a Case Study

As a case study, we adopted a prototype building consisting in an apartment block, already analyzed in a previous research project (Figure 10) [26,29].



Figure 10. Three-dimensional model of the simulated building.

The structure of this prototype building has no real evidence but is obtained from a process called "simulated design", based on the regulations in force before the introduction of the seismic and energy-efficiency legislation, on the construction uses of the time, and on the architectural peculiarities that distinguished the building boom between 1950 and 1980. This prototype is representative of a large share (almost 20%) [6] of the existing residential stock in the city of Catania, located in the eastern coast of Sicily and classified as seismic-prone in 1981. From a morphological point of view this building type is characterized by medium height, between 5 and 12 floors, elongated rectangular plant, central stairs and strong regularity both in plan and in elevation. The structure, characterized by resistant elements placed in only one direction and frames symmetrically arranged, is able to withstand only vertical loads. To aggravate its behavior under horizontal loads there is the lack of masonry infill on the ground floor, which may cause a soft story collapse [26,29].

More in detail the considered case study is characterized by a reinforced concrete frame. It has 10 residential floors, and a commercial ground floor. The building is 35 m high and has a rectangular plan ( $31.80 \times 13.80$  m), symmetrical along the transversal axis. On each floor there are three apartments, each of which has an area of approximately 140 m<sup>2</sup>. The two external apartments are perfectly mirrored, with the central one interposed between them. Most windows are on the long sides and only two small windows per floor are on the short fronts.

The elevations reflect the symmetry of the plan, and are characterized by a strong repetitiveness in height, due to the identical configuration of the spaces between the various levels.

On each floor there are balconies, placed symmetrically along the two axes. Each apartment therefore has a portion of its balcony and can enjoy an outdoor space.

Using a prototype building as case study is beneficial to perform structural analysis. Results, considerations and possible retrofit solutions may be replicated, in fact, on most apartment blocks having the same structural and constructive characteristics. However, to set up relations with the specific urban landscape and seismic area, it was decided to place the case study inside the so-called "sea zone" of Catania, a prestigious neighborhood built between the 1960s and the 1970s, during the economic and construction boom. The building stock of this area was largely realized before the enforcement of regulations for energy-efficient and seismic-resistant buildings, and the construction practice strongly clashes with the "rules of art", which up to some decades before, even in the absence of a theoretical awareness, allowed buildings to be respectful of the environment and often resistant to the earthquakes [30].

The selection of the site took place considering the characteristics of the lot in which it is placed as the primary criterion. In particular, in view of the construction of an external structure, we need a free space, on all fronts, to be used for laying foundation works. The choice fell on a large green area near the urban waterfront (Figure 11).



Figure 11. Rendering of the simulated building located in the selected site.

The advantages of the parametric approach to design include the possibility of easily creating multiple versions of the same project. Borrowing the term *versioning* from the jargon of software development, this indicates the opportunity offered by the management of files that are no longer static, but capable of evolving and generating variations when new forces and conditions come into play.

Two project ideas are proposed, each of which, due to the intrinsic nature of the parametric approach of generating multiple versions of the same concept, will return more than one final form in varying the parameters within the model and the definition of structural optimization. In a design of this type the design idea does not translate directly into the final shape of the exoskeleton, but in a starting geometric shape and in a series of variables that represent "what will happen to the model" during the optimization process.

The first project idea stems from the desire to clash, in order to cancel them, with the symmetry and regularity that characterize the considered building. Therefore, this leads to the creation of a strongly irregular structure-envelope, which, like a beneficial parasite, enters into symbiosis with the original structure, attacking it only where necessary and allowing the emergence of glimpses, through which to continue observing the existing building.

The first step is the one related to the creation of the starting form of the exoskeleton, which, once subjected to an optimization process according to rules imposed upstream, will return the final geometry of the structure. The starting structure is represented by a diagrid, of regular shape, that wraps the building parallel to its façades. The basic module of this structure, quadrangular with height equal to the floor (3.3m), is divided into four triangles deriving from the connection of the opposite nodes. The division into triangular modules enables the creation of a structure whose nodes are of the hinge type. Although more difficult to achieve in practice, these allow us to reduce the stresses on the rods to only axial stress, eliminating the bending momentum rates. In order to obtain a perfect closure of the mesh, at the corners of the building, the reticular system has been moved away from the façades of a distance equal to half the light of the span (2.25m). This is also convenient to avoid a "cage" effect, which risks generating feelings of oppression in the inhabitants of the building itself.

The starting diagrid system is modeled from four surfaces, each of which is parallel to a façade and divided into rectangular sub-areas measuring  $3.3 \times 2.25$  m. Each of these has been divided into the four triangular surfaces whose sides represent the linear elements to be converted into rods (Figure 12).



Figure 12. Geometries of the starting diagrid related to the first design idea.

As for the optimization variables, the model just described is allowed to "curl up" thanks to the movement, with respect to the initial position, in the range from one meter to the outside, one meter inwards, orthogonally at the respective façade, of the diagrid nodes. This transformation granted to the model is aimed at finding a configuration that returns the best resistance, exploiting the shape, to seismic actions. The movement of the nodes is accompanied by a process of removing the rods that do not contribute to the response to horizontal loads, in order to pursue the objective of minimizing the steel used. The strategy adopted for the removal of the rods is as follows: thinking of the building as a box inside which they have freedom to move points, in a number chosen by the user. Each of these represents the center of a sphere (Figure 13) capable of removing the triangular modules of which it incorporates the barycenters. The position of the spheres and their radius, as well as the value relative to the displacement of the nodes, fall within the optimization variables, and are therefore parameters that are automatically controlled by the solver of genetic algorithms, and in any case set manually by the 'user. The module for the geometric modeling of the exoskeleton also generates, automatically, and depending on the portions of the exoskeleton that are not removed, the connecting elements between the nodes of the new structure and those of the existing one. It was decided to operate with 20 spheres, each with a variable radius of 1 to 5 m.



**Figure 13.** Spheres, controlled by the solver of genetic algorithms, responsible for removing the modules during the optimization process.

The second project idea involves the creation of an exoskeleton which, while remaining regular in shape, aims at eliminating the symmetry that characterizes the existing building. While in the previously described solution, the solver of genetic algorithms is called to carry out a rod removal process, this time it aims to implement an additive strategy. The optimization, in fact, involves the insertion of structural elements in the areas of the exoskeleton that require greater rigidity.

Even in the case of the second design idea, the starting structure is a diagrid system, this time with quadrangular base module (Figure 14) and therefore formed by rods stuck to the ends. Also in this case, the structure is moved, on each front, at a distance equal to half of the light of the span (2.25 m) to allow for an optimal closure of the angular modules. Similarly to the first design idea, the geometric modeling involves the creation of four surfaces parallel to the façades, each of which is divided into rhomboid portions of height equal to the storey height (3.3 m) and width equal to the light of the span (4.5 m). The sides of these diamonds represent the linear elements to be converted into rods in the phase of creating the finite element model.



Figure 14. Geometries of the starting diagrid related to the first design idea.

The only transformation granted to the shape of this model consists in the possibility of each single rhomboidal module, to subdivide, in the optimization phase, into an additional four or 16 sub-modules, in order to obtain an increase in stiffness in the exoskeleton areas, under more stress caused by horizontal actions. Also in this case, the number of subdivisions represents one of the parameters of the optimization, and is automatically modified by the solver of genetic algorithms. As in the development of the first design idea, the portion of the algorithm relating to the modeling allows for the automatic generation of the connecting elements between the exoskeleton and the existing structure in reinforced concrete.

#### 7. Analysis and Results

The results are now illustrated and compared, in numerical and geometrical terms, returned by the optimization processes carried out. As already mentioned above, the configuration of the exoskeleton obtained is not unique. The approach used, in fact, provides the possibility to choose among several variants, deriving from the modification of the model parameters and the objectives defined in the fitness function.

In the present work, the first distinction is feasible since two different design ideas are taken into consideration. For both, an optimization process was carried out aimed at minimizing the weight of steel used and maintaining relative drift below the limits reported in Table 1. These are calculated by considering two thirds of the last minimum rotation of each storey of the concrete structure to obtain the value at the life-saving limit state.

Storey	h	Dlim	Dmax	2/3 Dmax
	[m]	-	[m]	[m]
1	4.3	0.003	0.0129	0.0086
2	3.3	0.003	0.0099	0.0066
3	3.3	0.003	0.0099	0.0066
4	3.3	0.003	0.0099	0.0066
5	3.3	0.003	0.0099	0.0066
6	3.3	0.003	0.0099	0.0066
7	3.3	0.003	0.0099	0.0066
8	3.3	0.003	0.0099	0.0066
9	3.3	0.003	0.0099	0.0066
10	3.3	0.003	0.0099	0.0066

Table 1. Displacement limits.

It was also required to verify all the rods by following the assignment of a cross-section included in a pre-established range. The solution obtained by imposing these conditions on the first design idea (variant 1), shown in Figure 15a, is not satisfactory since the removal of the rods is minimal. This is maybe caused by an incorrect interpretation of the problem by the genetic algorithm solver, and returns a design variant that is capable of maintaining the relative displacements below the limits. The result is a naked structural cage without any relationship with the context. In fact, the cage is neutral with respect to the surrounding environment: an indifferent isotropic metal structure that does not favor any relation. The minimization of steel weight resulting from this optimization process is also minimal, and for this reason this solution cannot be considered optimal. For this reason, this solution was compared with an equivalent that provided for a modification of the fitness function. Therefore, it is necessary for the optimization to point to the minimization, not more than the weight, but of the number of rods used. The exoskeleton obtained (variant 2), visible in Figure 15b, despite keeping the displacements below the imposed limits, most likely presents problems related to the earthquake response and it is therefore not adequate from a structural point of view. The elimination of rods in the angular zones of the building induces, in fact, an accentuation of the torsional response of the building-exoskeleton complex subjected to the earthquake. In order to avoid excessive irregular configurations, the movement of the 20 spheres capable of removing the diagrid modules has been limited. First extended to the whole building, it now takes place exclusively in the central area, ensuring the integrity of the angular zones of the exoskeleton. This solution (variant 3) is illustrated in Figure 15c. This optimization process led, in this case, to a considerable reduction of the weight of steel compared to the initial configuration, while maintaining relative displacements below the limits and a regular configuration in the corner areas of the building. Variant 3 is therefore characterized by a good balance between structural behavior and use of materials. It is interesting to analyze the solution (variant 4) obtained without imposing symmetry (Figure 15d), considering limit displacement values, greater than previous solutions. Given the fewer restrictions, the solver is given more freedom in removing the rods. This variant is purely aimed at experimentation and, given the exceeding of the values allowed in terms of displacements, certainly does not return an adequate structural behavior.

Numerical data about each variant, in terms of steel weight, rods number and relative displacements, are compared in Table 2.

Two distinct optimizations were carried out also for the option with embedded quadrangular modules, with the objective of minimizing the weight (variant 5) and the number of rods (variant 6). Because of the initial configuration, the resulting geometries, shown in Figure 15e, f, do not present substantial differences in terms of the overall image of the system, and they both harmoniously relate to the existing building. In contrast to the previous design idea, in this case the definition of the fitness function that aims at minimizing the weight, shows better results.



**Figure 15.** Variants obtained by modifying the parameters of the geometric models and the definition of the fitness function.

		DESIGN IDEA 1			DESI	GN IDEA 2	2		
		Pre- Optimisation	Version 1	Version 2	Version 3	Version 4	Pre- Optimisation	Version 5	Version 6
Weight	kg	336,134	319,900	284,133	269,704	206,155	103,512	225,200	229,124
Rods Number	-	3376	3201	2818	2673	2005	616	4456	4536
u1	m	0.0076	0.0074	0.0071	0.0076	0.0060	0.0100	0.0082	0.0083
u2	m	0.0048	0.0050	0.0058	0.0060	0.0086	0.0185	0.0066	0.0066
u3	m	0.0042	0.0047	0.0061	0.0054	0.0068	0.0266	0.0059	0.0060
u4	m	0.0044	0.0049	0.0063	0.0059	0.0071	0.0355	0.0063	0.0064
u5	m	0.0044	0.0053	0.0064	0.0053	0.0075	0.0457	0.0065	0.0066
u6	m	0.0042	0.0049	0.0065	0.0060	0.0078	0.0541	0.0065	0.0066
u7	m	0.0040	0.0045	0.0064	0.0056	0.0096	0.0661	0.0064	0.0064
u8	m	0.0037	0.0043	0.0058	0.0058	0.0099	0.0701	0.0054	0.0053
u9	m	0.0032	0.0037	0.0051	0.0052	0.0096	0.0848	0.0048	0.0048
u10	m	0.0026	0.0030	0.0048	0.0062	0.0070	0.0797	0.0040	0.0040

Table 2. Comparison between the versions in terms of steel weight, rods number and floors displacements.

Among all the obtained variants, numbers 3 and 5 are considered the best because of their structural responses in terms of displacements, obtained using a limited number of structural elements and a lower weight of steel compared to the other solutions.

The models of these variants were then exported within the SAP2000 finite element software to perform dynamic structural analyses of the existing building-exoskeleton assembly, confronting them with the bare building model. It should be noted that all data relating to materials, cross-sections of structural elements, nodal constraints and degrees of freedom of the rods are transferred automatically using the GeometryGym plug-in. In the SAP2000 environment we proceed exclusively to the introduction of the inputs for modal analysis and linear dynamic analysis.

The two variants, topologically different, but deriving from the same optimization process, respond in a similar way to the dynamic loads. From the modal analysis, a substantial increase in stiffness is denoted, as expected, which causes a reduction of the fundamental periods of oscillation (Figure 16). At the same time, from the examination of the results of the linear dynamic analyses, what emerged was that neither of the two geometric solutions obtained from the exoskeleton is able to maintain the displacements of the existing building within the fixed limits, even if, analyzing the overlap of the displacements over time, of the same point, for all the seven accelerograms chosen [26], a notable reduction of the maximum mean value can be seen. From the 400mm of the actual state, it goes to about 160mm in the project states (Figure 17). Although these are to be considered rough results, as deduced from a linear analysis applied to a problem that, with reference to the existing structure, is markedly non-linear, the results obtained glimpses of a positive response of the generated system.



**Figure 16.** Comparison, using design spectrum, between fundamental periods of the bare building and proposed solutions.



Figure 17. Comparison between the bare building and the two proposed solutions. Displacement diagrams in the most stressed direction. According to NTC18, the response in average terms is considered.

Among these are renderings (Figure 18) that show how further variations can be obtained by completing the exoskeleton with non-structural elements, such as shielding and panels of all kinds, vegetation and solar modules.



Figure 18. Renderings of the selected variants completed with non-structural elements.

#### 8. Conclusions

The urgency of improving the existing building stock in terms of seismic vulnerability offered us the opportunity to reflect both on the structural aspects and on the architectural image. Thus, the present work has tried to highlight the need to consider aesthetic issue as an important—albeit often neglected—dimension of the search for sustainability. Even if the concept of beauty is hard to tackle, almost impossible to objectify and impossible to be fully reached, it can be considered as a research horizon. And the issue of sustainability offers the opportunity to approach aesthetic research: an opportunity that should be seized since beauty affects all aspects of human life and experience.

The proposed solutions are just proof of the concept of a holistic renovation strategy. In fact, these do not represent the best solutions from an aesthetic point of view. They just demonstrate how, with parametric design, it is possible to produce many complex solutions easily and to choose that which better interprets the aesthetic canons selected by the designer.

In more detail, we have proposed the use of exoskeletons for the seismic retrofit of apartment blocks with reinforced concrete frames. Since the exoskeleton is applied from the outside, it has the advantage of not interfering with the regular operation of the building during the renovation works.

The adopted parametric approach allowed us to consider many variables (geometry and shape, weight, materials, structural elements, forces), leading to a set of solutions that contemplate a multiplicity of aspects of sustainable renovation. In fact, since the early architectural design stages, the creative process has integrated both form and structural functions, as well as information relating on materials and stress response.

This approach is a single tool that offers the possibility to generate multiple retrofitting alternatives based on variables drawn from multiple disciplines, opening the door to a great variety of integrated solutions. In the case presented, we verified the adaptability potential of different types of exoskeleton, which change their shape according to the variability of the external conditions: as occurs in the evolutionary process, reproduction and mutation operate on the genotype, while competition and selection occur at the level of the phenotype. In fact, the use of exoskeleton genotype in this work starts from the simple structural cage to achieve an elaborate structural texture on the façade: an exoskeleton with asymmetrical balance between empty and full spaces. Furthermore, the structure opens (with empty spaces) towards the most interesting landscape points of view and closes (with full ones) to protect itself from unfavorable climatic conditions. In this way we try to establish further relations with the environment.

Such a tool has taken the form of a genetic algorithm, within the Grasshopper graphic editor, aimed at generating and optimizing the form of the steel frames, based on their structural response to equivalent static seismic stresses. The generated geometries are optimized to keep the relative drifts below the limits, to verify all rods, and to minimize the steel weight and the number of rods. Both these parameters were decisive for choosing the best options among the multiple solutions obtained according to different design concepts, geometric inputs and variables of the objective function. These variables have been subjected to linear dynamic analysis that has shown a considerable

reduction of the structure displacements in comparison to the actual state. Due to the non-linear nature of the problem, the presented linear, static and dynamic analyses are to be considered preliminary. More detailed results could be expected from a non-linear analysis, thanks to the resistance contribution of steel in the plastic field.

Finally, the proposed solution, although limited to buildings whose footprint can be slightly enlarged, overcomes many of the limitations usually associated with seismic renovation actions including: (i) technical limits, because exoskeletons effectively minimize the renovation activities on the existing building and allow combining multiple interventions in a single stage (structural, energy efficiency, and architectural retrofitting); (ii) organizational limits, since occupants do not need to leave their apartments during renovation works, avoiding transfer disruption and costs, and facilitating condominium consensus; (iii) economic limits, since exoskeletons may reduce time and cost for renovation in comparison with other combined interventions.

The next development of this study will pursue the extension of the algorithm functionalities, introducing parameters related to thermal comfort and energy efficiency among the factors that influence the selection of the optimal exoskeleton shape. The introduction of new variables may further improve the holistic dimension of the proposed methodology, as well as its complexity and completeness.

Author Contributions: Conceptualization, aestethic issue and supervision, S.D.; introduction and state-of-the-art, S.D.; aesthetics and nature as a reference for parametric design, S.D.; the exoskeleton: shape and structure, S.D.; the prototyping algorithm, B.C.; the creation of structural exoskeletons, B.C.; case study, S.D. and B.C.; application to a case study, B.C.; analysis and results, S.D. and B.C.; conclusions, S.D.

**Funding:** This work has been funded by the University of Catania within the project "Piano della Ricerca Dipartimentale 2016-2018" of the Department of Civil Engineering and Architecture.

**Acknowledgments:** The authors would like to thank Ivo Caliò and Giuseppe Occhipinti, who supervised the structural aspects, offering a precious and relevant contribution. The authors would also like to thank Nikkei Architecture for the permission to use the last image of Figure 3.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- European Environmental Agency. Urban Sprawl in Europe. The Ignored Challenge; Report 10; EEA: Copenhagen, Denmark, 2006; ISSN 1725-9177.
- La Greca, P.; Margani, G. Seismic and Energy Renovation Measures for Sustainable Cities: A Critical Analysis of the Italian Scenario. *Sustainability* 2018, 10, 254. [CrossRef]
- ISI Ingegneria Sismica Italiana. Available online: https://www.ingegneriasismicaitaliana.com/it/24/ normative/ (accessed on 22 August 2018).
- Statistiche Istat. Censimento Popolazione Abtiazioni. Available online: http://dati-censimentopopolazione. istat.it/Index.aspx (accessed on 22 August 2018).
- Pasca, M. Il costruito italiano: Tipologie, problematiche, interventi pre e post sisma. *Tafter J.* 2012, 50. Available online: http://www.tafterjournal.it/2012/08/01/il-costruito-italiano-tipologie-problematicheinterventi-pre-e-post-sisma/ (accessed on 22 August 2018).
- Paradisi, R. I mostri di Ostia, le ceneri di Pasolini. Da Pier Paolo Pasolini, profeta del disincanto delle borgate, alla condizione di regresso cui assistiamo oggi in certe periferie. Una riflessione sul territorio italiano. *Domus*. 20 December 2017. Available online: https://www.domusweb.it/it/opinion/2017/12/20/i-mostri-di-ostiale-ceneri-di-pasolini.html (accessed on 22 August 2018).
- Zanfi, F. The Città Abusiva in Contemporary Southern Italy: Illegal Building and Prospects for Change. Urban Stud. Int. J. Res. Urban Stud. 2013, 50, 3428–3445. [CrossRef]
- Ferrante, A.; Mochi, G.; Predari, G.; Baldini, L.; Fotopoulou, A.; Semprini, G. A European Project for Safer and Energy Efficient Buildings: Pro-GET-onE (Proactive Sinergy of integrate Efficient Technologies on Buildings' Envelopes). Sustainability 2018, 10, 812. [CrossRef]
- 9. Altman, I.; Wohlwill, J.F. Human Behavior and Environment; Springer: Boston, MA, USA, 1976.
- 10. Hillman, J. City and Soul; Spring Publications: Putnam, CT, USA, 2006; ISBN 9780882145778.

- 11. Chandrasekhar, S. Truth and Beauty: Aesthetics and Motivations in Science; The University of Chicago Press: Chicago, IL, USA, 1987.
- 12. McAllister, J.W. Beauty & Revolution in Science; Cornell University Press: Ithaca, NY, USA; London, UK, 1996.
- 13. Bodei, R. Le Forme Del Bello; Il Mulino: Bologna, Italy, 1995.
- 14. Eco, U. Storia Della Bruttezza; Bompiani: Milano, Italy, 2010; ISBN 9788845265280.
- Rosenkranz, K. Aesthetics of Ugliness; Bloomsbury USA Academic: New York, NY, USA, 2015; ISBN 9781472568854.
- Greco, A.; Cannizzaro, F.; Pluchino, A. On the use of genetic algorithms to assess the seismic resistance of planar frame structures. In Proceedings of the COMPDYN 2017 6th ECCOMAS Thematic Conference on Computational Methods in Structural Dynamics and Earthquake Engineering, Rhodes, Greece, 15–17 June 2017.
- Pohl, G.; Nachtigall, W. Biomimetics for Architecture & Design. Nature. Analogies. Technology; Springer: Cham, Switzerland; Heidelberg, Germany; New York, NY, USA; Dordrecht, The Netherlands; London, UK, 2015.
- Marini, A.; Passoni, C.; Belleri, A.; Feroldi, F.; Preti, M.; Metelli, G.; Riva, P.; Giuriani, E.; Plizzari, G. Combining seismic retrofit with energy refurbishment for the sustainable renovation of RC buildings: A proof of concept. *Eur. J. Environ. Civ. Eng.* 2017. [CrossRef]
- Inarcos. I Segni Della Ricostruzione Post Terremoto Maggio 2012: L'adeguamento Sismico Della Palazzina Uffici Nello Stabilimento Magneti Marelli di Crevalcore, Inarcos, Ingegneri Architetti Costruttori. March 2014, pp. 53–62. Available online: http://www.teleios-ing.it/UserFiles/File/FRANCESCHINI/Articolo-Inarcos738\_Ricostruzione%20terremoto%202012\_Franceschini-Semproli-Secci.pdf (accessed on 22 August 2018).
- 20. Boake, T.M. *Diagrid Structures, Systems Connections Details;* BirkhÄuser: Berlin, Germany, 2014; ISBN 9783038215646.
- Kerdan, I.G.; Raslan, R.; Ruyssevelt, P. Parametric Study and Simulation-Based Exergy Optimization for Energy Retrofits in Buildings. ECOS. 2015. Available online: https://www.researchgate.net/publication/ 279767492\_Parametric\_study\_and\_simulation-based\_exergy\_optimization\_for\_energy\_retrofits\_in\_ buildings (accessed on 22 August 2018).
- 22. Rutten, D. Grasshopper-Algorithm Modeling for Rhino. Available online: http://www.grasshopper3d.com/ (accessed on 22 August 2018).
- 23. Tedeschi, A. AAD—Algorithm Aided Design: Parametric strategies using Grasshopper; Edizioni Le Penseur: Brienza, Italy, 2014.
- 24. Preisinger, C. Linking Structure and Parametric Geometry. Archit. Des. 2013, 83, 110–113. [CrossRef]
- Rutten, D. Evolutionary Principles Applied to Problem Solving. Available online: https:// ieatbugsforbreakfast.wordpress.com/2011/03/04/epatps01/ (accessed on 22 August 2018).
- Calio, I.; Marino, E.; Rossi, P.P.; Contrafatto, F.; Barbagallo, F.; Bosco, M.; Occhipinti, G. Definition of the Pilot Building (ANCE | Catania Project 2016); ANCE | Catania: Catania, Italy, 2017.
- 27. Mirtschin, J. GeometryGym. Available online: http://geometrygym.blogspot.com/ (accessed on 22 August 2018).
- 28. SAP2000. Available online: http://www.csi-italia.eu/software/sap2000/ (accessed on 22 August 2018).
- Occhipinti, G.; Caliò, I.; Izzuddin, B.; Macorini, L. Seismic Assessment and Rehabilitation of Existing RC Buildings Not Designed to Withstand Earthquakes. Ph.D. Thesis, Università degli Studi di Catania, Catania, Italy, 2017.
- 30. Margani, G. La Riqualificazione Dell'edilizia Residenziale Del Secondo Dopo Guerra: La "Zona A Mare" Di Catania; Aracne: Roma, Italy, 2014; ISBN 978-88-548-7585-2.



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





### Article Reducing Seismic Vulnerability and Energy Demand of Cities through Green Infrastructure

#### Riccardo Privitera and Daniele La Rosa \*

Department of Civil Engineering and Architecture, University of Catania, Via S. Sofia 64, 95123 Catania CT, Italy; riccardo.privitera@darc.unict.it

\* Correspondence: dlarosa@darc.unict.it; Tel.: +39-095-7382523/8

Received: 12 June 2018; Accepted: 20 July 2018; Published: 24 July 2018



Abstract: Historically, urbanization processes in Italy resulted in built environments with high levels of seismic vulnerability, low energy efficiency and a lack of green spaces. The latter represent the main providers of ecosystem services in cities and play a relevant role in reducing the effects of climate change by the regulation of microclimate and urban heat islands that are responsible for building energy consumption. Despite their importance in providing ecosystem services, the implementation of green infrastructure challenges limited financial resources for the public acquisition of private plots. This paper proposes a strategy to implement an urban green infrastructure aimed at generating a double positive effect on cities by triggering seismic retrofitting and the reduction of cooling energy demand of the existing urban fabric. This is proposed through a transfer of development rights program where landowners gain economic incentives to adopt seismic retrofitting interventions and, at the same time, public administrations implement the green infrastructure in the portion of areas transferred to the municipality. The energy efficiency of buildings closer to the green infrastructure, therefore, benefits from the cooling effects of this new greenery. The strategy is tested under different scenarios of acquisition of private land by public administrations in the metropolitan area of Catania (Italy).

Keywords: green infrastructure; seismic retrofitting; energy retrofitting; ecosystem services; urban planning

#### 1. Introduction

In urban contexts, there is a growing interest in using and deploying natural ecosystems to provide solutions to several urban issues and improve the overall sustainability of urban environments [1]. These nature-based solutions provide sustainable, cost-effective, multi-purpose, and flexible alternatives for various planning objectives and can significantly enhance the resilience of cities [2]. Furthermore, by reshaping the built environment, nature-based solutions can enhance the inclusivity, equitability, and livability of cities, regenerate deprived districts through urban regeneration programs, improve mental and physical health and quality of life for citizens, reduce urban violence, and decrease social tensions through better social cohesion (particularly for some vulnerable social groups, such as children, the elderly, and people of low socio-economic status) [3].

Among nature-based solutions, green infrastructure is a natural, semi-natural and artificial network of multifunctional ecological systems within, around and between urban areas, at all spatial scales [4]. This definition emphasizes the holistic ecosystem vision of urban environments (including the abiotic, biotic and cultural functions) and claims for multi-scale approaches able to take into account the scale-dependent relationships of ecological processes occurring in cities, with particular reference to human health and the well-being of citizens and residents.

Green infrastructure can comprise several urban ecosystems including parks and woodland, blue areas (lakes and streams), greenery, semi-natural areas and other urban features such as green roofs and street trees. In particular, for high-density urban contexts, green infrastructure aims at the following actions: (i) environmental protection and integration of agriculture into urban context, providing specific new agricultural land-use types such as agricultural parks, community supported agriculture, and allotment gardens; (ii) development of suburban green areas in order to provide a more equal distribution of public parks and gardens; and (iii) enhancement of current urban green spaces by improving quality, usability, and accessibility [5]. Ecosystems included in the green infrastructure provide important functions and relative services, such as CO<sub>2</sub> sequestration, production of O<sub>2</sub>, reduction of air pollutants and noise, regulation of microclimate and heat island effect, flood damage reduction, filtering water, pollination and supply of recreational value, and play a fundamental role in health, well-being, and social safety [6–8].

This paper focuses on the services of regulation of a microclimate, mainly provided by urban vegetation [9]. Urban vegetation can bring beneficial microclimatic effects, including air temperature reduction, which eases the urban heat island effect. The microclimatic benefits of trees are obtained through several physical processes [10,11]: (1) solar heat gains on windows, walls, roofs, and urban surfaces, including human bodies, are lowered through shading; (2) the buildings' long-wave exchanges are reduced at lower surface temperatures through shading; (3) the dry-bulb temperatures are lowered through evapotranspiration processes; and (4) latent cooling is increased due to the addition of moisture to the air through evapotranspiration. Estimating the cooling effect that can be obtained with different configurations of new greenery is, therefore, a crucial step in enhancing the regulating capacity of urban environments and therefore planning cities more resilient to climate change.

## 2. Implementation of a Green Infrastructure for Reducing Seismic Vulnerability and Enhancing Energy Efficiency

#### 2.1. Challenges for Planning a Green Infrastructure

Despite its importance in providing the aforementioned ecosystem services, green infrastructure faces several difficulties and limitations for its real development and management, especially in dense urban contexts. The most relevant of these challenges is the public acquisition of the land where green infrastructure will be designed and implemented. This is mainly due to the lack of available areas belonging to municipalities or other public bodies that could be used to develop the green infrastructure. This may dramatically hamper the economic feasibility of implementation, as the direct public acquisitions of land may be economically unsustainable for local administration and often face resistance from private landowners [12].

The issue of economic feasibility for managing public intervention and providing accessible public green spaces could be addressed through incentive-based approaches, including the transfer of development rights (TDR) [13,14]. TDR allows the sale and transfer of development rights from a specific parcel of land to other properties. Future use of the original parcel is then protected from development by a permanent conservation easement or deed restriction prohibiting development. A TDR programme defines an area to be protected from development (sending area) and one where development will be allowed to occur (receiving area) [15,16]. Landowners can transfer the rights to develop one parcel of land to another. As a consequence, the parcel from which the development rights are being transferred can no longer be developed, or developed only in a limited way [15]. Private landowners will get economic incentives from selling developing rights and the opportunity to adopt retrofitting measures aimed at the reduction of seismic vulnerability. This issue can be convenient for districts characterized by multi-storey apartment buildings and shared open setback yards.

#### 2.2. Seismic Vulnerability and Energy Efficiency of Existing Urban Fabric

The assessment of seismic risk of buildings is a complex task requiring the contributions of different scientific fields (from social science to engineering and economic sciences) to evaluate the three main parameters of exposure, vulnerability and seismic hazard [17,18].

Many urban contexts in Europe and especially in Italy face a strong seismic risk. The Italian Institute for Statistics indicates that more than 60% of Italian buildings were built without seismic provisions [19] and 50% of the residential stock is earthquake-prone [20]. Actions, norms and policies aimed at reducing seismic risk to target not only single buildings but, more significantly, be tailored for different portions of urban contexts, at different scales, from single buildings, to blocks, neighborhoods and entire urban areas [21].

To this end, urban policies and actions must be designed to minimise damage caused by earthquakes through reducing vulnerability and specifically, exposure to heritage buildings. In particular, the reduction of vulnerability is a key issue in the debate related to seismic risk reduction [18,22].

Different approaches are available to face seismic vulnerability, such the adoption of specific seismic binding norms, urban planning, and enhancing construction practices. However, in Italy building replacement is still limited [20] and, therefore, seismic retrofitting of existing vulnerable buildings is an essential way to reduce earthquake damage [23]. Retrofitting of historical and modern residential fabric can be an economically efficient approach if compared to post-earthquake re-construction or recovery [24].

An analog situation can be found when looking at the current energy efficiency of Italy's building stock, where more than 1/3 of the overall demand for energy can be attributed to the household and tertiary sector [20]. This high energy demand urges the adoption of energy retrofitting measures and policies at different political and administrative levels.

Many retrofitting technologies and approaches are available, and they can be categorised into three main categories: supply side management, demand side management, and change of energy consumption patterns due to human habits [25]. The choice of which technology, policy or measure should be used is a multi-objective problem with many variables, constraints and limitations. The optimal solution can be considered as a trade-off among a number of factors, such as energy, economics, technical, environmental, regulations, social, etc. [25].

However, seismic and energy retrofitting policies have to face a number of complex issues related to the required economic efforts, technical feasibility, number of actors involved, the political coordination among different administrative levels, the need of involving many disciplines and local requirements [20,23]. At the national level fiscal incentives are currently proposed, especially in the form of a tax credit (up to 65% in 10 years) to those willing to implement retrofitting actions. However, the required costs for the investment are still far from the possibilities of the majority of people, so alternative mechanisms to support seismic and energy retrofitting are still needed [26]. Urban planning strategies should, therefore, foster and support retrofitting measures for low-income people and to facilitate the engagement of building owners in retrofitting activities [20].

#### 2.3. A Green Infrastructure Strategy for Reducing Seismic Vulnerability and Increasing Energy Efficiency

We propose in this paper a strategy for implementing the components of a green infrastructure aimed at triggering seismic retrofitting and energy demand reduction of urban fabrics through a TDR programme. Typically, in a TDR programme, development rights are assigned to strips of private land parcels that are designated for accommodating the green infrastructure. Development rights can be then transferred from these portions of parcels to other urban zones and used for new developments. As a result of the sale of the development rights, the involved parcels of urban fabric can be seismic-retrofitted, the strips transferred to public property for placing new components of a green infrastructure, and the new greenery provide cooling effects and generate a considerable decrease of energy demand in adjacent buildings.

The amount of development rights to be assigned to strips of land parcels is identified according to an economic feasibility assessment tool (see Section 3.2.3). To ensure the economic feasibility of the seismic retrofitting, this tool enables the quantification of equitable development rights to be assigned to each private parcel taking into account the amount of current developed volume of buildings, size of open spaces to be potentially left, seismic retrofitting costs, and economic rate of development rights to be sold.

#### 3. Materials and Methods

#### 3.1. A Case Study in the Catania Metropolitan Area (Italy)

The case study investigated in this work is located in the metropolitan area of Catania, the largest in Sicily. Catania has experienced dramatic urban growth in recent decades, and is characterized by a high urban density and severe lack of public green spaces. In 40 years (1961–2001), the total population of 27 municipalities included in the metropolitan area grew more than 27% in terms of built-up areas. In 2017, approximately 60% of its total population lived outside the main city, indicating progressive population expansion beyond the city center. We chose the municipality of Tremestieri Etneo (Figure 1), as a representative area where urbanization processes have been characterized by multi-storey buildings urban development [27]. Furthermore, the entire metro area is one of the areas most exposed to seismic risk in Italy.



**Figure 1.** Localization and land-use the municipality of Tremesteri Etneo within the Metropolitan area of Catania (Italy) (source: Urban Atlas 2012).

#### 3.2. Method

The method is composed of four different phases: (1) urban growth and urban morphology analyses to identify building age classes and categorization of urban morphology types in order to select the most suitable type of urban fabric to apply the proposed strategy of green infrastructure; (2) a land-use/land-ownership analysis to identify the land-use and landownership asset of each compound; (3) An economic feasibility assessment tool to quantify the equitable development rights to be assigned to private land parcels and the size of open spaces to be transferred to public property, taking into account different economic costs of seismic retrofitting and urban development parameters;

(4) a microclimate model to evaluate the potential cooling effect of new vegetation in the green infrastructure, estimating the reduction of energy demand for typical multi-storey buildings.

#### 3.2.1. Urban Growth and Urban Morphology Analyses (First Phase)

The analyses of urban growth and urban morphology are a key phase because seismic vulnerability and energy demand options depend on the age and form of the urban fabric [28].

Urban growth analysis was developed at block and sub-block level through the examination of the mean building age of urban fabrics and identifying six age classes: buildings and fabric built until 1928, 1964, 1985, 2000, 2007 and 2015. These years represent the available topographic sources used to map urban growth. Buildings belonging to different age classes were characterised by different materials and construction technologies (stonemasonry, mix of stonemasonry and steel structures, reinforced concrete structures) that show different levels of resistance to seismic activity [29]. In particular, buildings built-up to 1928 were mainly stonemasonry buildings while reinforced concrete buildings appeared in the period 1928–1964. In Italy, the first regulation requiring the evaluation of the seismic actions on reinforced concrete buildings was introduced in 1975. The DM 3/3/1975 Act represented a fundamental innovation in the analytical procedures by taking into account the dynamic properties of the structures [30]. However, the very first set of technical indications for designing reinforced concrete buildings in seismic zones was introduced in 1986 by the DM 24/1/1986. This act provided the first requirements concerning structural maximum allowable deformation under seismic actions. Nevertheless, more comprehensive and reliable seismic design concepts only appeared in Italy in 1996 through the DM 16/1/1996 Code that introduced the capacity design concept, even based on non-structural damage limitation [30].

Thus, norms and regulations have significantly shaped the seismic features of reinforced concrete buildings over the past decades. Buildings built-up to 1985 (before the DM 24/1/1986 Act) can be considered as more vulnerable to seismic events compared to buildings constructed from 1986 and even more after the implementation of DM 16/1/1996 Code. In this study, according to the available six interval classes of building age, reinforced concrete buildings belonging to the age classes of 1964 and 1985 have been considered more vulnerable than buildings belonging to the age classes of 2000, 2007 and 2015 and, consequently, can be prior targets of seismic retrofitting policies.

Urban morphology analysis identified those buildings where the green infrastructure strategy could be implemented. This was developed at block and sub-block level taking into account results from the previous urban growth analysis and considered two additional features: type of buildings, such as detached or semi-detached houses, terraced houses, multi-storey apartments, towers; and the relationship among buildings, streets and public and private open spaces, that can be characterized by interior courtyards or front/rear/side setback yards, sidewalks, parking lots, or small urban gardens. Different combinations of these features produced varied morphological layouts and built-up density. Building coverage ratio and percentage of open spaces were measured to identify urban morphology types. The urban morphology analysis was based on a regional topographic map (1:10,000) and a more detailed municipal topographic map (1:2000). Here the green infrastructure strategy was implemented for regular and irregular blocks with multi-storey buildings of reinforced concrete frame structures. Regular and irregular blocks are composed of different compounds, defined as gated residential units of buildings and related open spaces. These compounds are characterized by very large residential population, multiple private properties (apartments, garages), spaces with shared assets such as lobbies, staircases, lifts and adjacent open setback yards (walkways, green spaces, and park plots) (Figure 2).



**Figure 2.** Landownership Analysis. Visual interpretation of high-resolution orthophoto (25 cm, 1:10,000) on a sample Compound (B = building, P = private open space, S = shared open space).

#### 3.2.2. Land-Use/Land-Ownership Analysis (Second Phase)

Since regular and irregular blocks with multi-storey buildings morphology type can be composed by a mix of both private and shared land uses, the land-use and the land-ownership asset of each compound were analyzed to verify the presence of open spaces that can be transferred to public property in the TDR programme. This was performed by visual interpretation of available high resolution orthophotos on sample areas for each of the selected three morphology types. Three land-use types were chosen and identified in each morphology type: buildings, private adjacent setback yards, and shared open spaces (walkways, green spaces, and park plots) (Figure 2). Among these land-use types, the shared open spaces represent those portions that could be left to public property against corresponding economic benefits in the TDR programme.

#### 3.2.3. Economic Feasibility Assessment Tool (Third Phase)

The implementation of a suitable and realistic TDR programme represents a crucial phase in the proposed strategy. This is based on the assignment of development rights to the shared open spaces within the compounds that could be designated for the green infrastructure. Development rights could be transferred from these private parcels to other urban zones and used for new development. Landowners of the compounds could also sell development rights to other private developers obtaining economic earnings for covering the costs of the seismic-retrofitting. Finally, the shared open spaces could be transferred to the municipality which will plant the new greenery, thus generating a cooling effect through the shading effect of trees on adjacent buildings and a consequent decrease in their energy demand.

In the third phase, the economic viability of the TDR programme was evaluated. This was accomplished through a tool that was developed to quantify the equitable development rights to be assigned to each compound for leaving to become public property a percentage of the shared open spaces, transferring to other privates and developers the virtual amount of development while gaining economic revenues for future seismic retrofitting (Table 1).

To evaluate the economic feasibility, the tool sets the values of three main input parameters. First, a virtual floor area ratio (VFA) of  $4.50 \text{ m}^3/\text{m}^2$  was identified over the land parcel of shared open spaces to be transferred (TFA) in terms of amount of development to be transferred (VT). This value of VFA represents the mean value in existing buildings calculated as the ratio between the developed volume (V) and the land parcel unit area (LPA). Second, a seismic retrofitting rate (SRR) was defined as  $70 \text{ €/m}^3$ , in accordance to the value indicated by the Italian Urban Developers National Association [31]. This allows evaluation of the total retrofitting costs of buildings within the compound taking into account the number of storeys (S) and the total amount of developed volume

(V). Finally, the provided economic rate of development rights ERT =  $150 \text{ €/m}^3$  allows the evaluation of potential economic revenues (ERV) to be gained after selling. ERT was been calculated according to the economic feasibility of new development that will occur in other parts of the urban settlement where the virtual development rights will be allocated. Indeed, developers buying these rights have to deal with further costs while striving for a final economic profit  $\geq 25\%$  that is considered by Italian Urban Developers National Association as a reasonable percentage value of investment profit [31]. According to the tool, the overall seismic retrofitting costs can be achieved when the final balance (B) between economic earnings from development rights (ERV) and the total retrofitting costs (SRC) is  $\geq 0$ .

Economic Feasibility Assessment Tool—Seismic Retrofitting			
	Buildings Area	BA	
	Land Parcel unit Area	LPA	
	Buildings Coverage Ratio	BCR = BA/LPA	
	gross height of single storey	h	
	number of Storeys	S	
Compound Morphology Parameters	total building Height	$H = h \times S$	
	amount of developed Volume	$V = BA \times H$	
	Floor Area Ratio	FAR	
	Open spaces floor area	OFA = LPA - BFA	
	% of open spaces to be transferred	% OFA	
	Floor Area to be transferred	$TFA = OFA \times \% OFA$	
	Seismic retrofitting rate	SRR	
	Seismic retrofitting cost	$SRC = SRR \times V$	
Innut Parameters	Virtual Floor Area Ratio	VFA	
input i araniciers	amount of development to be tranferred	$VT = VFA \times TFA$	
	% increasing of new development	% V = VT/V	
	economic rate of development rights	ERT	
	economic revenues from development rights	$ERV = ERT \times VT$	
Costs Coverage Parameters	percentage of costs Coverage	% C = ERV/SRC	
	final Balance	B = ERV - SRC	

Table 1. Economic feasibility assessment tool for seismic retrofitting intervention.

3.2.4. Potential Local Cooling Effect of Trees and Relative Building Energy Demand Reduction (Fourth Phase)

The interactions among urban vegetation, urban structures, and urban climate generate different microclimate conditions that also need to be evaluated for each urban context studied. Typically, vegetation can lower both the air temperature and wind speed of the surrounding microclimate, reducing the cooling load, but current research agrees that the cooling effect in small areas is obtained mainly through shading [10]. Other factors that inhibit the penetration of solar radiation, besides shading, may also play a role in determining the cooling effect of an urban green space. The geometric configuration may also affect temperature variations, as happens for non-wooded building structures [32]. Although a cooling effect has been reported even for small green areas [33], the extent of this effect on neighbouring buildings is less evident and subject to limited research [34].

We postulated that the implementation of green infrastructure in portions of private shared open spaces as a result of the TDR programme could result in a relevant energy saving through the shading effect of the new trees that are planted. To quantify the possible energy savings, we considered the model by [10], who analysed the dynamic interaction between microclimate and buildings. These authors simulated the optimal position of trees around a building through parametric design in order to estimate the energy saving that could be achieved through the shading effect of trees on the buildings. Results of the simulations resulted in an energy saving ranging from a minimum of 11% when locating only 1 tree to a maximum of 48.5% when locating 5 trees around buildings, therefore highlighting that a limited amount of greenery is able to achieve a relevant energy saving.

Figure 3 reports the possible configurations of the position of trees with respect to the buildings studied in this paper. Table 2 shows the relative energy savings (in %) according to those configurations as modelled by [35], for the configurations 1, 4, 6 and 7. Configurations 2, 3 and 5 were not taken into account as they were not assessed by [35].



 Table 2. Energy reduction for configuration of trees (Calcerano and Martinelli, 2016).

**Figure 3.** Possible configurations (W = West, E = East, S = South) of shading trees (in green) around buildings (in grey).

#### 4. Results

Urban growth analysis showed that in the tested municipality the majority of the urban fabric was developed after 1964 (56% in terms of building area, BA), with a first impressive boost in the period 1964–1985 and a second in the period 1985–2000 [32] (Figure 4, Table 3).

Buildings and Urban Fabric to	Area (m <sup>2</sup> )	Area (%)
1928	257,761	9.47
1964	213,667	7.85
1985	1,337,113	49.10
2000	774,885	28.46
2007	121,269	4.45
2015	18,294	0.67
Total	2,722,989	100.00

Table 3. Size and proportion of buildings and urban fabric alongside in the period 1928–2015.

Urban morphology analysis identified eight urban morphology types within the study area, sub-divided into two sub-classes: historical and modern urban fabrics. Modern urban fabric includes more categories, since a great variety of buildings typologies have been used since the 1960s. These are often characterised by urban patterns with multi-storey apartment buildings set back from streets and including private or semi-private open spaces. Urban morphology analysis was mapped in Figure 5

and summarised in Table 4 [36]. The category of Regular and Irregular blocks with multi-storey buildings represents almost 30% of the urban fabric, with a total of 132 compounds.



Figure 4. Urban growth map of the municipality of Tremestieri Etneo, Catania (Italy); (a) enclave of the municipality located at the south-east end of the territory.



**Figure 5.** Urban morphology map of the municipality of Tremestieri Etneo, Catania (Italy); (a) enclave of the municipality located at the south-east end of the territory.

Urban Morphology Types	Area (m <sup>2</sup> )	Area (%)
Historic monumental buildings	10,263.30	0.37
Grids	183,251.90	6.60
Linear	176,763.30	6.36
Irregular blocks with multi-storey buildings	440,140.00	15.85
Regular blocks with multi-storey buildings	345,091.70	12.42
Detached and semi-detached houses	1,324,738.70	47.69
Isolated buildings in farmlands	73,722.90	2.65
Industrial, commercial and public utilities areas	223,569.20	8.05
Total	2,777,541.00	100.00

Table 4. Size and proportion of each urban morphology type within the urban fabric.

15 patches were further detected and then excluded from the evaluation because they belonged to other morphological types and were mixed-up with the Regular and Irregular ones. The amount of shared open spaces (walkways, green spaces, and park plots) was also evaluated to understand the suitability of compounds for the proposed strategy. As a result, 49 compounds out of the remaining 117 were evaluated as not suitable because they lacked or showed an insufficient quantity of shared open spaces to be left as public property against the assignment of development rights as quantified by the strategy. Another 35 compounds belonged to the age classes 2000, 2007 and 2015 (see Figure 5) and, therefore, were considered as not a priority for seismic retrofitting actions compared to the reinforced concrete buildings belonging to the age classes of 1928, 1964 and 1985. As a final result, 33 compounds out of the initial 132 were evaluated as being seismically vulnerable and thus suitable for urban transformation and eligible for the proposed strategy (Figure 6).



**Figure 6.** Thirty-three selected compounds (from a to z.1, in grey) out of the initial 132 belonging to the morphology category of Regular (dark orange) and Irregular (light orange) Blocks with multi-storey buildings.

These compounds show different features in terms of land parcel unit area, building coverage ratio, amount of developed building volume, and open spaces floor area. The economic feasibility assessment tool presented in Section 3.2.3 assigned a virtual floor area ratio (VFA) =  $4.50 \text{ m}^3/\text{m}^2$  to the shared open spaces designated for being left to public property, and estimated an economic rate of development rights (ERT) =  $150 \text{ €/m}^3$ , and a seismic retrofitting rate SRR=  $70 \text{ €/m}^3$ . As an example, Table 5 reports the results of the application of the tool for the compound a.1.

Economic Feasibility Assessment—#Compound a.1				
Buildings area	BA	3867.00	m <sup>2</sup>	
Land parcel unit area	LPA	18,858.00	m <sup>2</sup>	
Buildings coverage ratio	BCR = BA/LPA	20.51	%	
Gross height of single storey	h	3.00	m	
Number of storeys	S	5.00		
Total building height	$H = h \times S$	15.00	m	
Amount of developed volume	$V = BFA \times H$	58,005.00	m <sup>3</sup>	
Floor area ratio	FAR	3.08	$m^3/m^2$	
Open spaces floor area	OFA = LPA - BFA	14,991.00	m <sup>2</sup>	
% of open spaces to be transferred	% OFA	40.42	%	
Floor area to be transferred	$TFA = OFA \times \% OFA$	6060.00	m <sup>2</sup>	
Seismic retrofitting rate	SRR	70.00	€/m <sup>3</sup>	
Seismic retrofitting cost	$SRC = SRR \times V$	4,060,350.00	€	
Virtual floor area ratio	VFA	4.50	$m^3/m^2$	
Amount of development to be tranferred	$VT = VFA \times TFA$	27,270.00	m <sup>3</sup>	
% increasing of new development	% V = VT/V	47.01	%	
Economic rate of development rights	ERT	150.00	€/m <sup>3</sup>	
Economic revenues from development rights	$ERV = ERT \times VT$	4,090,500.00	€	
Percentage of costs coverage	% C = ERV/SRC	100.74	%	
Final balance	B = ERV - SRC	30,150.00	€	

 Table 5. Economic feasibility assessment. Results from the compound a.1.

The tool showed that the overall seismic retrofitting costs can be covered at 95–100% (final balance  $B \ge 0$ ) when floor area ratio (FAR)  $\le 3.3 \text{ m}^3/\text{m}^2$  and % open spaces floor area (% OFA)  $\ge 40\%$ (compounds a, a1, i, t) (Figure 7a). Values of % OFA > 24% and FAR  $< 4.5 \text{ m}^3/\text{m}^2$  determine at least 2/3 of costs coverage (66%) and can cover up to a maximum of 94% that means a final balance < 0; this is the case of the compounds e, g (Figure 7b), d.1, j, n, r, u, v, w, z (Figure 7b). When FAR  $\leq 5.3 \text{ m}^3/\text{m}^2$  and % OFA  $\leq$  26% costs coverage are covered at least 50% and up to 65%: this is the case of the compound b.1, d, k, s, z.1 (Figure 7b). In all other remaining compounds (b, b.2, b.3, c, f, h, l, m, o, p, q, q.1, x, y), values of FAR and % OFA imply that potential economic earnings from development rights can no longer balance the total retrofitting costs with a percentage of cost coverage % C decreasing from 49% to about 8%. As a final result, in the 12% of the total amount of compounds (4 out of 33) seismic retrofitting is economically viable and can be fully funded (Table 6). For most of the compounds (30%, 10 compounds) retrofitting can be undertaken through covering 66% up to 94% of the total costs; 15% of the compounds (5) can get economic restoration and covering from 50% to 65% of the retrofitting costs. The remaining 47% of the compounds can be considered disadvantaged and the seismic retrofitting resulted to be not economically viable (Table 7). Plotting the values of the four seismic retrofitting costs coverage (% C) classes (>50%, 50–64%, 65–94%, >95%), Figure 8 shows that corresponding values of FAR and % OFA identify four clusters and that the economic feasibility of the compounds transformation increases according to the increasing values of FAR and the decreasing values of % OFA.



(a)

Figure 7. Cont.



(b)

**Figure 7.** (a) Compounds a, a.1, i, t (seismic retrofitting costs coverage = 95-100%); compounds e, g, h (seismic retrofitting costs coverage = 66-94%). Buildings in dark grey, shared open spaces in light green, private adjacent setback yards in white, trees line in dark green. (b) Compounds r, u, w, z (seismic retrofitting costs coverage = 66-94%); compounds b.1, k, s, z.1 (seismic retrofitting costs coverage = 50-65%); compounds j, q, q1 (seismic retrofitting costs coverage = 8-49%). Buildings in dark grey, shared open spaces in light green, private adjacent setback yards in white, treelines in dark grey.

#Compound	Floor Area Ratio (FAR) m <sup>3</sup> /m <sup>2</sup>	% Open Spaces to Be Transferred (% OFA)	% of Costs Coverage (% C)
а	3.29	42.91	102.68
a.1	3.08	42.91	100.74
b	3.18	12.41	27.68
b.1	4.46	35.60	54.16
b.2	3.87	24.00	44.44
b.3	4.88	24.60	32.80
с	2.94	15.88	39.23
d	5.29	43.30	55.76
d.1	4.49	42.21	68.10
e	3.74	37.97	77.69
f	4.42	28.84	47.51
g	3.21	32.37	79.80
h	6.08	20.24	21.25
i	3.13	48.05	117.15
j	3.44	36.24	78.30
k	4.22	33.72	61.54
1	9.74	30.13	16.00
m	8.44	12.16	8.30
n	4.20	41.16	68.02
0	7.12	31.58	25.88
р	6.04	26.68	28.33
q	3,38	22,78	46,63
q.1	3.53	13.63	26.24
r	2.44	24.25	69.91
S	3.13	25.96	52.21
t	3.12	47.43	95.93
u	2.39	31.18	92.62
V	2.46	29.86	85.17
W	2.61	30.61	80.45
х	2.65	19.22	49.42
У	3.99	35.02	47.04
Z	3.51	38.53	75.03
z.1	2.98	30.15	65.31

Table 6. Morphology parameters and percentage of costs coverage for the 33 selected compounds.

Table 7. Viability of seismic retrofitting (% of costs coverage) for each selected compound.

#Compound	Number of Compounds	% Compounds	% of Costs Coverage (% C)
a, a.1, i, t	4	12.12	95–100%
d.1, e, g, j, n, r, u, v, w, z	10	30.30	66–94%
b.1, d, k, s, z.1	5	15.15	50-65%
b, b.2, b.3, c, f, h, l, m, o, p, q, q.1, x, y	14	42.42	8-49%

Following the model reported in Section 3.2.4, the 33 selected compounds were analysed to evaluate the potential local cooling effect of the new trees in reducing the buildings' energy demand. Alongside the shared open spaces of each compound, seven configurations of shading trees were identified (Figure 3). The possible layouts of new trees to be planted around each building depend on the spatial relationship between buildings and the available shared open spaces. Among the 93 buildings in the 33 compounds, configuration #7 (S, South) resulted in being the most suitable for 27 buildings, configuration #6 (E, East) for 13 buildings, configuration #5 (W, West) for 14 buildings, and configuration #4 (E + W), configuration #3 (S + E), configuration #2 (S + W) and configuration #1 (S + E + W) were suitable for 3 buildings each. For the remaining 27 buildings, no configuration of shading trees can be proposed because of the lack of surrounding shared open spaces (Table 8).



Figure 8. Distribution of seismic retrofitting costs coverage (% C) according to the values of FAR  $(m^3/m^2)$  and % OFA (%).

#Configuration of Shading Trees	Number of Buildings	% of Buildings
1	3	3.23
2	3	3.23
3	3	3.23
4	3	3.23
5	14	15.05
6	13	13.98
7	27	29.03
/	27	29.03
Total	93	100

Table 8. Number and percentage of buildings per each configuration of shading trees.

The highest values of energy demand reduction (varying from 37.3–41.8% up to 44.4–48.5%) involve a total 6% of the buildings (respectively, configurations #4 and #1) and producing significant results in terms of cooling by shading and, therefore, decreasing the relative energy demands.

For most of the buildings (more than 29%), the range of energy demand reduction is 11.1–12.8% (configuration #7) while for almost 14% of the buildings the percentage of energy reduction varies from 19.2% to 21.2% (Table 9).

Table 9. Percentage of energy demand reduction per each building within the configuration of shading trees #1, 4, 6, 7.

Range of Energy Demand Reduction (%)	#Configuration of Shading Trees	Number of Buildings	% of Buildings
44.4-48.5	1	3	3.23
37.3-41.8	4	3	3.23
19.2-21.2	6	13	13.98
11.1–12.8	7	27	29.03

Within each compound, the number of buildings benefitting from the shading effect of the trees, varies according to the amount of surrounding shared open spaces (Table 10). It can be seen that for 15 compounds (45.45%), the total amount of existing buildings (100%) can benefit from reduced energy demand while the percentage of beneficiary buildings decreased from 83.33% up to 60% in another 5 compounds. Moreover, in 8 compounds (24.24% of the total), only 50% of the buildings can be surrounded by shading trees. While in the remaining 5 compounds (15.15%) the lack of surrounding shared open space does not allow any plantation of shading trees.

Table 10. Proportion of buildings benefiting from the shading effect of the trees within each compound.

% of Buildings Benefitting from Shading Trees	Number of Compounds	% of Compounds
100	15	45.45
83.33–60	5	15.15
50	8	24.24
/	5	15.15

#### 5. Discussions

#### 5.1. A Multi-Benefits Strategy Depending on Economic Viability

Overall, the benefits could be considered as the result of many intertwined attributes of the built environment, such as built-up density, building coverage ratio and percentage of open spaces within the blocks. Regular and Irregular blocks with multi-storey buildings include compounds that present different proportions and spatial arrangements of built-up areas and shared open spaces. Especially for dense urban contexts lacking green spaces, the peculiarities of these compounds may represent an opportunity for implementing a green infrastructure: they include open spaces with shared property assets that can be transferred to public property against the provision of development rights and can be used as areas in which to implement the green infrastructure.

The urban transformations of the future imply the potential achievement of new open spaces without financial efforts for local administrations. This could represent the basis for designing and implementing a new public green network of walking and cycling routes crossing-over the urban fabric. On the other hand, the landowners of the compounds, who have left their own shared open spaces to public property, could obtain as compensation a suitable amount of development rights to be sold to other private landowners and/or developers. The latter would be then able to transfer and use the development rights in other urban areas to increase the square footage and the height of existing buildings or developing new residential units.

The application of the strategy also returned a scenario of hundreds of small private open spaces scattered within the city that could be potentially transferred to public property and linked to existing roads, parking areas, playgrounds and existing small gardens. Even though the positive effects of just a few trees around each building was recognised, a green infrastructure strategy incorporating plantations of longer treelines could provide urban greenery and deliver a wider set of ecosystem services [37].

The multi-scale impact (from municipal level to building level) was the most significant feature of the proposed strategy. The possibility to improve the seismic safety and the energy efficiency of the built-up environment from neighbourhood level to the smallest scale of the single building, while enhancing the ecosystem performance of the overall urban fabric, represents a relevant opportunity for any planning strategy and transformation action. The urgent issue of making cities more sustainable, efficient, healthy and safer places [7] can only be addressed through a strategy which is grounded on an economic feasibility assessment of urban transformation and ensures the viability of the fragmented property assets balancing public/private costs and benefits [15].

Indeed, the economic feasibility of the proposed green infrastructure strategy might be affected by different market factors such as land and property prices that often depend strictly on local real estate dynamics [38]. Investigation of the economic land value of existing shared open spaces and the final market value of the potential up-zoning or new developments is crucial for better identifying the appropriate amount of development rights to be assigned to the compounds. Thus, the correct identification of the economic rate of development rights (ERT), through a specific local market prices survey, is required to trigger the transfer of development rights from compounds to other urban areas to be developed. Landowners of compounds would be willing to sell their own development rights whether the economic earnings will be enough to cover the seismic retrofitting costs. On the other hand, developers could buy the available development rights whether their cost will allow an adequate economic profit compared to the final market value of the new developments.

#### 5.2. Achieving Seismic Retrofitting and Reduction of Cooling Energy Demand

Concerning seismic retrofitting, the results highlight that half of the total compounds would need some seismic retrofitting but only 25% were suitable according to the availability of shared open spaces. More specifically, seismic retrofitting was found to be economically viable in more than 40% of these selected compounds, allowing landowners to be funded for more than 2/3 of the total costs. The availability of such funds to be shared among the landowners is a key issue for the viability of a seismic retrofitting due to the multi-fragmented ownership of each compound. The large amount of single private properties in the compound such as apartments and garages, mixed up with shared assets such as lobbies, staircases and lifts but also walkways, green spaces and park plots, represents a real obstacle for implementing any kind of intervention. A property asset with a high number of private landowners in the compound [39]. However, these results appear promising because efforts to reduce the seismic vulnerability of the existing real estate are limited and urban policies for seismic risk mitigation are still at an early stage of development [20].

At the building scale, results show that more than 50% could benefit from the cooling effect of the trees if planted alongside the public acquired open spaces. According to the results, planting a treeline simultaneously alongside east, west and south sides (or just east and west sides) of the buildings could lead to a reduction of energy demand of up to 48.5%: locating trees alongside these orientations represents the best option to obstruct solar radiation in summertime, provide a shading effect on buildings, while reducing the local temperature around.

The reduction of buildings' energy demand has been evaluated according to the study conducted by [35] that clearly showed the potential cooling effect of different tree layouts around buildings. Investigation in other similar research has detected different approaches and models for evaluating the trees' shade effect on cooling energy reduction. These studies confirmed and validated the trees shading effect on the reduction of building cooling energy demand with different results that depend on the different urban environments, land-use configurations, and micro-climate conditions. For example, [40] highlighted that shaded building by greenery have a greater inertia in warming up and demonstrated how shading can result in an important saving of money for cooling (up to 218 € in the case study of Akure, Nigeria). Reference [41] valuated that the savings associated with urban shade trees can be up to \$200 per tree (including the carbon sequestration effect). Reference [42] compared two identical buildings in Alabama (USA) and calculated that the unshaded building under full sunlight during the summer period required more than 2.6 times the amount of cooling energy than a shaded one. Moreover, a further but limited cooling effect could be provided by the evapotranspiration process of greenery, that could decrease local temperature in the surrounding of the buildings proportionally to the amount of tree canopy. Reference [43] quantified an energy saving of 15% when a scenario of shading and transpiration was considered.

Finally, tree species should be chosen and placed in order to properly shade the entire facades of the buildings and, therefore, maximise the reduction of energy demand: to this end, deciduous trees allowing for solar gain during the wintertime could be preferred, yet differences in cooling effects can be observed among different species [44,45].

#### 5.3. Limitations

Beside the positive relevance of the results, the proposed strategy also shows some limitations. Implementing a green infrastructure is not always viable and may be strongly affected by the quality and amount of the open spaces to be acquired. The size, shape and location of the shared open spaces within the compounds were not always considered suitable for conversion into components of a green network. Their reduced size and narrow shape could negatively affect the design of the green intervention while their location, remoteness and interrelationship with the private residential buildings could affect the real possibility to connect open spaces to existing public gardens and roads. Moreover, the amount and the geographical distribution of these open spaces (which could be concentrated in specific areas) could generate some inequality in the provision of new urban green spaces at municipal level [46].

Although not all the possible configurations of greenery around the buildings were assessed (Figure 3), we choose those configurations that limited the number of trees to be located and that, therefore, represented more economical solutions in terms of financial resources needed for their deployment.

The implementation of retrofitting actions would generate new development of more than +50% of the total current development volume of the compounds. This is dependent on the amount of development rights to be granted to private compounds as a compensation for leaving their shared open spaces. The total amount of development depends on the number of compounds which could be retrofitted. In this case study, seismic retrofitting is viable for a very small proportion of compounds (6% of the selected ones, more than 1% of the total compounds) and this means that a limited amount of new developments could be transferred to other areas to be developed. When the number of compounds to be retrofitted would be higher, this amount of development could exceed the real need of the municipality and generate unsustainable scenarios of excessive soil sealing. In these cases, to limit the new development, a priority plan should be used to reduce the number of compounds to be retrofitted by selecting those most vulnerable and exposed in terms of the number of residents.

We also have to underline that cost of retrofitting interventions could be dependent on the level of seismic risk and, therefore, not be constant when changing the geographical contexts or city. This is especially true for stonemasonry buildings that require higher retrofitting cost [18]. In this method we have used a value (the seismic retrofitting rate) derived by the Italian Urban Developers National Association [31] that is suitable for a high-risk seismic context, but this value could be different in other cities.

However, the use of a constant cost for a city represents a reasonable choice, as the green infrastructure strategy is supposed to be implemented in a single municipality. Within a single city, only the availability of high-resolution information on soils and other geological features (i.e., obtained by a detailed seismic zonation) or spatial survey of building types (which is not the objectives of our paper) would allow differentiated costs of retrofitting to be obtained.

The reduction of energy demand is limited to the summertime period and for a proportion of buildings within each compound. Results of the proposed method show that shading effects provided by tree plantation could not be extended to all buildings in the same compound. This would imply unbalanced benefits for only a few landowners—those benefiting from the energy demand reduction of their buildings—against the transfer of shared open spaces property belonging to all landowners in the compound.

Also, the method presents some specific limitations. The land-ownership analysis (second phase), based on the visual interpretation of available high-resolution orthophotos on sample areas for each of the selected three morphology types, could be more time-consuming at a wider scale than the municipal level, due the large amount of compounds to be analysed. Thus, the method, as proposed, is particularly effective only at a municipal scale. Moreover, the economic feasibility assessment tool (third phase) is based on the assumption of three basic input parameters. Different values of these parameters could affect the results and the final findings of the research. To overcome this issue, a next

step of the proposed study could include a sensitivity analysis of these input parameters for improving the quality and the reliability of the economic evaluation.

#### 6. Conclusions

Seismic vulnerability and energy inefficiency in existing urban fabric represent challenging issues for local spatial planning targeting more livable, healthy and safer cities. When urban fabric lacks public green space and is characterized by compounds with multi-storey buildings and several individual landowners sharing the same property, the real opportunities to undertake any sustainable urban transformation could be nearly null. The double nature of the compound property asset, characterised by single private properties mixed up to shared ownership properties, implies complex decision processes when planning an intervention at compound level. In these respects, the willingness and availability of financial means to contribute to any intervention could be different and conflicting among landowners concerned and could affect and even prevent the implementation of the compound transformation.

Such complexity calls planners and policy makers to have a better understanding of the morphological features and property assets of urban contexts while claiming new approaches and policies for managing urban transformation. In this perspective, the proposed study represents an innovative urban strategy able to combine the needs of implementing public green infrastructure for cities lacking green spaces with the need for seismic retrofitting and the reduction of cooling energy demand of private existing buildings. The strategy is based on an economic feasibility assessment of the urban transformation which ensures viability in fragmented property assets while balancing public and private costs and benefits.

The proposed strategy presents a scenario in which landowners of the compound leave portions of shared open spaces to public property and obtain, as a compensation, an equitable amount of development rights to be sold to other private landowners and/or developers. Economic earnings deriving from selling development rights constitute a shared budget for funding a seismic retrofitting intervention of the buildings belonging to the compound. The developers transfer the purchased development rights in other urban areas for increasing the square footage and the height of existing buildings or developing new residential units. Local administrators acquire new open spaces without financial efforts and implement a new green infrastructure.

Finally, new greenery to be located alongside these new public areas provides further benefits to private buildings in terms of a significant reduction of cooling energy demand through the shading effect. Such a multi-scale strategy, acting from municipal level to building level, would allow the regeneration of vulnerable, inefficient and seismically unsafe portions of the city and increase the number and quality of ecosystem services provided by the new green infrastructure.

Author Contributions: R.P. and D.L.R. defined the problem and designed the method. R.P. and D.L.R. wrote Sections 3 and 6. R.P. wrote Sections 4 and 5. D.L.R. wrote Sections 1 and 2.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- Cohen-Shacham, E.; Walters, G.; Janzen, C.; Maginnis, S. Nature-Based Solutions to Address Global Societal Challenges; IUCN: Gland, Switzerland, 2016; ISBN 978-2-8317-1812-5.
- Nesshöver, C.; Assmuth, T.; Irvine, K.N.; Rusch, G.M.; Waylen, K.A.; Delbaere, B.; Haase, D.; Jones-Walters, L.; Keune, H.; Kovacs, E.; et al. The science, policy and practice of nature-based solutions: An interdisciplinary perspective. *Sci. Total Environ.* 2017, 579, 1215–1227. [CrossRef] [PubMed]
- Fan, P.; Ouyang, Z.; Basnou, C.; Pino, J.; Park, H.; Chen, J. Nature-based solutions for urban landscapes under post-industrialization and globalization: Barcelona versus shanghai. *Environ. Res.* 2017, 156, 272–283. [CrossRef] [PubMed]

- Tzoulas, K.; Korpela, K.; Venn, S.; Yli-Pelkonen, V.; Kazmierczak, A.; Niemela, J.; James, P. Promoting ecosystems and human health using green infrastructure: A literature review. *Landsc. Urban Plan.* 2007, *81*, 167–178. [CrossRef]
- O'Brien, L.; DeVreese, R.; Kern, M.; Sievanen, T.; Stojanova, B.; Atmis, E. Cultural ecosystem benefits of urban and peri-urban green infrastructure across different European countries. *Urban For. Urban Green.* 2017, 24, 236–248. [CrossRef]
- 6. La Rosa, D.; Privitera, R. Characterization of non-urbanized areas for land-use planning of agricultural and green infrastructure in urban contexts. *Landsc. Urban Plan.* **2013**, *109*, 94–106. [CrossRef]
- Wild, T.; Henneberry, J.M.; Gill, L. Comprehending the multiple 'values' of green infrastructure-valuing nature-based solutions for urban water management, from multiple perspectives. *Environ. Res.* 2017, 158, 179–187. [CrossRef] [PubMed]
- Pappalardo, V.; La Rosa, D.; Campisano, A.; La Greca, P. The potential of green infrastructure application in urban runoff control for land use planning: A preliminary evaluation from a southern Italy case study. *Ecosyst. Serv.* 2017, 26, 345–354. [CrossRef]
- 9. McHale, M.R.; McPherson, E.G.; Burke, I.C. The potential of urban tree plantings to be cost effective in carbon credit markets. *Urban For. Urban Green.* **2007**, *6*, 49–60. [CrossRef]
- 10. Givoni, B. Impact of planted areas on urban environmental quality: A review. *Atmos. Environ.* **1991**, 25, 289–299. [CrossRef]
- 11. Bowler, D.E.; Buyung-Ali, L.; Knight, T.M.; Pullin, A.S. Urban greening to cool towns and cities: A systematic review of the empirical evidence. *Landsc. Urban Plan.* **2010**, *97*, 147–155. [CrossRef]
- Bengston, D.N.; Fletcher, J.O.; Nelson, K.C. Public policies for managing urban growth and protecting open space: Policy instruments and lessons learned in the United States. *Landsc. Urban Plan.* 2004, 69, 271–286. [CrossRef]
- 13. Brabec, E.; Smith, C. Agricultural land fragmentation: The spatial effects of three land protection strategies in the eastern United States. *Landsc. Urban Plan.* **2002**, *58*, 255–268. [CrossRef]
- Nelson, A.C.; Pruetz, R.; Woodruff, D.; Nicholas, J.C.; Juergensmeyer, J.C.; Witten, J. The TDR Handbook—Designing and Implementing Transfer of Development Rights Programs; Island Press: Washington, DC, USA, 2012; p. 344, ISBN 978-1597269810.
- 15. Martinico, F.; La Rosa, D.; Privitera, R. Green oriented urban development for urban ecosystem services provision in a medium sized city in southern Italy. *iForest* **2014**, *7*, 385–395. [CrossRef]
- 16. Pruetz, R.; Pruetz, E. Transfer of Development Rights Turns 40. Plan. Environ. Law 2007, 59, 3–11. [CrossRef]
- Scawthorn, C. A Brief History of Seismic Risk Assessment. In Risk Assessment, Modeling and Decision Support -Strategic Directions; Bostrom, A., French, S., Gottlieb, S., Eds.; Springer: Berlin, Germany, 2008; pp. 5–81, ISBN 978-3-540-71158-2.
- Crowley, H.; Colombi, M.; Borzi, B.; Faravelli, M.; Onida, M.; Lopez, V.; Polli, D. Comparison of seismic risk maps for Italy. *Bull. Earthq. Eng.* 2009, 7, 149–180. [CrossRef]
- ANCE/CRESME. Lo Stato del Territorio Italiano, Insediamento e Rischio Sismico e Idrogeologico 2012; ANCE: Roma, Italy, 2012; Available online: www.ance.it/docs/docDownload.aspx?id=8495 (accessed on 18 July 2018).
- 20. La Greca, P.; Margani, G. Seismic and Energy Renovation Measures for Sustainable Cities: A Critical Analysis of the Italian Scenario. *Sustainability* **2018**, *10*, 254. [CrossRef]
- Frolova, N.I.; Larionov, V.I.; Bonnin, J.; Sushchev, S.P.; Ugarov, A.N.; Kozlov, M.A. Seismic risk assessment and mapping at different levels. *Nat. Hazards* 2016, *8*, 43–62. [CrossRef]
- 22. Maio, R.; Ferreira, T.M.; Vicente, R.; Estêvão, J. Seismic vulnerability assessment of historical urban centres: Case study of the old city centre of Faro, Portugal. *J. Risk Res.* **2015**, *19*, 551–580. [CrossRef]
- Segal, E.; Negev, M.; Feitelson, E.; Zaychik, D. Devising 'policy packages' for seismic retrofitting of residences. Nat. Hazards 2017, 89, 497–519. [CrossRef]
- 24. Bernknopf, R.; Amos, P. Measuring earthquake risk concentration for hazard mitigation. *Nat. Hazards* 2014, 74, 2163–2192. [CrossRef]
- Ma, Z.; Cooper, P.; Daly, D.; Ledo, L. Existing building retrofits: Methodology and state-of-the-art. Energy Build. 2012, 55, 889–902. [CrossRef]
- Gooding, L.; Gul, M.S. Energy efficiency retrofitting services supply chains: A review of evolving demands from housing policy. *Energy Strateg. Rev.* 2016, 11–12, 29–40. [CrossRef]

- 27. Privitera, R.; La Rosa, D. Enhancing carbon sequestration potential of urban green spaces through transfer of development rights strategy. *Acta Geobalc.* 2018, *4*, 17–23. [CrossRef]
- Ko, Y. Urban Form and Residential Energy Use. A Review of Design Principles and Research Findings. J. Plan. Lit. 2013, 28, 327–335. [CrossRef]
- Ministero Delle Infrastrutture e Dei Trasporti. Linee Guida Per la Classificazione del Rischio Sismico Delle Costruzioni 2017. Available online: http://www.mit.gov.it/normativa/decreto-ministeriale-numero-65-del-07032017 (accessed on 18 July 2018).
- 30. Ricci, P.; De Luca, F.; Verderame, G.M. 6th April 2009 L'Aquila earthquake, Italy: Reinforced concrete building performance. *Bull. Earthq. Eng.* 2011, *9*, 285–305. [CrossRef]
- ANCE. Adeguamento Antisismico. In Sicurezza Strutturale Per Gli Edifici Residenziali in Cemento Armato Costruiti Prima Dell'entrata in Vigore Della Normativa Sismica; ANCE: Catania, Italy, 2012. Available online: http: //www.consultaingegnerisicilia.it/download/Studio%20adeguamento%20antisismico.pdf (accessed on 11 June 2018).
- 32. Ng, E.; Chen, L.; Wang, Y.; Yuan, C. A study on the cooling effects of greening in a high-density city: An experience from Hong Kong. *Build. Environ.* **2012**, *47*, 256–271. [CrossRef]
- Shashua-Bar, L.; Hoffman, M.E. Vegetation as a climatic component in the design of an urban street: An empirical model for predicting the cooling effect of urban green areas with trees. *Energy Build*. 2000, 31, 221–235. [CrossRef]
- Palme, M.; Inostroza, L.; Villacreses, G.; Lobato-Cordero, A.; Carrasco, C. From urban climate to energy consumption. Enhancing building performance simulation by including the urban heat island effect. *Energy Build.* 2017, 145, 107–120. [CrossRef]
- Calcerano, F.; Martinelli, L. Numerical optimisation through dynamic simulation of the position of trees around a stand-alone building to reduce cooling energy consumption. *Energy Build.* 2016, 112, 234–243. [CrossRef]
- Privitera, R.; Palermo, V.; Martinico, F.; Fichera, A.; La Rosa, D. Towards lower carbon cities: Urban morphology contribution in climate change adaptation strategies. *Eur. Plan. Stud.* 2018, 26, 812–837. [CrossRef]
- Graça, M.; Alves, P.; Gonçalves, J.; Nowak, D.J.; Hoehn Farinha-Marques, R.P.; Cunha, M. Assessing how green space types affect ecosystem services delivery in Porto, Portugal. *Landsc. Urban Plan.* 2017, 170, 195–208. [CrossRef]
- Linkous, E.R. Transfer of development rights in theory and practice: The restructuring of TDR to incentivize development. Land Use Policy 2016, 51, 162–171. [CrossRef]
- Webb, B.; Webber, S. The implications of condominium neighbourhoods for long-term urban revitalization. *Cities* 2017, 61, 48–57. [CrossRef]
- Balogun, A.A.; Morakinyo, T.E.; Adegun, O.B. Effect of tree-shading on energy demand of two similar buildings. *Energy Build*. 2014, *81*, 305–315. [CrossRef]
- Akbari, H. Shade trees reduce building energy use and CO<sub>2</sub> emissions from power plants. *Environ. Pollut.* 2002, 116, 119–126. [CrossRef]
- 42. Laband, D.; Sophocleus, V. An experimental analysis of the impact of tree shade on electricity consumption. *Arboric. Urban For.* **2009**, *35*, 197–202.
- Hsieh, C.H.; Li, J.J.; Zhang, L.; Schweglerd, B. Effects of tree shading and transpiration on building cooling energy use. *Energy Build.* 2018, 159, 382–397. [CrossRef]
- 44. Akbari, H.; Dan, K.M.; Sarah, B.E.; Hanford, J.W. Peak power and cooling energy savings of shade trees. *Energy Build*. **1997**, *25*, 139–148. [CrossRef]
- Lin, B.S.; Lin, Y.J. Cooling Effect of Shade Trees with Different Characteristics in a Subtropical Urban Park. HortScience 2010, 45, 83–86.
- La Rosa, D.; Takatori, C.; Shimizu, H.; Privitera, R. A planning framework to evaluate demands and preferences by different social groups for accessibility to urban greenspaces. *Sustain. Cities Soc.* 2018, 36, 346–362. [CrossRef]



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





## A European Project for Safer and Energy Efficient Buildings: Pro-GET-onE (Proactive Synergy of inteGrated Efficient Technologies on Buildings' Envelopes)

# Annarita Ferrante <sup>1,\*</sup>, Giovanni Mochi <sup>1</sup>, Giorgia Predari <sup>1</sup>, Lorenzo Badini <sup>1</sup>, Anastasia Fotopoulou <sup>1</sup>, Riccardo Gulli <sup>1</sup> and Giovanni Semprini <sup>2</sup>

- Department of Architecture, School of Engineering and Architecture, (UNIBO), Viale Risorgimento 2, 40126 Bologna, Italy; giovanni.mochi@unibo.it (G.M.); giorgia.predari@unibo.it (G.P.); lorenzo.badini3@unibo.it (L.B.); anastasia.fotopoulo2@unibo.it (A.F.); riccardo.gulli@unibo.it (R.G.)
- <sup>2</sup> Department of Industrial Engineering, School of Engineering and Architecture, (UNIBO), Viale Risorgimento 2, 40126 Bologna, Italy; giovanni.semprini@unibo.it
- \* Correspondence: annarita.ferrante@unibo.it; Tel.: +39-051-2093182

Received: 13 February 2018; Accepted: 8 March 2018; Published: 14 March 2018

Abstract: The paper describes the progress of the four-year European project Pro-GET-onE currently under implementation. This research and innovation project is based on the assumption that greater efficiency, attractiveness, and marketable renovation can only be achieved through an integrated set of technologies where all the different requirements (energy, structural, functional) are optimally managed. Thus, the project focuses on the unprecedented integration of different technologies to achieve a multi-benefit approach that is provided by a closer integration between energy and non-energy related benefits. The project aims to combine different pre-fabricated elements in a unified and integrated system resulting in a higher performance in terms of energy requirements, structural safety, and social sustainability. The project attempts to achieve this goal through the introduction of innovative solutions for building envelopes to optimally combine the climatic, structural, and functional aspects through a significant architectural transformation and a substantial increase of the real estate value of the buildings. This augmented value obtained through the application of the inteGrated Efficient Technologies (GETs) is extremely important when considering the necessity of creating an innovative and attractive market in the energy renovation of existing buildings towards the target of nearly zero energy buildings (nZEBs).

Keywords: building envelope; energy efficiency; seismic improvement; sustainability

#### 1. Introduction and State of the Art

The research project Pro-GET-onE is based on the integration of different technologies to achieve a multi-benefit approach through the closer integration between energy and non-energy related benefits, promoting a holistic vision based on the integration of different technologies where numerous requirements (energy, structural, functional) are managed as a whole. Thus, by implementing a same holistic and integrated system based on pre-assembled components, the research project aimed to achieve the highest performances in terms of:

- 1. energy requirements—by adding (or substituting the existing with) new prefab and plug and play high energy performing envelopes and HVAC (Heating, Ventilation, Air Conditioning) systems;
- 2. safety—using appropriate external structures to increase the overall structural capacity of the building while supporting the new envelope consisting of timber based components for opaque parts/surfaces, and aluminum, glass, PV photovoltaic, solar panels;



 social and economic sustainability—increasing the real estate value of the buildings and the desirability of retrofit options by providing tailored and customized solutions for users, owners and house managers, increasing safeness and minimizing disturbance to inhabitants.

The goal of this research project was to provide the market with an innovative, yet readily implementable system for the building envelope to be applied to an energy, structural, and user-oriented retrofit that would significantly increase the commercial value and the life cycle of buildings, involve the users in attractive and visible solutions and, ultimately, reduce the costs of energy retrofit options in the whole building life cycle. This goal will be attained through the application of solutions for the building envelopes, as well as through optimum climatic-structural-functional management, grounded on the substantial increase of the real estate value of the buildings through significant energy and architectural transformation. This incremented value will be obtained through the development and application of inteGrated Efficient Technologies (GETs) with the strategic aim of creating a new and attractive market in the deep renovation of existing buildings towards the target of nearly zero energy buildings nZEBs [1].

By coupling and combining these technologies, Pro-GET-onE aims to provide existing buildings with poor structural performance with improved, safer seismic performance to get as close as possible to the levels of the European standard EN 1998, Eurocode 8 [2].

Housing in the European Union (EU) represents a huge part of the building stock. EU dwelling stocks account for about 200 million units, representing around 27% of energy consumption in the EU: the potential reduction in  $CO_2$  emissions that energy efficient housing would provide cannot be underestimated. Three quarters of the buildings standing today including the residential stock are expected to remain in use in 2050. So far, only 1.2% of the EU's existing buildings are renovated every year [3,4]. The EU's energy efficiency challenge in buildings mainly concerns the energy efficient refurbishment and investments in its existing building stock. However, there is a clear investment gap in this sector with regards to the private housing market.

The cost-benefit assessments of retrofit actions in this sector have shown excessive payback times (payback times are up to 35–45 years). Furthermore, high investments are required up-front and are generally characterized by a high degree of risk with a potential limited return on investments.

In the Mediterranean and seismic areas of the EU, this gap is even exacerbated, being associated with a strong and generalized lack of confidence by the final users and owners and by weaker market conditions. In fact, the harder economic crisis that these areas are experiencing and the lack of confidence in the perceived sense of safeness in the majority of existing buildings are both major barriers when approaching the subject on building retrofit.

Information from the SHARE Project [5] indicates that Italy, Greece, Romania, and the Mediterranean countries of the European Union as the areas with the highest probability of an earthquake. In these areas, recent seismic events have shown how relevant the issue of seismic vulnerability for existing buildings of reinforced concrete is, given that many of these were designed without any reference to anti-seismic criteria. The evaluation of the vulnerability of the existing buildings and the subsequent assessment of the potential benefit provided by solutions and actions for seismic improvement is a much more complex topic than the design of new earthquake-resistant buildings.

Seismic improvement solutions for existing reinforced concrete buildings can be distinguished according to the number of resistant elements involved and the strategy of the intervention adopted. Local interventions that strengthen the structural elements (beams, columns), reinforce the seismic joints, and secure the vulnerable elements are commonly used. These interventions can increase stiffness, resistance, and eventually ductility at the expense of a significant invasiveness for the users of the building.

Pro-GET-onE proposes a technique that until now has not been commonly used and can be configured as an exoskeleton connected to the reinforced concrete frame of the existing buildings. This new structure can collaborate in order to resist the horizontal seismic actions. Outstanding projects that have adopted this idea are, for example, the Magneti Marelli factory offices and warehouse buildings in Crevalcore (Italy) by Teleios SrI [6,7]. In the office building, the external structure was composed of steel frames connected to the existing reinforced concrete building where the vertical elements had been released from horizontal loads, being completely assigned to the new structure. In the warehouse of the same factory complex, the steel frames were instead inserted inside the building, directly connected to the reinforced concrete portals. Both the interventions ensured full resistance to a designed earthquake according to the current Italian regulation. However, in the described cases, the exoskeleton does not provide integrated solutions for energy improvement and possible volumetric expansion, as in the case of this research project. Another case with a similar approach, and similarly limited to the structural aspect, was the seismic reinforcement of the rural and surveying engineering department of A.U.TH., Thessaloniki, Greece [8]. This project regarded the construction of a steel exoskeleton on the entire perimeter of the existing building, thus providing structural strengthening.

Regarding energy retrofit, several studies have been already carried out to overcome the barriers of high costs and time through technological solutions using prefab systems like TES FAÇADE [9], More Connect [10], and EU Prefabricated Systems for Low Energy Renovation. Most of these solutions are generally founded on the load bearing capacity of the existing buildings, conditions that are rarely applicable in the highly seismic areas of the Mediterranean countries. Thus, in the Mediterranean and seismic areas of the EU, it has become imperative to couple energy retrofits with the development of tools to increase confidence in safeness, and to make it clear that higher initial investments of retrofit are more interesting in the long-term than lower investments with higher paybacks.

#### 2. Multi-Benefit Solutions

As briefly discussed, renovation in buildings implies the solution of issues beyond the energetic sphere (namely structural and seismic safety, fire safety and functional, new technological networks as well as spatial and aesthetical amenities). Undeniable costs and long payback time of renovations led us to consider that an acceptable payback time for energy retrofitting was very difficult to achieve without considering the multiple benefits in economic, social, and environmental terms. In this frame, Pro-GET-onE focuses on the willingness to pay rather than the mere cost reduction. In fact, the core element of every redevelopment is the increase in value for the client (investor, building owner, and tenant), since focusing solely on the optimization of energy efficiency may result in failing to meet the overall requirements.

#### 2.1. Structural Requirements

Regarding the structural response under seismic loading, simulations using FEM software (EN 1998) performed for different residential buildings have shown an overall reduction of horizontal displacements and internal forces of the retrofitted structures. From the perspective of maximum compatibility and a minimum invasiveness, the overarching goal of the strategy adopted was to provide an intervention that increased the capacity of the building as a whole and only secondarily acted locally on existing vulnerabilities, minimizing and/or avoiding interventions that are not cost-effective and very invasive when applied extensively. External metal bracings like exoskeletons are a suitable solution to increase the capacity of existing structures when subjected to horizontal actions by increasing the rigidity, and consequently reducing the displacements (Figure 1).

More in-depth analyses were carried out. One of the virtual cases is reported in this paragraph. Following the Italian code and guidelines NTC 2008 (which refers to the Eurocode 8), modal analysis with response spectrum (or linear dynamic) was performed. In this case, the equilibrium was treated dynamically, and the seismic action modeled directly through the acceleration project spectra obtained from the seismic parameters related to a seismic zone selected by the authors (Table 1). The results shown below correspond to a seismic action determined with the following data:



Table 1. Seismic parameters of Bologna.

**Figure 1.** Model of the existing (**a**) and the new structures (**b**)—GET structure; (**c**)—GET structures connected on top—portal. This figure shows, as an example, the contribution of GETs on an existing reinforced concrete structure with columns and shear walls. The new structure (**b**)—in this example, steel columns and beams, steel stiffeners and XLAM plates when connected to the nodes of the existing building resulted in a reduction of displacements and internal forces in the existing structure (**a**). Further improvements resulted when the additional structures on each façade were connected on the top of the building (**c**).

The selected parameters allowed the evaluation of the seismic structural efficiency of the different design solutions of the GET system. As a direct consequence of earthquakes, the absolute displacements of the structure were used. As established by the code, the structure was analyzed with combined actions (SLV and SLD as indicated by the NTC 2008), with the displacements reported in both cases. Only the results of the SLV combination (coincident with the significant damage limit state for Eurocode 8) are shown since they were more severe. As part of the procedures for the analysis of structures, the fundamental period of vibration is an important feature in the evaluation of the stresses caused by the seismic action. The variations of the period depend on the mass and stiffness of the structures. The application of the steel external structure connected to the existing reinforced concrete building increases the rigidity of the structure k with a minimum mass increase, resulting in a decrease in the structure's period. In linear analysis methods, the identification of the period leads the estimation of the horizontal forces of the project. In general, it connects the capacity to seismic requirement, in order to determine the expected performance and therefore the safety design. Most common types of reinforced concrete construction were conceived and built largely between the 60s and the 90s when no clear characterization of the seismic territory had yet to be defined and horizontal earthquake actions were not considered. They were designed for only static loads or according to obsolete/poor seismic design criteria; thus, they resulted in small elemental sections and irregular rigidity distribution. To highlight the effectiveness of Pro-GET-onE, reinforced concrete buildings that had a geometric irregularity in the plan, with predominantly longitudinal development, were considered (Figure 2). Smaller section dimensions and different mechanical properties of the materials result in less resistance, leading to longer periods and displacements with a wide margin of improvement (Table 2).

The additional structure provided by the project consists of steel frame (two columns and a beam) for each floor, with bracings in the transversal direction, connected to the existing reinforced concrete frame at the column-beam joints. These frames are also connected in the longitudinal direction to create a space frame together with the existing one.
The first design solution provides a continuous addition on both longitudinal facades (Figure 3a) and therefore increases the depth of the building on the smaller side to give it a more regular geometry in plan and an increase in stiffness to cause a decrease in displacements and profile stresses.



Figure 2. Plan and finite element model of the virtual case study, below there are also the main analysis results.

Max	imum Displacen	nents	Structural I	Periods
Node	U1 (m)	U2 (m)	Vibration Mode	T—Period
127	0.1148	0.1618	1	2.761
			2	2.197
139	0.1108	0.1617	3	1.818
			4	0.896

 Table 2. Maximum displacements and periods of the initial state structure.

The second solution is instead an alternate addition on the same facades (Figure 3b). This attempt provides a reduction in the amount of steel used and therefore of material costs. The alternation of exploitable spaces allows an increase in the presence of bracing frames, resulting in a substantial transversal displacement reduction. The third design solution provides an improved connection between the two lateral frames made by trusses (Figure 3c).

SLV (SD) 1 I.S. -Δ- min. -Δ- MAX.	
	N N P N - N - N M W

|--|--|

SLV (SD)	U2 (m)	-Δ- U2 (%)	U1 (m)	-Δ- U1 (%)	(a)	T-Period	-Δ-T (%)
I.S.	0.162	ı	0.115	ı	I.S.	2.761	
-A- min.	0.099	38.75	0.080	30.75	-A- min.	1.499	45.71
:	:	:	:	:	:	:	:
-Δ- МАХ.	0.052	68.05	0.056	51.22	-Δ- MAX	1.144	58.57

SLV (SD)         U2 (m)         -Δ-U2 (%)         U1 (m)         -Δ-U1 (%)         (b)         T – Period         -Δ-T (%)           1.S.         0.162         -         0.115         -         1.S.         2.571         -           -Δ-min.         0.096         40.67         0.082         28.57         -Δ-min.         1.440         47.85           -Δ-MAX.         0.056         65.64         0.062         45.82         -Δ-MAX         1.382         57.19								
LS. 0.162 - 0.115 - L.S. 2.761	SLV (SD)	U2 (m)	-Δ- U2 (%)	U1 (m)	-Δ- U1 (%)	(p)	T-Period	-A-T (%)
- <b>Δ-min</b> 0.096 40.67 0.082 28.57 - <b>Δ-min</b> 1.440 47.85 	I.S.	0.162		0.115		I.S.	2.761	
······································	-Δ- min.	0.096	40.67	0.082	28.57	-Δ- min.	1.440	47.85
- <b>A</b> - <b>MAX</b> . 0.056 65.64 0.062 45.82 - <b>A</b> - <b>MAX</b> 1.182 57.19	:	:	:	:	:	:	:	:
	- <b>Δ- MAX</b> .	0.056	65.64	0.062	45.82	-A- MAX	1.182	57.19

SLV (SD)	U2 (m)	-Δ- U2 (%)	U1 (m)	-Δ- U1 (%)	(c)	T-Period	-A-T (%)
I.S.	0.162	ı	0.115	1	I.S.	2.761	
-Δ- min.	0.099	38.75	0.080	30.75	-Δ- min.	1.486	46.18
:	:	:	:	:	:	:	:
-Δ- MAX.	0.052	68.05	0.056	51.22	-A- MAX	1.149	58.39

Figure 3. Finite element models of the three design solutions (a-c) and the related analysis results. SLV, load combination of "stato limite di salvaguardia della vita" (NTC 2008) equivalent to *significant damage* (SD, Eurocode 8); I.S., initial state; -Δ-, difference; T, period of vibration. The range of results is determined by changes in the new structure as shown in Table 3 below.

Profiles used in the new structure	HEA 240 HEA 300	↓	Ļ	δ max	Ļ	$T_1$
Depth of the addition	1.5 m 2.5 m	↓	Ļ	δ max	1	$T_1$
Connection constraint between the two structures	Hinge joints Rigid joints	↓	Ļ	δ max	=	T <sub>1</sub>

Table 3. Variation and consequences obtained from FEM models.

In each of the designed solutions, these variations have the same effects. The increase in the size of the profile used clearly results in an improvement as it increases the overall structure's rigidity. The depth of the exoskeleton brings out conflicting effects: on the one hand, wider depths produce minor displacements as the resulting building (given by the existing and the added volumes) is more rigid; on the other hand, the same increase in depth comes across with a slightly worse period of the structure due to an increase in floor masses despite minor stiffness improvements.

The GET system introduces a metal structure with efficient stiffness; furthermore, the GET is applied externally to the existing building with a beneficial effect in terms of construction site management given that it does not require the performance of special operations inside the existing building. The installation is less complicated with respect to the usual insertion of new reinforced concrete structures within the existing building. Moreover, with respect to this insertion, the GET structure implies a significant reduction in terms of cost for the new foundations.

Regarding the connections, the use of rigid joints results in a reduction in both displacement and period. However, it should be highlighted that this connection is not easily implemented in the construction practice. This is especially true in the case of a connection between an existing reinforced concrete structure and a new steel frame. The element of great importance for the system is the connection linking the existing structure in reinforced concrete and the new metal structure "GET". To create an effective collaboration for horizontal actions while avoiding burdening the existing structure with vertical loads, this joint is assumed as a vertically sliding joint that allows only vertical movements.

### 2.2. Energy Requirements

The GET structure will be combined with energy and space requirements. To this aim, energy simulations performed have demonstrated that enclosing the structure with solar spaces that can be opened in summer may provide an energy reduction of up to the 75% in the cold winter season while reducing solar gains and increasing natural ventilation rates, thus achieving about 35% energy consumption. Simulations have been performed from northern climates to the Mediterranean area reaching nZEB performances with traditional thermal insulation coating combined with controlled mechanical ventilation (VMC). Many references in the literature have also confirmed this potential energy reduction [11]. Specific calculations have been performed for the three cases of the research project by using a monthly-based method according to the EN 13,790 standards (Figure 4).

Concerning technical HVAC plants, the GET system can be coupled with new network lines (thermal fluids, electricity, etc.) with the predisposition for future systems (i.e., water drainage pipes, telecommunication lines) to be integrated in the external structure for a "plug-and-play" connection with internal devices. External allocation of all main plant system (EHP, PV system, hydraulic pipes, electric lines) will allow for simple plant maintenance and/or substitution. As a whole, GET can be equipped with several installation plants. The structure may also be used as support for the telecoms infrastructure such as to ensure easy access to superfast broadband services as required by new EU directives for new and renovated buildings [12]. Thus, Pro-GET-onE proposes the highest transformation of an existing building shell with external strengthening structures that generate

energy efficient buffer zones (by reducing radiation in summer, providing solar heating in winter, and hosting flexible/adaptable plug-and-play installations), and increase the flat volume (with balconies, loggias, sunspaces, and extra rooms, according to the users' needs or expectations). See section of the GET system in Figure 5. These solutions have been designed and analyzed in a large set of existing buildings [11].



**Figure 4.** Simulations in three of the case studies show how combining the enclosure of volumes with insulation of existing envelopes will save up to 75% of the energy consumption.



**Figure 5.** Exoskeleton providing existing building (5) with: strengthening by GET structure (2), energy saving and plant distribution (1, 4, 6) increased comfort and living areas for residents, and additional new units (3).

Fotopoulou et al. [13] investigated an individual residential unit in a set of various hypotheses for targeted energy retrofitting interventions with different options both individually and in combination. The study was executed for three different climatic zones (Athens, Riga, Bologna) and showed that energy savings were larger during the winter period in southern climatic conditions while northern countries showed a larger energy saving during summer. Undoubtedly, in all three different climatic conditions, a zero energy building with the extension on the façade and with a standard retrofit seems to be an achievable goal.

Simulations of different scenarios of the additions on the existing building resulted in corresponding diverse energy performances, from the very low grade of performance in the "as built" scenario of the existing building, and up to nearly zero energy demand for selected technological solutions applied in specific climatic contexts. In different ways, the results proved that façade additions were very effective; therefore, the additional building envelope is a powerful technological solution combining the improved energy performance of the buildings with a new aesthetic/formal quality.

# 2.3. User Oriented Requirements

Last, but not least, the increased value of the buildings and the users that can benefit from the extended space, is quite clear. The resulting building may finally provide:

- comfort for users (combined with a proper ventilation system); and
- increased attractiveness even from social sectors that are usually more reluctant to change like elderly inhabitants, providing them with balconies and loggias for small individual gardens.

Potential problems that can be expected in terms of daylight reduction can be tackled by evaluating the appropriate depth of GET structure, the use of highly reflective surfaces for the internal coatings, the potential application of solar pipes to achieve visual comfort, and energy requirements while maintaining the seismic improvement. Thermal bridges are also foreseen. Thus, effort will be concentrated in searching for adequate materials and/or geometrical features for the structural joints to minimize thermal bridges while preserving the necessary structural cohesion. Depending on the original structural performance of the building, the structural frame can be designed according to different geometrical solutions and materials (aluminum, steel and wood). Figure 6 outlines different possible solutions of additional structures on the existing building's envelope.



**Figure 6.** Interactions between the existing façade and the steel aluminum structure (**a**); the possible positioning of ducts/pipes/storage (**b**); and the same external structure in timber or X-Lam structure (**c**). Different options for enclosures on the steel/aluminum.

To sum up, rather than the innovation in products, the Pro-GET-onE innovative aspect relies on the approach of the user and the building at the center of the energy retrofit to successfully implement energy strategies and solutions for deep renovation.

## 2.4. Economic Viability of the GET System

Undoubtedly, the GET systems will be characterized by higher up-front costs with respect to a deep energy renovation. Nonetheless, if we consider the associated costs for seismic retrofitting in the case of a standard seismic renovation, the GET system can produce a significant cost saving. In fact, in the proposed strategy, the avoided disturbance for residents and the moving costs for an average cost reduction of the GET system must be considered and compared to the case of the standard seismic and energy retrofit.

The comparison between the estimated unit costs/time (indicative costs) of a typical deep renovation and the GET system is presented in the following table (Figure 7). The various interventions were divided into the three main requirements of the project: the energy renovation, structural safety, and limited disturbance to the users.

Pro-GET-onE meets the target of 15% of cost reduction when compared to a typical renovation (i.e., renovation that meets the minimum energy and seismic safety requirement). In particular, this is achieved by summing up the construction costs of:

- 1. Energy renovation: the standard renovation costs are estimated at around  $360 \text{ euro/m}^2$  when compared to Pro-GET-onE where the renovation costs are  $380 \text{ euro/m}^2$
- 2. Structural safety: the standard renovation costs are estimated around 390 euro/ $m^2$  when compared to Pro-GET-onE where the renovation costs are 330 euro/ $m^2$
- 3. Inhabitants' relocation: the standard renovation costs include a quota of 100 euro/m<sup>2</sup>; this cost is avoided through Pro-GET-onE, which allows the inhabitants to stay in the building during renovation. The actual cost reduction is therefore about 16.5%. Moreover, it is also significant to consider the added value in economic terms consisting of the extra surface generated by the Pro-GET-onE system. The real estate increased unit value has been evaluated to be around 130–180 euro/m<sup>2</sup> depending on the different regional markets. This consideration reduces

the payback time and increases the impact of the project on the economical side. It can be stated that Pro-GET-onE achieves a unit cost reduction of up to 32–38% when compared to a typical renovation.

	TYPICAL DEEP RENOVATION			PRO-GET-ONE SYSTEM RENO	VATION	
MEET ENERGY	INTERVENTIONS	Cost €/m <sup>2</sup>	Days	INTERVENTIONS	Cost €/m <sup>2</sup>	Days
REQUIREMENTS	External thermal insulation + finishing systems	60	90	PRO-GET-ONE standard system (structural not included)	90	60
	Windows replacement	70	30	Windows replacement	80	30
	HVAC and water heating system improvements/replacements	80	90	HVAC and water heating system improvements/replacements, plug and play	80	60
	Related demolitions and reconstructions	30	30	Related demolitions and reconstructions	0	0
	Scaffoldings and safety installations	30	240	Scaffoldings and safety installations	10	0
	New renewable energy systems	100	30	PRO-GET-ONE standard renewable energy systems	100	30
	TOTAL CONSTRUCTION COSTS AND DURATION	360	240	TOTAL COSTRUCTION COSTS AND DURATION	380	60
	Maintenance and replacements (25 years cycle, heating/cooling running costs not included)	135		Maintenance and replacements (25 years cycle, heating/cooling running costs not included)	115	
MEET SAFETY REQUIREMENTS	INTERVENTIONS	Unit Cost €/m²	Days	INTERVENTIONS	Unit Cost €/m²	Days
	New reinforced concrete structures (e.g., shear walls) + foundations	350	180	PRO-GET-ONE steel and wood structure + foundations	320	60
	Demolitions and reconstructions related to new structures (e.g., floor replacement)	40	60	Demolitions and reconstructions related to new structures	10	10
	TOTAL CONSTRUCTION COSTS AND DURATION	390	240	TOTAL CONSTRUCTION COSTS AND DURATION	330	70
	Maintenance and replacements (25 years cycle)	5		Maintenance and replacements (25 years cycle)	25	
MEET USER REQUIREMENTS	INTERVENTIONS	Unit Cost €/m2	Days	INTERVENTIONS	Unit Cost €/m²	Days
	Inhabitants relocation (no tailored design)	100	360	Inhabitants relocation (user-oriented design)	0	0
ALL REQUIREMENTS	TOTAL CONSTRUCTION COSTS	850		TOTAL CONSTRUCTION COSTS Per m <sup>2</sup> of existing UFA	710	
				TOTAL CONSTRUCTION COSTS Per m <sup>2</sup> of existing UFA plus extra surface (+20% of UFA)	560	
	LIFE CYCLE COSTS (after 25 years, excluding energy running costs)	990		LIFE CYCLE COSTS (after 25 years, excluding energy running costs)	850	
	EXPECTED REAL ESTATE VALUE AFTER INTERVENTION	+15%		EXPECTED REAL ESTATE VALUE AFTER INTERVENTION	+50%	

Figure 7. Comparison of construction unit costs between a typical and GETs deep renovation.

### 3. Results of a Case Study

One of Pro-GET-onE feasibility studies was located in Greece, more precisely in Peristeri, a suburban municipality of Athens in the Attica region. It has a population of 146,000 inhabitants and is located at a distance of 5 km in the western part of Athens and is the biggest and the second densest suburb of the Attica region. The pilot case of the Peristeri compound (Athens, Greece) is a typical social housing development from the late 1960s.

The main structure of the buildings is reinforced concrete (pillars and beams), concrete slabs, and hollow brick external walls. This is a typical construction typology and is globally presented in all Attican suburbs and the city center. It also has a common structure with similar building blocks all over Europe.

Each building block has a centralized heating system plant. Existing windows are made of an aluminum or wooden frame with single glass although part of the external windows has already been replaced. The energy performance of the buildings is very low and in need of energy retrofit ( $160 \text{ Kwh/m}^2 \times \text{y}$  (winter)/ $110 \text{ Kwh/m}^2 \times \text{y}$  (summer)).

The majority of these buildings are residential use with the only exception being the building block (B6) where small businesses shops, offices, and retail are located at the ground floor level. Ownership is 100% by private owners. In total, there are 550 apartments while the average heated residential area per unit is 85 m<sup>2</sup> (Figure 8).



Figure 8. Peristeri urban compound where is located the feasibility study.

A retrofitting project that can guarantee a high level of energy performance can be a source of savings for the inhabitants and municipalities. High-energy costs have led the majority of the residents to choose alternative and less efficient heating solutions such as kerosene, electricity, coal or wood, increasing environmental pollution. The large number of standardized multi-apartment residential blocks leads to the possibility of adopting similar solutions to improve energy efficiency, thus ensuring an economy of scale.

The building identified in compound (A7) is the one outlined in red in Figure 8.

The application of the project could guarantee a substantial energy improvement and would not be limited to this aspect. As we have already seen, the benefits of the "GET" structure would span the improvement of earthquake performance, which is a fundamental aspect to increase the value of the intervention.

Regarding the seismic classification of Greece, as for the Italian case, the territory is divided into zones. The four zones that were created are characterized by a probability of excess of 10% in a reference period of 50 years and a return period of 475 years (PGA values assigned to areas with soil type A). Figure 9 and Table 4 show the seismic subdivision of the territory as reported in the Greek standards, EAK 2000 [14].



Figure 9. Seismic areas of Greece, as reported in EAK 2000 [14].

Table 4. Subdivision of the seismic areas of Greece.

Seismic Zone	Anchor Acceleration Values
Ι	0.12 g
Π	0.16 g
III	0.24 g
IV	0.36 g

Following this classification, Athens is located in Zone II with an anchor acceleration value of 0.16 g to be applied in the definition of the response spectrum to carry out the checks. In the analysis phase, the structure was subjected to a greater acceleration to highlight the results of the system.

### 3.1. Seismic Analysis

The building is from the 60s and has a longitudinal shape with a reinforced concrete structure. It was built in the period after World War II when there was a "boom" in construction of this type in the suburbs of all European cities. Unlike the Italian case, despite the construction period, this building testifies to the already present conception of seismic design, as demonstrated by the dimensions of the structural elements.

It has a reinforced concrete structure with a mainly longitudinal development. It is composed by frames arranged in the direction of the shorter side. The concrete slabs lie on these frames. There are also two secondary frames on the edges and one in the middle characterized by the presence of flat beams. A twenty-centimeter reinforced concrete wall can be identified near the stairwells and the elevators.

The original geometrical and architectural data of the initial state were provided to the authors by the municipality of Peristeri. The dimensions and the structural schemes of the beams were obtained by photographic survey, while those of the columns and of the concrete walls were taken from the original architectural plans.

The structure is composed of six units with an average span 6.60 m in the longitudinal development, (the two external ones are of about 6.55 m while the four internal ones have a distance of 6.8 m). Transversely, the space is divided into two zones of a spacing of 3.25 m and 6.05 m. On the ground floor is the *pilotis* while the six upper floors are dedicated to residential units (Figure 10).



Figure 10. Structural plan—Type plan.

The simulations carried out were done with finite elements software SAP2000 [15]. Linear dynamics (modal analysis with response spectrum) was chosen for the seismic analyses. Table 5 shows data regarding the modeling phase, while Table 6 indicates the seismic parameter for the response spectrum definition for the analyses and for the verifications.

Table 5	. Modeling	data.
---------	------------	-------

Analysis	Linear Dynamics
Modal combination	CQC (§ 4.3.3.2 Eurocode 8)
Eccentricity value	5%
Directional combination	$1.00 \cdot E_x + 0.30 \cdot E_y + 0.30 \cdot E_z$
Limit state	Significant Damage—SD
Behavior factor	<i>q</i> = 2.00
Reference parameters in assessments	Maximum absolute displacements—δmax Structural period—T1

Table 6. Seismic parameters for the definition of the elastic response spectrum.

Analysis
SLV (SD)— $P_{VR}$ = 10%; $T_R$ = 475 years; $V_R$ = 50 years
• $a_g/g = 0.259$
• $F_0 = 2.363$
• $T_{\rm C} = 0.342$
Verification—Seismic Zone II (EAK 2000)
SLV (SD)— $P_{VR}$ = 10%; $T_R$ = 475 years; $V_R$ = 50 years
• $a_g/g = 0.16$

In this section, we refer to two parameters regarding the results of the proposed structure: the absolute displacements at the top of the existing building, and the period of the structure.

The assumed structure composed briefly as shown above in the structural plan was modeled in SAP2000 [15] through the use of linear elements (for beams and columns) and bilinear elements, shells (for concrete walls). Figure 11 reports the main results of the linear dynamics analysis of the initial state.



Figure 11. Displacements and structural period of the linear dynamics analysis of the initial state.

Regarding the building's seismic response, it is evident that it was designed to withstand seismic actions. In fact, despite the high level of the applied seismic action, there are limited displacements considering the height of the building. From the analyses carried out after the application of the "GET" system, it was therefore predictable to obtain very limited improvements when compared to the virtual cases previously described.

The additional structure provided for the project consists of steel frames (columns and beams) for each floor, braced in the transversal direction due to the architectural requirements, and linked to the joints of the existing reinforced concrete frame (created from the intersection of beams and pillars). These frames are also connected in the longitudinal direction to create a spatial frame interconnected with the existing structure.

The depth increase of the smaller side of the building allows it to obtain more regularity in plan and to increase the stiffness, causing a reduction of displacements and internal forces in the elements.

The first steel structure solution (solution A, shown in Figure 12) is continuous on the whole façade and presents transversal bracing at the middle and at the ends of the new structure.

Here we show the results due to the variation of the profiles that compose the structure (columns and beams). The improvement is calculated on the comparison before and after the application of the steel structure.

In this phase, several profiles were examined initially by varying the type at the same weight and subsequently by varying the type with the same height in order to obtain a general scheme. Based on these analyses, the incidence of the profiles has been verified.

It has been verified that the choice of the columns is decisive in the improvement and that the types of profiles that have a different inertia module in the two directions (e.g., IPE type) are inconvenient. In fact, in some cases, these profiles aggravate the displacements in the longitudinal direction. The other way around, profiles with equivalent stiffness or almost in the two directions are ideal for the intervention (HE and pipes). Clearly, HE profiles are preferable during the assembly procedure.



**Figure 12.** Finite element model of solution A, structural schemes of the new external structure and results in displacements and periods.

The following analyses were carried out with the use of HEB 240 and  $\Phi$ 323.9/12 profiles. The results obtained using the last indicated profile are shown below, but follow different structural configurations as described in Section 2.1:

- A—Continuous addition to the longitudinal façades. It presents transversal bracing at the middle and at the ends of the new structure.
- B—Alternate addition to the longitudinal façades. It presents transversal bracing at each span.
- C—Alternate addition to the longitudinal façades combined with continuous addition on the transversals.
- D—Alternate addition to the longitudinal façades, continuous addition on the transversals plus top connection with reticular beams.

Looking at the results obtained from the analyses (Figure 13), design solution B had more diagonal bracing on the transversal planes and smaller displacements in the Y (U2) direction. Moreover, this solution turned out to be the least expensive by using lower quantities of steel in the project.



**Figure 13.** Finite element model of solution A, structural schemes of the new external structure, and results in displacements and periods.

Solution C had an alternate addition to the longitudinal facades and a continuous addition on the transversals. The addition on the short sides was not connected to the existing structure, but only to the longitudinal ones. This helped to improve the performance of the structure in both directions to the detriment of an increase in costs countered by a small increase in useful area.

The last solution of the analysis phase provided a superior connection between the lateral additions, which produced a decrease in displacements against an increase in the period due to the rise in building height.

Finally, by also combining an assessment of the costs of the structure, a check of the elements of the new steel frame was carried out. In this phase, all the profiles constituting the new structure were defined and differentiated based on internal stress and on the cost calculation dependent on the weight of steel. The aim was to find a fair compromise between the construction costs and performance in terms of improvement achieved on the existing structure.

Another fundamental aspect was represented by a parallel evaluation of the added surfaces due to the volumetric external addition. The added value given by these areas reduced the expenses.

As reported at the beginning of the section regarding the verifications (Eurocode 3 and 8), the seismic load relating to Athens was used based on the EAK2000 [14].

Two structural solutions were considered due to different benefits:

- B1—Alternate addition to the longitudinal façades (Figure 14a). This solution involves the greatest performance benefit with minimal cost; however, it has a small contribution in terms of added area.
- E1—Alternate addition to the longitudinal façades, continuous addition on the transversals plus top connection with Vierendeel beams (Figure 14b). A different solution to the previous ones is illustrated below. Thanks to the possibility of raising the structure of a floor, this allows the greatest contribution in terms of added surface, a higher cost with equal performance benefit on displacements.



**Figure 14.** Axonometric drawings of the two selected solutions. On the left (**a**) is presented the alternate addition to the longitudinal façades B1, while on the right (**b**) there is the alternate addition to the longitudinal façades, continuous addition on the transversals plus top connection with Vierendeel beams E1.

Regarding the design solution E1, Figure 15 shows the schemes and the analyses output data.

Furthermore, the actual benefit that the Pro-GET-onE implies for the existing construction in terms of earthquake response was verified. Various design solutions were analyzed that involved variable quantities of added area, and used material (steel), which produced different responses to the earthquake. Overall, an improvement in terms of displacements was always obtained due to the increase in stiffness given by the addition of the new structure. The analysis carried out initially focused on the maximization of the benefits on the existing construction and afterwards, on the most valid design solutions (in terms of performance), a compromise has been proposed to ensure a seismic improvement at the lowest possible construction cost (considering only the structural components of the project).

Two proposals have been made (B1 and E1), which guarantee good performance in terms of transversal displacements (16–17% improvement), a substantial indifference for the displacements in the longitudinal development of the construction, and a large increase in added surface that reduced the cost of construction. Additionally, the assessments on the construction cost showed that both solutions had the same cost per square meter of added surface.

In conclusion, as far as regarding seismic safety, this case study verified that the GET system could provide improvements even for buildings that have already been designed with an adequate performance for horizontal loads. Certainly, the value of improvement for the previous buildings mentioned, was limited when compared to the structures that have been designed to withstand only vertical loads, in which cases the GET system could prove better safety results.



Figure 15. Finite element model of solution E1, structural schemes of the new external structure, and results in displacements and periods.

## 3.2. Architectural Verification—The Abacus

Parallel to the seismic improvement conferred by the external steel structure, a series of possible architectural solutions have been developed to incorporate the skeleton initially formed by the single structural component. In this phase, it is necessary to consider several factors that determine the appearance and the way of using the additional space by choosing constructive solutions, materials, and the functional types. Highlights of this design typology include the versatility of the additional volumetric units in relation to the possibilities and the choices of the user, and the constant search of energy improvement aimed at the goal of bringing the existing building toward the nZEB.

Currently, one of the main shortcomings of deep retrofitting towards nearly zero energy is that they generally rely on separate clusters of technologies that are difficult to integrate. To overcome these barriers and create a roadmap for cost effective renovation through a well-balanced strategy of mass customization, the research project has envisaged an integrated modular system, composed of components manufactured off-site, that can be customized and optimized for different cases in a user-oriented perspective (by adding balconies, loggias, sunspaces according to the users' needs and expectations). The integration of the system will focus on the interfaces between the different components to ensure their collective performance according to the project requirements. Standardized interfaces will also ensure the flexibility of the system, as different components can be interchanged and adjusted as a function of different climate conditions and urban context, as well as according to the inhabitants' requirement.

Regarding the Greek building, different architectural hypotheses have been realized. In each of these solutions, several additional volumetric units were hypothesized and divided into three functional types: sunspace, extra-room, and balcony. Figure 16 shows a possible functional and therefore architectural variation of the same external volumetric addition.



**Figure 16.** Possible functional variation of the same external volumetric addition made by the alternate combination of sunspace, balcony, or extra-room. Every plan solution determines a different façade conformation. Solution (**a**) refers to an extension with the option of a balcony, where (**b**) is a combination of an extra room with a balcony, (**c**) refers to the combination of a sunspace with a balcony and finally (**d**) is the combination of an extra room and a sunspace.

Performed cost-benefit analysis in a large set of reference buildings in the context of another EU project ABRACADABRA [16] that considered the hypothetic investment in additional units on top of GETs showed that the potential economic gains obtained through the sale would largely compensate the energy retrofit cost including RES to set the energy demand of the whole building to zero. The GET

system, in fact, could be used to support additional loads on top of buildings that were not structurally conceived for addition. This aspect could implement and accelerate the market penetration of deep renovation within the private sector, which is the most challenging sector to overcome the existing barriers in energy retrofit market uptake. In fact, energy-retrofitting actions are very often implemented in over-imposed actions by the main ownership, with no direct benefits to the final users.

To overcome this limit, it is necessary to focus on the local private owners of real built environments where owners may directly benefit from the economic and spatial gains. Different options of façade adds-on to be integrated on the vertical surfaces of the existing buildings will be studied and categorized in a comprehensive abacus containing the different solution along with the variable measures/materials/technologies to be adopted. The possible modifications in the façade modules will be studied according the main structural frame and the residential units' utilities.

They will be grouped in one abacus of possibilities that will become one of the main design tools for planners and professionals involved in the GET process. In fact, the abacus can be tailored and customized as a function of different construction elements and architecture in the different case studies and it will represent the catalogue of a new production line for a possible joint participation between SME partners. The abacus is designed to define classification criteria, to launch an open energy performance and architectural repository to be used as an unlocked resource where energy professionals and major users like home-owners, tenants, condominium's administrators, etc. may find technical tools to deep renovate housing residential buildings.

In terms of the architectural solutions, the development phase of the technical solutions for the realization of the horizontal and vertical partitions of the GET system is under process. Therefore, the Peristeri case application is the first approach to this detailed phase of design that will lead to the definition of the integrated technical solution. Figure 17 shows drawings of another example of application.



Figure 17. Façade, plan and detailed section of a typical application of the GET system on the Peristeri building.

Different material, technical, and functional choices involve different compositional solutions. On the basis of the design hypotheses carried out on the Peristeri case, perspective views have been made of what may result following the realization of the Pro-GET-onE (Figure 18).



Figure 18. Possible façade verification of the application of the GET system on the Peristeri building.

Outstanding examples in architecture building practice so far that could be considered as the inspiration behind the GET strategy include the transformation of the Tour Bois Le Pretre by the French architect Frédéric Druot [17,18]. This is a significant example of deep renovation combining energy retrofit with architectural quality and social sustainability. Another significant example is the extension and refurbishment of the residential Tower Weberstrasse, Winterthur, an existing 12 level apartment tower, built at the 1960s by the architect H. Isler. The project refers to an extension of the rear facade planned by Bulkhalter Sumi architekten [19].

In both examples, the additional space led to the variation of the existing apartments, the increase of a greater sense of security, and at the same time upgraded the social life of the community with the active participation of the owners during the whole procedure. New "envelopes" often consist of architectural spaces and units: the new volumes with the winter gardens, the extra balconies and galleries create a transition zone between the existing building envelope and the external climatic conditions and that results, as reported in the reference cases, in a consistent decrease in the initial energy consumption.

The architectural solutions here described, starting from a non-energy related objective like the increase of the rentable surface and, more generally, the increase of the asset value, prove that technological and architectural transformation in buildings do have the highest potential to decrease energy consumption in the existing ones. Regarding the case study of the Tower Weberstrasse [19], the measured energy consumptions calculated on the heated surface before the renovation (3887 m<sup>2</sup>) was calculated up to 604,244 kWh/year. This resulted in 155 kWh/m<sup>2</sup> per year in terms of gas consumption. After the renovation, the calculated energy consumption accounted for a global 61.5 kWh/m<sup>2</sup> per year, considering a total increased surface of 4.830 m<sup>2</sup>, thus, including the addition.

# 4. Field of Application and Potential Impact

The field of investigation and design is limited to existing buildings from the 1950s and 60s onwards. Indeed, this is not an actual limit, since these buildings represent the large majority of the EU building stock and are the biggest source of energy loss. Taking into account that today, only 1% per year of the existing buildings are renovated, it is obvious that these buildings embed a great potential in terms of impact on the building construction sector for energy, architectural, and economical reasons.

The project, by setting the ambitious target of achieving nZEBs in the most critical cluster of buildings (the building blocks from the 1970s onwards represent the most inefficient buildings located in the poorer and seismic areas of the Mediterranean) addresses:

- The large majority of urban areas in the EU; as reported in noticeable studies in the EU 27 the peri-urban areas represent the larger majority with respect to the central urban areas. Indeed, recent studies [20] have revealed how dense peripheral contexts preserve large areas for possible densification.
- The large majority of the EU building stock. As a matter of fact, about 70% of buildings in the EU have been built after the Second World War (60s to 90s) and well before the entry into force of regulatory measures on energy consumption reduction [21].
- The most inefficient buildings in all their different sizes and type (single family houses, multifamily houses and high-rise buildings);
- Buildings where change and transformation are feasible (due to free or available areas) and beneficial for energy, architectural, economical, and social reasons.

Furthermore, Pro-GET-onE aims at addressing a large sector of the residential stock: the ones owned by owner-occupiers, private landlords, and social housing companies. In particular, the proposal largely aims to address the accommodation of almost all EU citizens: indeed just over seven out of 10 (70.6%) people in the EU-28 lived in owner-occupied dwellings, while 18.5% were tenants with a market price rent, and 10.9% were tenants in reduced-rent or free accommodation.

The information required in the design, construction, and operation of facilities from their inception onward, will be based on computer-generated models containing accurate geometry and relevant data needed to support the whole lifecycle of buildings. This concept is referred to as Building Information Modelling, as pointed out by Charles Eastman [22] and by many other scientific works [23–29].

The production process of components in off-site factories, their supply to the construction site, and the onsite assembly procedures will have to be optimized by a BIM-based process to maximize the workflow and project efficiency.

Many firms are already using BIM design to collaborate with general contractors and construction managers, to automate production and prefabricate building components such as mechanical equipment and curtain-wall systems. In Pro-GET-onE, BIM will be exploited as a pipelined process among researchers, designers, managers, engineers, architects, and contractors, all sharing a common language made of digital representations. Three kinds of knowledge will be characterized to properly model the facility: knowledge about object shapes (survey), knowledge about objects identities (metadata implementing), and knowledge about the relationships between elements (BIM semantic modelling aimed at analysis).

This will enable both the minimization of the embedded energy of the system, and the minimization of its costs. The planning, simulation, and optimization of the different processes occurring in building construction will be enabled through the following tools and methodologies:

- A Building Information Modelling (BIM) software platform for pre-production design coordination, interface checking, and clash detection;
- A Process Information Modelling (PIM) framework that combines BIM with an applied kinematics
  assembly simulation system to simulate and optimize the production processes taking into account
  the material attributes and production speed. The PIM framework will be descriptive, prescriptive,
  and explanatory. While BIM is a good approach for these aspects, a wider system of connections
  has to be layered to become PIM-compliant, so during the executive stage, a proper BIM execution
  plan will be delivered to implement the advantages of PIM.

### 5. Expected Future Impact

The current impact of the research project is based on 24 dwelling units of about 100 sqm. each that will be renovated in the Groningen area. Other case studies consist of the project demo-building in Athens (about 2600 sqm.), and a building in the Reggio Emilia district (about 1000 sqm.). In the following years, more than 113 dwelling units will be used as regional specific buildings in the Groningen area. However, to calculate the large scale potential impact of the project, new calculation techniques and methods based on parametric modeling will be developed to improve the speed and efficiency of the calculations for individually different residential buildings and the surrounding available area for different seismic target regions.

The integrated and multi-purpose nature of Pro-GET-onE determines the high level of effectiveness of the proposal. Moreover, the consolidated knowledge used in the seismic technical solution proposed and its combination of different products already available on the market in the definition of a new system are the major outcomes. By assessing and answering—in one prefabricated solution—the energy, structural, and fire safety needs, together with the possibility of integrating the personalization of different components within the same mass-produced product, Pro-GET-onE opens a methodological revolution in the retrofitting practice. This highly innovative and effective technology offers the possibility of re-launching the renovation sector and foster its application on a broader scale in Europe.

The studied solutions aim at enabling the conditions to create attractive, self-financing schemes to support deep renovation actions; in fact, the GET system represents a possible standardized solution with a highly replicable strategy, especially for the Mediterranean countries of the EU and all the induced seismic areas of the EU. It is the authors' considered opinion that this strategy could more easily convince the users, the urban dwellers, and investors in the energy regeneration and major architectural revamp of the existing buildings. This ambitious idea is based on the willingness of creating completely retrofitted buildings that can be admired and looked-for from other condominiums in the surrounding areas and, from them, to many other buildings in the Mediterranean and EU. Moreover, Pro-GET-onE will put in place the legal/economic and social conditions where energy savings, combined with the increased real estate value, can be mobilized to repay a significant part of the energy investments.

The research project aims at ensuring the proper exploitation of a ground-based knowledge to boost the European strategic aim of mobilizing investment in the energy renovation of the existing building stock.

Acknowledgments: This project has received funding from the European Union's Horizon 2020 Innovation action under grant agreement No. 723747. The authors acknowledge the following master's students in engineering and architecture from the Department of Architecture of the University of Bologna: S. Gammino, E. Ledda, L. Conti, C. De Viti, and G. Turicchia.

Author Contributions: As project coordinator, Annarita Ferrante was responsible for the overall research concept, for the definition of the possible architectural choices and for the structure envelope of the GET system. She particularly dealt with the integration between, structural, energy and user oriented requirements related with the architectural improvement of the building. Giovanni Mochi participated in the development of the structural components of the façade addition and he also dealt with the economic viability of the GET System, providing a comparison of construction unit costs between a typical deep renovation and the new GETs. Giorgia Predari was responsible for defining the characteristics of the structural scheme to be adopted to achieve seismic improvement, as part of the multi-benefit solutions. Furthermore, she contributed to the modeling of the existing structure and the steel exoskeleton, with the interpretation of the results as regards the structural part. Lorenzo Badini dealt with the bibliographic research and the definition of the state of the art; additionally he worked on the structural simulations using the SAP2000 calculation software, applying the proposals of the structural schemes to the case study of the residential building in Peristeri. Anastasia Fotopoulou was responsible for defining the energy and user oriented requirements of the GET structure, and she contributed to the drafting of the abacus for the case study of Peristeri, with the definition of the possible functional variation of the same external volumetric addition made by the alternate combination of sunspace, balcony, or extra-room. Riccardo Gulli made its contribution in defining the applicability of the GET system to the existing building made by reinforced concrete and in its expected future impact. Giovanni Semprini was responsible of the plant integration in the volume of the

GET system and how to satisfy the nearly zero energy demand, with energy simulation implemented through calculation software.

Conflicts of Interest: The authors declare no conflict of interest.

### References

- European Commission. Innovative Financing for Energy Efficiency and Renewable 28/04/2015; Executive Agency for SMEs: Brussels, Belgium, 2015.
- EN 1998: Design of Structures for Earthquake Resistance. Available online: http://eurocodes.jrc.ec.europa. eu/showpage.php?id=138 (accessed on 12 March 2018).
- 3. Buildings Performance Institute Europe (BPIE), Belgium. Available online: www.bpie.euwww.buildingsdata.eu (accessed on 12 March 2018).
- The Coalition for Energy Savings. EU EED (2012/27/EU) Guidebook for Strong Implementation; The Coalition for Energy Savings: Brussels, Belgium, 2013.
- SHARE Project (Seismic Hazard Harmonization in Europe). Available online: http://www.share-eu.org/ (accessed on 12 March 2018).
- Inarcos, I segni della ricostruzione post terremoto maggio 2012: l'adeguamento sismico della palazzina uffici nello stabilimento Magneti Marelli di Crevalcore, Inarcos, Ingegneri Architetti Costruttori. marzo 2014, pp. 53–62. Available online: http://www.teleios-ing.it/UserFiles/File/FRANCESCHINI/ Articolo-Inarcos738\_Ricostruzione%20terremoto%202012\_Franceschini-Semproli-Secci.pdf (accessed on 12 March 2018).
- Material supplied by Teleios S.r.l. (an engineering Italian company engaged in an interdisciplinary process of designing civil works, system and energy), via Salvatore Quasimodo, 44 40013 Castel Maggiore [BO]. Available online: www.teleios-ing.it (accessed on 12 March 2018).
- Avramidis, I.; Mistakidis, E.; Morfidis, K. Seismic Strengthening of the Building of the Rural and Surveying Engineering Department of A.U.TH. Through External Steel Structure. Available online: http://eeme.ntua. gr/proceedings/7th/130.pdf (accessed on 12 March 2018).
- Technical University of Munich, TES EnergyFacade. Available online: http://www.holz.ar.tum.de/ forschung/tesenergyfacade/ (accessed on 12 March 2018).
- 10. MORE-CONNECT (2014-2018). Available online: www.more-connect.eu/ (accessed on 12 March 2018).
- Hilliaho, K.; Mäkitalo, E.; Lahdensivu, J. Energy saving potential of glazed space: Sensitivity analysis. Energy Build. 2015, 99, 87–97. [CrossRef]
- Directive 2014/61/EU of the European Parliament and the Council of 15 May 2014 on Measures to Reduce the Cost of Deploying High-Speed Electronic Communications Networks. Available online: https://eur-lex. europa.eu/legal-content/en/TXT/?uri=celex%3A32014L0061 (accessed on 12 March 2018).
- Fotopoulou, A.; Semprini, G.; Cattani, E.; Schihin, Y.; Weyer, J.; Gulli, R.; Ferrante, A. Deep Renovation in Existing Residential Buildings through Façade Additions. A Case Study in a Typical Residential Building of the 70's, 2008. Available online: https://doi.org/10.1016/j.enbuild.2018.01.056 (accessed on 12 March 2018).
- 14. Ministry of Environment Planning and Public Works, Greece (OASP) Organization for Earthquake Resistant Planning and Protection, Greek code for Seismic Resistant—EAK2000, 2000 and Enhanced in 2003, the English Version Is Printed in the World List 2008. Available online: http://iisee.kenken.go.jp/worldlist/21\_ Greece/21\_Greece\_Overall.pdf (accessed on 12 March 2018).
- 15. Computer & Structures Inc. SAP2000—Integrated Finite Element Analysis and Design of Structures; Computers & Structures Inc.: Berkeley, CA, USA, 2000.
- 16. Abracadabra-project.eu. Available online: http://www.abracadabra-project.eu (accessed on 12 March 2018).
- Frederic Druot Architecture; The "Bois le Pretre" Tower Metamorphosis: Conversion of 100 social dwellings, operation in occupied site and high environmental quality, 5 Bld du Bois le Prêtre, Paris 17ème. Available online: http://www.druot.net/Book-Tour-Bois-le-Prêtre-EN.pdf (accessed on 12 March 2018).
- 18. Frederic Druot Architecture. Available online: http://www.druot.net/ (accessed on 12 March 2018).
- 19. Burkhalter Sumi Architekten. Projects. Available online: http://www.burkhalter-sumi.ch/projects (accessed on 12 March 2018).
- 20. Ferrante, A. *Towards Nearly Zero Energy. Urban Settings in the Mediterranean Climate;* Elsevier: Amsterdam, The Netherlands, 2016, ISBN 978-0-08-100735-8.

- 21. IMPRO-Building. Improvement Potential of Residential Buildings; Institute Prospective Technological Studies: Seville, Spain, 2008.
- 22. Eastman, C.; Teicholz, P.; Sacks, R.; Liston, K. BIM Handbook: A Guide to Building Information Modeling for Owners, Managers, Designers, Engineers and Contractors; John Wiley and Sons: Hoboken, NJ, USA, 2008.
- Young, N.W., Jr.; Jones, S.A.; Bernstein, H.M. BIM: Transforming Design and Construction to Achieve Greater Industry Productivity; McGraw-Hill: New York, NY, USA, 2008.
- Garagnani, S. Digital semantic modelling for the Cultural Heritage: Integrated representation of real architecture. In *Handbook of Res. on Emerging Digital Tools for Architectural Surveying, Modelling, Representation;* IGI Global Publishing: Hershey, PA, USA, 2015; pp. 292–310.
- Bogus, S.M.; Molenaar, K.R.; Diekmann, J.E. Concurrent Engineering Approach to Reducing Design Delivery Time. J. Constr. Eng. Manag. 2015, 131, 11. [CrossRef]
- Kensek, K.; Noble, D. Building Information Modeling: BIM in Current and Future Practice; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2014.
- Fai, S.; Graham, K.; Duckworth, T.; Wood, N.; Attar, R. Building information modeling and heritage documentation. In Proceedings of the 18th International Conference on Virtual Systems and Multimedia, XXIII CIPA Symposium, Milan, Italy, 2–5 September 2012.
- Gu, N.; London, K. Understanding and facilitating BIM adoption in the AEC industry. J. Autom. Constr. 2010, 19, 988–999. [CrossRef]
- Tang, P.; Huber, D.; Akinci, B.; Lipman, R.; Lytle, A. Automatic Reconstruction of As-built Building Information Models from Laser-scanned Point Clouds: A Review of Related Techniques. *Autom. Constr.* 2010, 19, 829–843. [CrossRef]



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





# Article Renovation of a School Building: Energy Retrofit and Seismic Upgrade in a School Building in Motta Di Livenza

# Tiziano Dalla Mora \*, Maria Pinamonti, Lorenzo Teso, Giosuè Boscato, Fabio Peron and Piercarlo Romagnoni

Department of Design and Planning in Complex Environments, University IUAV of Venice, Santa Croce, 191, 30135 Tolentini Venice, Italy; pinamontimaria@gmail.com (M.P.); teso.lorenzo@gmail.com (L.T.); gboscato@iuav.it (G.B.); fperon@iuav.it (F.P.); pierca@iuav.it (P.R.)
\* Correspondence: tdallamora@iuav.it; Tel.: +39-041-257-1485

Received: 15 February 2018; Accepted: 20 March 2018; Published: 26 March 2018



Abstract: The main part of Italian building stock was built before the energy and seismic regulations, so most of buildings need comprehensive refurbishment to achieve the performance required by laws that are in force. This paper presents an experimental study for an energy and structural upgrade methodology, applied to an existing school building in the north-east of Italy. The methodology is based on the International Energy Agency–Energy in Buildings and Communities Programme (IEA-EBC) Annex 56 project guidelines. For the energy retrofit, a set of interventions is defined concerning the building envelope and systems. Among these interventions, the optimal cost is identified: this minimizes the energy demand and the CO<sub>2</sub> emissions, and reduces the financial commitment. The analysis of the seismic retrofit is developed using innovative techniques of intervention and high-performance materials. The proposed interventions are evaluated in terms of efficacy and cost. The results show that it is possible to identify a comprehensive energy retrofit at optimal cost, thanks to high energy saving and subsidies. For the seismic retrofit, the intervention with the higher cost-effectiveness ratio is determined, but the related investment does not have a payback time. The union of the two retrofits permits the combination of benefits and has a payback time for both the interventions. It is possible to state that the cost of a combined intervention is lower than the costs of two different interventions; therefore, when a single retrofit is needed, the possibility of a combined intervention should be evaluated.

Keywords: energy retrofit; seismic analysis; nearly zero-energy buildings (nZEB); Annex 56; cost-effective; optimization

# 1. Introduction

The European Parliament approved the recast of the Energy Performance of Buildings Directive (EPBD recast) in 2010 [1]. As in the previous European Directive 2002/91/EU [2], this concerns the energy efficiency of buildings, but the new directive reinforces the requirements regarding the energy performance of existing and newly constructed buildings and fixes the target of nearly zero-energy buildings (nZEB) for new constructions by 2021. According to the EPBD recast, member states (MS) should consider cost-optimality to establish minimum energy performance requirements in buildings at the lowest cost. In Italy, the national school stock represents a strategic sector to promote interventions for energy retrofits and environmental impact reduction on existent buildings or to convert them into nZEB. In general, non-residential buildings are around 13% of the Italian building stock [3], where around 51,000 buildings are used entirely or partly as schools which, however, have higher energy

consumption: in fact, schools use about 1 million TEPvear (70% for heating and 30% for electricity), representing 2% of the 50 million TEP of total civil use in Italy [4]. So, acting on schools does not have a huge effect from an energy-saving point of view, but from the point of view of training citizens [5]. Most of existing school buildings have inefficient heating systems and old technologies. Space heating is still the main end-use, with 43% of heating needs met using natural gas in 2012. In most school buildings, heating generation mainly uses gas-/oil-fired boilers [6] and radiator heating systems are installed. Moreover, the vast majority of schools are in public ownership and, as consequence, the possibility for comprehensive renovations is limited due to lack of funds for public administration. Howqever, national and European projects represent a way to find funding and incentives for the redevelopment of existing schools with the aim of spreading strategies and best practice among MS, such as the ENTRANZE, RENEW SCHOOL, ZEMeds, School of the Future [7] and VERYSchool [8] projects. Among these, a selection aims to achieve to a kind of retrofitting according to the nZEB target. The ENTRANZE project (2012–2014) supported policy makers by providing the required data, the analysis, and the guidelines to achieve a fast and strong penetration of the nZEB goal within the existing national building stocks by connecting building experts from European research and academia to national decision makers and key stakeholders [9]. The RENEW SCHOOL project (2014–2017) aims at retrofitting a large number of school buildings to highest nZEB standards, by promoting appropriate tools and measures, helping to downsize the energy use significantly as well as create and secure comfortable conditions for pupils and teachers [10]. The ZEMedS project (2013–2016) [11] focuses on the refurbishment of Mediterranean schools to nZEB. The project covers a complete renovation path, tackling strategies for the envelope, the systems and renewable-energy applications as well as the energy management and users' behavior. In this context, the first results are presented with case studies of school buildings that have been analyzed in terms of the energy efficiency and cost optimality so as to define a detailed renovation action plan [12].

Historically and in recent years, Italian territory has been the site of some seismic events that have affected ancient and newly constructed buildings [13,14]. Italian technical standards deal in depth with the procedures for the seismic analysis of buildings [15–21]. Mainly these are structures built in the past which have undergone various adaptation and expansion works, in most cases without considering the situation from a construction point of view, and in combination with techniques and materials that tend to weaken the overall response of the building in case of a seismic event. The study of the seismic vulnerability of existing buildings translates into a determination of the structure's ability to withstand horizontal stresses such as seismic events; in particular, as regards masonry constructions, it is important to study the non-linear behavior of the structure to know the level of safety and structural deficit [22–26].

The main part of the Italian building stock was built before energy and seismic regulations, so the majority of constructions need a comprehensive refurbishment in both these aspects to achieve the EU requirements. This paper proposes a method for defining and comparing different measures for energy retrofitting and seismic improvement, as interventions on the building envelope and the heating system, as long as the decision-making process advances those measures. The overall strategy consists in considering energy and seismic topics at the same time in order to achieve benefits from several points of views, such as economic, time-saving and logistics management during the building-intervention process.

# 2. Materials and Methods

### 2.1. Methodology

The methodology foresees a comparison in terms of costs [27] and energy performance [28] of construction alternatives; the aim is to define the cost-optimal level, i.e., to propose the solution presenting the lowest total costs. The solutions found are shown in a graph presenting a comparison between global costs ( $\varepsilon m^{-2}$ ) and primary energy consumption (kWh m<sup>-2</sup>·y<sup>-1</sup>).

The required benchmarks concern the achievement of nZEB targets and the calculation of incentives. The study to derive cost-effectiveness from a technical and economic perspective is carried out in accordance with the EPBD recast, the Delegated Regulation n. 244/2012 [29] and its Guidelines [30]. The methodology consists of several steps: definition of reference buildings; definition of energy-efficiency measures (measures based on energy from renewable-energy sources (RES) and/or packages and variants of such measures for each reference building); calculation of primary energy demand resulting from the application of the previously selected measures and/or packages of measures; calculation of global costs in terms of net present value for each reference building; sensitivity analysis related to cost data; and identification of cost-optimal levels in each reference building.

Subsequently, seismic retrofit interventions are based on the identification and analysis of the structural vulnerabilities of the building through modal and spectral analysis carried out with commercial finite element analysis (FEA) software. Required improvements are determined, and the best intervention is identified in terms of the cost–benefit ratio, aiming to find the cost-optimal seismic upgrade. The proposed method is divided into the several phases: definition of seismic action, load analysis, linear dynamic analysis, definition of interventions and cost–benefit calculation.

The methodology (Figure 1) described above is applied to a school building, as a case study, to verify its effectiveness: the school building was built in the 1930s and extended twice in the 1960s and 1970s. Thus, to date it presents non-homogeneous masonry and floors that need an energy retrofit and seismic upgrading.



Figure 1. Flow chart showing the adopted methodology for the study.

### 2.1.1. Energy Retrofitting, Optimization and Nearly Zero-Energy Building (nZEB) Targets

A first step for the optimization process is the definition of a reference building and this study used the definition proposed by the IEA Annex 56 Cost-Effective Energy and CO<sub>2</sub> Emissions Optimization in Building Renovation [31]. In the reference case (Figure 2), the renovation consists only of measures carried out for maintaining the building and its functionality. In this kind of renovation, so called «anyway» measures strive for the renewal of building elements or building parts which have arrived at the end of their service life, not deliberately endeavoring to attain higher energy performance [32]. This anyway renovation solution, comprising the so-called anyway renovation measures, identifies a reference situation for determining and assessing the impacts of an energy-related renovation solution on energy use, carbon emissions, materials, costs and possible benefits. The energy-related solution comprises, on the one hand, those retrofit measures of the anyway renovation which are not changed by the energy-related measures. On the other hand, it comprises additionally the energy-related measures by measures which also improve energy performance and do not only restore the original functionality of the particular building element. Building renovation comprising energy-related measures is compared to the anyway reference case to determine the effects of the energy-related measures.



**Figure 2.** "Anyway renovation" vs. "Energy-related renovation" in the case of an anyway necessary building renovation due to functional reasons or building elements at the end of their service life, as defined by the International Energy Agency–Energy in Buildings and Communities Programme (IEA–EBC) Annex 56 Program.

The retrofit solutions are defined through different steps (Table 1). In the first step, the thermal envelope is analyzed: all the elements (external wall, roof, basement, windows) are considered in terms of thermal losses and percentage incidence of surfaces. Then, three groups of energy-efficiency measures (EEMs) are defined according to the standard values of thermal transmittance given by the Conto Termico 2.0 regulation (D.M. 16/02/16) [33,34]: interventions of thermal insulation on each technological element of the envelope; interventions on several technological components in accordance with their percentage incidence of thermal surface, first from higher to lower area and then from lower to higher area. In this way, 10 EEMs on envelope are defined and successively associated with the electrical lighting and wiring system interventions, for a total of 40 combinations: installation of photovoltaic system to cover the 50% of electrical need; and substitution of light bulbs with high-efficiency light-emitting diode (LED) lamps. Finally, all these solutions are combined with the substitution of the existent energy generator with three different boilers: installation of a gas condensing boiler; installation of a biomass boiler; installation of an electrical heat pump.

The EEMs, as they have been defined, number 120 and they will be analyzed in terms of primary energy use and annualized global cost for a lifecycle of 30 years.

After the definition of the reference buildings and the energy-efficiency retrofit measures, primary energy demand is calculated using software for an energy dynamic simulation, such as Energy Plus [35] with Design Builder [36] as a graphic interface. Heating, cooling, ventilation, domestic hot water, lighting and auxiliary demands have been estimated in accordance with the Italian technical specifications UNI 11300 [37], which implement the European standards [38,39]. The characteristics of energy production, distribution, emission and control, as well as the energy carrier, are inserted to derive the final primary energy consumption, according to the conversion factor given by the national normative [40]. The model is calibrated by means of the energy consumption of the last few years.

Measure Code	Envelope Interventions	Measure Code	System Intervention	Measure Code	Thermal Generator Substitution
M 1	First technological component (higher percentage incidence of thermal surface)	Photovoltaic			Installation of
M 2	Second technological component	M 11 to M 20	(PV) system installation	M 1.1 to M 40.1	condensing boiler
M 3	Third technological component	-	mounder		
M 4	Last technological component		Light-emitting diode (LED) M 1.2 +		Installation of biomass boiler
M 5	M1 + M2	M 21 to M 30		M 1.2 to M 40.2	
M 6	M1 + M2 + M3	- 101211010130	installation	WI 1.2 to WI 40.2	
M 7	M1 + M2 + M3 + M4	_			
M 8	M4 + M3				
M 9	M4 + M3 + M2	M 31 to M 40	PV + LED M 1.3 to M 4	M 1.3 to M 40.3	Installation of electrical heat pump
M 10	M4 + M1	-			ciccu icai neat punip

Table 1. List of energy retrofit interventions and their organization on different combined measures.

In this study, the financial level is defined as the global cost: the sum of the initial investment, the sum of the annual costs for each year (energy, maintenance, operation and any additional costs), the replacement of systems and components, the final value, and the costs of disposal, as appropriate. All costs are actualized to the starting year, considering a lifespan of 30 years and the interest rate, through Equation (1):

Annualized global cost = NPV × r × 
$$[1 - (1 + r)]^n$$
 (1)

The equation for the annualized global cost (GC) calculation considers the NPV (net present value)  $(\ell)$ , r as an annual real discount rate (%), and n as the lifetime (year).

The financing framework methodology is based on the net present value (global costs) calculation, carried out according to standard EN 15459:2007 [27], which provides a method for considering the economic aspects of the application of heating systems and other technical systems that affect the energy consumption of the building.

A sensitivity analysis is carried out considering the global cost and the primary energy consumption for each EEM compared to the reference scenario, to find out the optimal cost solution [41].

The main goal of the study is the transformation of a school building into nZEB. As defined by the EPBD recast, a nZEB is a building characterized by a very high energy performance; the nearly zero or very low amount of energy required should be largely covered by renewable energy produced on-site or nearby. A building is considered an nZEB when the following requirements are met: the energy performance (EP) is lower than the cost-optimal level, the differential GC with reference to the building before the refurbishment is negative (nZEB is cost effective) and the national minimum energy targets for nZEBs are achieved. Thus, the nZEBs should have a primary energy consumption lower than the cost-optimal range, and the GC in between the cost-optimal cases and the current reference building (Figure 3).



**Figure 3.** Global cost (GC) curve after renovation with energy-efficiency measures (EEMs) in comparison with reference situation; Identification of nearly zero-energy building (nZEB) solutions (red area) in sensitivity analysis of global cost and primary energy consumption. The graph shows the global cost curve after renovation (yearly costs for energy, operation and maintenance): the curve starts from the reference situation A (anyway renovation). Point O represents the cost-optimal renovation option and point N represents the cost-neutral renovation option with the highest reduction of primary energy.

The Italian law D. Leg. 63/2013 [42] and D. M. 26/06/2015 [28] defined minimum values to achieve nZEB targets for the following performance parameters: the overall heat transfer coefficient (H't), the solar transfer coefficient ( $A_{sol,est}/A_{sup utile}$ ), the energy performance index ( $EP_{H,nd}$ ,  $EP_{C,nd}$ ,  $EP_{gl,tot}$ ), the efficiency for generating thermal and electrical energy ( $\eta_H$ ,  $\eta_W$ ,  $\eta_C$ ), and the integration of renewable energy sources, according to national minimal requirements [43]. With reference to the aforementioned parameters, this study considers the proposed EEMs which followed nZEB targets, identifying the cost-optimal solutions between them. Even if the EPDB Directive recast does not consider the application of financial incentives, this study considers the calculation of the Conto Termico 2.0 program, developed by GSE (Gestore dei Servizi Energetici, Rome, Italy), that regulates the incentive for interventions of small dimensions for increasing energy efficiency and for the production of thermal energy from renewable sources. Both public administrations and private owners are admitted, and the incentive duration varies from 2 to 5 years depending on the type of intervention. The characteristics of the envelope and system for retrofit measures are based on the benchmarks fixed by the Conto Termico 2.0 to evaluate the possible application of incentives to the different kind of interventions.

For solutions according to nZEB parameters, the bonus is equivalent to 65% of total investment cost of the intervention and conversion into nZEB (Table 2), where  $I_{a tot}$  is the annual subsidy in euros,  $E_i$  is the yearly thermal-energy production;  $C_i$  is the thermal-energy valorization coefficient as prescribed by Conto Termico 2.0 in Table 7; while  $I_{a tot}$  is the annual subsidy in euros;  $P_n$  is the system's rated power;  $h_r$  is system's functioning hours as prescribed by Conto Termico 2.0 in Table 10;  $C_i$  is the thermal-energy valorization coefficient as prescribed by Conto Termico 2.0 in Table 9; and  $C_e$  is the rewarding coefficient related to particulate emissions as defined by Conto Termico 2.0 in Tables 11 and 12. After defining the costs for each intervention with the calculation of incentives, the optimal cost is selected among all possible interventions and among those which comply with the nZEB parameters.

Intervention Typology	Subsidies Related to Investment (%)	Maximum Investment (€·m <sup>-2</sup> )	Maximum Subsidy (€)
I-Horizontal opaque structures: roof insulation from outside	55%	200.00	
II-Horizontal opaque structures: flooring insulation from inside	55%	100.00	$(\mathrm{I} + \mathrm{II} + \mathrm{III}) \leq 400,000.00$
III-Vertical opaque structures: wall insulation from outside	55%	100.00	
Windows substitution with installation of thermoregulation systems	55%	450.00	100,000.00
Substitution of existing lamps with LED lamps	40%	35.00	70,000.00
Building modification in nZEB building	65%	500.00	1,750,000.00
Condensing boiler installation	55%	160.00	3000.00
Electrical heat pump installation	To be determined with the formula: $I_{a \ tot} = E_i \ \times \ C_i$		nula:
Biomass boiler installation	$ \begin{array}{l} \mbox{Biomass boiler installation} \\ \mbox{To be determined with the formula:} \\ I_a \mbox{tot} = P_n \times h_r \times C_i \times C_e \end{array} $		nula:

Table 2. Incentives regulation for small-scale energy-retrofit interventions.

## 2.1.2. Definition of Seismic Action and Analysis of Structural Vulnerability

Seismic action is defined by several characteristics of the buildings as: its use, nominal life  $(V_N)$ , reference period  $(V_R)$ , the seismic hazard of the construction site, and other parameters related to the typology of the soil. These values are defined by the Italian Building Code, NTC 2008.

The seismic hazard of the site is defined by referring to the ultimate limit states (ULS) and the serviceability limit states (SLSs), as defined by the NTC 2008. The seismic hazard for ULSs, referring to the case study analyzed in this work, is characterized by the parameters shown in Table 3.

**Table 3.** Parameters used to define the elastic response spectrum of the case study of this work:  $T_R$  represents the period return;  $A_g$  is the maximum horizontal acceleration;  $F_0$  is the horizontal acceleration spectrum amplification coefficient; and  $T_C^*$  represents the period of the stroke at constant speed of the horizontal components.

Limit State	T <sub>R</sub> (Year)	A <sub>g</sub> (g)	F <sub>0</sub>	T <sub>C</sub> * (s)
SLSs	712	0.146	2.531	0.366
ULSs	1462	0.186	2.602	0.377

Based on these characteristics, the design response spectra for ULS, for both horizontal components, are calculated by NTC 2008 (Figure 4). For the determination of the spectra, a damping factor of 5% is assumed.



Figure 4. The design response spectra for ultimate limit states (ULS) along the horizontal and vertical components. The spectrum is defined related to the site and the characteristics of the case study in Motta di Livenza.  $S_d(g)$  is equal to the design seismic acceleration, while T (s) is the vibration period.

The elastic response spectrum allows identification of the elastic displacement response spectrum. Through the elastic displacement response spectrum, it is possible to compare the "displacement capacity" of the structure with the "displacement demand" required by the site, according to the reference state limit. The response spectrum is used to perform the linear dynamic analysis of the building. This analysis allows maximum displacement values to be obtained and the most stressed point of the structure to be located. Starting from these results, the intervention proposals are defined. The aim of the possible actions is to improve the critical aspects by lightening the horizontal elements, improving the resistance to stress of the vertical elements, and preserving the masonry box-like behavior.

Concerning energy issues, this research has achieved four cost-optimal solutions, selecting them among all proposed measures and among those that achieve nZEB targets, considering or not the calculation of incentives. All the outputs are summarized in charts representing the most efficient measure, considering primary energy use and the annualized global cost.

With regard to seismic assessment, intervention measures are simulated and compared with the behavior of a recent seismic event in the building. The value of performance and cost of intervention are collected to verify the effectiveness and validity of the proposal.

## 2.2. Applications—Case Study

The methodology is applied on a case study: the primary school A. Manzoni (Figure 5), located in the Italian municipality of Motta di Livenza (TV). The school was built in 1930 and later expanded in the 1960s and 1970s. The floor area of the building is 415.38 m<sup>2</sup>. The thermal surface is 947.98 m<sup>2</sup> and the heated volume is 2070 m<sup>3</sup>. The geometry is compact and regular, with a compactness surface-area-to-volume ratio (S/V) to 0.46. The building consists of several types of masonry and floor, corresponding to different construction phases; the features of the external envelope are summarized in Table 4 describing the thermal losses behavior and the relative percentage incidence.

**Table 4.** Area, thermal transmittance, thermal dispersion and percentage distribution of the envelope elements % by surface and % by thermal losses.

Element	Area (m <sup>2</sup> )	Thermal Transmittance (W⋅m <sup>-2</sup> ⋅K <sup>-1</sup> )	Thermal Dispersion (W∙K <sup>-1</sup> )	% By Surface	% By Thermal Losses
Brick wall with 2 heads	213.79	1.76	376.27	22%	26%
Double brick wall UNI	143.34	0.90	129.00	15%	9%
Alveolar block wall	107.17	0.90	95.45	12%	6%
Total wall	464.30	-	601.72	49%	40%
Windows	68.30	3.19	217.88	7%	15%
Roof	207.69	1.75	363.46	22%	25%
Basement	207.69	1.93	288.69	22%	20%
Total	947.98	-	1471.75	100%	100%

The mechanical characteristics of the structural elements are derived from in situ test results [44] and are summarized as follows (Table 5):

**Table 5.** Mechanical characteristics of the structural elements: d represents the thickness of the element;  $\gamma$  the specific weight; E the elastic module; and  $G_1$ ,  $G_2$  and  $Q_k$  the permanent and variable loads.

Construction Year	Description	М	echanical Characterist	ics
1930	Brick wall with 2 heads Concrete slab	$d = 0.29 \text{ m} \\ G_1 = 3.00 \text{ kN} \cdot \text{m}^{-2}$	$\begin{split} \gamma &= 18.00 \ k N{\cdot}m^{-3} \\ G_2 &= 1.25 \ k N{\cdot}m^{-2} \end{split}$	$\begin{split} & E = 2227 \; Nm {\cdot}m^{-1} \\ & Q_k = 3.00 \; kN {\cdot}m^{-2} \end{split}$
1960	Double brick wall UNI Concrete and masonry flooring system	$d = 0.29 \text{ m} \\ G_1 = 2.95 \text{ kN} \cdot \text{m}^{-2}$	$\begin{split} \gamma &= 12.00 \ k N {\cdot} m^{-3} \\ G_2 &= 1.50 \ k N {\cdot} m^{-2} \end{split}$	$ \begin{split} & E = 3600 \; Nm \cdot m^{-1} \\ & Q_k = 3.00 \; kN \cdot m^{-2} \end{split} $
1970	Alveolar block wall Concrete and masonry flooring system	$d = 0.29 \text{ m} \\ G_1 = 1.30 \text{ kN} \cdot \text{m}^{-2}$	$\begin{array}{l} \gamma = 15.00 \ k N {\cdot} m^{-3} \\ G_2 = 3.40 \ k N {\cdot} m^{-2} \end{array}$	$ \begin{split} & E = 3500 \; Nm {\cdot}m^{-1} \\ & Q_k = 1.20 \; kN {\cdot}m^{-2} \end{split} $

The structure is analyzed by the linear dynamic analysis carried out through the modal analysis to identify the potential failure mechanisms and the spectral analysis to define the maximum displacement values and to locate the most stressed point caused by the seismic action. The finite element calculation software SAP2000 [45] is used to undertake this structural analysis. A 3D model of the school is built in the software, based on the available data regarding the geometry of the structural components and the characteristics of the materials (Figure 6).

The vertical structural elements are divided into rectangular sub-portions that in the software correspond to two-dimensional shell elements. The shells are connected to each other by nodes in the corners. Three different types of shells are modeled to represents the three masonries of the building, as shown in Figure 7 through the use of different colors. The openings are represented as empty shells. Then, the flooring system is modeled as rigid diaphragms sealed to the masonry with full fixed constraints (black elements in Figure 6). The diaphragms are "thin" elements, without thickness, thus

they are subjected only to bending deformations. At the base of the building, the nodes of the shells are tied with rigid constraints, represented in Figure 6 by the green elements, as foundations of the building. Every foundation element is connected to a node blocking its vertical and horizontal translation as well as its rotation. In this way, the whole base of the building is rigid: only the above-ground part is permitted to deform.



Figure 5. Front and back view of the A. Manzoni school.



Figure 6. Finite element 3D model of the A. Manzoni school in the software SAP 2000; (a) wireframe view; (b) solid view.

The horizontal elements of the model are stressed with the load calculated through the load analysis. The load analysis defines the elementary load conditions of the structure and their combinations, as defined by NTC 2008. These loads are categorized as permanent (G) or variable (Qi) based on the variation of intensity over time.

Following the steps of the linear dynamic analysis, the modal analysis is carried out to define the proper vibration modes of the structure. The results of the modal analysis are shown in Figure 8.

Modes 1 and 2 show a torsional behavior of the structure, with a participant mass to the rotation of respectively 95% and 5%. Although the building is symmetrical, the torsion is caused by the asymmetry of the mechanical characteristic of the elements built in different periods. The torsion activates a considerable percentage of mass, even if the frequencies are low. The higher modes until the 7th do not activate a relevant quantity of mass. The 7th mode shows prevalent displacement in the Y direction. The displacement is larger in the oldest portion of the structure. Here, the walls are thinner and not braced because of the absence of internal partitions. The modes from the 8th to the 12th show displacement in both X and Y directions, with an activated mass of 66–72%.



**Figure 7.** View of the model façades in the software SAP2000: (**a**) south façade; (**b**) west façade; (**c**) east façade; (**d**) north façade. The different colors represent the different kind of masonries: brick wall with 2 heads in pink, double brick wall according to the UNI specifications in yellow, and alveolar block wall in blue.

In succession, the seismic analysis is carried out. In the software, the seismic action is applied to the structure as a dynamic load. Two different load cases are defined for the two X and Y directions. The input data for the seismic analysis are the masses of the structure and their barycenter, the proper vibration modes defined by the modal analysis, and the spectral function defined by the acceleration values of the elastic response spectrum. Following the NTC 2008 indications, the acceleration values of the two X and Y directions are associated to scale factors to define the two different load cases, as shown in Equations (2) and (3):

$$X = F_x + 0.3 F_v$$
 (2)

$$Y = 0.3 F_{\rm X} + F_{\rm y} \tag{3}$$

The  $F_x$  and  $F_y$  are the seismic component in both directions. For the analysis, the structural factor of 2 and the damping factor of 5% are considered.

The seismic combination in ULS is given, as prescribed by NTC 2008, by Equation (4):

$$F_d = E + G_K + \Sigma \psi_{2i} Q_{ik} \tag{4}$$

In the equation, G1 and G2 are the dead loads of the structural and non-structural elements respectively; Q is the accidental load; and E is the seismic action. The recommended value of the  $\psi$  factor is extracted by Table 2.5.1 of NTC 2008.

For the two load combinations, the maximum frame displacements (Figure 9), the maximum floor displacements (Figure 10), the intermediate shifts, and the maximum stress (Figure 11) values are

defined. These values are then checked following the prescription of NTC 2008 to determine the most vulnerable elements of the building.



**Figure 8.** Results obtained by the modal analysis, for the most significant modes. Ux are the masses participating in the *X* direction. Uy are the masses participating in the *Y* direction. Rz are the masses participating in the rotation around the *Z* axis. The color scale shows the zone of greater reaction, and its values are expressed in mm.



**Figure 9.** The maximum frame displacements for the load combination in the *X* direction (**a**) and in the *Y* direction (**b**). The displacement values are defined by referring to the quiet state of the structure.



**Figure 10.** The maximum floor displacements for the load combination in the *X* direction (**a**) and in the *Y* direction (**b**). The maximum interpolation displacements are calculated through the difference between the floor displacements.



**Figure 11.** The maximum stress values for the load combination in the *X* direction (**a**) and in the *Y* direction (**b**).

Analyzing the results, the element of the structure that is more vulnerable is located on the second floor near the four openings of the south façade. This element is characterized by the maximum values of displacement and stress for the load combination in the *Y* direction.

To ensure the study is complete, some characteristics of the structure are identified for which a level of uncertainty was found about their mechanical characteristics. The foundation elements and the connection status of the different masonry present no comprehensive information, so different hypothesis of degradation of the elements are defined:

**Hypothesis 1.** *State of good conditions of the foundation system and a good connection between the different types of masonry;* 

Hypothesis 2. State of degradation of foundations and the consequent yielding support of the building on the ground;

**Hypothesis 3.** *State of degradation of the foundation system as defined in the previous case, assumption of the loss of connection between the different masonry;* 

# Hypothesis 4. Implementation of energy-retrofit intervention.

In relation to these hypothesis, four different 3D models were built in the software. The linear dynamic analysis, as shown before, is carried out for each them. In this case, only the results of the four analyses are presented (Table 6).

The results of the linear dynamic analysis show the main weakness of the building: the high interpolation movement, the frame movement and the instability of one of the walls on the second floor, which is characterized by significant thinness. In Hypothesis 1, the building has a good response to the seismic action not presenting any particular fragility. The weaknesses are larger in Hypothesis 2. Hypothesis 3 presents slightly better behavior than the second hypothesis because of the loss of connection between the three different types of masonries that cause a better response to seismic action. Hypothesis 4 presents reduced fragilities over Hypotheses 2 and 3 because of the augmented mass of the vertical envelope elements given by the application of the insulating material. For all the hypotheses, the stress values are considered insignificant since those are lower than the minimum value admissible for the compressive strength of a brick wall ( $f_{m,min} = 2.4$  MPa) as prescribed by the instructions for the application of NTC 2008; they will not be examined in the following analyses.

Hypothesis	T <sub>1</sub>	Maximum Interpolation Displacement (mm)	Interpolation Displacement Test <sup>1</sup>	Maximum Frame Displacement (mm)	Maximum Stress (MPa)
1	29.59	3.85	Verified	17.39	0.80
2	38.77	16.07	Not Verified	23.69	0.44
3	38.77	15.98	Not Verified	23.69	0.44
4	33.94	19.67	Not Verified	23.01	0.44

Table 6. Results of the linear dynamic analysis of the different cases.

<sup>1</sup> The admissible interpolation displacement is calculated through the formula  $\Delta x_{max} = 0.003$  h f<sub>loor</sub>, as prescribed by NTC 2008. For the case study, the admissible interpolation displacement values for the ground and first floors are 11.88 mm and 12.63 mm.

The definition of energy-retrofitting interventions is structured considering the envelope first. Following the methodology proposed in this paper, the interventions on the envelope are defined in this case according to the amount of thermal surfaces of the envelope (Table 7). For each intervention, the percentage of the envelope affected by the retrofit is shown.

Table 7. Definition of EEMs for energy retrofitting on building envelope.

Intervention Code	Description	Percentage of Envelope Considered for the Intervention
M 1	External wall	49%
M 2	Roof	22%
M 3	Basement	22%
M 4	Windows	7%
M 5	M 1 + M 2	71%
M 6	M 1 + M 2 + M 3	93%
M 7	M 1 + M 2 + M 3 + M 4	100%
M 8	M 4 + M 2	29%
M 9	M 4 + M 2 + M 3	51%
M 10	M 4 + M 1	56%

The adoption of characteristics for the envelope and heating system considers the benchmarks provided by the Conto Termico 2.0, developed by GSE (Rome, Italy). As previously described, EEM interventions have been combined with the replacement of the three generators and the installation of a photovoltaic system and LED lamps (Table 8).

Intervention	Description	Current State Value	After Intervention
External wall	External insulation of the wall of mineral wool 14 cm	$\begin{array}{l} U = 1.76 \; W{\cdot}m^{-2}{\cdot}K^{-1} \\ U = 0.90 \; W{\cdot}m^{-2}{\cdot}K^{-1} \end{array}$	$\begin{array}{l} U = 0.22 \ W {\cdot} m^{-2} {\cdot} K^{-1} \\ U = 0.22 \ W {\cdot} m^{-2} {\cdot} K^{-1} \end{array}$
Roof	External insulation of the second slab of mineral wool 16 cm	$U = 1.75 \ W{\cdot}m^{-2}{\cdot}K^{-1}$	$U=0.21\;W{\cdot}m^{-2}{\cdot}K^{-1}$
Basement	Insertion of insulating layer of mineral wool 14 cm below the floor level	$U = 1.93 \ W{\cdot}m^{-2}{\cdot}K^{-1}$	$U = 0.20 \; W {\cdot} m^{-2} {\cdot} K^{-1}$
Windows	Installation of double glazing with argon cavity and low-emissivity coating, polyvinyl chloride (PVC) frame with thermal break	$U = 3.19 \text{ W} \cdot \text{m}^{-2} \cdot \text{K}^{-1}$	$U = 1.21 \ W {\cdot} m^{-2} {\cdot} K^{-1}$
	Condensing boiler with buffer storage tank		$\eta = 0.98$
Heating system	Biomass boiler	$\eta = 0.62$	η = 0.92
	Electric water-heating pump		COP = 3.9
Lighting system	Lighting substitution with 16 W LED lamps	Use = 7.05 kWh $\cdot$ m <sup>-2</sup>	Use = $3.13 \text{ kWh} \cdot \text{m}^{-2}$
RES production	Installation of photovoltaic panels with 2.4 kW peak power	-	Prod. = 1828 kWh $\cdot$ y <sup>-1</sup>

**Table 8.** Description of proposed EEMs for the case study, with the characteristics for the envelope and heating system after the intervention according to what is prescribed by the Conto Termico 2.0.

The seismic retrofit consists of four possible interventions (Table 9) and the definition is based on the most vulnerable properties of the building: excessive displacement movements, deformations, and the instability of the sloping wall located on the second floor of the original building structure.

Code	Intervention	Description	Characteristic
1	Roof relief	Replacing the existing soleplates with new elements composed of armored plates in autoclaved aerated concrete	Own weight= 1.44 kN/m <sup>2</sup> , Thermal transmittance = 0.67 W m <sup>-2</sup> K <sup>-1</sup>
2	Anchorage with tie rods	Inserting a tie system to prevent the most vulnerable wall collapse mechanisms	Bar type: FeB32K, Tie diameter: 20 mm, Plate type: FeB32K, Plate dimension: $0.5 \times 0.5 \times 0.5$ m, Installation height: 8.5 m
3	Reinforcement with fiber-reinforced polymer (FRP)	Stiffening of the slim wall at the openings with FRP bands with bonded carbon fibers in perfect adherence to the masonry	Tensile modulus= 240–640 MPa, Tensile strength= 4200–4800 MPa
4	Reinforcement with double layer of FRP	Stiffening of slim wall with double layer of FRP bands	Tensile modulus = 240–640 MPa, Tensile strength = 4200–4800 MPa

Table 9. Definition of the proposed seismic retrofit strategies of interventions for the case study.

After the analysis of each energy and seismic retrofitting interventions, the assessment for the Manzoni school presents cost-optimal solutions as follow in the next section.

### 3. Results

With the application of the methodology described in the previous section, three different cost-optimal interventions are presented and it is possible to evaluate the results on energy performance and global costs in comparison to the reference case (Table 10).

Table 10. Energy use and cost for the business-as-usual building.

	Heating Primary Energy Use	Total Primary Energy Use	Investment Cost	Annualized Global Cost
Reference	$50.66 \text{ kWh} \text{ m}^{-2}$	$121.93  \rm kWh  m^{-2}$	200.42 € m <sup>-2</sup>	21.00 € m <sup>-2</sup>
# 3.1. Cost-Optimal Among All the Interventions

The intervention M 21.1, as presented in the description in Table 1, represents the optimal solution among all the interventions (Figure 12). It consists on the realization of the external insulation of mineral wool in the envelope and the installation of a condensing boiler and LED lamps. The outcome results of primary energy use and annualized global cost are, respectively, 57.32 kWh m<sup>-2</sup> and 16.30  $\notin$  m<sup>-2</sup> (Table 11).



**Figure 12.** Cost-optimal solutions concerning primary energy consumption and global cost for the proposed EEMs, among all the interventions.

Table 11. Energy use and	l cost for the identified	cost-optimal measure.
--------------------------	---------------------------	-----------------------

Measure	Heating Primary Energy Use	Total Primary Energy Use	Investment Cost	Annualized Global Cost
M 21.1	$50.66 \text{ kWh} \text{ m}^{-2}$	$57.32 \text{ kWh m}^{-2}$	$200.42 \notin \mathrm{m}^{-2}$	16.30 € m <sup>-2</sup>

#### 3.2. Cost-Optimal Among All the Interventions that Achieve the nZEB Targets

According to the nZEB benchmarks, the cost-optimal solution is M 31.2 (Figure 13), including the installation of a biomass boiler and LED lamps, the realization of external insulation, and the installation of a photovoltaic system. The results for M 31.2 show a primary energy use of 10.38 kWh m<sup>-2</sup> for years and an annualized global cost of 17.23  $\in$  m<sup>-2</sup> during the lifecycle (Table 12).

Table 12. Energy use and cost for the identified cost-optimal measure.

Measure	Heating Primary Energy Use	Total Primary Energy Use	Investment Cost	Annualized Global Cost
M 31.2	$10.27 \text{ kWh} \text{ m}^{-2}$	$10.38  \rm kWh  m^{-2}$	$247.57 € m^{-2}$	17.23 € m <sup>-2</sup>



**Figure 13.** Cost-optimal solutions concerning primary energy consumption and global cost for the proposed EEMs, among the interventions that achieve the nZEB targets.

# 3.3. Cost-Optimal Among All the Interventions with Incentives

The incentive calculation (Table 13) gives evidence of the M 37.2 (Figure 14) as the cost-optimal solution among all the interventions: each envelope element is retrofitted (insulation on external walls, basement, roof, and replacement of windows), a biomass boiler and photovoltaic system are installed and a light system with LED lamps is considered, saving overall energy use and global cost during the lifespan. The measures allow a primary energy use of 4.85 kWh m<sup>-2</sup> for a year and an annualized global cost of 9.10  $\notin$  m<sup>-2</sup> during the lifecycle (Table 14).

Intervention Typology	Intervention Price (€)	Subsidies Related to Investment (%)	Subsidies (€)	Maximum Subsidy (€)
I-Horizontal opaque structures: roof insulation from outside	41,146.06	55%	22,630.33	
II-Horizontal opaque structures: flooring insulation from inside	24,669.25	55%	13,568.09	$(\mathrm{I} + \mathrm{II} + \mathrm{III}) \leq 400,000.00$
III-Vertical opaque structures: wall insulation from outside	24,892.26	55%	13,690.74	
Window substitution with installation of thermoregulation systems	44,754.00	55%	24,614.70	100,000.00
Substitution of existing lamps with LED lamps	919.25	40%	367.70	70,000.00
Building modification in nZEB building		65%		1,750,000.00
Condensing boiler installation	9870.56	55%	5428.81	3000.00
Electrical heat pump installation	39,274.63	-	25,528.51	-
Biomass boiler installation	23,001.58	-	14,951.03	-

Table 13. Intervention price and incentive amount for every foreseen renovation action.

Table 14. Energy use and cost for the identified cost-optimal measure.

Measure	Heating Primary Energy Use	Total Primary Energy Use	Investment Cost	Annualized Global Cost
M 37.2	$4.74 \; \rm kWh \; m^{-2}$	$4.85 \ \rm kWh \ m^{-2}$	$139.74 \in m^{-2}$	9.10 € m <sup>-2</sup>



**Figure 14.** Cost-optimal solutions concerning primary energy consumption and global cost for the proposed EEMs applying incentives calculation.

## 3.4. Cost-Optimal Among All the Interventions that Achieve the nZEB Targets with Incentives

The same measure of the previous section, M 37.2, complies with the nZEB parameters, resulting as the best proposed solution that allows a primary energy use of 4.85 kWh m<sup>-2</sup> per year and an annualized global cost of  $9.10 \text{ fm}^{-2}$  during the lifecycle (Figure 15).

In the case study, a single intervention of thermal insulation on the external walls is proposed as the cost-optimal solution, corresponding to the best compromise of energy saving at lower global cost. The conversion to nZEB allows higher energy saving in comparison to the cost-optimal solution (from 53% to 91%), showing a similar annualized global cost; the application of the current national subsidy program allows the global cost to be halved during the 30-year life span (-56%), even if the renovation measures regard all the building elements of the envelope, corresponding to the highest investment cost due to the number of technical elements and intervention areas.

## 3.5. Seismic Retrofitting Interventions

For the purpose of this study, the evaluation of seismic interventions considers only hypothesis number 4. This hypothesis permits examination of the effects of the combinations of the energy and seismic retrofit. The linear dynamic analysis is carried out for all the interventions described in the previous chapter. The results (Table 15) are expressed as a variation of the values compared to the current situation (Table 6), to show the structural benefits.

All the interventions allow improvement of the structural behavior of the building. The only exception refers to the acceptance criteria of the displacement, related to the load combination in the Y direction; in this case, the interpolation displacement values are referred to the nodes of the north façade, which is not affected by the interventions because it is in a good condition, and the values are the same as in the actual situation. Analyzing the results through a cost–benefit evaluation, the best solutions consider as interventions anchorage with tie rods (n. 2) and reinforcement with fiber-reinforced polymer (FRP) (n. 3). In fact, solution n.1 results in being very expensive and ineffective,

**Table 15.** Results of the seismic analyses of the proposed interventions for hypothesis no. 4. The results are expressed in terms of variation from the values of the current situation. 1-2: values referred to the load combinations in the *X* and *Y* directions.

Intervention	Interpolation Displacement Value Variation	Interpolation Displacement Test (Dir. $X^{1}$ )	Interpolation Displacement Test (Dir. $\gamma^2$ )	Maximum Frame Displacement Value Variation	Investment Cost
1. Roof relief	-0.08 mm	Verified	Not verified	-0.15 mm	70,464.80 €
<ol><li>Anchorage with tie rods</li></ol>	-0.00 mm	Verified	Not verified	-33.62 mm	2205.58 €
3. Reinforcement with FRP	-0.43 mm	Verified	Not verified	-2.42 mm	9741.54 €
4. Reinforcement with double layer of FRP	-0.37 mm	Verified	Not verified	-2.44 mm	14,547.66 €



**Figure 15.** Cost-optimal solutions concerning primary energy consumption and global cost for the proposed EEMs applying the incentives calculation, among all the interventions that achieve the nZEB targets with incentives.

# 4. Discussion

This paper identifies the interventions to be adopted for the optimal energy and seismic retrofit of the A. Manzoni school in Motta di Livenza. From the energetic point of view, it is convenient to intervene in the envelope by creating an exterior insulation and finishing system, isolating the upper floor and the floor on the ground and replacing the window frames, and on the systems by means of the installation of LED lamps, a photovoltaic system, and the replacement of the existing generator with a biomass boiler.

Regarding the seismic upgrading intervention, it is assumed that the installation of two tie rods to the intrados of the last floor, and the reinforcement of the most vulnerable wall by installing FRP on the outer side of the masonry, will be undertaken.

Finally, the cost of intervention was calculated in the case of a joint energetic and seismic retrofit renovation, and the expected cost is 193.31  $\text{C·m}^{-2}$ . The expected cost is slightly lower than that of separate interventions, thanks to the possibility of reducing site labor and construction costs.



**Figure 16.** Cost of interventions graph. The blue bar represents the cost for the energy retrofit M 37.2; the orange bar is the cost for the seismic retrofit (tie rod); the green bar represents the cost for the seismic retrofit (one FRP layer); the grey bar is the cost for three separated interventions (M 37.2, tie rod, FRP layer); and the yellow bar represents the cost for three joint interventions (M 37.2 + tie rod + FRP layer).

The annualized global costs for the energy retrofit interventions, considering a lifecycle of 30 years, and the intervention costs for the seismic renovation have been calculated. After that, the cost of intervention was assessed in the event that the two redevelopment interventions were carried out jointly.

For the energy retrofit intervention, in the 30-year life span considered, the economic gain is 56%, and the energy saving 96%, compared to maintaining the current situation. For the seismic retrofit intervention, the annualized global costs, always referred to over 30 years, increase by 7% and the primary energy consumption remains unchanged.

Combining the two interventions, we obtain a reduction in the total annualized cost of 3%, thus passing from 56% to 53% of economic gain (the annualized global cost increases from  $9.10 \text{ } \text{ } \text{e} \text{ } \text{m}^{-2}$  to  $9.86 \text{ } \text{ } \text{e} \text{ } \text{m}^{-2}$ ) with 96% of primary energy saved.

Eventually, the same considerations are made for the other two cost-optimal cases previously examined: with and without transforming the school into a nZEB building, both without considering any kind of subsidies.

It can be noted that even in those cases where the use of incentives is not foreseen, it is possible to aim at an economic saving in a 30-year life span even combining the energy and seismic retrofit action. It can be assumed that the realization of an energy refurbishment can also provide the possibility to adapt the building from the seismic point of view, aiming at acceptable periods of return for the investment. Moreover, if a building needs a seismic retrofit it is possible to carry out an energy requalification that makes the investment economically viable, thanks to the reduction of energy consumption and operating costs.



Figure 17. Cost-optimal and nZEB solutions compared with the reference case.

The data collected define the possibility of combining energy and seismic retrofit interventions in an effective and convenient way for investors, also referring to other buildings than the case study.

Considering the Italian building stock, it can be assumed that most of the artefacts still have high energy consumption or are not in compliance with the new seismic standards defined in recent years. It is, therefore, considered reasonable, both in one case or another, to evaluate the hypothesis of a joint intervention that can bring the buildings back to a suitable condition, aiming, in a brief time, for a total economic return on the investment.

In future renovation works, the designer of the building refurbishment should consider joining the retrofit works at the beginning of the project. In particular, the best choice should be retrofitting a single envelope element, from both seismic and energetic aspects, thus reducing the cost of interventions, materials and building-site impacts.

The seismic retrofit intervention that does not generate an economic saving and, therefore, a return on investment can be more economically sustainable if combined with an energy requalification.

## 5. Conclusions

The paper presents optimal measures of intervention in the cases of both energy and seismic retrofit for a public school building. The study is developed according to the application of the last European Union norm that concerns the achievement of nZEB targets for cost-effective energy-reduction measures, considering also seismic procedures.

The energy cost-optimal measure with subsidies allow an energy saving of 56% and an energy-use reduction of 96% in comparison to the reference situation during the 30-year life span. Analyzing the seismic intervention during the same period, the annualized global cost is calculated with an increase of 7% in comparison to the reference case while the energy consumption is unchanged. With the combination of energy and seismic retrofit, the cost saving is 53% and energy consumption is reduced by about 96%.

The methodology described in this paper points to several results. First of all, a cost-optimal measure for the energy retrofit of the case study could be found in different ways in relation to the conversion to a nZEB scenario and the application of subsidies. Considering these cases, the cost-optimal solution consists of a deeper upgrade with higher energy savings. The methodology could be applied to different building typologies in order to evaluate the different kinds of intervention in relation to the financial and economic availability and feasibility.

The cost-optimal measure for the seismic retrofit is based on the evaluation of several interventions on the building envelope; the paper compared structural benefits and costs for each intervention in order to define the best measure for a global seismic upgrade.

At last the two cost-optimal measures (energy and seismic) were combined. In this case, the results show that the economic saving in 30 years is far lower in respect to an intervention that considers only an energy retrofit. In fact, generally seismic retrofit measures do not create a money saving and a favorable payback time. In this case, the energy saving permits the seismic intervention to be paid back in 12 years (Figure 18).



**Figure 18.** Return time graph for the investment cost of the energy and seismic retrofit. As prescribed in the Conto Termico 2.0, the subsidies are calculated to be provided during the first 6 years. Red line represents the payback flow M 37.2 investment cost in respect to the blue line of the Reference case.

The global cost of a combined intervention is lower than the costs of two different interventions. A single intervention reduces the period of the works and the consequent troubles for the school activities. When a single retrofit is needed, the possibility of a combined intervention should be analyzed.

Acknowledgments: The authors would like to thank the Municipal Administration of Motta di Livenza (TV), for the availability of documents about the building.

Author Contributions: All authors have seen and approved the manuscript and have contributed significantly for the paper in all phase and section.

Conflicts of Interest: The authors declare no conflict of interest.

#### Abbreviations

EEMs	Energy-efficiency measures
EPBD	Energy performance of building directive
EU	European Union
FEA	Finite element analysis
FRP	Fiber-reinforced polymer

GC	Global cost
GSE	Gestore Dei Servizi Energetici
IEA-EBC	International Energy Agency–Energy In Buildings and Communities Program
MS	Member state
NTC	Norme Tecniche per le Costruzioni
nZEB	Nearly zero-energy building
RES	Renewable energy source
SLS	Serviceability limit states
TEP	Tonne of oil equivalent
TV	Treviso province
ULS	Ultimate limit states
V <sub>N</sub>	Nominal life
VR	Reference period

# References

- European Union. Directive 2010/31/EU of the European Parliament and of the Council on the Energy Performance of Buildings (EPBD Recast). 19 May 2010. Available online: https://eur-lex.europa.eu/ legalcontent/EN/TXT/?uri=celex%3A32010L0031 (accessed on 10 October 2017).
- European Union. Directive 2002/91/EC of the European Parliament and of the Council on the Energy Performance of Buildings (EPDB). 16 December 2002. Available online: https://eur-lex.europa.eu/ legalcontent/EN/TXT-/?uri=celex%3A32002L0091 (accessed on 10 October 2017).
- Eurostat. Final Energy Consumption by Sector. 2014. Available online: http://ec.europa.eu/eurostat/web/ products-datasets/-/tsdpc320 (accessed on 10 October 2017).
- JRC Science and Policy Report, Energy Renovation: The Trump Card for the New Start for Europe. 2015. Available online: http://publications.jrc.ec.europa.eu/repository/bitstream/JRC92284/eur26888\_ buildingreport\_online\_2015-03-25.pdf (accessed on 10 October 2017).
- 5. CRESME. Guida per Il Contenimento del Consumo Energetico Nelle Scuole; ENEA-Fire: Rome, Italy, 2011.
- De Santoli, L.; Fraticelli, F.; Fornari, F.; Calice, C. Energy performance assessment and retrofit strategies in public school buildings in Rome. *Energy Build*. 2014, *68*, 196–202. [CrossRef]
- 7. School of the Future. Available online: http://www.school-of-the-future.eu/ (accessed on 10 October 2017).
- 8. VERYSchool. Available online: http://www.veryschool.eu/ (accessed on 10 October 2017).
- 9. ENTRANZE Project. Available online: http://www.entranze.eu/ (accessed on 10 October 2017).
- 10. Renew School. Available online: https://www.renew-school.eu/ (accessed on 10 October 2017).
- 11. ZEMeds. Available online: http://www.zemeds.eu/ (accessed on 10 October 2017).
- 12. Gaitani, N.; Cases, L.; Mastrapostoli, E.; Eliopoulou, E. Paving the way to nearly zero energy schools in Mediterranean region-ZEMedS project. *Energy Procedia* **2015**, *78*, 3348–3353. [CrossRef]
- D'Agostino, N.; Mantenuto, S.; D'Anastasio, E.; Avallone, A.; Barchi, M.; Collettini, C.; Radicioni, F.; Stoppini, A.; Fastellini, G. Contemporary crustal extension in the Umbria–Marche Apennines from regional CGPS networks and comparison between geodetic and seismic deformation. *Tectonophysics* 2009, 476, 3–12. [CrossRef]
- Michele, M.; Di Stefano, R.; Chiaraluce, L.; Cattaneo, M.; De Gori, P.; Monachesi, G.; Latorre, D.; Marzorati, S.; Valoroso, L.; Ladina, C.; et al. The Amatrice 2016 seismic sequence: A preliminary look the mainshock and aftershocks distribution. *Ann. Geophys.* 2016, 59. [CrossRef]
- Ministry of Infrastructures and Transports. Ministerial Decree "Norme Tecniche per le Costruzioni", 14 January 2008. Gazzetta Ufficiale n. 29, 4 February 2008. Available online: http://www.gazzettaufficiale.it/eli/ id/2017/04/4/17G00055/sg (accessed on 10 October 2017).
- Ministry of Infrastructures and Transports. Ministerial Circular n. 617 "Istruzioni per l'Applicazione delle Nuove Norme Tecniche per le Costruzioni", 2 February 2008. *Gazzetta Ufficiale n. 47*, 26 February 2008. Available online: http://cslp.mit.gov.it/-index.php?option=com\_content&task=view&id=79&Itemid=1 (accessed on 10 October 2017).

- Cecchi, R.; Calvi, M. Guidelines for Evaluation and Mitigation of Seismic Risk to Cultural Heritage with Reference to Technical Standard for Construction; Gangemi, G., Ed.; Ministry for Cultural Heritage and Activities, Department of Civil Protection Agency: Rome, Italy, 2010. (In Italian)
- Priestley, M.J.N.; Seible, F. Design of seismic retrofit measures for concrete and masonry structures. Constr. Build. Mater. 1995, 9, 365–377. [CrossRef]
- Shrive, N.G. The use of fibre reinforced polymers to improve seismic resistance of masonry. Constr. Build. Mater. 2006, 20, 269–277. [CrossRef]
- Lourenço, P.; Poças Martins, J.P. Strengthening of the Architectural Heritage with Composite Materials. In Proceedings of the 1st International Conference on Composites in Construction—CCC2001, Porto, Portugal, 10–12 October 2001.
- 21. Burdette, J.Q.; Hamid, A.A.; Larralde, J. Development of plastic-reinforced wall ties. *Masonry Soc. J.* **1996**, 4, 37–44.
- Sayed-Ahmed, E.Y.; Lissel, S.L.; Tadros, G.N.G. ShriveCarbon fibre reinforced polymers (CFRP) post-tensioned masonry diaphragm walls: Prestressing, behaviour and design recommendations. *Can. J. Civ. Eng.* 1999, 26, 324–344. [CrossRef]
- Nanni, J.G. Tumialan, Fibre-reinforced composites for the strengthening of masonry structures. *Struct. Eng. Int. IABSE* 2003, 13, 271–278. [CrossRef]
- 24. Chopra, A.K. Dynamics of Structures, 3rd ed.; Pearson Prentice Hall: Upper Saddle River, NJ, USA, 2007.
- Fajfar, P. Approximate Methods for Nonlinear Seismic Analysis of Building Structures. In Proceedings of the 2rd International Conference on Seismology and Earthquake Engineering, Lisboa, Portugal, 21–25 June 1999; Seco e Pinto, P., Ed.; IISEE: Ibaraki, Japan, 1999; pp. 649–658.
- Tso, W.K.; Moghadam, A.S. Pushover procedure for seismic analysis of buildings. *Prog. Struct. Eng. Mater.* 1998, 1, 337–344. [CrossRef]
- European Committee for Standardization, CEN. EN 15459, Energy Performance of Buildings. Economic Evaluation Procedure for Energy Systems in Buildings; European Committee for Standardization: Brussels, Belgium, 2007.
- 28. Ente Italiano di Normazione, UNI. UNI/TS 11300-3, Energy Performance of Buildings; Ente Italiano di Normazione: Milano, Italy, 2010.
- European Parliament. Commission Delegated Regulation (EU) No 244/2012 of 16/01/2012 Supplementing Directive 2010/31/EU of the European/Parliament and of the Council on the Energy Performance of Buildings by Establishing a Comparative Methodology Framework for Calculating Cost-Optimal Levels of Minimum Energy Performance Requirements for Buildings and Building Elements. *Off. J. Eur. Union* **2012**, 55. Available online: http://eur-lex.europa.eu/legal-content/EN/ALL/?uri=CELEX:32012R0244 (accessed on 10 October 2017).
- 30. European Parliament. Guidelines Accompanying Commission Delegated Regulation (EU) No 244/2012 of 16 January 2012 supplementing Directive 2010/31/EU of the European Parliament and of the Council on the energy performance of buildings by establishing a comparative methodology framework for calculating cost-optimal levels of minimum energy performance requirements for buildings and building elements. *Off. J. Eur. Union* 2012, *55.* Available online: https://eur-lex.europa.eu/legal-content/EN/ALL/?uri= CELEX%3A52012XC0419%2802%29 (accessed on 10 October 2017).
- Ott, W.; Bolliger, R.; Ritter, V.; Citherlet, S.; Favre, D.; Perriset, B.; Almeida, M.; Ferreira, M. Methodology for Cost-Effective Energy and Carbon Emissions Optimization in Building Renovation (Annex 56)—Methodology and Assessment of Renovation Measures by Parametric Calculations; EBCP—IEA, University of Minho: Guimarães, Portugal, 2014.
- 32. EBC Energy in Buildings and Communities Programme, International Energy Agency. Annex 56. Cost Effective Energy and Carbon Emissions Optimization in Building Renovation. Available online: http://www.iea-annex56.org/ (accessed on 10 October 2017).
- Ministry of Economic Development. Decree "Aggiornamento Conto Termico", 16 February 2016. Gazzetta Ufficiale n. 51, 2 March 2016. Available online: http://www.gse.it/it/Conto%20Termico/Conto% 20Termico%202.0/-Pagine/default.aspx (accessed on 10 October 2017).

- 34. Ministry of Economic Development. Annex I-II "Conto Termico 2,0", 16 February 2016. Gazzetta Ufficiale n. 51, 2 March 2016. Available online: https://www.gse.it/documenti\_site/documenti%20gse/servizi% 20per%20te/conto%20termico/normativa/allegato%20decreto%20interministeriale%2016%20febbraio% 202016.pdf (accessed on 10 October 2017).
- U.S. Department of Energy, DOE. *EnergyPlus Documentation—Getting Started*; U.S. Department of Energy: Washington, DC, USA, 2017. Available online: https://energyplus.net/ (accessed on 10 October 2017).
- Design Builder Software Ltd. DesignBuilder SBEM v5.0.3 Manual (SBEM v5.3a); Design Builder Software Ltd: Gloucester, UK, 2016; Available online: https://www.designbuilder.co.uk/download/documents (accessed on 10 October 2017).
- 37. Ente Italiano di Normazione, UNI. UNI/TS 11300-1, Energy Performance of Buildings; Ente Italiano di Normazione: Milano, Italy, 2014.
- European Committee for Standardization, CEN. EN 15316-4-1, Heating Systems in Buildings—Method for Calculation of System Energy Requirements and System Efficiencies; European Committee for Standardization: Brussels, Belgium, 2008.
- European Committee for Standardization, CEN. EN 15243, Ventilation for Buildings—Calculation of Room Temperatures and of Load and Energy for Buildings with Room Conditioning Systems; European Committee for Standardization: Brussels, Belgium, 2007.
- Ministry of Economic Development. Decree "Applicazione Delle Metodologie di Calcolo delle Prestazioni Energetiche e Definizione delle Prescrizioni e dei Requisiti Minimi Degli Edifici", 26 June 2015. *Gazzetta Ufficiale n. 162*, 15 July 2015. Available online: http://www.gazzettaufficiale.it/eli/id/2013/ 08/03/13A06688/sg (accessed on 10 October 2017).
- 41. Buildings Performance Institute Europe, BPIE. Cost Optimality—Discussing Methodology and Challenges within the Recast Energy Performance of Buildings Directive; BPIE: Brussels, Belgium, 2010.
- President of Italian Republic. Decree n. 63/13 "Disposizioni Urgenti per il Recepimento della Direttiva 2010/31/UE", 4 June 2013. Gazzetta Ufficiale n. 130, 5 June 2013. Available online: http://www. gazzettaufficiale.it/eli/id/2013/08/03/13A06688/sg (accessed on 10 October 2017).
- 43. President of Italian Republic. Decree n. 28/11 "Attuazione della Direttiva 2009/28/CE sulla Promozione dell'Uso dell'Energia da Fonti Rinnovabili, Recante Modifica e Successiva Abrogazione delle Direttive 2001/77/CE e 2003/30/CE", 3 March 2011. Gazzetta Ufficiale n. 71, 29 March 2011. 2011. Available online: http://www.gazzettaufficiale.it-/eli/id/2011/03/28/011G0067/sg (accessed on 10 October 2017).
- Bisetto, M. (Motta di Livenza, Treviso, Italy). Verifica Statica dei Solai, Comprensiva di Indagine Diagnostica delle Strutture Orizzontali della Scuola Primaria A. Manzoni di S. Giovanni. Personal comunications, 2016.
- Computer and Structures Inc., CSI. Analysis Reference Manual SAP2000. Structural Software for Analysis and Design; Computer and Structures Inc.: Berkeley, USA, 2016; Available online: http://docs.csiamerica.com/ manuals/etabs/Analysis%20Reference.pdf (accessed on 10 October 2017).



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





# Article Seismic Retrofit Measures for Masonry Walls of Historical Buildings, from an Energy Saving Perspective

# Mariangela De Vita\*, Antonio Mannella, Antonio Sabino and Alessio Marchetti

Construction Technologies Institute CNR, Via G. Carducci 32 C, 67100 L'Aquila, Italy; antonio.mannella@itc.cnr.it (A.M.); antonio.sabino@itc.cnr.it (A.S.); alessio.marchetti@itc.cnr.it (A.M.) \* Correspondence: ing.mariangeladevita@gmail.com; Tel.: +39-329-1750844

Received: 25 January 2018; Accepted: 23 March 2018; Published: 27 March 2018



Abstract: The planning of energy saving and structural retrofit interventions on masonry buildings are usually two separate projects; combined interventions are rare. Solutions tackling both aspects can reduce total refurbishment costs and improve global building performance. However, heavy interventions on the envelope have to be carefully designed in order to improve both seismic and energy performance whilst mitigating environmental impact. National energy saving regulations are generally less strict for historical buildings, making this category of buildings very interesting not only because of their widespread diffusion across Italy but also because of the possibility of planning interventions that will allow significant improvements by combining building efficiency and safety. This research looks at these aspects and proposes new methods for refurbishing masonry buildings combining seismic improvements and energy saving interventions. Among those mostly commonly applied on masonry buildings in Italy, and described in this paper, are various combined retrofit interventions, and we report the results of these interventions on thermal transmittance reduction and global costs.

**Keywords:** historical building; seismic and energy retrofit; combined interventions; U-value; historical masonry

# 1. Introduction

This article analyses the results of a wide-ranging study investigating the effects on energy performance of masonry buildings with historical/heritage value that have undergone typical seismic improvement interventions. The most frequently used techniques have been examined by analyzing, for each of them, the resulting effect on energy performances of the envelope in order to evaluate the variation of thermal properties of the materials employed for structures and finishes.

A significant part of Europe's building heritage is made up of old buildings [1–4] constructed using poor quality insulation materials. Therefore, recent regulations, such as the EU Directive 2010/31 [5] on buildings' energy efficiency, aim to increase energy efficiency standards, by considering both the single components and the entire building. In Italy, the energy improvement intervention regulated by the Legislative Decree [6] takes into account the European Directive 2002/91/CE [7]; actual regulations have been updated several times to comply with new European directives. The current energy saving standards have been recently integrated with a set of NZEB objective norms; last updates of 90/2013 law have been published in the Official Journal of the Italian Republic, 15 July 2015, n.39 [8]. However, the application of the aforementioned legislation is subordinate to Italy's own cultural conservation principles. Rules concerning the Italian energy performance certificate (APE), to check thermal installations, are still valid, whilst any other intervention must be evaluated by the local office

of Ministry of Cultural Heritage and Activities and Tourism (MiBACT). The latter office evaluates, case by case, the adequacy of interventions. Despite their best intentions [5,8], current regulations have led to an increase in the gap in required performance between new buildings and historical buildings. Furthermore, a substantial absence of general rules about better compatibility among available energy retrofit technologies and other eligible interventions can lead to differences in treatment of similar cases in several contexts [9].

In recent years, a number of investigations have looked at energy saving and structural aspects. Different approaches have been adopted, ranging from holistic ones [10], dynamic simulations of entire buildings [11], to specific solutions such as the adoption of a thermal, vegetal based, insulating plaster [12].

An interesting multidisciplinary approach was investigated by Ascione et al., Mannella et al. and De Berardinis et al. [13–15]: they proposed a replicable methodology for improving the performance of historic buildings based on preliminary historical analysis, structural diagnosis and in-situ investigations. This multidisciplinary approach to building structural and energy diagnoses was applied to a case study obtaining a model to predict the structural safety of the building and its energy consumption. Tiberi and Carbonara [16] explored aspects relating to retrofitting interventions for energy saving and the financial costs, introducing a case study and finding four solutions for the envelope; Calvi et al. [17] presented an integrated approach to seismic resilience and energy efficiency assessments. Marques et al. and Calvi [18,19] performed an in-depth cost–benefit analysis of the strengthening solutions, comparing the economic benefit gained by reducing the seismic damage against the intervention cost. More recently, smart and innovative integrated systems have also been designed in order to achieve important energy and environmental benefits regarding historical buildings [20,21].

The state of the art approach leans towards combined interventions and multidisciplinary approaches in the refurbishment process; however, it is important to understand the interactions going on. This study highlights how the execution of some of the most adopted typologies of structural improvement intervention on the outer walls typically cause an increase of the envelope thermal transmittance and, consequently, a general worsening of the energy performance of the building. However, an appropriate selection of materials and the techniques employed, with a very small increase in working times and costs, allows a rebalancing and even, in some cases, a notable reduction in thermal conductivity whilst respecting the original historical values of the buildings. The analyses carried out permit us to evaluate the impact of current regulation on energy efficiency in buildings that constitute Italy's cultural heritage [22].

## 2. Materials and Methods

The impact on buildings' energy efficiency produced by structural improvement interventions is analyzed below focusing on conductivity and thermal transmittance of the envelope. This work analyzes some of the most common structural improvement interventions on various kinds of masonry and typically occurring in the historical buildings of central Italy (a territory characterized by important seismic events and by historical buildings, as shown in Figure 1).

The investigated interventions are often used for the seismic retrofit of buildings damaged by earthquakes [23]. In general, restoration projects that include seismic improvements use several different types of intervention. Figure 2 illustrates the distribution of the principal interventions carried out on masonry buildings located outside the historical center of L'Aquila after the earthquake (Mw = 6.3) that occurred in the Abruzzi region on the 6 April 2009 [24,25]. Figure 2a shows the distribution of the most diffuse interventions and Figure 2b shows the distribution of secondary interventions: main intervention, for each building, refers to more widespread and costly work typology; secondary interventions, on the other hand, are complementary to the main ones.



**Figure 1.** Historical buildings heavily damaged by the earthquake that occurred on 30 October 2016 in Central Italy: (a) Exterior view of the residential building; (b) Fortified residence.



**Figure 2.** Type and frequency of techniques for (**a**) main and (**b**) secondary interventions to historical buildings after the Abruzzo earthquake of 6 April 2009 in Abruzzo. Red: strengthening of masonry walls (i.e., reinforced plastering mortar, injections, diàtonoi); yellow: single, individual interventions; green: application of tie-rods in connections and walls; blue: interventions related to the foundations.

# 2.1. Methodology

The methodology for this research involves seven steps and terminates with an optimization of the interventions carried out. The methodology of the study is the following:

- Step 1: execution of preliminary investigations on the morphology of the masonry and evaluation of the thermal physical state of the envelope;
- Step 2: application of the most common interventions for the structural improvement of the investigated masonries;

- Step 3: application of the most common energy efficiency interventions, compatible with the structural interventions chosen in the previous step;
- Step 4: determination of the new thermal-physical characteristics of the masonries using thermal equivalent conductivity (λeq);
- Step 5: comparison between the thermal transmittance (U-value) of the masonries after seismic retrofitting intervention and after energy retrofitting interventions combined with seismic retrofitting interventions;
- Step 6: economic analyses of each of the solutions analyzed, to show the advantages and disadvantages of using combined solutions;
- Step 7: evaluation of the design process to optimize the balance between cost and work, to achieve structural improvement as well as adequate energy saving.

# 2.2. Masonry Types and Thermal Properties

This research relates to the variations of the thermal properties of each material making up the walls after seismic retrofitting interventions: the analyses carried out aim to establish the new values of the U-value of the masonry.

In the case of heterogeneous materials, like masonry, it is not formally correct to use thermal conductivity because the flux is not solely due to conduction but also to convection and radiation in the internal cavities of the wall. For this reason, the study was conducted by varying the thickness of each layer of masonry analyzed (masonry cut stones with possible rubble in-fill). Table 1 shows the values of  $\lambda$ eq for different value of thicknesses.

Table 1. Main properties of investigated masonries. The thickness values refer only to the thickness of the masonry, not the total wall section.

Material	ho (kg/m <sup>3</sup> )	λeq (W/mK)
Masonry 20 cm thickness	2100	2.4
Masonry 26 cm thickness	2100	2.28
Masonry 36 cm thickness	2100	2.14
Masonry 56 cm thickness	2100	2.02
Masonry 76 cm thickness	2100	1.94
Rubble	1500	0.7

To calculate the U-value of heterogeneous structures, reference was made to the UNI EN ISO 6946:2008 and EN ISO 12567-1 [26,27], that also specify a method to measure the thermal conductivity of homogeneous materials. For the reference values of thermal conductivity for common building materials, references used are in the following standards: UNI 10351:2015 (homogeneous materials) and UNI 10355:1994 (not homogeneous materials) [28,29].

The next step was to define the whole section of the wall using the equivalent conductivity of the masonries described in Table 1. Thus, it was possible to determine, in six typical configurations investigated, the impact of the most common structural improvement interventions on the masonry thermal properties. The most recurrent envelope thicknesses in the Italian historical buildings of the Central Apennines were considered. Table 2 shows the outer walls investigated in some selected different thicknesses. In the table below, the middle column refers to the wall section layout characterized in terms of the constituting layers: e.g., section "plaster (2) masonry (20) rubble (8) masonry (20) plaster (2)" refers to a section with five layers, namely a 2-leaf wall with main layers made of stone masonry, with 20 cm thickness for each one, a filling layer of rubble stone with 8 cm thickness, and the plaster on both faces with an average thickness of 2 cm.

The execution of seismic retrofitting interventions generally involves the removal and then remaking of one or both of the external finishing plasters. In this case, the new plasters are made with high mechanical performance pre-mixed materials based on pozzolanic binders or mixed lime/cement.

Then a final finishing layer, made of lime-based plaster for the interior surface of the wall and cement-based plaster for exteriors, is spread on the masonry surface. In Table 3, various types of most common used plasters are shown, including some high thermal performance plasters.

Table 2. Wall thickness composition (in cm) and the U-value for all the samples analyzed. The finishes are lime/gypsum plasters.

ID	Wall Section Layout (cm)	U (W/m <sup>2</sup> K)
MS4	plaster (2) masonry (36) plaster (2)	2.485
MS6	plaster (2) masonry (56) plaster (2)	1.955
MS8	plaster (2) masonry (76) plaster (2)	1.598
MR4	plaster (2) masonry (20) rubble (8) masonry (20) plaster (2)	1.941
MR6	plaster (2) masonry (26) rubble (8) masonry (26) plaster (2)	1.734
MR8	plaster (2) masonry (26) rubble (24) masonry (26) plaster (2)	1.242

Table 3. Water vapor resistance coefficient µ	$\iota$ and conductivity $\lambda$ of finishes.
---	---

μ	$\lambda$ (W/mK)
10.7	0.7
22.7	0.9
22.7	0.9
22.7	0.9
<9	0.09
4.5	0.075
4	0.045
	μ 10.7 22.7 22.7 22.7 <9 4.5 4

<sup>1</sup> The thermal characteristics of the insulating plasters derive from the average of the values declared in technical sheets of the most used commercial products.

Often the plasters in Table 3 replace the historic lime finishes, generating a further change of the envelope thermal performance; for this reason, the choice of finishes with specific characteristics is very important in the refurbishment process. Planners must pay close attention to the choice of materials and their combination in the design phase of the retrofit intervention because this strongly determines the performance of the enclosure. Different combinations of the materials described above will be shown in the next sections in order to define the most frequent types of structural intervention on the masonry and their new thermal characteristics.

# 3. Case Study

The structural retrofit of historical buildings has achieved commendable results in the last few decades from both a regulatory and an applicative point of view [30] and is the result of research as well as historical and technical traditions, which have been fully adopted by practitioners and designers. Research suggests that when planning structural interventions, the one solution fits all approach is not effective and each intervention must be assessed with a case by case approach.

This methodology is the result of an interdisciplinary dialogue that has been taking place for some time among engineers, restorers and technicians. This interdisciplinary dialogue has not, unfortunately, been extended to the energy installations sector, which still today proceeds in a somewhat ad hoc style notwithstanding recent significant developments in this field. Indeed, an absence of specific studies on the relationship between restoration, structures and installations often results in a design that is aware of neither its own limits nor its potential.

## 3.1. Seismic Retrofit on Vertical Structures

The study focuses on the analysis of stone masonry that characterizes the building technology most often found in historical and valuable buildings. Although a variety of materials and techniques are

used, masonry presents recurrent problems with regards to seismic vulnerability and the applicability of the reinforcement techniques most frequently used. Generally the vertical structures examined are constituted by irregular shaped, stone elements (unworked stone, often of irregular shape, of different dimensions and sometimes even of different materials) that have been used to build one or more leafs. The connections between single stone blocks constituting the wall are made with low quality mortar in terms of their composition and strength. For this type of masonry, often poorly connected to the horizontal structures, the effects of seismic forces may create conditions of instability. These are related to the disintegration of the wall or part of it (expulsion of the outer leaf), to activation of kinematic mechanisms (out-of-plane or other mechanisms of local collapse) or to second mode damage due to forces acting in the plane of the wall (in-plane shear or buckling).

Interventions on masonry vertical structures are designed to increase the mechanical strength and ductility of the walls, improving the behaviour of the building subjected to static and seismic loads. The materials used are selected so as to obtain the chemical-physical and mechanical characteristics as compatible as possible with the original materials. Technical standards, mostly normalized in EU, give specific instructions about the mechanical qualification and the durability of the materials treated, which can be used for structural applications only if CE marked. Specific procedures are also defined for the classification of materials with specific characteristics of thermal conductivity, employable in the context of energy efficiency interventions. However, commercial products for which the thermal characteristics are specified and also appropriate for structural purpose are very few, and in particular, until now, no grout or structural plaster fall into this category.

The most common structural interventions can be classified into three main categories: localized repairs; localized improvement interventions that interest single discontinuity of the wall structure (such as filling of the cavities); and general improvement of the masonry characteristics, with specific actions taken to mitigate vulnerabilities associated with the type of brickwork and/or its structural elements and mortar. These interventions range from local rebuilding methodology: scuci-cuci (substitution of damaged elements with new ones, reestablishment of the structural continuity), that aims to restore the wall continuity along cracking lines; to mortar bed-joint repointing that improves deteriorated joints (both reinforced or not); to insertion of artificial diatonoi (= stones which run through the thickness of the wall and bind it together as described by Vitruvius in [30]), that consists in the improvement of the connection between multi-leaf masonry walls via the insertion of small tie beams across the wall in a regular pattern; and to external reinforcement (jacketing with Fiber Reinforced Polymers, Glass Fiber Reinforced Polymers or grids of steel) [31,32].

Particular attention is given to the consequences that interventions of mortar injections (Figure 3) facing of masonries with reinforced plaster and the insertion of artificial diàtonoi have on the thermophysical characteristics of the wall surface. The decision to focus on these kinds of intervention was because of their frequent use and massive and diffuse application.



**Figure 3.** Example of grout injected masonry in an historical building in L'Aquila: (a) cannulae distribution on the wall; (b) details.

# 3.2. Energy Efficiency Interventions on Outer Walls in Historical Buildings

Preliminary evaluations of interventions for structural improvement allow us to define appropriate combined solutions for the energy efficiency of the building. In fact, the possibility of adopting solutions that are more or less invasive will depend on the degree of invasiveness of the structural intervention. Insulating materials can be used both inside and outside buildings, and external insulation, if carried out without interruption on the surface and with suitable materials, is capable of creating a continuous barrier to the heat transfer and hence controlling humidity on the internal side of the wall. It follows that where possible, the application of an external insulation is preferable to an internal insulation [22].

In the Table 4, all the retrofit interventions analyzed (either seismic and/or combined with energetic interventions) are described. The last intervention shows the characteristics of exterior insulation finishing system (EIFS) used in energy efficiency interventions. By way of comparison, Table 9 illustrates the results of applying EIFS in three different thicknesses: 3 cm, 6 cm and 10 cm.

ID	Interventions
GI1	Grout injections and 2 cm of finishing cement plaster ( $\lambda = 1$ )
GI + RP	Grout injections + reinforced plaster and 3 cm of finishing cement plaster ( $\lambda = 1$ )
RP	Reinforced plaster and 3 cm of finishing cement plaster ( $\lambda = 1$ )
D32	Artificial diàtonoi of 32 mm diameter and original finishing plaster
D100	Artificial diàtonoi of 100 mm diameter and original finishing plaster
TP1	3 cm of thermal insulating plaster ( $\lambda = 0.075$ ), exterior side
TP2	3 cm of thermal insulating plaster ( $\lambda = 0.075$ ), exterior and interior side
EIFS	Ext. Insulation Finishing System ( $\lambda$ = 0.04) and 1 cm of finishing plaster ( $\lambda$ = 0.7)

Table 4. Definition of the seismic and energetic interventions.

## 4. Non-Destructive Testing

The thermal characteristics of stone walls are directly related to the amount of voids in the masonry. This parameter is also related to the resistance of the masonry, in terms of bearing and shear

strength. The amount of voids inside the masonry also affects its thermal characteristics, especially with an increasing regularity of the stone elements; thus with a minor percentage of voids, it is possible to observe an increase of the equivalent conductivity. Given the lack of specific experimental results, non-destructive testing was used to obtain velocity measurements and injectability measurements of different sample walls in order to estimate the voids' percentage and consequently the thermal characteristics variations of the masonry walls as a result of grout interventions.

#### 4.1. Sonic Tests to Evaluate the Percentage of Cavities in Rubble Masonries

The sonic test is a non-destructive test that measures the speed of propagation of elastic waves within a wall. These waves are produced by a pulse generator (usually an instrumented hammer) and recorded by a receiver. The wave velocity increases with the bulk density of material. Thus, the tests are able to provide indications of a qualitative nature on the strength of the masonry and the presence of cavities, cracks or material heterogeneity intercepted along the wave path. For research purposes, knowing the percentage of injected voids is fundamental to estimate the physical properties of the wall, such as thermal insulation and breathability.

Results of the experimental testing by Artioli et al. [33] carried out on rubble masonry walls of historic buildings affected by the 2009 L'Aquila earthquake, in Italy, were used for the characterization of rubble masonry. That study has identified six severely damaged buildings in the urban centers of Onna, Tempera and Sant'Eusanio Forconese, in the Aterno Valley (L'Aquila, Italy). From these buildings 21 masonry portions were obtained and investigated with sonic tests before and after the injection of grout. These walls are representative of the rubble masonry widespread in Central Italy. The rubble between the two faces is of limited size, and consists of small pieces of stone mixed with the same mortar used for laying the stones.

During the interventions, the injection time of each single hole was monitored to estimate approximately the amount of grout injected and the percentages of injected cavities. Through sonic tests, a map of sonic speeds, made up of a square grid of measuring points with distances  $0.20 \text{ m} \times 0.20 \text{ m}$ , was obtained.

The research [33] shows that although the percentage of injected voids is placed between 7% and 18%, and on average 12%, the increases in terms of sonic speeds reach values of approximately 2000 m/s, corresponding to 380% of the original speeds. This result could depend on the sonic wave path that decreases substantially after the injections because the waves do not intercept cavities and the wave is propagated through the densest part of the masonry, regardless of the percentage of the injected voids. Seeing as it is possible to assimilate the propagation of the heat flow through the wall to the sonic wave, it is also right to suppose a very high increase of the conductivity values in the post-intervention. However, in this study the increase of the equivalent thermal conductivity of the injected rubble masonry has been determined by assigning to it the same value of the average amount of percentage of injected voids, equal to 12%.

#### 4.2. Grout Injection Tests to Evaluate the Percentage of Voids on Irregular Stone Masonry

For the characterization of irregular shape, stone masonry, reference was made to the values provided by a grout injection test performed during the post-earthquake reconstruction in L'Aquila, in two buildings located in Via S. Marciano and in Via dei Vetusti (Tables 5 and 6). The injected masonry has a 60 cm thickness and is made of hewn stone in a single wall. The injection was performed by the execution of 1.4 cm diameter holes. The holes length was approximately 2/3 of the wall thickness and 1.2 cm diameter cannulae with 1 cm diameter internal hole were used (the cannula was inserted approximately 30 cm into the interior of the wall and sealed with mortar). The injections were planted 50 cm  $\times$  50 cm in the traditional staggered pattern. The testing allowed us to estimate that internal cavities were around three percent of the gross volume of the masonry; therefore, for this masonry typology the same value was assumed for the equivalent thermal conductivity increase.

**Table 5.** Sample 1 (3.20 m  $\times$  1.5 m), from a masonry wall reinforced with injections in the post-earthquake reconstruction. The building is located in Via S. Marciano, L'Aquila, Italy.

Sample 1, Via San Marciano	Survey Results		
total area covered	4.8 m <sup>2</sup>		
number of holes	19		
total injected material	150 kg		
average material injected per unit area	30 kg/m <sup>2</sup> (approximately 60 kg/m <sup>3</sup> wall)		
number of holes not responsive	5		
number of little responsive holes	7		
number of very responsive holes	7		

**Table 6.** Sample 2 ( $1.9 \text{ m} \times 1.6 \text{ m}$ ), from a masonry wall reinforced with injections in the post-earthquake reconstruction. The building is located in Via Vetusti, L'Aquila, Italy.

Sample 2, Via Vetusti	Survey Results		
total area covered	3 m <sup>2</sup>		
number of holes	14		
total injected material	87.5 kg		
average material injected per unit area	28 kg/m <sup>2</sup> (approximately 58 kg/m <sup>3</sup> wall)		
number of holes not responsive	4		
number of little responsive holes	7		
number of very responsive holes	3		

## 5. Results

The analyses on the walls were carried out by calculating the variation of the equivalent conductivity as a result of three main interventions: grout injections, grout injections and plaster reinforced with welded wire mesh, and insertion of artificial cement diatonoi with a thickness of 32 mm and 100 mm. The new equivalent conductivity obtained from the analyses is related to the several masonry types investigated and is shown in Table 7. The subscripts refers to the intervention IDs defined in Table 4. All values have been obtained without considering the finishing.

Masonry	$\lambda_{eq}$	$\lambda_{GI1}$	$\lambda_{RP}$	$\lambda_{GI + RP}$	$\lambda_{D32}$	$\lambda_{D100}$
MS4	2.14	2.20	2.145	2.204	2.16	2.149
MS6	2.02	2.08	2.025	2.085	2.04	2.025
MS8	1.94	1.99	1.96	2.002	1.97	1.95
MR4	1.71	1.91	1.724	1.914	1.75	1.75
MR6	1.75	1.96	1.764	1.974	1.79	1.79
MR8	1.331	1.49	1.34	1.502	1.37	1.38

Table 7. Values of  $\lambda eq$  (W/mK) after different kind of interventions.

The first case (GI1) is related to a retrofit intervention with grout injections: the new conductivity value was calculated according to the results provided by the experimental tests and in particular for the rubble masonry, reference was made to the non-destructive sonic tests (12% of the masonry volume is empty before the intervention and filled with grout after the intervention). For the irregular stone masonry, reference was made to the grout injection testing (three percent of the masonry volume is filled of air before the intervention and filled with grout after the intervention). The second case (RP) regards the strengthening of masonry with reinforced plaster. The third type of consolidation investigated (GI + RP) concerns combined interventions of grout injections and reinforced plaster: in this case the value of the post-intervention conductivity is higher than the two previous cases, because of both the injection and the steel roads inserted into the masonry. The latest investigated configurations are related to diàtonos interventions (D32 and D100): because of the lower invasiveness

of the intervention the conductivity increase is small. For the cases D32 and D100, four diàtonoi per unit area were considered and, in the analyses of the thermal properties of the reinforced surface, the U-value was computed in parallel. Each diàtonos was armed with a  $\phi$ 16 bar. The conductivity of the reinforced masonry through RP was obtained considering the presence of five  $\phi$ 8 parallel bars per unit area of wall. The results of the analyses carried out in order to obtain the new masonry U-values corresponding to GI1, RP and GI + RP interventions and related to the whole section of wall, are shown in the table below.

Despite the seismic intervention analyzed in Table 8, diàtonos interventions do not modify the U-value significantly (see Figure 4); for this reason, in the next steps of the analyses carried out, no energy efficiency interventions were applied.

Once the equivalent conductivity of the structural part of the masonry was calculated (Table 7), the possibility of producing variations in the U-value relating to the whole section of wall (finishing included) was investigated (see Table 9).

In the first column of Table 9, the masonry type is reported; in the second column, the structural intervention applied to the wall combined with thermal plaster; the third column gives the new U-values; and in the fourth column another energy retrofitting intervention (EIFS) has been associated to the same structural intervention. The fifth column shows the thermal transmittance relative to the combined solution (seismic and energy retrofitting interventions) which has been calculated taking into account three feasible thicknesses of EIFS solution.



Figure 4. Effects of diàtonos interventions on U-value: (a) Diameter = 32 mm; (b) Diameter = 100 mm.

ID	U-Value (W/m <sup>2</sup> K) Pre-Intervention	Intervention	U-Value (W/m <sup>2</sup> K) after Intervention
MS4	2.485	GI1	2.545
MS6	1.955	GI1	2.006
MS8	1.598	GI1	1.636
MR4	1.941	GI1	2.081
MR6	1.734	GI1	1.868
MR8	1.242	GI1	1.353
MS4	2.485	GI1 + RP	2.658
MS6	1.955	GI1 + RP	2.077
MS8	1.598	GI1 + RP	1.688
MR4	1.941	GI1 + RP	2.160
MR6	1.734	GI1 + RP	1.935
MR8	1.242	GI1 + RP	1.391
MS4	2.485	RP	2.627
MS6	1.955	RP	2.043
MS8	1.598	RP	1.665
MR4	1.941	RP	2.036
MR6	1.734	RP	1.808
MR8	1.242	RP	1.282

Table 8. Effects on U-value of the structural interventions GI1, RP and GI + RP.

Table 9. Effects on the energy efficiency interventions on the U-value.

ID	Intervention	U-Value (W/m2K)	<b>*</b>	U-Value (W/m2K)		
ID			Intervention	3 cm <sup>1</sup>	6 cm <sup>1</sup>	10 cm <sup>1</sup>
MS4	GI1 + TP1	1.564	GI1 + EIFS	0.859	0.522	0.344
MS6	GI1 + TP1	1.342	GI1 + EIFS	0.788	0.495	0.332
MS8	GI1 + TP1	1.166	GI1 + EIFS	0.723	0.469	0.320
MR4	GI1 + TP1	1.375	GI1 + EIFS	0.799	0.500	0.344
MR6	GI1 + TP1	1.279	GI1 + EIFS	0.765	0.486	0.328
MR8	GI1 + TP1	1.014	GI1 + EIFS	0.662	0.442	0.308
MS4	GI1 + RP + TP1	0.882	GI1 + RP + EIFS	0.859	0.523	0.344
MS6	GI1 + RP + TP1	0.807	GI1 + RP + EIFS	0.788	0.495	0.332
MS8	GI1 + RP + TP1	0.741	GI1 + RP + EIFS	0.725	0.469	0.320
MR4	GI1 + RP + TP1	0.820	GI1 + RP + EIFS	0.800	0.500	0.334
MR6	GI1 + RP + TP1	0.785	GI1 + RP + EIFS	0.767	0.487	0.328
MR8	GI1 + RP + TP1	0.678	GI1 + RP + EIFS	0.664	0.443	0.308
MS4	RP + TP1	0.879	RP + EIFS	0.856	0.521	0.344
MS6	RP + TP1	0.802	RP + EIFS	0.783	0.493	0.331
MS8	RP + TP1	0.737	RP + EIFS	0.720	0.568	0.320
MR4	RP + TP1	0.801	RP + EIFS	0.782	0.493	0.331
MR6	RP + TP1	0.763	RP + EIFS	0.746	0.478	0.325
MR8	RP + TP1	0.651	RP + EIFS	0.638	0.432	0.302

<sup>1</sup> The values refer to the thickness of EIFS.

It is worth noticing that not all the types of finishes analyzed in Table 3 and not all the energy efficiency solutions adopted in Table 9 are always compatible with the protection needs of historically important buildings. In Section 6, the aspects concerning the choice of better solutions will be discussed taking into account that restoration issues are a real restriction in case of high value finishes. The impact of the different finishes on the inner side, outside or on both faces of the wall for each intervention analyzed, was evaluated and is shown in Figures 5–7.



Figure 5. Comparison of the masonry thermal U-value before the interventions, after grout injections and after the application of finishing solutions for energy saving.



**Figure 6.** Comparison of the masonry thermal U-value before the interventions, after the steel mesh reinforced plastering mortar and after the application of finishing solutions for energy saving.



**Figure 7.** Comparison of the masonry thermal U-value before the interventions, after the combined reinforced system and after the application of finishing solutions for energy saving.

Finally, the intervention cost variations, caused by the use of more high performance materials in place of those used in current practice, were evaluated. These assessments were made using the regional list of Price Guidelines for Construction work in the Abruzzo Region "Prezzi Informativi Opere Edili della Regione Abruzzo". The prices indicated in this document are automatically applied in the execution of public works carried out in the Abruzzo region and, in particular, in the seismic retrofit interventions following the earthquake of 6 April 2009. The cost increment, due to the use of more high performance materials and to the different techniques adopted in the execution of finishes, was determined for each masonry analyzed, according to the following:

- Identification of all the techniques necessary for the execution of the structural improvement intervention; the cost of these techniques is constant and independent of the type of finishes adopted. The possibility of using structural materials with a higher energy performance in place of the traditional ones was not considered because it was not possible to find structural materials for which these performances were suitably certified in commerce. Lacking a reference performance standard, it is also difficult to justify the differences in cost of the materials themselves;
- Identification of the finishing techniques for the completion of the intervention, adopting finishing materials that do not possess particular energy performances;
- Application of high performance materials in place of standard materials. The possibility of adopting special plasters or EIFS of different thicknesses was considered. In Figures 8 and 9, the increment in cost relative to the use of more thermally performing materials is compared to the reduction of the U-value of the wall examined. The variation of the cost of the interventions only refers to the structural intervention.



Figure 8. Transmittance reduction vs. costs increase for stone masonry.



Figure 9. Transmittance reduction vs. costs increase for rubble masonry.

# 6. Discussion

In the previous analyses, the impact of structural interventions on masonry buildings was looked at from an energy efficiency perspective and in fact, the thermal characteristics of the structural materials analyzed are strictly connected to the building energy performance. This research demonstrates that these kinds of intervention can contribute to a worsening of the thermal performance of a wall, as Tables 8 and 9 highlight. The technological solutions suggested by this research are the

adoption of thermal insulating plaster and exterior insulation finishing system (EIFS) to compensate this worsening: the use of thermal insulating plasters or EIFS on the reinforced masonry allows an improvement of the thermal characteristics of the opaque outer walls, considerably reducing the heat flow through the wall.

The analyses carried out also show that the use of diàtonoi in the masonry does not significantly worsen the thermal characteristics of the envelope: this means that energy saving can be achieved by optimizing heating and cooling systems rather than carrying out work specifically on the envelope. This evaluation is very useful in the case of historic buildings with valuable finishes that need to be conserved. On the contrary, the injection of grout and the reinforced plaster, applied separately or in a combined solution, have a significant impact on the thermal characteristics of the masonry, leading to an increase in terms of thermal transmittance, as shown in Table 8.

Table 9 shows the reduction of the U-value through the use of thermal insulating plaster ranges from 25% to 39% in the case of grout injection interventions, from 49% to 67% in the case of reinforced plaster and from 51% to 67% in the case of combined solutions (GI1 + RP). The use of EIFS solution can even reduce the U-value from 86% (10 cm thickness) to 50% (3 cm thickness). It is worth pointing out that the application of heat-insulating plaster delivers better results in masonry of less thickness, while the performances of EIFS only varies according to the thickness with which it is applied and almost independently of masonry type.

In the case of historically valuable buildings in which finishes often have to be preserved on one or on the two sides of the wall, the structural intervention has to be designed accordingly and the use of more extreme solutions such as EIFS are not always possible: the best architectural solution has to be arrived at by following a knowledge process that includes historical and non-destructive surveys to identify the valuable elements to be preserved [15]. The applicability and the compatibility of these more invasive interventions (GI1 + RP) must be verified case by case. It is evident that where an invasive structural intervention on the masonries that includes diffuse grout injections or even the removal of the historical plaster in favor of a mesh-reinforced plaster or FRP is adequate, other more advantageous solutions for the energy performance of the envelope are equally applicable.

The graphs in Figures 8 and 9 show the transmittance variations percentage and the relative costs increase, highlighting the variation of the combined intervention (structural and energy) with respect to the structural intervention alone. The examination of the diagrams shows the economical sustainability of combined interventions: for example thermal plaster application together with a structural injection and reinforced plaster, results in a reduction in the masonry U-value of 66%, compared to a cost increase of less than 20%. Similar results characterize other combined solutions such as the application of thermal insulating plaster and mortar injections (-32% U-value reductions, 10% costs increment) or a limited thickness of EIFS application (-45% U-value, +30% costs increment) in the case of rubble masonry.

In general, the execution of energy efficiency interventions, in conjunction with structural improvement interventions, results in economic savings linked to the reduction of demolitions and restoring of the finishing components. However, it should be noted that as the insulating capacity increases, the ratio between costs and benefits is reduced because the cost of raw materials gains an ever increasing weight in the overall burden of the intervention.

# 7. Conclusions

From the analyses carried out on the masonry typology considered, structural interventions inevitably result in an increment of the thermal conductivity of the wall. This increment is significant in those cases (that are often very frequent) which utilize grout injections or grout injections and reinforced plaster. However, it is negligible when using reinforcement techniques that include insertion of artificial diàtonoi for increasing the connection between multi-leaf masonry walls.

It was possible to constitute a catalogue of outer walls that have up-to-date thermal properties in function of the structural intervention and of the energy efficiency improvement associated with it.

From the analyses, it is evident how an increment in U-values, due to the intervention of structural improvement, is drastically reduced by the use of technological solutions that are energy saving, such as EIFS or thermal insulating plaster. The use of structural materials with specific thermal properties that are currently little used would also reduce intervention costs.

Often solutions for finishes capable of compensating the increase of conductivity caused by the structural intervention or structural material with certified thermal characteristic are not frequently adopted because availability on the market is still limited. However, the use of lime, grout and mortar with low thermal conductivity could greatly reduce the thermal U-value of walls undergoing interventions to improve seismic vulnerability.

Acknowledgments: The authors would like to thank C.E.V. spa for providing the results of their grouting injection test employed in the analyses carried out in Section 4.2.

**Author Contributions:** Antonio Mannella conceived the research; Mariangela De Vita wrote the paper and performed the analyses on the masonries; Antonio Sabino and Antonio Mannella produced the graphics, analyzed the intervention cost variation and wrote Sections 6 and 7 with M.D.V.; Alessio Marchetti coordinated the research.

Conflicts of Interest: The authors declare no conflict of interest.

#### **References and Note**

- Prez-Lombard, L.; Ortiz, J.; Pout, C. A review on buildings energy consumption information. *Energy Build.* 2008, 40, 394–398. [CrossRef]
- 2. Building Performance Institute Europe. *Europe's Buildings under the Microscope*; Building Performance Institute Europe (BPIE): Brussel, Belgium, 2011.
- 3. Buratti, C.; Belloni, E.; Palladino, D. Evolutive Housing system: Refurbishment with new technologies and unsteady simulations of energy performance. *Energy Build*. **2014**, *74*, 173–181. [CrossRef]
- Buratti, C.; Belloni, E.; Lunghi, L.; Borri, A.; Castori, G.; Corradi, M. Mechanical characterization and thermal conductivity measurements using of a new 'small hot box' apparatus: Innovative insulating reinforced coatings analysis. J. Build. Eng. 2016, 7, 63–70. [CrossRef]
- European Commission. Directive 2010/31/EU of the European Parliament and of the Council of 19 May 2010 on the Energy Performance of Buildings (Recast). *Off. J. Eur. Union* 2010, *18*. Available online: https://eur-lex.europa.eu (accessed on 20 March 2018).
- Legislative Decree 22 January, 2004 No. 42. Off. J. Ital. Repub. 2004, 28. Available online: http://www. gazzettaufficiale.it (accessed on 20 March 2018).
- European Commission. Directive 2002/91/EC of the European Parliament and of the Council of 16 December 2002 on the Energy Performance of Buildings. *Off. J. Eur. Communities* 2002. Available online: https: //eur-lex.europa.eu (accessed on 20 March 2018).
- 8. Italian Ministry of Economic Development. Ministerial Decree 26 June, 2015. *Off. J. Ital. Repub.* **2015**, *39*. Available online: http://www.gazzettaufficiale.it (accessed on 20 March 2018).
- Mazzarella, L. Energy retrofit of historic and existing building. The legislative and regulatory point of view. Energy Build. 2015, 95, 23–31. [CrossRef]
- 10. Vieites, E.; Vassileva, I.; Arias, J.E. European initiatives towards improving the energy efficiency in existing and historic buildings. *Energy Procedia* 2015, *75*, 1679–1685. [CrossRef]
- Kramer, R.P.; Maas, M.P.E.; Martens, M.H.J.; van Schijndel, A.W.M.; Schellen, H.L. Energy conservation in museums using different setpoint strategies: A case study for a state of the art museum using building simulations. *Appl. Energy* 2015, *158*, 446–458. [CrossRef]
- 12. Zagorskas, J.; Zavadskas, E.K.; Turskis, Z.; Burinskiene, M.; Blumberga, A.; Blumberga, D. Thermal insulation alternatives of historic brick buildings in Baltic Sea Region. *Energy Build*. **2014**, *78*, 35–42. [CrossRef]
- Ascione, F.; Ceroni, F.; De Masi, R.F.; de Rossi, F.; Pecce, M.R. Historical buildings: Multidisciplinary approach to structural/energy diagnosis and performance assessment. *Appl. Energy* 2017, 185, 1517–1528. [CrossRef]
- Mannella, A.; De Vita, M.; Sabino, A. Interventi combinati di miglioramento sismico ed efficientamento energetico di edifici esistenti in muratura. In Proceedings of the 17th Anidis International Congress L'ingegneria sismica in Italia, Pistoia, Italy, 17–21 September 2017.

- De Berardins, P.; Bartolomucci, C.; Capannolo, L.; De Vita, M.; Laurini, E.; Marchionni, C. Instruments for Assessing Historical Built Environments in Emergency Contexts: Non-Destructive Techniques for Sustainable Recovery. *Buildings* 2018, *8*, 27. [CrossRef]
- 16. Tiber, M.; Carbonara, E. Comparing energy improvements and financial costs of retrofitting interventions in a historical building. *Energy Procedia* **2016**, *101*, 995–1001. [CrossRef]
- Calvi, G.M.; Sousa, L.; Ruggeri, C. Energy Efficiency and Seismic Resilience: A Common Approach. In *Multi-Hazard Approaches to Civil Infrastructure Engineering*; Gardoni, P., LaFave, J.M., Eds.; Springer International Publishing: Cham, Switzerland, 2016; pp. 165–208. [CrossRef]
- 18. Marques, R.; Lamego, P.; Lourenço, P.B.; Sousa, M.L. Efficiency and cost-benefit analysis of seismic strengthening techniques for old residential buildings in Lisbon. J. Earthq. Eng. 2018. [CrossRef]
- 19. Calvi, G.M. Choices and criteria for seismic strengthening. J. Earthq. Eng. 2013, 17, 769-802. [CrossRef]
- Stazi, F.; Gregorini, B.; Gianangeli, A.; Bernardini, G.; Quagliarini, E. Design of a smart system for indoor climate control in historic underground built environment. *Procedia Eng.* 2017, 134, 518–527. [CrossRef]
- Pisello, A.L.; Petrozzi, A.; Castaldo, V.L.; Cotana, F. On an innovative integrated technique for energy refurbishment of historical buildings: Thermal energy, economic and environmental analysis of a case study. *Appl. Energy* 2016, *162*, 1313–1322. [CrossRef]
- 22. Italian Ministry of Cultural Heritage and Activities and Tourism. *Guidelines for Energy Efficiency Improvements in the Cultural Heritage;* MiBACT: Rome, Italy, 2015.
- Mannella, A.; Martinelli, A. April 2009 Earthquake in Central Italy: Initial considerations about reconstruction costs and procedure. In Proceedings of the 19th International CIB World Building Congress, Brisbane, QSD, Australia, 5–9 May 2013.
- Ludovico, M.D.; Prota, A.; Moroni, C.; Manfredi, G.; Dolce, M. Reconstruction process of damaged residential buildings outside historical centres after the L'AQUILA earthquake—Part 2: Heavy Damage, Reconstruction. *Bull. Earthq. Eng.* 2017, 15. [CrossRef]
- Ludovico, M.D.; Prota, A.; Moroni, C.; Manfredi, G.; Dolce, M. Reconstruction process of damaged residential buildings outside historical centres after the L'Aquila earthquake—Part 1: Light Damage, Reconstruction. *Bull. Earthq. Eng.* 2017, *15*, 667–693. [CrossRef]
- Building Components and Building Elements—Thermal Resistance and Thermal Transmittance—Calculation Method; EN ISO 6946:2008; European Committee for Standardization: Brussels, Belgium, 2008.
- Thermal Performance of Windows and Doors—Determination of Thermal Transmittance by the Hot-Box Method—Part 1: Complete Windows and Doors; EN ISO 12567-1:2010; European Committee for Standardization: Brussels, Belgium, 2010.
- Building Materials and Products—Hygrothermal Properties—Procedure for Determining the Design Values, CTI—Thermal Insulation—Calculation and Test Methods (UNI/TS 11300-1); UNI 10351:2015; CTI (Italian Thermotechnical Committee): Milan, Italy, 2015.
- Walls and Floors. Thermal Resistance Values and Calculation Method, CTI—Thermal Insulation—Calculation and Test Methods (UNI/TS 11300-1); UNI 10355:1994; CTI (Italian Thermotechnical Committee): Milan, Italy, 1994.
- Vitruvius, P. *De Architectura Libri Decem*; Krohn, F., Ed.; B.G. Teubneri: Lipsiae, Germany, 1912; Volume 2, Chapter 8, Section 7; Available online: http://www.perseus.tufts.edu/hopper/text?doc=Perseus%3Atext% 3A1999.02.0072%3Abook%3D2%3Achapter%3D8%3Asection%3D7 (accessed on 20 March 2018).
- 31. Italian Council of Ministers, Directive 9 February 2011 of the Council of Ministers. Off. J. Ital. Repub. 2011, 47.
- 32. Binda, L.; Modena, C.; Baronio, G.; Abbaneo, S. Repair and investigation techniques for stone masonry walls. *Constr. Build. Mater.* **1997**, *11*, 133–142. [CrossRef]
- Artioli, G.; Casarin, F.; Benetta, M.D.; da Porto, F.; Secco, M.; Valluzzi, M.R. Restoration of historic masonry structures damaged by the 2009 Abruzzo earthquake through injection grouts. In Proceedings of the 9th Australasian Masonry Conference, Queenstown, New Zealand, 15–18 February 2011; pp. 15–18.



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





# Article Seismic and Energy Retrofit of the Historic Urban Fabric of Enna (Italy)

# Tiziana Basiricò <sup>1,\*</sup> and Daniele Enea <sup>2</sup>

- <sup>1</sup> Faculty of Engineering and Architecture, University of Enna Kore, via delle Olimpiadi 4, 94100 Enna, Italy
- <sup>2</sup> S.P.I.C. Engineering Ltd., via Croce Rossa, 90146 Palermo, Italy; daenea23@yahoo.it
- \* Correspondence: tiziana.basirico@unikore.it; Tel.: +39-0935-536431

Received: 12 February 2018; Accepted: 5 April 2018; Published: 10 April 2018



Abstract: This paper exemplifies several methods for retrofitting existing housing through four case studies, representative of the historical centre of Enna, a small town in Sicily, according to the requirements of static safety, typological adaptation and indoor comfort. These buildings were mostly built in the nineteenth century, up to three floors based on load-bearing masonry, iron joists and hollow tile floors and wooden roofs. A typological and construction analysis of these buildings was carried out to identify the stratigraphy of the different technical elements. Static and energy audits had been previously undertaken to understand the gap between the current state of the buildings and Italian standards and to develop appropriate interventions taking into account the site characteristics and the energy and seismic risk class pre- and post-retrofit intervention. The analyses and the retrofit interventions were performed in compliance with Italian standards and laws and strove to reach the minimum level. The study supports the planning of structural and energy retrofit interventions designed for historic load-bearing masonry buildings. Finally, the study simulates a strategy of action to provide subsidies and tax relief related to effective seismic and/or energy improvement that could be relevant for owners/builders as well as for local authorities.

Keywords: seismic retrofit; energy retrofit; historic urban fabric

## 1. Introduction

The static safety of existing buildings in Italy is increasingly important, especially due to the intensity of recent earthquakes (L'Aquila, 6.3 magnitude on the Richter scale, in 2009; Emilia Romagna, 6.0 in 2012; and Umbria, 6.6 in 2016), that caused the loss of human lives [1]. This is true also in other countries (Sumatra, 9.1, 2004; near the coast of Sendai, Japan, 9.1, 2011).

Compared to other earthquakes of higher magnitude, which occurred in areas where constructions were built according to anti-seismic criteria, the number of victims was considerably higher due to the vulnerability of the Italian buildings in historical centres impacted by the seism. For example, as a result of the 1968 Belice earthquake in Sicily, which had a magnitude of 6.1, 296 people were killed and 563 injured, compared to the last eight earthquakes that occurred in California from 2000 to 2014, with magnitudes between 5.0 and 7.2, where seven people were killed and less than 500 people were injured [1].

Of the 7.7 million residential buildings built before the first Italian anti-seismic law—Law No. 64 of 2 February 1974 [2]—almost 55% were built using the load-bearing masonry technique and mainly placed in historical centres [3].

Energy efficiency is also a relevant subject of current debate, considering the European regulations, e.g., Directive 2012/27/EU, and the proposed revision that was discussed and presented on 30 November 2016, proposing energy targets for 2030, including a revised Energy Performance of Buildings Directive (EPBD) [4].

On 19 December 2017, a provisional political agreement was subscribed to by negotiators from the Council of the European Union, the European Parliament, and the Commission, to update the EPBD and introducing new measures, particularly oriented to the renovation of existing buildings [5].

The preservation of buildings in historical centres from damage due to earthquakes, the need to make these buildings safe, especially for their historical value, became essential, to allow their repopulation. The current seismic, energy and functional inefficiency, in fact, has led to the abandonment of some of these historical centres.

Only in Sicily, there are sixteen "ghost" villages and in Italy overall the last census showed 190 abandoned villages [6]. Together with the abandonment of whole villages, the historic urban fabric, which is the core of the cultural identity of the Italian countryside, does not meet the current standards. The south of Italy shows the largest number of abandoned buildings. Table 1 provides some details. Milan has a similar number of abandoned residential buildings as the cities of the south of Italy [7].

Municipality	Population	Residential Buildings	Abandoned Residential Buildings	Abandoned Buildings Percentage
Palermo	672,398	44,499	2068	4.65
Catania	315,769	26,755	1272	4.75
Reggio Calabria	182,323	26,368	2010	7.62
Naples	969,490	40,344	1107	2.74
Rome	2,874,529	107,332	2365	2.20
Milan	1,353,467	42,628	1791	4.20
Turin	885,651	35,814	1119	3.12

Table 1. The most populous cities of Italy with data on residential buildings and abandoned buildings.

This has often led to degenerative instabilities and collapses and an increasingly widespread architectural and urban deterioration.

The intervention into these buildings supports the conservation of local territory, both in terms of reducing the risks of decay and collapse and abandonment [8,9]. It is also a tool to avoid the loss of cultural and technological knowledge of a place.

It is becoming increasingly important to incentivize and study methodologies and techniques for the regeneration of these buildings that are able to combine the achievement of current performance standards with the preservation of their peculiarities.

The two above-mentioned approaches are often planned and performed separately, considering the seismic aspect a priority, thus ignoring the importance of having energy-efficient spaces to live, or vice versa, considering the energy aspect first, and ignoring the importance of having safe spaces to live.

Besides this, the interest of communities is more frequently pointed towards monuments or public buildings due to their social value, such as town halls [10], schools [11], and churches [12], not considering the two approaches combined.

Even when the interest is focused on vernacular architecture, the analysis is carried out of building types and spatial organization, not considering seismic and energy improvement interventions [13].

Belleri and Marini strongly recommend integrated interventions, combining structural, architectural and energy improvement interventions, resulting in a more sustainable approach [14].

Integrated interventions, particularly energy-efficient interventions, reduce the environmental impact of buildings (in terms of energy consumption, operating costs and CO<sub>2</sub> emissions) and even it increases the life expectancy of the building.

Some recent studies of the "minor historical centres"—with historical buildings of little value, often located in small towns—of the Abruzzo area, in the centre of Italy—particularly of the village of Sant'Eusanio Forconese [15]—focused on both seismic and energy improvements, and on the importance of the conservation of the characteristics of the vernacular architecture.

This study, consistent with the work carried out by several research groups in recent years, especially in Italy, aims to further enrich technical knowledge and to demonstrate that it is possible to meet the essential requirements of structural safety and energy improvement, with the preservation of cultural values [11,16–18].

Particular attention was paid to the choice of interventions adopted in the various cases, choosing, where possible, non-invasive, reversible techniques compatible with traditional materials. The approach to intervention was that of "minimal intervention", according to the basic principles of the NIKER project (2010–2012) [19], co-funded by the European Commission, which sees in "minimum intervention" the only intervention method to guarantee compatibility and low intrusion.

The study, therefore, aims to develop integrated seismic and energy intervention strategies for the minor construction of historic centres, starting from the analyses performed on the selected case studies.

Thus, the selected target was the historic centre of Enna, and among this building stock, the regeneration of four representative buildings was studied.

Several intervention methodologies were identified, being potentially standardizable for similar building types, in other similar historical centres, to obtain those requirements according to "contemporary" use (static safety, typological adaptation, environmental comfort).

These intervention methodologies require a deep knowledge of the building in its current state from a typological, construction and energy point of view.

Through the application of recent Italian regulations on energy efficiency and seismic improvement, published respectively in 2015 and 2017, the seismic risk class and the energy class were identified before and after the intervention, which was necessary to formulate optimal strategies.

# 2. Methods and Case Studies

To define an intervention methodology for the historic urban fabric, it was necessary to carry out an analytical phase, which led to knowledge of the typological, construction, seismic and energy characteristics of the current state of the building. The next step was calculation, taking into account the different alternatives for seismic and energy improvement intervention, leading to the development of the various intervention strategies.

The methodology for the analytical phase was performed on two different levels. On the urban scale, it concerned the urban fabric, the shape and orientation of the blocks and their relationship with public spaces and contiguous buildings. At the architectural scale, the study provided a systematic collection of relevant data on the building techniques, materials and finish, state of conservation, use and spatial organisation [20]. Interest was focused on the four main technical elements of these buildings (walls, slab-on-ground floor, roofing and doors and windows).

# 2.1. Seismic Analysis

Since most of the buildings in the historic centres mainly consist of private residential buildings (not protected by law), in order to carry out a large-scale seismic improvement interventions, an Italian law was necessary to encourage private people to carry out these types of intervention.

In accordance with these objectives, the guidelines for the classification of the seismic risk of buildings (Decree of the Minister of Infrastructure and Transport no. 58, dated 28 February 2017, as modified by the Decree of the Minister of Infrastructure and Transport no. 65, dated 7 March 2017) were approved, combining the seismic improvement of buildings and economic incentives, called "sismabonus" [21].

These guidelines are important because they define for the first time the subdivision into classes of the seismic risk of buildings (from A+ to G), before and after any anti-seismic interventions; the seismic risk is defined as a measure to assess the expected damage following a possible seismic event. This depends on the interaction of related factors, namely the seismic hazard of the area, divided into high (zone 1 and zone 2), medium (zone 3) and low (zone 4), the vulnerability of buildings and the exposure of the various contexts.

The seismic risk class depends on two parameters: the average annual expected loss (PAM, the Italian acronym reported in the abovementioned law), which takes into account the economic loss associated with the damage and refers to the cost of reconstruction of the building; the IS-V safety index of the structure, defined as the ratio between the peak ground acceleration (PGA), which determines the achievement of the limit state for safeguarding life, and the value of the PGA that the law indicates for the site where the building is located.

The PGA represents the maximum acceleration value of the soil measured during an earthquake. It takes into account the influence of any amplification effects of the seismic motion, depending on the subsoil characteristics or topography.

The guidelines include two methods of seismic diagnosis, the "simplified" method, applicable only to load-bearing masonry constructions and local reinforcement surveys and interventions, taking into account only the PAM parameter, and the advanced method, the so-called "conventional" method.

The simplified method was applied to the selected case studies, introduced in §2.3, namely those built using load-bearing masonry.

Based on the type of masonry, which is divided into seven classes by the EMS-98 [22], the average vulnerability class is individuated by a table provided by the abovementioned Italian law.

Once the average vulnerability class is known and the seismic hazard area is identified, according to the ordinance of the Italian Prime Minister No. 3274 on 20 March 2003, the risk class is defined by the provided table and the PAM class is assigned consequently.

## 2.2. Energy Audit

The energy analysis was carried out through analytical calculations, as it is not possible to perform an in situ experimental analysis (e.g., thermal resistance measurement by means of guarded hot plate and heat flow meter methods), because the analysed buildings are abandoned, with partial collapses, and lacking doors and windows, and therefore there is no difference in temperature between the inside and outside.

Two types of energy analysis were conducted simultaneously. The first considered the entire building according to Italian standards, thus calculating the energy class of the buildings. The second considered the local type through the calculation of the thermal transmittance of the technical elements of the envelope according to UNI EN ISO 6946:2008.

While in order to be able to connect the thermal loss to the building types, the surface percentages of the various technical elements of the envelope were calculated with respect to the total surface.

The energy audit for the determination of the energy class of the buildings in the historic centre of Enna consisted of two parts: the analysis of energy performance in the current situation and the analysis of energy performance after the interventions.

All the characteristics (typological, geometrical and construction) found in the first phase were summarized in a mathematical model that shows the energy performance of the building.

The building energy simulation was performed by means of the open access software DOCET 3 ITC—ENEA (Italian National Agency for New Technologies, Energy and Sustainable Economic Development). This software conforms to the set of Italian technical standards UNI/TS 11300 series [23], which implements the European standards EN 15316 series and EN 15243:2007 [24], and is appropriate for evaluating the energy class.

Performing the energy analysis before and after the intervention, the software verifies how much each intervention influences the energy performance of the building. Only interventions on the elements of the envelope that could increase the energy performance were considered, neglecting any improvements in the performance of the plants, considering the absence of the latter in the selected case studies.

In the discussion part of this study, the choice of the energy performance improvement intervention is evaluated in relation to the interventions necessary to improve the seismic efficiency.

# 2.3. Four Case Studies in the Historic Centre of Enna

Enna is the highest Italian provincial capital at 931 m above sea level, a small town in the hinterland of Sicily, south of Italy, and due to this position it earned the nicknames *Urbs Inexpugnabilis* (unassailable town) by Romans, *belvedere* (panoramic viewpoint) and *ombelico* (navel) of Sicily.

The historic centre of Enna presents features common to many other Sicilian and Italian historical centres, such as a narrow street network, load bearing masonry buildings of not more than three levels, plaster and/or fair-faced stone masonry, Sicilian tiles as the roofing finish and green or brown wooden doors and windows.

Large parts of the historic centre still maintain their identity and the original typological and morphological characteristics of the building fabric, not altered by new construction.

This is also due to the orography of the land, with the historic centre of Enna on a mountain (so-called "Enna Alta") and a consequent expansion of the city downstream ("Enna Bassa"), as shown in Figure 1, dating back to 1950.



Figure 1. View of the historic centre of Enna.

However, this expansion has in part caused the abandonment of some buildings in the historic centre.

Many people preferred to move to new buildings better suited to current housing needs.

The lack of use of most building in the historic centre eventually caused partial collapses (Figure 2).



Figure 2. Examples of abandoned and partially collapsed buildings in the historical centre of Enna.

The last census of residential buildings in Enna showed 4791 buildings, of which 3376 were built with load-bearing masonry (937 were in a bad condition, partly abandoned, while 144 were in complete decay and abandonment) before 1970, thus without anti-seismic criteria [25].

From the historic urban fabric of Enna, four case studies were identified with different morphological, typological and construction characteristics. The buildings were chosen in the historical fabric of "Enna Alta".

The typological analysis followed a procedure that considered only the morphological characteristics detectable from the outside and from the cartography [26].

The survey started with research into archival documents on related similar studies and on the contribution made by the present and past urban plans. In particular, we followed the same method of subdivision into building types used by the Detailed and Recovery Plans of the Municipality of Enna from 1990.

The urban morphology of the historic centre of Enna is very complex. Its urban fabric is characterized by a great variety of building types, presenting very different configurations within the different urban meshes in which they are set.

The building types that can be found are linked to the urban hierarchy deriving from the road relationship and the morphology of the site, distinguishing buildings of "major" and "minor" importance.

According to this hierarchy, terraced buildings of considerable consistency were found, overlooking the main streets and monumental buildings (churches, convents and noble palaces) or the squares.

In addition, along the cross streets of the main axes and behind the squares, other significant building typologies were identified: "corner buildings", almost exclusively single-family, "at court" and some cases of "tower" houses.

Excluding the recent and large volume residential buildings (4–6 floors), present in a limited number, residential buildings are characterized by buildings with two or three levels above ground.

In order to gain knowledge of the typology examined, axonometric schemes were produced reproducing the fundamental aspects of the four building types: a corner building (Figure 3a), terraced building (Figure 3b), tower building (Figure 3c) and at court building (Figure 3d).



Figure 3. Axonometric scheme of the four building types.

From these schemes, such characteristics emerge as, for example, the ratio between horizontal and vertical development, the division into living cells and the contiguity between the buildings.

Plans and schematic sections were produced, showing how the typology is generally inserted within the urban fabric and whether, usually, this type is placed inside or on the front portion of a building block, whether it has a courtyard or not and whether it faces a road axis.

The selected case studies are representative of the following types (Figure 4):

- 1. Corner building: a building consisting of the union of two contiguous living cells, placed at the corner of a quarter, constituted by irregular plots. The plots at the rear saturated the space. (Case No. 1 located in Orfanotrofio St.).
- 2. Terraced building: the building is placed on the front portion of a building block (Case No. 2 located in Portosalvo St.).
- 3. Tower building: the building is defined by its height and it breaks the alignment of the block to which it belongs. It is characterized by one single room for each floor, all connected by stairs to each other (Case No. 3 located in Zacche St.).
- 4. At court building: the building is set on the edge of a "court" building block (Case No. 4 located in Colajanni St.).



Figure 4. Four case studies in Enna (Sicily).

As is common for Italian minor centres, Enna is characterized by a spontaneous "architecture without architects" style [27] developed by local builders, according to local topography and climate.

These were mainly built between the end of the 19th century and the beginning of the 20th century [28], with load-bearing masonry (mainly irregular stones), girder and hollow brick floors or wooden beams with gypsum vaults floors and wooden roofing. All the buildings selected were not currently in use and in a state of abandonment [29].

In relation to the size and shape of the stone elements and the quality of the mortar and the stone, the walls present several textures with different strengths, lacking an effective connection between them and the horizontal elements. In addition, floors and roofing mainly made of iron and wooden elements have low in-plane stiffness and strength. These buildings have very limited resistance to seismic actions for both horizontal stresses in the masonry (shears stress) and out-of-plane (buckling stress), therefore requiring seismic retrofitting. Besides, the lack of structural integrity due to a poor connection between structural elements is one of the main causes of earthquake damage in vernacular buildings, where the "box-behaviour" is not guaranteed [13].

The Case No. 1 building is in a state of abandonment and presents the partial collapse of the roof and floors. It consists of load-bearing irregular stone masonry, consisting of 34 cm for external walls and 20 cm for internal walls, finished outside with 2 cm cement-based plaster and inside with 1.5 cm gypsum plaster, wooden floors with gypsum vaults, according to traditional construction techniques. It has non-thrusting roofs with wooden beams arranged parallel to the eave line, completed with planks and curved tile roofing as shown in Figure 5a.

The Case No. 2 building is in a good state of conservation and has undergone some energy improvement interventions in recent years. It is also made of irregular stone load-bearing masonry with an average thickness of 70 cm, with an insulating inner layer made of 4 cm extruded polystyrene panels, an internal finish of 1.5 cm gypsum plaster and an external finish of 2 cm hydraulic lime plaster. The first floor was rebuilt in the 70s and is a hollow brick floor. The internal space is divided into three levels, one of which is the basement, and the two original floors consist of wooden beam floors to support the bedding mortar and the flooring; the intrados are plain with a cane mesh ceiling finished with plaster mortar. A similar construction technique with wooden beams and an upper and lower layer of 1.5 cm cane mesh was found in the roof, finished with Sicilian curved roof tiles, as shown in Figure 5b.

The Case No. 3 building is abandoned and shows collapse of the inter-floor slab. The load-bearing structure is made of irregular stonewalls, 60 cm thick, finished externally with hydraulic lime mortar and internally with gypsum plaster. The inter-floor slab is made of iron beams and bricks, the roof of wooden beams, joists and Sicilian curved tiles as shown in Figure 5c.

The Case No. 4 building is in good condition. It shows a load-bearing structure of irregular stone, 85 cm thick, an inter-floor slab with iron beams and brick tiles and a wooden roof, finished with Sicilian curved tiles and wooden beams covered by 1.5 cm of plasterboard as shown in Figure 5d.

All the case studies show similar characteristics: the slope of the roofing is between 35% and 40%, the height of each floor is between 375 cm and 420 cm, and the shape of the interior rooms is polygonal.

In order to develop adequate seismic and energy improvement interventions, static and energy checks were performed accounting for the characteristics of the site—geographical coordinates lat. 37.34, long. 14.16, seismic zone 2 and climate zone E, degree-day 2.248—identifying the seismic risk class and the energy class.



Figure 5. Structural cross section of the case studies.
# 3. Results

# 3.1. Seismic Analysis

For the case studies analysed, the simplified method was applied for the calculation of the seismic risk class, suitable for masonry buildings, considering the PAM parameter.

For each case study, by applying the guidelines the vulnerability class of the structure was identified based on the European Macro-Seismic Scale (EMS), which includes six classes of vulnerability from V1 to V6, based on the type of masonry, as reported in Figure 6. In all the four case studies, based on "irregular stone masonry", the most probable class of vulnerability (mean class) is V5. However, in cases No. 1 and No. 3, there is a deviation from the mean class with an increase in vulnerability to V6, due to the conditions of high degradation and collapse (collapse of the floors and part of the roof).

and the second second		CLASS OF VULNERABILITY						
L	TYPE OF STRUCTURE	V <sub>6</sub> (mA <sub>EMS</sub> )	Vs (mB <sub>EMS</sub> )	V4 (=C <sub>EMS</sub> )	V <sub>3</sub> (=D <sub>EMS</sub> )	V2 (=EEMS)	V1 (=FEMS)	
	Irregular stone without mortar	0	1					
	Raw earth brick	0-	-					
×	Rough-hewn stone	+	-0					
MASONR	Massive stone for monumental construction		+	-0-	-1			
	Square ashlars and bricks	+	0-	1				
	Bricks and high stiffness floors		-	-0-				
	Confined and/or reinforced			+	-0-	-1		

Figure 6. Class of vulnerability of the structure [21].

By relating the vulnerability class and the danger of Enna's location (seismic zone 2), the seismic risk class was defined among the eight classes provided by the guidelines. According to the table of guidelines shown in Figure 7, the buildings of case No. 1 and No. 3 can be classified within the F risk class, while the buildings of cases No. 2 and No. 4 can be classified within risk class E.

Risk Class	PAM	Zone 1	Zone 2	Zone 3	Zone 4
A+*	PAM ≤ 0,50%	1			$V_1 \div V_2$
A*	0,50% <pam≤1,0%< td=""><td>1</td><td></td><td><math>V_1 \div V_2</math></td><td><math>V_3 \div V_4</math></td></pam≤1,0%<>	1		$V_1 \div V_2$	$V_3 \div V_4$
B*	1,0% <pam≤1,5%< td=""><td>V1</td><td>V1 ÷ V2</td><td>V<sub>3</sub></td><td>V5</td></pam≤1,5%<>	V1	V1 ÷ V2	V <sub>3</sub>	V5
C*	1,5% <pam≤2,5%< td=""><td>V<sub>2</sub></td><td>V3</td><td>V4</td><td>V<sub>6</sub></td></pam≤2,5%<>	V <sub>2</sub>	V3	V4	V <sub>6</sub>
D*	2,5% <pam≤3,5%< td=""><td>V<sub>3</sub></td><td>V4</td><td>V5 ÷ V6</td><td></td></pam≤3,5%<>	V <sub>3</sub>	V4	V5 ÷ V6	
E*	3,5% <pam≤4,5%< td=""><td>V4</td><td>V<sub>5</sub></td><td></td><td>1</td></pam≤4,5%<>	V4	V <sub>5</sub>		1
F*	4,5% <pam≤7,5%< td=""><td>V5</td><td>V<sub>6</sub></td><td></td><td></td></pam≤7,5%<>	V5	V <sub>6</sub>		
G*	7,5%≤PAM	Ve			

Figure 7. Seismic risk class from the Italian guidelines [21].

From these examples, it is clear that for existing buildings with load-bearing masonry it is not possible to obtain a seismic risk class higher than E or D for areas with high seismic risk, not adding into the calculation the geometric and construction characteristics of the building.

## 3.2. Energy Analysis

The energy analysis performed before and after the various recovery interventions allowed us to develop optimal energy intervention strategies. From the global thermal analysis, considering the various buildings without winter and summer air conditioning systems and accounting only for the energy required for lighting and domestic hot water, it was found that all the buildings studied were in class G.

Since the buildings in the case studies are single family residences, the energy improvement interventions affect the entire building envelope and therefore a percentage higher than 50% of the gross surface of the entire building.

These interventions fall in the typology defined by the D.M. of 26 June 2015, important first level renovations [30], and therefore the energy performance requirements to be verified concern the entire building.

The verification of the thermo-physical characteristics of the various elements of the building envelope, on the other hand, was useful to identifying the elements where the heat loss was higher, based on the thermal conductivity of every layer, calculated according to UNI 10,351:2015 [31], UNI EN ISO 10,456:2008 [32] and UNI EN ISO 13,370:2008 [33] (Tables 2–5), and thus guiding the energy regeneration strategies.

Type of Structure	Original Layers	Thickness [m]	$\lambda [W/mK]$	R [m <sup>2</sup> K/W]	U [W/ m <sup>2</sup> K]
	Internal hydraulic lime and gypsum plaster	0.015	0.70	0.021	
F 1	Solid brick	0.340	1.30	0.226	
Façade Wall	External cement-based plaster	0.020	0.90	0.022	
	*			0.440 *	2.271
Roofing	Fir plank	0.020	0.12	0.167	
	Sicilian curved tiles	0.010	1.00	0.010 0.317 **	3.158
Slab-on-ground floor	Ceramic floor tiles	0.012	1.00	0.012	
	Bedding cementitious mortar	0.030	1.30	0.023	
	Concrete layer	0.200	1.90	0.100	
	Rubble drain	0.400	1.80	0.220	
				0.355 ***	0.876

Table 2. The layers of the envelope's elements and their relative thermal resistance for Case No.1.

\* indicates the values of thermal resistance, R, for walls, considering the addition of the superficial internal thermal resistance equal to 0.13 m<sup>2</sup>K/W and the superficial external thermal resistance equal to 0.04 m<sup>2</sup>K/W; \*\* indicates the values of R, for roofing, considering the addition of the superficial internal thermal resistance equal to 0.1 m<sup>2</sup>K/W and the superficial external thermal resistance equal to 0.1 m<sup>2</sup>K/W and the superficial external thermal resistance equal to 0.1 m<sup>2</sup>K/W and the superficial external thermal resistance equal to 0.1 m<sup>2</sup>K/W and the superficial external thermal resistance equal to 0.04 m<sup>2</sup>K/W; \*\*\* indicates the values of R<sub>f</sub>, the thermal resistance of the slab-on-ground floor.

Thermal transmittance in the case of the slab-on-ground floor was calculated in accordance with UNI EN ISO 13370:2008:

$$U = \frac{2\lambda}{\pi B' + dt} ln \left(\frac{\pi B'}{dt} + 1\right)$$

 $B' = \frac{2A}{P}$ , characteristic dimensions of the floor, being A its surface and P its perimeter;  $dt = w + \lambda \left( R_{si} + R_f + R_{se} \right)$ , being w the total thickness of walls;  $\lambda = 2 \text{ W/m}^2\text{K}$ , the thermal conductivity of ground (sand or gravel);  $R_{si} = 0.17 \text{ m}^2\text{K/W}$ , the internal superficial thermal resistance for descending flux;  $R_{f}$ , = the thermal resistance of the floor;  $R_{se} = 0.04 \text{ m}^2\text{K/W}$ , the external superficial thermal resistance. In order to be able to connect the thermal loss to the building types, the surface percentages of the various technical elements of the envelope were calculated with respect to the total surface, as in Figure 8a.

Type of Structure	ype of Structure Original Layers		$\lambda [W/mK]$	R [m <sup>2</sup> K/W]	$U \left[ W / m^2 K \right]$
	Internal hydraulic lime and gypsum plaster	0.015	0.70	0.021	
Façade wall	Extruded polystyrene panels	0.040	0.04	1.000	
	Calcarenite irregular stone masonry [34,35]	0.700	1.50	0.467	
External NHL plaster		0.020	0.90	<u>0.022</u> 1.680 *	0.595
Roofing Internal hydraulic lime and gypsum plaster		0.010	0.70	0.014	
	Cane mesh ceiling	0.015	0.06	0.273	
	Cavity	0.100	0.25	0.400	
	Cane mesh ceiling	0.015	0.06	0.273	
	Cementitious screed	0.030	1.30	0.023	
	Sicilian curved tiles	0.010	1.00	<u>0.010</u> 1.132 **	0.883
Slab-on-ground floor Bedding cementitious mortar		0.050	1.30	0.038	
	Rubble drain	0.800	1.80	<u>0.444</u> 0.482 ***	0.638

Table 3. The layers of the envelope's elements and their relative thermal resistance for Case No.2.

### Table 4. The layers of the envelope's elements and their relative thermal resistance for Case No.3.

Type of Structure	Original Layers	Thickness [m]	$\lambda [W/mK]$	R [m <sup>2</sup> K/W]	U [W/ m <sup>2</sup> K]
	Internal hydraulic lime and gypsum plaster	0.015	0.70	0.021	
Façade wall	Calcarenite irregular stone masonry	0.600	1.50	0.400	
	External NHL plaster	0.020	0.90	$\frac{0.022}{0.614}*$	1.630
Roofing	Fir plank	0.020	0.12	0.167	
	Sicilian curved tiles	0.010	1.00	<u>0.010</u> 0.317 **	3.158
Slab-on-ground floor	Bedding cementitious mortar	0.030	1.30	0.023	
	Concrete layer	0.200	2.00	0.100	
	Rubble drain	0.800	1.80	0.444 0.567 ***	0.589

Table 5. The layers of the envelope's elements and their relative thermal resistance for Case No.4.

Type of structure	Original Layers	Thickness [m]	$\lambda$ [W/mK]	R [m <sup>2</sup> K/W]	U [W/ m <sup>2</sup> K]
	Internal hydraulic lime and gypsum plaster	0.015	0.70	0.021	
Façade wall	Calcarenite irregular stone masonry	0.850	1.50	0.567	
	External NHL plaster	0.020	0.90	<u>0.022</u> 0.780 *	1.282
Roofing	Plasterboard false ceiling	0.015	0.21	0.071	
	Cavity	0.015	0.25	0.600	
	Fir plank	0.020	0.12	0.167	
	Sicilian curved tiles	0.010	1.00	0.010	
				0.988 **	1.012
Slab-on-ground floor	Marble grain floor tiles	0.010	2.80	0.004	
	Bedding cementitious mortar	0.020	1.30	0.015	
	Concrete layer	0.200	2.00	0.100	
	Rubble drain	0.400	1.80	0.222	
				0.337 ***	0.762

The reduced percentage of door and window surfaces between 3% and 8%, typical of traditional buildings of historic centres, which should make it possible to consider them negligible for the calculation of thermal transmittance.

Calculations show that under more unfavourable conditions, the increase found—considering the thermal input of doors and windows—never exceeded 2.21%.

From the comparison of the thermal transmittances obtained for the various technical elements of the envelope with the performance requirements, according to the current regulations, the thermal transmittance values were found to be considerably higher than the threshold values, as shown in Figure 8b.



Figure 8. (a) Surface percentages of the various technical elements of the envelope on the total surface; (b) Thermal transmittance of the technical elements of the envelope of the considered case studies.

The elements with higher heat losses were the roofing (maximum value  $3.22 \text{ W/m}^2\text{K}$  for cases No. 1 and 3) exceeding from 4 to 15 times the threshold thermal transmittance, equal to  $0.25 \text{ W/m}^2\text{K}$ , and the walls (maximum value  $2.25 \text{ W/m}^2\text{K}$  for case No. 1), exceeding from four to seven times the threshold value of  $0.3 \text{ W/m}^2\text{K}$ . Only in case No. 2, for the walls ( $0.61 \text{ W/m}^2\text{K}$ ), which exceeded by only two times the threshold value, as this building had experienced energy improvement interventions with internal insulation.

Dealing with the walls of all of the case studies, a reduced thickness determined an increase in thermal transmittance.

The maximum value was U =  $2.271 \text{ W/m}^2\text{K}$ , for case No. 1 this was s = 0.375 m. The U percentage reduction for case No. 2 was equal to 73.80%, with an increase in the wall thickness of 106.67%. For case No. 3, the U percentage reduction with respect to case No. 1 was equal to 28.25%, with an increase in the wall thickness of 69.33%. For case No. 4, the U percentage reduction with respect to case No. 1 was equal to 43.57%, with an increase in the wall thickness of 136.00%.

For the floors on the ground, the thermal transmittance was calculated between  $0.589 \text{ W/m}^2\text{K}$  and  $0.876 \text{ W/m}^2\text{K}$  (according to UNI EN ISO 13370:2008).

## 3.3. Strategies for Seismic and Energy Improvement of Historical Buildings

Once the risk class of the case studies was defined, it was possible to evaluate any risk improvement and mitigation measures. However, using the simplified method it was possible to choose only local interventions, which do not produce substantial changes in the behaviour of the structure as a whole, addressing only the passage to the higher risk class.

Following the indications of the interventions provided in the guidelines, on the basis of similar solutions adopted in other European countries [36] and the findings regarding recent seismic events in central Italy, a list of possible actions for the case studies analysed was derived (Table 6). These seismic improvement interventions are directed to pursue an overall "box" behaviour, which eliminates or rather reduces local mechanisms outside the plane of the walls. Thus, these interventions aimed at

the improvement of the relative mechanical resistance and the strengthening of mutual connections between them and the horizontal elements (floors and roofs).

The interventions on the walls consisted of the improvement of the mechanical strength in two ways: consolidation by means of mortar injections or glass fibre reinforced polymer (GRFP) net and hydraulic lime or thermally insulating mortar (with perlite or polystyrene), 3 cm thick [37].

To increase the wall-to-wall connection, tie-rods were considered on both sides of the wall (Figure 9a), while in the case of adjacent buildings, the solution chosen was the realization of an internal iron hooting (Figure 9b). The effectiveness of the tie-rods connecting the walls to each other was also demonstrated by the analysis of the damage undergone by a building in L'Aquila during the 2009 earthquake, since it was a seismic retrofit intervention in 2003, as demonstrated in the study by Lucibello [38].



Figure 9. (a) Intervention in an isolated building by tie-rods and (b) on a terraced building by iron hooping.

Site of Intervention	Retrofit Solution	Description
Walls	S <sub>W1</sub>	Mortar injections
	S <sub>W2</sub>	GRFP
	$S_{W3}$	Wall to wall connection improvement through tie-rods or internal hooping by means of iron profiles
Floors	S <sub>F1</sub>	Stiffening and connection with the load-bearing walls
	S <sub>F2</sub>	Reconstruction
Roofing	S <sub>R1</sub>	Steel or reinforced brick bond-beam with steel tie-rods or carbon fibre
	S <sub>R2</sub>	Stiffening of the existing roof with steel tie-rods in the intrados
	S <sub>R3</sub>	Connection of the existing roofing with the masonry through tie-rods or internal hooting by means of iron profiles

Table 6. Possible actions provided for the selected case studies.

Interventions on the floor depend on its static conditions. If in good status, the intervention consists of the stiffening of the existing floor and the connection with the load-bearing walls. Steel ties are the recommended intervention. The intervention with steel ties is less invasive, not requiring the demolition of the existing pavement, as reported by Branco and Guerreiro [39].

Chaining with steel profiles, moreover, has been implemented every time it has been necessary to reconstruct floors or roofing, with glued laminated timber or cross-laminated timber panels. These "C" or "L" steel profiles make it easier to set up new floors as well as creating a connection between vertical and horizontal elements. The connection between parallel walls and the floor/roof, using steel plates and wooden beams as connecting elements is one of the techniques of seismic reinforcement proposed by Diz and Costa for supporting masonry buildings in the Azores [40].

In the case of the maintenance of existing steel floors, stiffening interventions have been provided from the extrados by means of reinforced concrete slabs, connected to the masonry through injections with stainless steel bars (Figure 10a).



Figure 10. Connection interventions between walls and slabs.

All the selected interventions were compatible with the chemical–physical characteristics of natural stone materials and were in most cases reversible [41–43].

Otherwise, the reconstruction of the floor is provided for with lamellar wood beams or cross-laminated timber panels, both making a connection to the walls with steel profiles (Figure 10b).

The lamellar wood beams reproduce the original static scheme, which also acts as a chaining, facilitating its installation; the cross-laminated timber panels ensure greater stiffness of the plane.

Interventions into roofing consist of the stiffening of the existing roof with steel tie-rods in the intrados and its connection with the masonry through similar interventions to those proposed for the floors or the steel or reinforced brick bond-beam with steel tie-rods or carbon fibre in the case of roofing reconstruction.

The results of the energy analysis demonstrated the need for energy improvement. Considering the climatic conditions of the place, it was more important to provide interventions to reduce consumption related to winter heating. Excluding an intervention on the plant, only passive insulation measures (internal and/or external) of the envelope with traditional and innovative materials, with high-energy performance and small dimensions were considered.

In order to provide optimal energy performance intervention strategies [44], checks and calculations were carried out, through the combination of the following four interventions on the insulation of external walls, roofing and slab-on-ground floor, doors and windows.

From the various simulations performed for the four case studies, the best performance was found to occur either on the walls or the roofing or both, in relation to the percentage extension of one or the other with respect to the entire surface.

Dealing with energy interventions on walls, the choice between external and internal insulation was basically dictated by the presence or absence of the external plaster or any stone elements in the façade, as it was permitted to intervene from the outside if the building was finished by plaster.

It was preferred, if possible, to use the system with the external insulation, guaranteeing better living comfort both in summer and in winter.

The choice of the insulating materials, using the current technological solutions available, was carried out in compliance with the studies carried out by De Bernardis on a series of buildings in Abruzzo with characteristics similar to those of Enna [15].

From these studies we found that there were about 50% energy savings using vacuum insulation panels (VIP), better set outside than inside; about 44% energy savings using aerogel; about 42%

using EPS (extruded expanded polystyrene); and 30% with external TIM (heat reflective materials), considering the same thickness of the insulation material.

For cases No. 1, No. 2 and No. 3, dealing with plastered masonry buildings ( $\lambda = 1.3 \text{ W/mK}$ ), where the walls were available for seismic improvement, an external 12 cm cladding insulation (L = 0.04 W/mK) was provided (cases No. 1 and No. 3), thus enhancing the limits of the law, and 8 cm in addition to the 4 cm existing internal insulation (case No. 2). For case No. 4, due to the presence of fair-faced stone around the doors and windows, a different choice was made.

Internal insulation was provided with 80 mm spray polyurethane foam ( $\lambda = 0.035 \text{ W/mK}$ ) directly to the back of the existing masonry, the application of a 50 mm calcium silicate sheet ( $\lambda = 0.094 \text{ W/mK}$ ) and 20 mm NHL plaster.

This solution was able to reach the performance standards required by law (U =  $0.279 \text{ W/m}^2\text{K}$ ) and did not cause condensation problems in winter and mould growth.

To verify this, a calculation by means of a Glaser diagram was performed and Figure 11 shows the results.



Figure 11. Glaser diagram of the proposed retrofit solution for Case No. 4.

This solution was also validated by Straube et al. in a report sponsored by the U.S. Department of Energy [45].

Strategy 1 requires intervention on only one of the four technical elements of the envelope identified (external walls, roofing, floor slab and doors and windows). Strategies 2–4 provide for the combination of two interventions on two technical elements of the envelope. Strategies 5 and 6 provide for the combination of three interventions on the technical elements of the envelope. Table 7 summarizes the six retrofit strategies and for each case study, indicated in subscript, the relative energy class is reported.

Table 7. Retrofit strategies of the selected case studies and relative energy class.

Strategy	Walls	Roofing	Slab-on-ground Floor	Doors and Windows
1	$F_1 E_2 E_3 D_4$	$G_1 G_2 G_3 G_4$	$G_1 G_2 G_3 G_4$	$G_1 G_2 G_3 G_4$
2 (Walls)	-	$D_1 D_2 D_3 D_4$	$E_1 D_2 D_3 D_4$	$E_1 D_2 D_3 D_4$
3 (Roofing)	$D_1 D_2 D_3 D_4$	-	$G_1 G_2 G_3 G_4$	$G_1 G_2 G_3 G_4$
4 (Slab-on-ground floor)	$E_1 D_2 D_3 D_4$	$G_1 G_2 G_3 G_4$	-	$G_1 G_2 G_3 G_4$
5 (Walls + roofing)	-	-	D <sub>1</sub> D <sub>2</sub> D <sub>3</sub> D <sub>4</sub>	D <sub>1</sub> D <sub>2</sub> D <sub>3</sub> D <sub>4</sub>
6 (Walls + slab-on-ground floor)	-	$D_1 D_2 D_3 D_4$	-	$E_1 D_2 D_3 C_4$

In case study No. 1, the best combinations always provide an intervention on the walls. We moved from energy class G, detected in the actual state, to energy class F, by means of only interventions on walls (26%), and to class D by intervening on the walls and roofing (71%), and to class E by intervening on the walls and floors on the ground or on the doors and windows. Combining three interventions between them, in no one case was the energy class higher than D.

In cases No. 2 and No. 3, on the other hand, by only intervening on the walls, it changed from class G in the actual state to class E, since the walls had a high percentage extension over the entire surface (75%). By intervening both on the walls and on any other closing element (roofing, slab-on-ground floor, doors and windows), it passed from class G to D. As regards the combination of several interventions, similar observations can be made to those of case No. 1.

In case No. 4, by only intervening on the walls, it passed from class G to D, and adding an intervention on another element of the envelope did not produce any class increase. Only using strategy 6 was class C obtained.

### 4. Discussion of the Results

The choice of interventions, accounting for the peculiarities on a case-by-case basis, was made by preferring interventions that involved the same technical element to produce both seismic and energy improvement. By means of the analyses in the previous paragraphs, the elements that showed the best performance were the walls and the roofing.

In fact, the sixth energy strategy, on the basis of the increase in energy class, led to the exclusion of the energy improvement of both the slab-on-ground floor and doors and windows.

For the walls, thin reinforced plaster interventions were planned for walls with limited thickness (30–40 cm), allowing an increase of the shear resistance in the plane and the flexural strength with limited thickness (about 3 cm) reducing the masses and weights on the structure.

For walls between 60 and 80 cm thick, mortar injections were planned to increase the mechanical resistance and the connection of the wall panels to each other by means of steel C-shaped profiles.

Table 8 shows the seismic and energy improvement interventions selected for the various case studies and the related class increases that can be obtained.

Case Studies	Seismic Improvement Interventions	Energy Improvement Intervention	Seismic Risk Class	Energy Class
No. 1	$S_{W2} S_{R3}$	External insulation, 12 cm thick, and $L = 0.04$ W/mk Roofing insulation, 15 cm thick, and $L = 0.04$ W/mk (Figure 12a)	$F \rightarrow E$	$G \rightarrow D$
No. 2	$S_{W1} S_{R2}$	External insulation, 8 cm thick, and $L = 0.04$ W/mk and integration of internal insulation, 4 cm thick Roofing insulation, 11 cm thick, and $L = 0.04$ W/mk (Figure 12b)	$E \rightarrow D$	$G \rightarrow D$
No. 3	$\mathrm{S}_{\mathrm{W1}}\mathrm{S}_{\mathrm{W3}}\mathrm{S}_{\mathrm{R2}}$	External insulation, 12 cm thick, and $L = 0.04$ W/mk Roofing insulation, 15 cm thick, and $L = 0.04$ W/mk	$f \to E$	$G \rightarrow D$
No. 4	$S_{W1} S_{R1}$	Internal insulation, 8 cm thick SPF, and $L = 0.035$ W/mk + 5 cm thick calcium silicate sheet, and $L = 0.094$ W/mk Roofing insulation from outside or intrados with insulation, 12 cm thick and $L = 0.04$ W/mk	$E \rightarrow D$	$G \rightarrow D$

Table 8. Seismic and energy improvement interventions.



**Figure 12.** Detail of seismic and energy interventions on roofing and outer vertical walls (**a**) Case No. 1; (**b**) Case No. 2.

# 5. Conclusions and Future Research Lines

This research proposes a design tool to improve the energy efficiency and the seismic resistance of historic buildings, considering their own architectural value, thus using a "case-by-case" approach.

Through the analysis of the case studies, based on a deep knowledge of their peculiarities, we found that it is possible for existing buildings in the historic centres to develop intervention methodologies on the same technical elements-generally walls and roofing-that combine seismic and energy improvements. These strategies allow the optimization of interventions and above all the possibility to obtain relevant tax relief, combining the two incentives: "sismabonus" (for seismic improvement) and "ecobonus" (for energy improvement) [Italian law No. 205/2017]. While the "sismabonus" is proportionate to the seismic improvement obtained with the intervention, equal to 70% in the case of a single class improvement and 80% for an improvement of two classes, the "ecobonus" refers to the mere total cost of the intervention without accounting for the results of the energy improvement obtained. With the application of the simplified method of the Italian guidelines, we found that for existing masonry buildings, a seismic risk class higher than E or D could not be achieved for areas with high seismic risk, except through the application of the conventional method and the single seismic class improvement, corresponding to the 70% incentive. Besides this, the simplified method for buildings in historic centres is much cheaper than the more conventional and complex method, which only allows an additional 10% financial incentive. It would be desirable that the economic incentive for energy efficiency would be proportionate to the energy class obtained. In particular, the incentive for single-family buildings, as are most buildings in historic centres, would be equivalent to that provided for condominiums that can reach up to 70% for interventions on the envelope with an incidence of more than 25% of the building's surface, and to 75% for improving winter and summer energy performance. Moreover, the possibility for single-family buildings to obtain credit from banks would also make retraining interventions accessible to low-income people.

Further analysis of other case studies is needed both in the town of Enna and in other historic centres to support this methodology.

Furthermore, the authors consider it important to perform a quantitative seismic analysis, e.g., based on the interpretation of the data provided by the limit states design methods, to verify the real improvement of the seismic class, which is assumed to be superior to that obtained with the qualitative method.

Author Contributions: T.B. conceived the research; D.E. wrote Section 1; T.B. wrote Section 2; T.B. and D.E. wrote Sections 3 and 4.

Conflicts of Interest: The authors declare no conflict of interest.

### References

- 1. Available online: https://earthquake.usgs.gov/earthquakes/map/ (accessed on 6 February 2018).
- Legge 2 Febbraio 1974, n. 64. Provvedimenti per le Costruzioni con Particolari Prescrizioni per le Zone Sismiche. Available online: http://www.gazzettaufficiale.it/atto/serie\_generale/ caricaDettaglioAtto/originario?atto.dataPubblicazioneGazzetta=1974-03-21&atto.codiceRedazionale= 074U0064&elenco30giorni=false (accessed on 2 April 2018).
- Cresme Ricerche SpA; Consiglio Nazionale degli Architetti, Pianificatori, Paesaggisti e Conservatori. Chi ha Progettato l'Italia? Ruolo Dell'architettura nella Qualità del Paesaggio Edilizio Italiano; Conferenza Nazionale degli Ordini degli Architetti, Pianificatori, Paesaggisti e Conservatori: Padova, Italy, 2017.
- Available online: http://www.europarl.europa.eu/legislative-train/theme-resilient-energy-union-witha-climate-change-policy/file-energy-efficiency-directive-review/ (accessed on 8 January 2018).
- Available online: https://ec.europa.eu/energy/en/topics/energy-efficiency/buildings (accessed on 19 March 2018).
- 6. Available online: http://www.paesifantasma.it/Paesi/italia.html (accessed on 20 March 2018).

- Available online: http://italia.indettaglio.it/eng/statistiche/statistiche\_edilizia.html (accessed on 19 March 2018).
- Maietti, F. Centri Storici Minori: Progetti di Recupero e Restauro del Tessuto Urbano fra Identità Culturale e Salvaguardia; Maggioli Editore: Santarcangelo di Romagna, Italy, 2008; ISBN 978-88-387-4656-7.
- Coletta, T. I Centri Storici Minori Abbandonati Della Campania—Conservazione, Recupero e Valorizzazione; Edizioni Scientifiche Italiane: Naples, Italy, 2010; ISBN 9788849520613.
- Righi, A.; Dalla Mora, T.; Peron, F.; Romagnoni, P. Historical buildings retrofit he city hall of the city of Motta di Livenza (TV). *Energy Procedia* 2017, 133, 392–400. [CrossRef]
- Nardi, I.; De Rubeis, T.; Taddei, M.; Ambrosini, D.; Sfarra, S. The energy efficiency challenge for a historical building undergone to seismic and energy refurbishment. *Energy Procedia* 2017, 133, 231–242. [CrossRef]
- 12. Brandonisio, G.; Lucibello, G.; Mele, E.; De Luca, A. Damage and performance evaluation of masonry churches in the 2009 L'Aquila earthquake. *Eng. Fail. Anal.* 2013, *34*, 693–714. [CrossRef]
- Ortega, J.; Vasconcelosa, G.; Rodrigues, H.; Correia, M.; Lourenco, P.B. Traditional earthquake resistant techniques for vernacular architecture and local seismic cultures: A literature review. *J. Cult. Heritage* 2017, 27, 181–196. [CrossRef]
- Belleri, A.; Marini, A. Does seismic risk affect the environmental impact of existing buildings? *Energy Build*. 2016, 110, 149–158. [CrossRef]
- De Berardinis, P.; Rotilio, M.; Marchionni, C.; Friedman, A. Improving the energy-efficiency of historic masonry buildings. A case-study: A minor centre in the Abruzzo region, Italy. *Energy Build.* 2014, 80, 415–423. [CrossRef]
- Modena, C.; da Porto, F.; Valluzzi, M.R.; Carapezza Guttuso, F.; Iannelli, P.; Rubino, C. Sustainable approaches to the assessment and mitigation of seismic risk and of the effects of earthquake induced damages to historic urban centers. In *Proceedings of the 10th International Conference on Structural Analysis of Historical Constructions*. *Anamnesis, Diagnosis, Therapy, Controls, SAHC 2016, Leuven, Belgium, 13–15 September 2016*; Van Balen, K., Verstrynge, E., Eds.; CRC Press: London, UK, 2016; ISBN 9781138029514.
- Cardinale, T.; Colapietro, D.; Cardinale, N.; Fatiguso, F. Evaluation of the efficacy of traditional recovery interventions in historical buildings. A new selection methodology. *Energy Procedia* 2013, 40, 515–524. [CrossRef]
- La Greca, P.; Margani, G. Seismic and energy renovation measures for sustainable cities: A critical analysis of the Italian scenario. In *Proceedings of the International Conference on Seismic and Energy Renovation for Sustainable Cities, SER4SC, Catania, Italy, 1–3 February 2018*; Margani, G., Rodonò, G., Sapienza, V., Eds.; EdicomEdizioni: Gorizia, Italy, 2018; pp. 93–104. ISBN 978-88-96386-56-9.
- 19. Available online: http://www.niker.eu/ (accessed on 10 January 2018).
- Vitale, M.R.; Salerno, M.; Billeci, B.; Dessì, M.; Callea, L.; The historical centre of Marignane (France). Survey, diagnostic campaign and structural assessment. In *Proceedings of the 10th International Conference on Structural Analysis of Historical Constructions. Anamnesis, Diagnosis, Therapy, Controls, SAHC 2016, Leuven, Belgium, 13–15 September 2016;* Van Balen, K., Verstrynge, E., Eds.; CRC Press: London, UK, 2016; ISBN 9781138029514.
- Decreto Ministeriale n. 65 del 07/03/2017. Sisma Bonus—Linee Guida per la Classificazione del Rischio Sismico delle Costruzioni e i Relativi Allegati. Modifiche all'articolo 3 del Decreto Ministeriale Numero 58 del 28/02/2017. Available online: http://www.mit.gov.it/normativa/decreto-ministeriale-numero-65-del-07032017 (accessed on 2 April 2018).
- 22. Conseil de l'Europe. *Cahiers du Centre Européen de Géodynamique et de Séismologie*; European Macroseismic Scale; Grünthal, G., Ed.; Conseil de l'Europe: Strasbourg, Italy, 1998; Volume 15.
- 23. UNI/TS 11300-1, Energy Performance of Buildings—Part 1: Evaluation of Energy Need for Space Heating and Cooling; Ente Italiano di Normazione, 2014.
- 24. UNI EN 15243, Ventilation for Buildings—Calculation of Room Temperatures and of Load and Energy for Buildings with Room Conditioning Systems; Ente Italiano di Normazione, 2008.
- 25. Available online: http://italia.indettaglio.it/eng/sicilia/enna.html (accessed on 7 February 2018).
- Santos, C.; Ferreira, T.M.; Vicente, R.; Mendes da Silva, R. Building typologies identification to support risk mitigation at the urban scale. Case study of the old city centre of Seixal, Portugal. *J. Cult. Heritage* 2013, 14, 449–463. [CrossRef]
- 27. May, J.; Reid, A. Architettura Senza Architetti; Rizzoli: Bologna, Italy, 2010; ISBN 9788817037853.
- 28. Severino, C.G. Enna. La Città al Centro; Gangemi Editore: Rome, Italy, 1996; ISBN 9788874486816.

- Acampa, G.; Campisi, M.T.; Zarbo, I. Fra conservazione e riuso: Strumenti di valutazione per una progettazione critica. In *Proceedings of the International Conference on Documentation, Conservation and Recovery* of Architectural Heritage and Landscape Protection, ReUSO 2016, Pavia, Italy, 6–8 October 2016; Parrinello, S., Ed.; Edifir-Edizioni Firenze: Florence, Italy, 2016; pp. 284–292. ISBN 978-88-7970-810-4.
- Decreto Interministeriale 26 Giugno 2015—Adeguamento Linee Guida Nazionali per la Certificazione Energetica Degli Edifici. Available online: http://www.sviluppoeconomico.gov.it/index.php/it/normativa/ decreti-interministeriali/2032968-decreto-interministeriale-26-giugno-2015-adeguamento-linee-guidanazionali-per-la-certificazione-energetica-degli-edifici.html (accessed on 20 March 2018).
- UNI 10351:2015. Building Materials and Products—Hygrothermal Proprieties—Procedure for Determining the Design Values; UNI, Ente Italiano di Normazione: Milan, Italy, 2015.
- UNI EN ISO 10456:2008. Building Materials and Products—Hygrothermal Properties—Tabulated Design Values and Procedures for Determining Declared and Design Thermal Values; UNI, Ente Italiano di Normazione: Milan, Italy, 2008.
- UNI EN ISO 13370:2008. Thermal Performance of Buildings. Heat Transfer via the Ground. Calculation Methods; UNI, Ente Italiano di Normazione: Milan, Italy, 2008.
- 34. Grabau, A.W. Paleozoic coral reefs. GSA Bull. 1903, 14, 337-352. [CrossRef]
- 35. Genova, E.; Kilian, R. Thermal and hygrometric properties of traditional calcarenite stones in the area of Palermo. *Energy Procedia* 2017, 122, 1081–1086. [CrossRef]
- Maio, R.; Estêvão, J.M.C.; Ferreira, T.M.; Vicente, R. The seismic performance of stone masonry buildings in Faial island and the relevance of implementing effective seismic strengthening policies. *Eng. Struct.* 2017, 141, 41–58. [CrossRef]
- 37. Gattesco, N.; Gubana, A.; Melotto, M. GFRP to strengthen masonry walls: Numerical analysis and evaluation of the different mechanical parameters role. In *Proceedings of the 10th International Conference on Structural Analysis of Historical Constructions. Anannesis, Diagnosis, Therapy, Controls, SAHC 2016, Leuven, Belgium, 13–15 September 2016;* Van Balen, K., Verstrynge, E., Eds.; CRC Press: London, UK, 2016; ISBN 9781138029514.
- Lucibello, G.; Brandonisio, G.; Mele, E.; De Luca, A. Seismic damage and performance of Palazzo Centi after L'Aquila earthquake: A paradigmatic case study of effectiveness of mechanical steel ties. *Eng. Fail. Anal.* 2013, 34, 407–430. [CrossRef]
- Branco, M.; Guerreiro, L.M. Seismic rehabilitation of historical masonry buildings. *Eng. Struct.* 2011, 33, 1626–1634. [CrossRef]
- Diz, S.; Costa, A.; Costa, A.A. Efficiency of strengthening techniques assessed for existing masonry buildings. Eng. Struct. 2015, 101, 205–215. [CrossRef]
- 41. Suriya Prakash, S.; Alagusundaramoorthy, P. Behaviour of masonry load bearing walls retrofitted with GFRP composites. J. Struct. Eng. 2009, 36, 397–405.
- Sivaraja, S.S.; Thandavamoorthy, T.S.; Vijayakumar, S.; Mosesasaranganathan, S.; Rathnasheela, P.T.; Dasarathy, A.K. GFRP Strengthening and Applications of Unreinforced Masonry wall (UMW). *Proc. Eng.* 2013, 54, 428–439. [CrossRef]
- Proença, J.; Sousa Gago, A.; Cardoso, J.; Cóias, V.; Paula, R. Development of an innovative seismic strengthening technique for traditional load-bearing masonry walls. *B Earthq. Eng.* 2012, 10, 113–133. [CrossRef]
- 44. Penna, P.; Prada, A.; Cappelletti, F.; Gasparella, A. Multi-objectives optimization of Energy Efficiency Measures in existing buildings. *Energy Build.* **2015**, *95*, 57–69. [CrossRef]
- Straube, J.F.; Ueno, K.; Schumacher, C.J. Measure Guideline: Internal Insulation of Masonry Walls; Building Science Corporation: Somerville, MA, USA, 2012.



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





# A Methodology for an Integrated Approach for Seismic and Energy Refurbishment of Historic Buildings in Mediterranean Area

# Angela Moschella <sup>1,\*</sup>, Antonio Gagliano <sup>2</sup>, Alessandro Lo Faro <sup>1</sup>, Attilio Mondello <sup>1</sup>, Angelo Salemi <sup>1</sup> and Giulia Sanfilippo <sup>1</sup>

- <sup>1</sup> Department of Civil Engineering and Architecture, University of Catania, Via Santa Sofia 64, 95123 Catania, Italy; alessandro.lofaro@darc.unict.it (A.L.F.); amondello@darc.unict.it (A.M.); angelo.salemi@darc.unict.it (A.S.); giulia.sanfilippo@darc.unict.it (G.S.)
- <sup>2</sup> Department of Electric, Electronic and Computer Engineering, University of Catania, Viale A. Doria 6, 95125 Catania, Italy; agagliano@dii.unict.it
- \* Correspondence: angela.moschella@darc.unict.it; Tel.: +39-095-738-2532

Received: 28 May 2018; Accepted: 10 July 2018; Published: 13 July 2018



Abstract: Energy savings and seismic risk mitigation are the main issues regarding the refurbishment of traditional buildings. Referring to historic buildings, we have to take into account, as design requirements, the cultural sustainability, which means the respects of the cultural value of the built heritage. Therefore, for historic buildings is not acceptable to adopt the conventional design choices applied to newer buildings on energy efficiency and seismic risk mitigation. Generally, the design on the built heritage requires a careful cognitive phase for firstly to identify the performance deficits and subsequently to define which actions are compatible with the cultural value of each building, according to a "case by case" approach. In Italy, specific guidelines have been elaborated on cultural heritage but such guidelines are not integrated into a single methodological process. This paper, through the study of two historic buildings, aims to identify the relationship between the two specialisms, seismic and energy, within an integrated approach. As a result, this study proposes an innovation process characterized by the integration of these two protocols within the cognitive phase and, especially in the pre-diagnosis phase; this phase is identified in the standard diagnosis (Energy approach) and in the LV2 knowledge (Seismic empirical approach).

**Keywords:** historic massive envelope; cultural value; energy performances; damage mechanisms; pre-diagnostic process; seismic improvement; energy savings; Eastern Sicily

# 1. Introduction

Built heritage tells of the close relationship between man and environmental resources; this has defined the image of our historic cities in the material and formal aspects. According to ISTAT data (Italian National Institute of Statistics), the residential historic buildings (built before 1919) account for 19.2% of the Italian built heritage. Considering the masonry buildings, this percentage increases to 61.5%, if we think that this type of construction had also been used up to the 1950s.

Every building is an open system that interacts with its environment continuously. We refer to the climatic conditions, which, in the past, determined the shape of the urban fabric and the effects of earthquakes. Within the Mediterranean climate environment, we have directed our attention to Catania, whose seismicity degree is very high; in fact, the city was rebuilt after the violent earthquake of 1693. Its historic centre is therefore made up of massive buildings, with a loadbearing masonry structure.

Traditional buildings are characterized by vertical opaque closures with high thermal transmittance (U), which do not guarantee a good energy performance in the winter season [1].







# Article Energy and Seismic Recovering of Ancient Hamlets: the Case of Baia e Latina

# Marina Fumo<sup>1</sup>, Antonio Formisano<sup>2</sup>, Giulia Sibilio<sup>1</sup> and Antonella Violano<sup>3,\*</sup>

- <sup>1</sup> Department of Civil, Construction and Environmental Engineering DICEA—University of Naples "Federico II", Piazzale V. Tecchio, 80, 80125 Napoli, Italy; marina.fumo@unina.it (M.F.); uni-file@libero.it (G.S.)
- <sup>2</sup> Department of Structures for Engineering and Architecture (DIST)—University of Naples "Federico II", Piazzale V. Tecchio, 80, 80125 Napoli, Italy; antoform@unina.it
- <sup>3</sup> Department of Architecture and Industrial Design DADI—University of Campania "L. Vanvitelli", Via S. Lorenzo, 81031 Aversa (Caserta), Italy
- \* Correspondence: antonella.violano@unicampania.it; Tel.: +39-347-6861477

Received: 6 May 2018; Accepted: 7 August 2018; Published: 9 August 2018



Abstract: This research proposes the development of a diagnostic tool to separately inspect the energetic and seismic behaviour of buildings in the small hamlet of Baia e Latina (district of Caserta) in order to evaluate and implement retrofitting interventions from seismic, energetic, and functional points of view. Methods, approaches, and tools relating to the minimisation of seismic vulnerability and energy consumption have been increasingly used and tested in order to ensure both sustainability and safety, with a connection that may improve the performances of both cultural and environmental heritage. The diagnosis method, stemming from the energy audit and the energy imprint evaluations of the buildings system (and the envelope above all), aims to redesign the whole construction or some of its parts within an energetic framework. With reference to the seismic behaviour of building aggregates, the basic methodology that has been conceived for isolated masonry buildings through a survey form has represented the starting tool for the application of an appropriate quick evaluation form considered for the aggregated structural units of historical centres. Finally, the methodology employed is aimed at obtaining an Energy Performance Certificate for the structural units of examined masonry aggregates without neglecting their seismic behaviour, which has been assessed in terms of vulnerability and damage.

**Keywords:** technological design; energy performance; seismic vulnerability assessment; risk analysis; masonry building aggregates

# 1. Introduction

In the latest ICOMOS document: "Guardians of Heritage, Finders of Meaning" (February 2017), the value of choral architectural heritage, in its global conception of human, social, economic, and environmental ecosystems, is reaffirmed with wise attention to its resilience and safety features. The future development model of historic urban tissue can change its individual units by responding to functional needs; however, it must keep its form and identity intact. In fact, the urban landscape, including buildings and their relations with external open spaces (streets, avenues, squares, courtyards, gardens ... ), is a "system" that we have to preserve with its authenticity and integrity, in order to maintain the perception of its history and of social and economic urban changes through the times.

The Italian territory has been significantly damaged by many seismic events over the last 10 years; the seismic issue is finally becoming a priority in the country. Affected populations asked for the preservation of ancient hamlets, and not just for emergency solutions: timber houses offered and





# Article Mechanical Tests on Innovative BIPV Façade Components for Energy, Seismic, and Aesthetic Renovation of High-Rise Buildings

# Rossella Corrao 1,2

- <sup>1</sup> Dipartimento di Architettura, Scuola Politecnica, Università degli Studi di Palermo, 90128 Palermo Italy; rossella.corrao@unipa.it; Tel.: +39-912-389-6144
- <sup>2</sup> SBskin, Smart Building Skin s.r.l., 90128 Palermo, Italy

Received: 30 September 2018; Accepted: 26 November 2018; Published: 30 November 2018



Abstract: The paper shows the results of mechanical tests carried out on prototypes of a new Building Integrated Photovoltaic (BIPV) component developed by the author and SBskin Smart Building Skin s.r.l. This patented innovative component is able to merge structural function, insulation proprieties, and production of clean energy for retrofit actions and/or the construction of translucent façades in high-rise buildings located in different climatic contexts. Due to colored PV cells integrated into 3 Dimensional (3D) glass components and the dry-assembly system used for assembling them into precast and pre-stressed panels, an easy and creative customization of the product is allowed. Green energy production, safety, and energy efficiency of buildings can be assured in accordance with the environmental conditions and users' needs. The pre-stressing force used to improve the mechanical resistance of the panel toward horizontal forces due to winds and earthquakes guarantees the construction of secure translucent and active building envelopes. The paper summarizes the features of this innovative and patented BIPV product by focusing on its mechanical behavior. Laboratory tests are described and commented for underlining the benefits derived from the use of the dry-assembly system and of the supporting structure made of plastic for the construction of the panels. Bending and breaking strength tests have been carried out on two sq.m of panel prototypes, which have been dry-assembled through a supporting structure made of Polypropylene (PP) in order to compare the results with the theoretical calculations derived from the Finite Element (FE) simulations. Cyclic mechanical testing of the panel has been also carried out to verify its behavior under cyclic loading and understanding its ability to counteract the actions of the wind and earthquake.

**Keywords:** building envelope; innovative product; translucent panel; multifunctional component; energy efficiency; BIPV (Building Integrated Photovoltaic); dry-assembly system; mechanical resistance; high-rise building; façade

# 1. Introduction

SBskin. Smart Building Skin s.r.l. (www.sbskin.it) is an innovative start-up and academic spin off of the University of Palermo co-founded by the author (the CEO) with the aim to give a small and ambitious contribution to make our planet greener and our buildings sustainable and beautiful at the same time. By spreading the use of Renewable Energy Sources (RESs) and, in particular, of third generation PV technologies as well as by improving the building sector in Europe and abroad through the use of an innovative solution for the construction of translucent, active, and safe building envelopes even in high-rise buildings.

Buildings account for about 40% of the European Union's and the world's  $CO_2$  emissions largely due to the operations of air conditioning, heating, and electricity systems [1]. The "2020 Climate and

MDPI St. Alban-Anlage 66 4052 Basel Switzerland Tel. +41 61 683 77 34 Fax +41 61 302 89 18 www.mdpi.com

Sustainability Editorial Office E-mail: sustainability@mdpi.com www.mdpi.com/journal/sustainability

